Electronic Thesis and Dissertation Repository

4-24-2023 11:45 AM

Music Making in Elderly Community Program for Korean Immigrants in Canada

H. Elisha Jo, Western University

Supervisor: Veblen, Kari K., The University of Western Ontario

A thesis submitted in partial fulfillment of the requirements for the Doctor of Philosophy degree in Music

© H. Elisha Jo 2023

Follow this and additional works at: https://ir.lib.uwo.ca/etd



Part of the Music Education Commons

Recommended Citation

Jo, H. Elisha, "Music Making in Elderly Community Program for Korean Immigrants in Canada" (2023). Electronic Thesis and Dissertation Repository. 9692.

https://ir.lib.uwo.ca/etd/9692

This Dissertation/Thesis is brought to you for free and open access by Scholarship@Western. It has been accepted for inclusion in Electronic Thesis and Dissertation Repository by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

Abstract

This integrated-article dissertation explores music making in a community program for Korean older adult immigrants in Canada. Drawing upon WHO's active aging framework and theoretical concepts of active aging, Erickson's 8th stage, and lifelong learning, this research investigates social participation and engagement through music making among seniors.

The first article uses ethnographic research tools to explore a culturally specific community program for Korean elderly immigrants called Canada Enoch Seniors' College. Through engaging musically in familiar songs in their mother tongue, participants redefined/claimed identity and group membership. Music plays an important role as participants collectively remember their shared culture.

The following article is an ethnography of five Korean Canadian first-generation immigrant seniors. After surviving great historical turmoil, including oppression through Japanese rule and the Korean War, these elders chose to move to a foreign land, where they faced many barriers, in order to offer their children a better life. Now participants in the Enoch College, these older adults happily connect and recreate memories of their Korean homeland through music and shared activities in a church-based context. Their stories illuminate how music has been integral to their lives.

The third article assesses the program's impact using a health-related quality of life survey SF-36v. This survey measures health -related quality of life and overall well-being. The results are statistically significant and suggest that the program provides positive effects to these elderly participants.

This research considers the power of music to promote wellness and active ageing among Korean Canadian seniors. Having a strong sense of meaning and purpose that music can afford is not only associated with well-being in old age, but also relates positively to successful coping with pain and illness. Although limited to a specific population in a particular time and context, this study contributes to growing body of research, filling. Voices of these Korean Canadian seniors offer valuable insights for music educators, community musicians, researchers, community program directors, health care providers as well as those guiding government policy.

Keywords: music, older adults, community program, community music, lifelong learning, aging, well-being, identity, immigrant, culture, engagement, Korean immigrants, Asian diaspora

Summary for Lay Audience

This integrated-article dissertation explores music making in a community program for Korean older adult immigrants in Canada. Drawing upon WHO's active aging framework and theoretical concepts of active aging, Erickson's 8th stage, and lifelong learning, this research investigates social participation and engagement through music making among seniors.

The first article uses ethnographic research tools to explore a culturally specific community program for Korean elderly immigrants called Canada Enoch Seniors' College. Through engaging musically in familiar songs in their mother tongue, participants redefined/claimed identity and group membership. Music plays an important role as participants collectively remember their shared culture.

The following article is an ethnography of five Korean Canadian first-generation immigrant seniors. After surviving great historical turmoil, including oppression through Japanese rule and the Korean War, these elders chose to move to a foreign land, where they faced many barriers, in order to offer their children a better life. Now participants in the Enoch College, these older adults happily connect and recreate memories of their Korean homeland through music and shared activities in a church-based context. Their stories illuminate how music has been integral to their lives.

The third article assesses the program's impact using a health-related quality of life survey SF-36v. This survey measures health -related quality of life and overall well-being. The results are statistically significant and suggest that the program provides positive effects to these elderly participants.

This research considers the power of music to promote wellness and active ageing among Korean Canadian seniors. Having a strong sense of meaning and purpose that music can afford is not only associated with well-being in old age, but also relates positively to successful coping with pain and illness. Although limited to a specific population in a particular time and context, this study contributes to growing body of research, filling. Voices of these Korean Canadian seniors offer valuable insights for music educators, community musicians, researchers, community program directors, health care providers as well as those guiding government policy.

Acknowledgments

Grace - a gift freely given - is perhaps another word to sum up this experience.

I have been blessed beyond measure to receive the best gift of all: a relationship built on trust and patience.

Dr. Kari Veblen has been a living example to me, walking the very path I aspire to follow. With a smile like a gentle wind, so refreshing that it can even beautify some of the darkest moments. She has always been there for me, sincere and kind, embodying the wisdom I hope to emulate someday. I wouldn't be here if it hadn't been for her faithful support when others, including myself, doubted. More than words can express, I want to extend my heartfelt gratitude to you, dear Kari, for your *help* in every possible way.

'Help' holds a significant meaning for me. The story takes us back to a time when my dear sister and I ventured through what felt like a dark abyss in our lives - a period marked by financial, physical, and emotional struggles. Seated across from each other at a café, our words were few. Perhaps as an effort to break the silence, I impulsively directed a question at my sister, but in truth, I was posing that question to myself. I asked, "What is your deepest wish?" After a brief pause, she responded softly, "To be a helper." No further explanation was necessary; our emotions nearly overwhelmed us. "Me, too..." I whispered to myself. In that moment, it felt like a prayer, stemming from our feelings of hopelessness.

Since then, I have learned the truth from the Word of God and the wise words of elderly Koreans like Yong, Kay, Kim, and Min. They have reminded me that it is not about the circumstances themselves, but rather about how we respond and learn from what happens to us.

This journey with elderly Koreans was a delightful surprise in many ways. Spending every Thursday – Enoch day – with these wonderful individuals, their faces etched vividly in my heart, brought an unparalleled joy to my life. It was an immense privilege to have crossed paths and spent time with each and every elderly Korean:

Elder Choi, the grandpa I had never known, who still plays his harmonica in the depths of my heart; his perfect life partner, Kay, who possesses timeless grace; Elder Song, whose undying passion for music continues to inspire; Hyo, whose quiet demeanor speaks volumes beyond words; Pastor Kim, a young man at heart, brimming with the same passion; Min, who blossomed like Spring flowers; CESC director/pastor Kim, your passion for older adults and gentle leadership have contributed to making CESC what it is today, and her faithful supporter and husband, Elder Kim, alongside numerous other elders who willingly opened up their hearts to embrace and welcome me – this experience was genuinely humbling.

I am grateful to Dr. Patrick Potter, whose invaluable contribution and expertise have truly enriched this study.

Dr. Stephen Messenger, your kindness and generosity have made this entire endeavor possible. Thank you for your help.

Dr. Ruth Wright, I can't thank you enough for your thoughtful questions that ensured the message is well thought-out and crystal clear.

Dr. Betty Anne Younker, for your concise insights were instrumental in the study, and I am appreciative of your input.

Dr. Carol Beynon, your encouragement and open-mindedness have been a constant source of motivation as I moved forward with this research. Thank you for your support. Dr. Victor Fung, your cultural insights added a depth to the study, and I'm grateful for the broader perspective you brought.

To each and every one of you, thank you from the bottom of my heart for your unyielding support. Your contributions have made a significant impact on this journey, and I couldn't have done it without you.

To Dad and Mom, whose prayers have brought me this far, I am deeply grateful and so humbled by your love. I am so blessed to be your daughter, and I couldn't have asked for better Appa and Umma. Your sincere and devoted heart to God is the best gift that parents can give to their child. Your continuous support and patience have carried me on this journey. My dearest and lovely sis, Jungsuna~ you are my biggest support, and I thank God for you each and every day. You know that you will always have a special place in my heart.

Dearest Sam, your faithful prayers filled with tears will never be forgotten; they will live on in my heart. I am indebted to your love and incredibly grateful for the greatest lesson in life: always teaching the Word of truth through your unwavering obedience to God. Your compassionate heart and your journey of faith are what I aspire to follow. Thank you dearly for being a source of inspiration and love.

Thanks to JDC family: Jenny unni, Robin, Ruth, Amy, Uncle Paul, Mr. C. Lee, Mrs. J. Jung, Mrs. E. Lee, Mom Lee, Soo unni, EH unni, Yes-i & HY unni, JH unni, Hyerin, Soyeon, Eddy and the list goes on. For all your help, love and prayers, never without coffee and chocolates. Thank you.

Above all, Jesus Did it! Jooha!

More than words can express, more than I can ever know, it is only by grace I am what I am.

Table of Contents

Abstract	i
Summary for Lay Audience	iii
Acknowledgments	iv
Table of Contents	vii
Chapter One: Introduction and Overview	1
Study Overview	2
My Background	3
Coming to the Enoch Senior's College (CESC)	6
Purpose of the Study	8
Research Questions	9
Theoretical Framework	9
Overview of the Research Framework	9
Data Collection and Analysis	10
Design and Method	11
Ethnography	11
Narrative Inquiry	12
Principles of a Mixed Methods Approach	13
Overview of the Complementary Study Methodologies	13
Mixed Methods Research	13
Three Research Designs	14
Quantitative and Qualitative Research Methods	15
Data Collection and Analysis	17
Health Survey	18
Data Collection and Analysis	19
Dissertation Overview	20
Integrated-article Format	20
Study Overview	21

Chapter Two: Review of Literature	23
Wellness as Understood Through the Concept of Active Ageing	23
WHO's Active Ageing Framework	23
Life Course Perspective to Active Ageing	28
The Demographics of Ageing in Canada	28
Life Review and Reminiscence	30
Korean Community in Canada	35
Health Disparities and Research Needs of Visible Minority Groups in Canada	37
Role of Ethnic Faith-based Organizations	40
Korean Church's Role in Welcoming Immigrants	40
Historical Background of Korean Immigrant Seniors	42
Korean Elder Identity	45
Lifelong Learning	48
Promotion of Lifelong Learning: World Trends	50
Older Adults' Perspectives of Learning	52
Learning Approach	53
Formal Older Adult Learning Programs	54
Lifelong Music Learning in Community Contexts	55
Community Contexts for Lifelong Music Engagement	58
Community Music: Definitions and Negotiations	59
Historical Background of Community Music	60
Community Music in Canada	60
Social Networking in Community Music	61
Missing Research in Community Music Literature	61
Chapter Three: Korean Immigrant Seniors' Music Making in an Ethno-Cultural Comm	unity
Program in Canada	
Introduction	64
Methodology	65
Program Structure	66
Shared Musical Knowledge Plays a Significant Role in the Program	69

Health and Wellness	69
"Well-being lunch" at Enoch	70
Valuing and Belonging	71
Discussion and Conclusions	71
Reflection	72
References	75
Chapter Four: Enoch Senior's College for Korean Immigrant Seniors: Quality of Life I	Effects
	79
Résumé	79
Abstract	80
Introduction	80
Background on the CESC	85
Methods	86
Results	89
Quantitative Evaluation	89
Qualitative Inquiry	90
CESC as Promoting Health and Vitality	90
CESC as Fostering Renewed Perspectives on Aging and Self	91
CESC as Providing Emotional Support and a Sense of Belonging	93
Discussion	95
Reflection	
References	105
Chapter Five: Remembering a Hard Life with Joy: Music-Making among Korean Elde	rs . 116
Literature Review	116
Method	118
Narrative Approach	118
Continuing Ethnographic Framework	119

Five Narratives	119
1. Song's story: "Wind from a mountain and wind from a river."	119
2. Yong's story: Harmonica and "Heidenröslein"	121
3. Kay's story: Joy of singing, thirst for learning	122
4. Kim's story: Music in a life for others	124
5. Min's story: Grief opens the heart	126
Conclusion	130
Reflection	130
References	134
Chapter Six: Summary and Conclusion	141
Revisiting Research Questions	141
Research Question 1) What Factors Contribute to Participation in the Program?	141
Focus on Wellbeing and Healthy Ageing	142
Music's Part in the Program	142
Korean Traditional Culture and Music	143
Familiar Music	146
Language Considerations	146
Cultural Understandings of Ageing	147
Summary of Research Question 1	148
Research Question 2) How Does Participating in the Community Program Affect	ct Health and
Active Aging among Elderly Korean Participants?	148
Health Survey	149
Findings from Health Survey	149
Discussion of Program Features	150
Research Question 3) What Does Music Mean and How Important is it in the Li	ves of
Elderly Koreans?	151
Shared Cultural Identity	152
Music Making	152
Life Narratives	152

Music assigns Meaning and Value to Life	
Hwa-Byung and Need for Social Connection	
Further Study Findings	156
Ageing and Quality of Life	156
Wellness and Active Ageing	
Importance of Social Support	
COVID-19 Pandemic and Social Isolation	
Senior Korean Immigrants in North America	
Government Initiatives and Local Stopgap Measures	161
Canadian Lifelong Learning Programs for Seniors	
Researcher's Stance	
Implications	
Recommendations for Practice and Future Research	
Summary	166
References	
Appendix A: Major Terminology	222
Appendix B: The SF-36v2 Health Survey	225
Curriculum Vitae – Elisha Jo	229

Chapter One: Introduction and Overview

As countries and cultures experience demographic shifts, wellness and active aging within a population become more and more crucial concerns. Active aging is a global term promoted by the World Health Organization (WHO). It has been employed in diverse policy frameworks which seek to ensure health and well-being for senior citizens in countries which have rapidly aging populations. The WHO defines active aging as "the process of optimizing opportunities for health, participation and security in order to enhance quality of life as people age" (World Health Organization [WHO], 2002, p. 12). Here, "active" refers to lifelong engagement in social, economic, cultural, spiritual, and civic areas, rather than simply a person's employability or ability to remain physically active.

The United Nations has reported that, globally, the number of international migrants increased by nearly 48 million between 2000 and 2010 (United Nations, 2020). In 2020, the number of these persons who live outside of their country of origin reached 281 million. Despite declines in migration flow since 2020 due to the COVID-19 pandemic, this number has been steadily increasing.

In Canada, international migration is the main source of population growth (Statistics Canada, 2022a). Canada has just recently recorded its highest population growth rate—0.3% in the first quarter (January 1, 2022, to April 1, 2022)—and has welcomed 113,699 immigrants (Statistics Canada, 2022a). Immigration represents two-thirds of Canada's population growth, and immigrants make up more than 21% of Canada's population; permanent and non-permanent immigration accounted for over 80% of Canada's population growth in 2019 (Government of Canada, 2021). The proportion of immigrants is estimated to reach between 24.5% and 30.0% in 2036. Moreover, it is estimated that a large proportion of these immigrants, between 55.7% and 57.9%, have been born in Asia, while between 15.4% and 17.8% have been born in Europe (Statistics Canada, 2017a).

My research focuses on visible minority seniors who have a Korean cultural background. Their Korean background makes them the minority among minority groups¹ in Canada. The present study uses the theoretical underpinning of the WHO's active-aging framework. It was conducted in the Greater Toronto Area (GTA), one of the few places in Canada where many immigrant seniors reside and receive support from ethnic organizations. As Canada prepares to address the diverse needs of an increasingly older population and puts policies concerning aging into practice, community programs rooted in ethnic communities can provide worthwhile, efficacious exemplars. Situated community programs are uniquely positioned to adapt, tailor, and implement culturally appropriate and relevant learning opportunities for the elderly.

Study Overview

This dissertation focuses on wellness and active-aging among Korean-Canadian immigrant seniors who were participating in a community-based, ethno-cultural music-making program for seniors at the Enoch Senior's College (CESC) in the Greater Toronto Area. This chapter introduces the study, highlights the significance of the study, and provides an overview of the research framework and thesis structure.

The dissertation takes the form of three integrated articles which, from complementary perspectives, examine wellness and active-aging through music in the selected cohort of Korean-Canadian immigrant seniors. The first article studies the group via an ethnography, providing an introduction and backdrop to the study. Understanding the context of the cohort, how it is situated and planned, and who is involved helps to provide insight into the group.

The second article presents personal narratives of self-selected members who offer their unique experiences of musical engagement. Only a few stories are shared here because participation was voluntary and only a few volunteered. The narratives add depth and richness to the study and help to create a broader perspective that is helpful in the analysis of other data.

¹ According to the most recent census data available, Korean population accounted for 0.5% (188,710) of the total Canadian population in 2016 (Statistics Canada, 2017b).

The third article uses a health survey to analyze well-being. The health survey showed the effectiveness of the program and how music played a positive role in the lives of the Korean seniors. The survey findings are especially helpful for community workers who are looking for ways to improve the lives of community-dwelling seniors. The present study may indeed provide a resource to inform similar community-based seniors' programs. Overall, although the three articles were written separately with three different views and designs and with different purposes, they complement each other as a whole.

My study of wellness and active-aging among Korean-Canadian immigrant seniors began with an ethnography of the Enoch Senior's College (CESC) where I acted as a participant observer to collect data from multiple sources. "Enoch," as it is called by the elderly Korean participants, is a community program for seniors offered by Westside Presbyterian church. The first of its kind in Ontario, the program began in 2003. The history of CESC and positive comments from the seniors themselves about the program sparked my interest. News of CESC spread by word of mouth among Korean seniors in Korean communities in the Greater Toronto Area as CESC became a popular gathering place for seniors. As its regard grew, the program developed more classes for seniors and helped other churches implement similar programs.

I joined the CESC music program in 2013 as a volunteer, first as an accompanist for small groups, then in a variety of roles from photographer to greeter to program assistant. Hearing the stories of elderly Korean immigrants' hardships, resilience, and wisdom, first-hand, has been a significant, life-changing experience for me.

My Background

I was born in Korea into a loving family with one sibling, my younger sister. We have an extended family of aunts, uncles, and cousins and, since members of my family are devout Christians, many of the friends we have met through church have become like family to us. It was also probably in the church where I was first introduced to music. The musical activity I most enjoyed was singing in a Sunday school choir. I would recall songs that I had learned previously and sing them on my way to school, and during time alone. Looking back, I think I found singing entertaining and enjoyed learning new songs every Sunday. I participated in a music class once a week in day school where we focused mainly on musical literacy which did

not excite me. We had an upright piano at home and my mother occasionally sang and played hymns. As a toddler I would sit on the piano bench with my mother who gave me my first piano lesson. My formal piano education began when I started primary school at age six at the local hagwon (after-school class).

Hagwons are scattered around most Korean cities as the most common way of offering education that meets the educational demands and needs of Korean parents for their children in subjects ranging from academics to aesthetics. The piano hagwon I attended was like many others: There was a main room where the head teacher taught lessons; there were a few small private piano practice rooms and a common area where students learned music theory. Other teachers were available to teach piano and theory under the direction of the head teacher.

I went straight to hagwon after school four or five times a week and spent about an hour there. Upon arriving, students could choose to complete a theory worksheet assigned for the day or practice piano and prepare for the lesson, until the teacher called them into the main room. After their lessons, students were given assignments to complete before being dismissed to go home. (Students in Korea can attend several hagwons. The most popular are hagwons for core school subjects such as math, English, and Korean, but there are others for the arts, such as drawing and piano.)

Although I liked music, the hagwon was not an unmixed experience. I did not like to be confined alone for an unknown length of time in the practice rooms, and I struggled to play in front of people. I felt self-conscious and embarrassed to have a lesson in an open room where others could judge my playing. I would volunteer to go for the lesson last after most students had gone home. My teacher was nice but strict and would strike my hands with a baton whenever I made mistakes, a common teaching practice then. She later told me that I was mostly punished because I played from memory instead of looking at the music score. While I may have developed a fear of performing, I did continue to pursue music.

My family moved to Canada when I was in high school, and we settled in a small village where my parents owned a convenience store. We lived above the store, and my sister and I helped out when we could. Living in Canada was especially difficult because of the language and cultural barriers, and because we felt the hostility of Canadians toward foreigners and strangers.

In Canada, ethnic churches have played a vital role throughout the history of immigration in the process of acculturation (Hurh & Kim, 1990; Tirrito & Choi, 2004). Newcomers usually have no family members to help them and it can be almost impossible for anyone with limited English to find the help they need. Because they do not have credibility in their new country, recent immigrants need someone to co-sign to rent an apartment, send children to school, and apply for social services. Like many of our first-generation immigrant predecessors, my family first landed in Canada without any extended family or acquaintances. For us, too, the Korean church and its members became an important resource providing practical support and linking us to a family doctor, schools, and jobs. The church helped us navigate the foreign Canadian system.

After high school, I obtained a BMus degree in Organ Performance at McGill. This was an extension of my musical study of piano. It happened that across the street from the high school where I attended was a church with a pipe organ. The music director, an organist, was willing to take on a student. Since I was one of few Asians who could not speak English, the school was a difficult place and lunch time was especially hard. Having no one to talk to, I retreated to the church to practice the organ.

As I contemplated applying to university, I was most interested in music education but part of me wanted to continue to play and learn the organ because it represented the dominant western culture. I may have thought that studying the western art canon would soften the barriers of prejudice I experienced in high school and that I would feel more accepted. Given encouragement, I went on to complete the Master's degree in organ performance. From my early experience at the piano hagwon where I had watched other students play, I had developed a perception of what a 'performer' is and I concluded that I did not have the natural flare to be a performer. However, I pursued performance studies in graduate school, taking the opportunity to broaden my musical milieu. There was always a part of me that felt I was lacking in cultural competence for western music, so I took this opportunity to delve further into the culture I to which I had always felt foreign.

My interest in music education began while taking an elementary music education course in graduate school; that course led me to a practicum in a pre-K (Little Musicians') music class.

Because of my high school experience, I could not bear the thought of facing high school students again, but working with little children was different. Being brought up in a non-western culture and studying western music, I saw the benefit of early exposure to music and the importance of early music education.

In an elective course I took at McGill, I became interested in the question of identity and how it is closely linked to culture. The elective course, called Soul and Soul Music, was offered by the Faculty of Religion. The professor had once witnessed a student committing suicide upon receiving a failing final grade and this had provided the genesis for his course. This professor believed education should be about students finding who they are through experience. Through engagement in the arts (attending art exhibitions, music concerts, and films), we engaged in a series of reflective writings. The course made me aware that some of my colleagues shared my search and feelings of being 'other.' My interests expanded and I continued questioning: What is the influence of music and how is it expressed? What is the 'culture' and 'identity' of Korean Canadians? What is the link between theories of identity and its practice in the community? These questions have led me to my current doctoral program and my present research.

Coming to the Enoch Senior's College (CESC)

In 2007 my father retired from his business and devoted more of his time to volunteering at our church. He found a sense of achievement and fulfillment through this work and gladly volunteered to help the director of Enoch when she asked for his help. This move became significant for me when I responded to my dad's request to help him there.

I vividly remember my first days at Enoch because they were filled with music, especially singing. Well-remembered songs like "Home in the Springtime" made me feel at home; it turns out that Korean elders and I share the same heart songs from our childhoods. "We all become children," one participant told me. Elderly Korean participants become children again at Enoch in the sense that they can let go of themselves—can be open to change, learning, loving, and most of all, hope—despite their perceptions of old age. In my mind, Erikson's words, "(a) return to childlikeness seasoned with wisdom" (Erikson, 1964, p. 133), capture the essence of Enoch participants.

Although most of the Enoch elders shared health concerns and issues which often disturbed their nights, they were able to provide empathic support for each another. Perhaps, at Enoch, Korean seniors felt valued, and appreciated by the supportive team of volunteers who worked closely with Pastor Kim, a senior herself. The team led by Pastor Kim demonstrated an attitude towards serving elders as if the seniors were their own parents. Lunches, for example, were planned with care to deliver nutritionally balanced meals that met digestive concerns yet had flavors to make the meal enjoyable. The Korean greeting "how are you" can also be expressed by "have you eaten?" "Have you eaten?" is an expression of care and concern. Eating is a value-laden word in Korean culture and treating someone to a nice meal is one way to show appreciation.

The significance of music in the CESC program cannot be over-emphasized. Pastor Kim, who had been planning the seniors' program, clearly knew that music would play a central role. Perhaps largely because of the importance of music to Korean seniors' cultural identity, music accompanied all programs and sessions seamlessly. I particularly remember my very first observation of the Harmonica ensemble. Mr. Choi, who was also a participant, volunteered as a facilitator/teacher. The informal class was filled with impromptu musical bursts, such as solo singing of "Heidenroselein" in German by Mr. Choi himself. The ensemble had no need for a conductor. When Mr. Choi used the magic words, "Let's play like we sing," the group would instantly fix any musical issues and play perfectly in unison.

To learn more about the prevalence of music at Enoch, I asked the seniors about the meaning of music in their lives. The question, however, was redirected back to me with blunt looks. "How can anyone not like music?" they asked. Most shrugged and treated my inquiry as a rhetorical question. Perhaps less well-articulated but obviously deeply felt was how music was always remembered and shared by the Korean seniors. While musical recollections elicited smiles, other complex emotions surfaced. For example, one favorite song is "Home in the Springtime" ² which began as a poem about a remembered home which is lost. The poem was

² The lyrics of the Home in the Spring Time: 'Flowers bloomed in my home town, long time ago. Peaches, apples and apricots...and pink blossoms too. Red and violet of the rainbow, flowers paint the town. I still long to go back to my hometown in the sun'.

put to melody as a song during the time when Koreans were under Japanese colonial rule. Although not directly expressed, the song implied longing for home by Korean resisters who went to fight for the independence of the nation against Japan. The elders at Enoch understood the history of the song and had indeed learned it in their childhood. Like the resisters fighting the Japanese, the seniors could not return to the home of their childhood. That place no longer exists. Despite this loss, the singers remember their childhood filled with happiness that will not come again. "Home in the Springtime" is both a joyful and a sorrowful song, tinged with defiance and memories of the resisters fighting for independence.

Purpose of the Study

Broadly conceived, my study is an investigation of the role of music and musical community in the lives of community-living older adults. Major themes concern wellness, the role of music role in lifelong learning, community music, and immigrant experiences. Emerging themes include life review and reminiscence.

More specifically, the aim of my study is to examine the role of music and active aging in the lives of community-living Korean seniors in a particular Canadian community program. I make observations and explore the seniors' perspectives on their experience. I review recent and relevant studies regarding aging trends in Canada, and health disparities and the needs of visible minority groups in that country. I also consider the role of faith-based organizations in immigrants' lives within a brief overview of the Korean community and the history of Korean immigration to Canada.

The specific objectives of my study are to

- i. document a successful ethno-cultural program—one with a focus on music—called "Canada Enoch Senior's College" (CESC) for Korean-Canadian seniors;
- ii. identify factors that contribute to the program's success and sustainability and evaluate its impact on the participants' health-related quality of life;
- iii. explore narratives of aging among Korean-Canadian seniors through the

experience of music by examining individual experiences and perceptions of music-making and their influences on participants' overall sense of well-being.

Research Questions

My study is designed to offer a closer look at the role of music in a community program as it relates to active aging in community-living Korean seniors in the Canadian immigrant context. My main question concerns how musical engagement shapes the identity and influences the wellbeing of Korean elders. I explore the following sub-questions:

- 1) What factors contribute to participation in the program?
- 2) How does participating in the community program affect health and active aging among elderly Korean participants?
- 3) What does music mean and how important is it in the lives of elderly Koreans?

Theoretical Framework

My study is constructed upon three interlocking notions that serve as a theoretical framework for the research: 1) Active aging is the main concept, with 2) Erickson's Stages of Psychosocial Development and 3) lifelong learning as subset concepts. These topics are addressed in the literature review in Chapter 2.

Overview of the Research Framework

The first phase of my study documents the unique features of CESC through ethnographic methodology, providing an exposition of the program while examining (a) the barriers to program participation as well as (b) the support provided by CESC that enabled continued program participation on the part of seniors.

The next phase of my research was informed by findings from the first phase of my research. I explored narratives acquired through in-depth interviews with five seniors. I used a musical lens to better understand the importance of music in their lives. In this phase, I also conducted an explorative inquiry to determine health-related benefits of participation, if any, using a health survey tool, the Short Form 36 Version 2 (SF-36v2), to measure the health-related quality of life (HR-QOL).

In the third and final phase of my research, I used a mixed-methods approach (which was informed by the first phase) to evaluate the health impact of the program on participants. The quantitative findings were supported by in-depth qualitative interviews of the participants.

Data Collection and Analysis

As a participant observer, I conducted most of the ethnographic fieldwork during two consecutive terms between April and October 2014³ while I attended classes and performed the roles of greeter, photographer, and sectional accompanist. A researcher may gain a deeper understanding and knowledge of the single (case) culture-sharing group by participating as a volunteer in the program (Stake, 1995; Yin, 2009). I gathered data from a combination of field notes, informal conversations with the participants and class instructors, and self-reflections (Denzin & Lincoln, 2005).

In order to observe and document the participants' manners of involvement with music (engaging, singing, and performing) in the program, I also used video recording to supplement data collection as I could not observe, participate, make notes, and fully engage in the environment all at the same time (Bogdan & Biklen, 2003; Goldman-Segall, 1998). Video-based data provided visual contextual information, including the ambiance or natural setting and verbal and non-verbal behaviors of participants that aided in cross-checking the data.

I collected additional data from public records (newspaper articles) and photographs as well as official documents such as course resources and materials, schedules, and internal publications, to acquire a more contextualized and holistic examination of the program. I collected data from multiple sources to enable triangulation and strengthen validity of findings. I transcribed and translated official documents and informal conversations as necessary. I am a native Korean-speaker. As well, I collaborated with a bilingual translator (Korean-English) to ensure I correctly conveyed nuances, innuendoes, cultural/ethical assumptions, and cultural expressions as needed.

³ The CESC Spring term ran for 11 weeks from April 10 to June 19, 2014, and the Fall term ran for 8 weeks from September 4 to October 23, 2014.

The process of data analysis was done by reading and re-reading transcribed texts and watching video clips to determine emerging themes (Green & Thorogood, 2009). Analysis of this ethnographic case study was conducted in five parts:

- 1) organization of data: reducing data into manageable and meaningful chunks or files;
- 2) coding: developing themes that involve making sense of the textual data by assigning codes to chunks of data;
- 3) description of context: taking a narrative approach to provide a holistic picture of the context (for example, describing the activities in the CESC programs);
- 4) interpretation: describing the findings in a meaningful way, adding significance to the analysis, and illustrating patterns and relationships;
- 5) representation: employing narrative to represent the findings and taking themes from data analysis with direct quotes to support and link the themes.

These steps allowed to make a detailed and holistic description and analysis of the case and the context (Creswell & Creswell, 2018).

Design and Method

Here I outline general principles of mixed-methods qualitative and quantitative research. I present the types of research used in the order in which they were employed in the articles that comprise this dissertation.

Ethnography

Ethnography arose from the field of anthropology and seeks to understand cultural phenomena as a whole. The word itself comes from *ethnos*, meaning "folk or people," and *graphos*, meaning "to write." In an ethnographic study, the researcher, as a participant observer, may immerse herself in the culture - living, interacting, and engaging - to come to a holistically understanding of a way of life in a natural setting (Bresler, 1995; Creswell & Creswell, 2018; Fetterman, 1998). Through detailed observations and explorations of a cultural group, the

researcher creates an understanding of a complex woven fabric of cultural and social factors and influences (Denzin & Lincoln, 2005; Creswell & Creswell, 2018).

Becoming a part of the daily life of a group fosters "thick descriptions" (Geertz, 1973) or multilayered descriptions of the actions or events that help seek cultural and social meanings, beyond the mere observation or semiotic reading of the culture. As a result, a deeper understanding of local setting and cultural knowledge of a community may be acquired. The ethnographic case study design is employed in this research to provide a contextualized description and understanding of the CESC program and a unique cohort of first-generation Korean immigrants who are engaged in the program. According to Clandinin and Connelly (2000), "Context is necessary for making sense of any person, event, or thing" (p.32). A qualitative report filled with rich descriptions of people, places, and events, draws readers closer to a more complete account of participants' views of the reality and the contexts in which motivations and meaning construction are created.

Narrative Inquiry

Narrative inquiry (Freeman, 1997; Josselson, 2006) relies on "telling a life story, retelling a life story, and reliving a life story," a story from which our identities are constructed and from which more knowledge can be gained (Clandinin & Connelly, 2000, p.418). This method allows a researcher to investigate how the participants construct self-understanding and how "their way of being" itself is "storied" (Connelly & Clandinin, 1990, p.12).

Participants decide what is meaningful to them; they decide how the story is told, in what way and with what emphasis (Bell, 2005). Following the notion of social constructivism, narrative inquiry allows access to embedded social and cultural differences as well as forces and ambiguities that help contour the concept of self, and help one make sense of the world. This approach is particularly relevant in uncovering the cultural narratives of this cohort of Korean older adults who are pioneer first-generation Korean immigrants in Canada and have endured a great deal of conflict and instability in their lives, such as that caused by the Korean War and its aftermath.

I used the narrative inquiry approach to capture detailed and deep personal accounts of participants' experiences related to music. The approach permitted me to preserve the integrity of their narratives (Bell, 2005; Clandinin & Connelly, 2000) while revealing the culturally and historically embedded experiences they found meaningful.

Principles of a Mixed Methods Approach

Overview of the Complementary Study Methodologies

Mixed methods research is an emerging research approach in the social and health sciences that combines both statistical data and narratives to study human and social problems. Mixed method research involves the merging, integrating, linking, or embedding of qualitative and quantitative techniques within the study (Creswell & Hirose, 2019). The potential contribution of two distinctive approaches can strengthen the study because one method can offset weaknesses of the other method (Creswell & Creswell, 2018).

An example of the complementary advantage to using mixed methods research techniques can be illustrated with an analogy to a recent natural phenomenon. Media have been reporting a series of earthquake in Turkey. While news reports emphasize factual data, including number of people who died, number of injuries, and possible losses, a more complete understanding of the phenomenon is gained from survivors of the earthquake through interviews and personal accounts.

Mixed Methods Research

Mixed methods research is one kind of multimethodology that includes, combines, or integrates elements of qualitative and quantitative methods. Researchers may employ mixed methods to gain a deeper understanding of a given phenomenon (Creswell & Hirose, 2019). Creswell and Creswell (2018) maintains that mixed methods research is not just having both quantitative and qualitative data. While mixing is central to the mixed methods approach, the studies must integrate, link, and connect both kinds of research.

Conceptualization of integration occurs through linking the methods of data collection and analysis (Creswell & Plano Clark, 2018). Creswell and Plano Clark (2018) suggest that

linking can occur in several ways: connecting, building, merging and embedding. Connecting takes place when one type of data is linked with another type of data through the sampling process. For example, in this study, in-depth qualitative interviews of the participants were selected from the respondents to the SF-36v2 Health Survey. Integration through building occurs when results from one data collection method inform and shape the data collection approach of the other method, with the latter method expanding upon the former. For example, ethnographic data on the cultural community program suggested potential health impacts on elderly Korean participants and the importance of language barriers and cultural understanding in interpreting the wording of the questions. This allowed for cultural adaptation in the interpretation of the SF-36v2 survey. Integration through merging of data involves the combination of two databases for analysis and comparison. In this study, merging takes place after the statistical analysis of the SF-36v2 and the qualitative analysis of textual data. Integration through *embedding* may occur when data collection and analysis involve any combination of connecting, building, or merging. A distinguishing feature of this approach is the linkage of qualitative data collection with quantitative data collection at multiple points. This approach is especially important in interventional advanced designs.

Three Research Designs

Three designs—three ways to integrate data—characterize mixed-method research (Creswell & Plano Clark, 2018). One a is a 'convergent mixed-methods design.' Collection and analysis of qualitative and quantitative data occur roughly at the same time so that one database can be checked against the other. A mixed methods approach captures two different perspectives such as, for example, one perspective from a survey and one from interviews; then the data and analysis are brought together.

A second mixed method research design is an 'explanatory sequential mixed-methods design.' Research occurs in two phases. First, a quantitative measure such as a survey is administered and analyzed. With insights gained from the survey, a researcher then conducts a qualitative phase, perhaps an ethnography or interviews. Each phase of the study propels the next phase and allows for interpretation. For example, a researcher may use a survey to collect data

from a large group of people, and then later may select members of that group to interview to gain insight into the survey results. The qualitative data help interpret the quantitative data.

A third design, the reverse of the second design, is called 'exploratory mixed-methods design.' It is also a two-phase design but starts with qualitative data collection such as interviews that explore the context and individual views, and then, in light of the findings, is followed with quantitative techniques such as a questionnaires or experiments. This method is most often used when developing and testing a new instrument.

Quantitative and Qualitative Research Methods

Quantitative and qualitative research methods are different with distinct features. Perhaps a basic way to distinguish the two is that data collection in qualitative research involves words, and quantitative research involves numbers (Greene, 2007). Quantitative research methods focus on promoting an objective investigation of observable data using a variety of tools. Numerical and statistical analyses give statistical results which may be generalized to the larger population. Quantitative research tools can be efficient and help answer some kinds of questions. However, this type of research has limitations in that quantitative tools do not reveal the reasoning behind participants' responses, and they often do not reach underrepresented populations.

Qualitative research methods depend on the use of open- or closed-ended responses which are possible in interviews, observations, or both. Data are open-ended and allow for the voices of the research participants to be heard; they are subjective, and the results are presented as a narrative report. Qualitative research arises from and is most often used in social science disciplines such as sociology, psychology, and anthropology (Creamer, 2018). The research questions posed in these areas center on human interactions, perceptions, cultural norms, and beliefs. Qualitative research is most useful when trying to understand people's behavior and the meaning they give to their experiences. Qualitative research typically involves interviews, focus groups, or observations to provide data that are rich in detail and context. Information sources may include ephemera, notes, verbal reports, books, works of art, maps, interview transcripts, and photos or videos. The amount of information gleaned from these data sources can be exhaustive.

Qualitative researchers rely on several analytic strategies: coding, pattern thematic analysis, and content analysis. Coding analysis happens when the researcher associates main themes revealed through phrases and words and interprets their meaning. In pattern thematic analysis, information is sorted into patterns as the primary basis for organization. Through content analysis, the researcher makes valid inferences linking the data to the context (Krippendorf, 1980). Computer tools may aid such analysis.

Qualitative research has limitations. One is that the vast amount of material generated during an investigation necessitates careful reflection and analysis. Interpretation is necessarily subjective as the researcher's experiences are part of the inquiry. The relationship between researcher and participant is pivotal and may be volatile. Although the information is rich and descriptive, the sample in qualitative research is usually small. Given that interpretation then extends to a select group, results are unique.

I have used a mixed-methods approach for my study, employing both quantitative and qualitative strategies, to collect and analyze data. Creswell & Creswell (2018) explain that a mixed-methods design is useful to capture the best of both quantitative and qualitative approaches as it may allow a researcher to generalize the findings to a population and develop a detailed, personal account of a phenomenon or experience and what it means to the members of the cohort. Among the three designs mentioned in the previous section, I have chosen the explanatory sequential mixed-methods design. Their two-phase design works well with my research as a large collection of quantitative data are taken and analyzed followed by a qualitative phase interview to gain further insights into survey result and aid in interpretation. A mixed methods approach works well for my study's objective—to understand the role of music in the lives of Korean seniors in an active aging program. Moreover, a careful design may help neutralize or cancel the biases inherent in any single method (Creswell & Creswell, 2018). For example, a health survey can quantitatively measure the quality of life (QOL) whereas semistructured interviews allow the researcher to explore and understand participants' perceptions. As QOL is influenced by socially constructed contexts as well as personal characteristics, this construct may be difficult to explore solely through quantitative research methods. Therefore, the study adopted in-depth interviews with Korean elderly participants and ethnography to gain an

understanding of participants' perceptions and experiences within their context. These detailed voices complemented the quantitative findings.

Triangulation or increasing credibility and validity of research findings is made possible through employing several complementary approaches (Creswell & Creswell, 2018). To this end, my study draws on ethnography and narrative inquiry as well as the SF-36 health survey with semi-structured interviews.

Data Collection and Analysis

I interviewed five Korean seniors during a four-month period starting at the beginning of the fall term in 2014. The five interviewees had known me for some time through my involvement in CESC as a volunteer. Establishment of rapport with interviewees is an essential component of narrative inquiry (Bell, 2005; Clandinin & Connelly, 2000; Fetterman 1998). This allows a dialogic relationship between the researcher and the participants as they engage actively, interacting morally, emotionally, aesthetically, and intellectually in the exploration of meaning (Bochner, 2001; Clandinin & Connelly, 2000).

As a dialogic and collaborative practice, a narrative approach is sensitive to culture through listening, telling, and sharing (Bochner, 2001; Bochner & Ellis, 2003); narrative inquiry not only validates the analysis but also elicits insightful accounts as it allows a time of reflection, learning, and discovering who the narrator is and how he or she sees the world.

My interviews began by my inviting seniors to talk about their earliest experiences related to music. I initiated interviews as conversations which the interviewee guided. These interviews, conducted face-to-face at a location chosen by the interviewees, lasted over two hours, and continued through emails, phone conversations, or in person during school hours at CESC, as suited each person. Follow-up questions facilitated the narrative.

I recorded all interviews, transcribed them into Korean, and translated the Korean into English. I checked with each individual to ensure that the interview was correct and complete. This was a further opportunity to clarify and talk further.

A narrative analysis may start at the very moment in which the researcher organizes the descriptions of events into a coherent story. The narrative approach accepts that life experiences are not simple; they are complex, multidimensional, and at times, chaotic. Stories, therefore, may swing between the past, the present, and the future. As well the stories can interweave between past, present, and future time periods to reflect evolving identities. A non-linear order of narrative can pose challenges in the data analysis process (Clandinin & Connelly, 1999; Josselson, 2006; Polkinghorn, 1995). Thus, narrative analysis is traditionally begun by organizing descriptions of events into a coherent story. A temporal framework organizes pivotal moments to establish initial coherence (Polkinghorne, 1995). The analysis continues to a paradigmatic analysis, which includes a more explicit interpretation of the narrative. During the process, themes emerge to be checked and crosschecked against all the interviews (L. Cohen et al., 2007; Creswell & Creswell, 2018).

Health Survey

In order to gain a more complete understanding of participants' health and to measure the effectiveness of the program related to health status, I performed a quantitative evaluation of health-related quality of life using the SF-36v2 Health Survey to assess participants' physical and mental health outcomes after participating in the CESC program. Health-related quality of life assessment was an addendum to this study to provide outcome measures for physical and emotional well-being of participants in the CESC program.

The Medical Outcome Study-Short Form (SF-36), originally developed for the Rand Corporation's Health Insurance Experiment (Ware & Sherbourne, 1992; Ware & Gandek, 1998), measures health across eight domains. The SF-36 survey is the world's most widely evaluated generic health outcome measurement (Garratt et al., 2002); it has been translated for use in more than forty countries, including Korea (Han et al., 2004). I chose the SF-36v2 because it is a validated evaluation tool of health using multiple criteria such as body function, mood, symptoms, and social interaction that are relevant to participating seniors and has been effectively used as a community-based health survey for older adults (Walters et al., 2001). Even a physical symptom such as chronic low back pain is known to have psycho-social factors such

as emotional distress and has been found to be a strong predictor of outcomes (Pengel et al., 2003).

The survey comprises thirty-six items across eight domains of status and perceptions of health. These can be summarized in two large measurements, one concerning physical health (Physical Functioning, Role-Physical, Bodily Pain, and General Health), the other concerning mental health (Social Functioning; Role-Emotional; Vitality and Mental Health). The physical functioning domain uses ten items to examine limitations in physical activities such as walking and other vigorous to mild activities. The role-physical domain uses four items to measure problems with activities as a result of physical health. The bodily pain domain contains two items to assesses limitations caused by pain, and the general health domain uses five items to assesses perceptions of self-rated health such as expectations.

The social functioning domain uses two items to examine physical or emotional effects on daily social activities. The role-emotional domain uses three items to examine limitations due to emotional problems that impinge upon daily activities. The vitality domain uses four items to assess energy and tiredness, and the mental health domain uses five items to assess feelings of happiness and nervousness that are related to psychological symptoms. The survey was translated into Korean, and pilot tested with three participants to ensure clarity and to determine if changes were necessary (Williams et al, 2003).

Data Collection and Analysis

During two consecutive terms in 2014, a total of 79 participants completed survey questionnaires.⁴ The survey was distributed twice per term. The pre-test was distributed in the second week of the term because the first week of the program is spent on greetings and special events to welcome previous and new members. Participation was voluntary, and while some participants opted to fill out the surveys at the site, most took them home and returned them to me the following week. The program is structured and organized in such a way that little time is

⁴A subset of participants (n=25) completed questionnaires in Spring term (April 10 to June 19) and fifty-four completed the questionnaires twice in Fall term (September 4 to October 23) of 2014.

provided for any other activities than the scheduled classes and meal; it is a highly active program, so many participants chose to complete the survey when they had extra time at home.

The second (or post-) test was distributed during the second to last week of the program and most completed tests were collected in the following week. Data were analyzed by conducting paired t-tests to determine the statistical significance of the difference between the baseline and the result after seven to ten weeks of participation at CESC for each health measure. A total sample of 79 pairs was analyzed. I conducted semi-structured interviews with eleven participants in the fall term. I recorded all interviews then transcribed them into Korean after which I translated them into English. This phase of the survey used ethnographic data for triangulation to increase the validity of the study results.

Dissertation Overview

This dissertation is written in the integrated-article format. Here, I discuss and justify my choice of this format.

Integrated-article Format

The integrated-article thesis (or multiple journal article thesis) has gained popularity over the traditional monograph dissertation as a requirement for the Ph.D. degree. This format consists of a collection of scholarly papers. A selection of separate articles (e.g., submitted articles, unpublished but publishable articles, articles already published in a journal or conference publication series, or a combination of such articles) make up individual chapters of the dissertation.

An integrated-article dissertation has several pragmatic advantages over the traditional dissertation. Because each chapter may stand on its own as a separate piece of research, they can be ready to submit for publication quickly, thereby facilitating the transfer of knowledge to a wider audience of practitioners and colleagues in a timely manner. Although dissertations are accessible to the general public through university libraries, the length, style, and format of the dissertation may challenge the lay reader, and readership is likely be limited largely to members of the academy. Moreover, the integrated article format provides doctoral candidates experience in writing for journals, thereby better preparing them for an academic research career after their

degrees have been granted and facilitating a smooth transition from student to published researcher.

The integrity of the traditional dissertation is maintained in the integrated article format through the establishment of an overarching logical connection that provides overall coherence to the work. An introductory chapter and a concluding chapter describe the research and synthesize the chapters as a whole. The individual papers thus are given the unitary and cohesive focus of a single program of research much like the monograph thesis format.

Study Overview

Three integrated articles, all of which have been published in peer-reviewed publications, form the heart of my research. Chapter 3, the first of the integrated articles, presents an ethnographic study of a seniors' college in Mississauga, Ontario, Canada, where first-generation Korean immigrants come together to connect, engage, and learn on a weekly basis over the two-semester school year. The college serves as a social hub for many Korean seniors living in the Greater Toronto Area. Ethnographical findings reported in this article serve as a background for the next two articles. Chapter 4, the second of the integrated articles, examines the effects of the community-based senior's college program on the quality of life of the older Korean immigrants. It employs semi-structured interviews, self-reported health measure questionnaires, and field notes as data collection tools. The third of the integrated articles presented in chapter 5 explores the salience of music in the lives of five seniors who are members of the college. This case study may be seen as a more in-depth approach to uncovering each person's understandings of the complex systems of their own histories.

Following the present introductory chapter, my dissertation is organized in six chapters. Chapter 2 provides relevant background literature in wellness, the immigrant experience, music's role in lifelong learning, and community music programs. Chapters 3, 4, and 5 present published peer-reviewed articles that comprise the integrated-articles of this dissertation. Chapter 3 presents the findings from the ethnography of CESC which provides an overview of the program in which salience of music is evident. Chapter 4 presents the findings from the quantitative phase of this research in which the CESC

program was evaluated using a health survey supplemented by interviews. Chapter 5 presents the narratives of five Korean-Canadian seniors of their life stories through musical experiences. Chapter 6 concludes this dissertation by presenting findings, summarizing main outcomes and contributions of this research, discussing implications, and offering recommendations for practice and future research. Supplementary material is contained in two Appendices: 1) Terminology, 2) the SF-36v2 Health Survey.

Chapter Two: Review of Literature

This literature review comprises an analysis of themes that are relevant for the three articles at the heart of this investigation. Four main topics propel this study: 1) wellness and active ageing, 2) the immigrant experience, 3) music's role in lifelong learning, and 4) community music programs. The literature review is organized as follows:

- 1) Models of wellness are examined through the concept of active ageing, an overview of the World Health Organization's (2002) active ageing framework with its underpinning theory, and the demographics of ageing in Canada.
- 2) Immigrant experiences in Korean communities in Canada are examined. Health disparities of immigrant communities are identified as is the need for research in this area. The role of ethnic faith-based organizations in active ageing, the historical background of seniors in Korean society, and Korean identity are all examined.
- 3) Lifelong learning and music's role in lifelong learning is traced through current international research.
- 4) Literature concerning recent and relevant developments in the emerging field of Community music is examined.

Wellness as Understood Through the Concept of Active Ageing

The conceptual model for the present study is partially provided by the World Health Organization's active ageing framework and the life-course perspective inherent to that framework.

WHO's Active Ageing Framework

The concept of active ageing has been established as the leading policy response related to population ageing in European countries, gaining its widespread currency from international governmental organizations including the World Health Organization (WHO), the United Nations (UN), and the Organization for Economic Co-operation and Development (OECD) (Foster & Walker, 2013; 2015; Walker, 2015). The adoption of the term active ageing by the WHO in the late 1990s played a major role in its rapid diffusion (Walker, 2002). The WHO

defines active ageing as "the process of optimizing opportunities for health, participation and security in order to enhance quality of life as people age" (WHO, 2002, p. 12).

"Active" refers to lifelong participation in social, economic, cultural, spiritual, and civic areas, encompassing a person's employability and ability to remain physically active. This concept has been widely applied not only in policy but in research and practice and has helped to distance the discourse of ageing from the traditional portrayal of old age as a period of inactivity and dependency (Boudiny, 2013). Active ageing derives from the activity perspective promoted by the gerontology of the 1950s and 1960s, which described a connection between various forms of physical activity and well-being (Walker, 2015). This perspective came about as the antithesis to disengagement theory which portrayed old age as an inevitable period of withdrawal from roles and relationships (Cumming & Henry, 1961).

Similar concepts such as healthy ageing, successful ageing, and productive ageing have also surfaced in the past few decades as have divergent views about their meanings. Active ageing is sometimes used synonymously with these terms, but even active aging has been defined differently by various authors (Boudiny, 2013; Hung et al., 2010; Mendes, 2013), and none of the terms are without criticism. Successful ageing is the most common term found in the literature (Hung et al., 2010; Peel et al., 2004). The term "successful ageing" connotes a focus on productivity or success related to western cultural values; it has been criticized as having too narrow a focus (Boudiny, 2013). Some authors (e.g., Rowe and Kahn, 1997) have broadened the notion of activity by labeling an activity as productive "if it creates societal value, whether or not it is reimbursed" (p. 434), a notion which can include care and voluntary work in addition to paid labour. Rowe and Kahn's (1997) definition nevertheless fails to include leisure activities as it focuses on economically and socially productive activities. Further, productive ageing also has variations in definition but is restricted in its application because most definitions are related to the promotion of economic contributions to society through labour market participation (Davey, 2002).

Active ageing as proposed by the WHO "goes against the trend towards a detrimental view aimed only at employability" and takes a more inclusive approach that considers factors from both personal and social spheres (Mendes, 2013, p.180). Active ageing is broader in scope

than its precursor terms, productive ageing and successful ageing, in that it refers to activity that does not focus exclusively on productiveness. Emphasizing productivity can be problematic because older women who are involved in domestic tasks, including care for others, would be considered active, but whether such activity contributes to their well-being is questionable—the context and the nature of the work, and whether the activities are health-sustaining activities, would have to be considered, particularly in immigrant contexts in Canada. Seniors who are sponsored by family members to come to Canada may be asked to take care of their grandchildren and can experience social isolation even while living with the family; further, these seniors may be more susceptible to elder abuse as they do not have financial independence (Matsuoka et al., 2013). In fact, studies show that seniors who live with their spouses and adult children have lower overall psychological well-being than those who live alone (Han et al., 2007; Wong et al., 2007).

Active ageing is a more inclusive term than healthy ageing as the definition goes beyond the issue of health care to involve various factors that affect how individuals and populations age (Kalache & Kickbusch, 1997). The comprehensive nature of the WHO's active ageing framework has its roots in the health promotion models designed by the Ottawa Charter for Health Promotion (WHO, 1986) and the Jakarta Declaration on Leading Health Promotion into the 21st Century (WHO, 1997). In this respect, the active ageing framework has similarities to the "healthy ageing" term adopted by Health Canada. Unlike other definitions, Health Canada's understanding of "healthy ageing" derives from a population health approach and is inclusive of health promotion. This approach emphasizes the importance of broad determinants of health, including key personal health practices and environments that support healthy behaviours (Health Canada, 2002).

Health Canada defines healthy aging as "a lifelong process of opportunities for improving and preserving health and physical, social, and mental health" (Health Canada, 2002). Health Canada's adoption of the "healthy aging" concept focuses on health care as part of a lifestyle that can help people maintain optimal health (Healthy Aging and Wellness Working Group, 2006). This approach may be seen as similar to the active ageing framework adopted by WHO (2002). Both approaches endorse similar actions for healthy aging: maintaining social connections, increasing the level of physical activity, maintaining healthy eating habits, reducing the risk of

falls, and quitting smoking (Healthy Aging and Wellness Working Group, 2006; WHO, 2002). Supportive environments and age-friendly communities promote positive outcomes (Smith et al., 2013). Such supportive environments may include a variety of features—such as opportunities for social connection through increasing interactions and civic engagement. As well, accommodating urban design might provide safe places to walk and be active, accessible transportation, appropriate housing, and neighborhood access to services as well as restaurants and grocery stores welcoming to seniors. Such adaptations reduce inequities as they enhance independence and quality of life.

The WHO suggests that, for active ageing to be realized for people, actions are required in three spheres: participation, health, and security. Participation refers to active involvement of people not only in their civic duties but also in their involvement in social and other aspects of community life. Having social, economic, and physical security enables people to participate and live freely. As people age, autonomy and independence become increasingly important to maintaining the quality of life (WHO, 2002). Health affects one's ability to lead an independent life which, in turn, affects how one experiences one's surroundings and interactions with the environment, including participation in many areas of life. The WHO highlighted eight main determinants of active ageing in which actions are necessary to bring about the changes that promote active ageing: culture, gender, health, social service, behavior, physical environment, social environment, and economic determinants of those related to the individual (WHO, 2002).

Within the three spheres of active ageing, my study focuses on participation. First-generation immigrants often face multiple challenges as they adjust to the host country. Limited cultural and language understanding pose barriers to information, access to services, and community participation (Ahmed et al., 2016; Chai, 2023; Rohr et al., 2019; Zeidan et al., 2023). Being able to maintain good health enables people to participate in leisure activities. For immigrants, their official language proficiency and knowledge of the social and health systems of the host country affect their participation in the host society in many ways. Research suggests that level of participation varies by immigration generation status, and is lower for the first generation, especially for leisure activities such as involvement in sports clubs or hobbies like dance and the arts, involvement which in turn can affect quality of life (Statistics Canada, 2003). Moreover, without having met social, financial, and physical security needs through supportive

policies and programs, a person cannot have the opportunity to achieve good health in the first place or engage in meaningful activities or contribute to society.

Multiple agencies provide services within the provinces and their communities. The Ontario Ministry of Citizenship and Immigration is an umbrella organization that offers resources for settlement, and information about language programs (Ontario Council of Agencies Serving Immigrants, 2020). The ministry also provides aid for housing, health, education, recreation, and community regional health services. Each region has its own community-based multicultural agency to provide settlement related services. For example, in the Peel region, which is the site for this study, COSTI (not an acronym) provides services such as housing, employment, education, health and social engagements for all immigrant families and individuals (COSTI Immigration Services, 2022).

Some scholars have suggested that active ageing can be an unattainable ideal if people are not supported by the resources needed to implement their choices (Asquith, 2009; Mendes, 2013). There is also a concern that WHO and local health frameworks lack clarity regarding their interpretation (Clarke & Warren, 2007; Ranzijn, 2010). While the strength of the framework is in its comprehensive nature, this also can be a source of weakness. The challenge lies in how the framework can be operationalized so as to reach all citizens, including vulnerable groups of older adults such as those who are frail or have disabilities (Boudiny, 2013; Walker 2002). The implementation of a comprehensive strategy requires a collaboration of many distinct but closely linked disciplines and sectors, such as "employment, health, social protection, pensions, social inclusion, technology, economic policy and research" (Walker, 2015, p. 6). The inclusion of cross-cutting themes of culture, social protection, and gender are particularly pertinent to older populations such as single seniors who are disproportionately women. 27% of single women had a lower income than 5.2% of coupled seniors as reported in a 2013 study (Jackson, 2016). Foster (2010) argues that "the gendered nature of poverty in older age reflects women's constrained opportunities across the life course including the unequal provision of care, its impact on employment, and, subsequently, women's greater likelihood of reaching retirement with inadequate pensions" (Foster & Walker, 2013, p. 3). This is true in the Canadian context; for senior women from visible minority groups, the gap is likely to be even greater. Thus, an ageing

framework needs to consider the specific challenges that women face in the ageing process as well as their economic disadvantages.

Active ageing and healthy ageing are terms I use interchangeably throughout this thesis. Regardless of which term is used, both represent a common goal of wellbeing, and the expansion of opportunities for older populations to achieve physical, mental, and social health.

Life Course Perspective to Active Ageing

A life course perspective is inherent to the concept of active ageing. Recognizing the importance of the effects of earlier life experiences on people's health as they age, the WHO's active ageing framework is rooted in a preventative life course orientation (Walker, 2015). Evidence is accumulating that chronic health conditions such as diabetes and heart disease may begin in early childhood. The risk of these diseases is subsequently shaped and modified by socio-economic status, experiences, and other factors across the whole life span. This is a useful view to consider, not only for biological reasons, but because older adults are not a homogeneous group, and individual diversity tends to increase as one ages. For ethnic minority seniors their unique cohort's historical experiences as well as their individual life stories can inform us who they are as individuals and as a group, and how their health fares in comparison to the native-born population.

The Demographics of Ageing in Canada

According to the United Nations, a society with a proportion of seniors at 14% or higher is regarded as an "aged society" whereas a nation with a proportion of seniors at 7% or higher is categorized as an "ageing society" (United Nations Department of Economic and Social Affairs, Population division, 2015). Statistics Canada (2022a) reports that the number of seniors aged 65 and older rose to 18.3% or 7.0 million in 2021, accounting for nearly one-fifth of the population. Canada would therefore be regarded as an aged society. Children aged 14 and under actually decreased by 0.3% during the same period (Statistics Canada, 2022a). In 2022, the only age group with a growth rate higher than 20% was above age 60 and in the five-year groups, the 60-to 64-year-old group saw the greatest increase at 29.1% (Statistics Canada, 2012). All projections indicate that Canada will enter an era as a "super-aged society" in the next few decades, where more than 20% of the population is over the age of 65.

The far-reaching social and economic consequences of an increasing senior population raise issues about demand for health care resources and social supports. While Canadians are now living longer, with an average life expectancy of 81.7 years, the research shows that their functional health⁵ decreases as they age; seniors tend to decline more rapidly after age 65 (Decady & Greenberg, 2014). The latest estimate, using the Healthy Utility index which uses health-adjusted life expectancy,⁶ shows that an average Canadian may live roughly 10.5 years with some level of disability which limits activities (Decady & Greenberg, 2014). The greater pressure on social and health services that is expected as a result of more Canadians living longer presents a challenge for society to find a way of prolonging individuals' healthy years while enhancing the quality of their lives.

Even with improvements in quality of life and longer life expectancies, today's senior citizens are more likely to suffer deteriorating physical and mental health conditions than in the past. In fact, four out of five Canadian seniors have reported chronic health conditions such as arthritis, hypertension, back pain, heart disease, and cataracts (Health Canada, 2002; Kembhavi, 2012). Chronic pain is indeed a major health issue for seniors. It is estimated that 27% of seniors living in private households experience pain on a regular basis, and the proportion of seniors affected by chronic pain is even higher, at 38% (Fitzpatrick et al., 2005), for institutionalized seniors. An increase in pain over a two-year period is associated with a higher likelihood of being unhappy or having negative self-perceived health (Fitzpatrick et al., 2005). Any efforts to reduce pain, independent of the presence of chronic conditions, would then have the potential to improve the well-being of seniors.

Canada has experienced a significant increase in its visible minority population in the past decade. The 2016 National Household Survey shows that approximately 22.3% of the Canadian population are members of a visible minority group, approximately a three-point percentage increase from 19% in 2011 (Statistics Canada, 2017a). Ethnic and cultural diversity

⁵ Functional health is measured using a scoring system based on self-reported performance on eight key health attributes: vision, hearing, speech, mobility, dexterity, feelings, cognition, and pain (Decady & Greenberg, 2014).

⁶ Health-adjusted life expectancy refers to the equivalent number of years a person can be expected to live in good or "full" health (Decady & Greenberg, 2014).

have important implications in policies and programs affecting Canadian seniors. Recent estimates show that more than one-quarter of seniors born between 1947 and 1951 and between 1957 and 1961 are immigrants. This proportion of immigrants is estimated to increase in subsequent cohorts to at least 36% of the 1967-1971 cohort and to reach nearly 47% of the 1997-2001 cohort (Carrière et al., 2016). Immigration has played a significant role in determining the size and diversity of various cohorts of seniors. As immigration from primarily European countries shifted to immigration from primarily Asian nations (including the Middle East) in the past few decades, the socio-demographic characteristics of the Canadian senior population has changed to show a significant increase in the share of seniors who belong to a visible minority group (Carrière et al., 2016; Statistics Canada, 2013b).

By 2032 when the remainder of the 1967-1971 immigrant cohort should begin turning 65, more than one in four seniors in Canada will belong to a visible minority group, compared to 13% of the 1947-1951 cohort (Carrière et al., 2016). This increase will widen the cultural spectrum of Canadian seniors. Furthermore, it is estimated that when the 1997-2001 cohort reaches 65 years of age, which will be between 2062 and 2066, nearly half (44%) of Canadian seniors will belong to a visible minority group (Carrière et al., 2016). The shifting profile of the Canadian senior population, with its expected increase in ethnic and cultural diversity, calls for special attention directed toward understanding and serving seniors who belong to visible minority groups that are markedly different in cultural and linguistic background from the mainstream culture.

Life Review and Reminiscence

Life review refers to the recollection and attribution of meaning to past memories throughout one's lifespan. Older adults often review their life experiences. Studies show that earlier life experiences affect one's later life and that change occurs continuously throughout the life span, even in the final stages of life (Elder & Giele, 2009; Vaillant, 2012).

Erikson's psychosocial development theory (1958, 1964) proposes that personality develops in eight stages throughout a person's life. Influenced by Freud at first, Erik Erikson with the aid of his wife Joan Erikson, moved beyond Freud's psychosexual theories to develop a broader perspective of self and other. This broader theory maintains that social context and

interactions are valuable throughout the lifespan. Each stage of life involves a crisis or developmental task that can be identified as a turning point in life. These developmental stages revolve around what are termed syntonic and dystonic dispositions such as trust versus mistrust (Erikson, 1964, p.139). A positive resolution of crises from each stage achieves psychological virtues or inherent strengths necessary for growth and thus are a foundation for the following stages. Developing a healthy personality and sense of self associated with each stage contributes to overall psychological wellbeing. As Erikson's theory evolved, it became clear that the stages were fluid. See Table 1 for an overview.

Table 1: Erikson's Psychosocial Stages Summary Table

Stage	Basic Conflict	Important Events	Key Questions to be answered	Outcome
Infancy (0 to 18 months)	Trust vs. Mistrust	Feeding/ Comfort	Is my world safe?	Children cultivate a sense of trust when caregivers consistently offer reliability, care, and affection. Conversely, the absence of such nurturing attributes can result in the development of mistrust.
Early Childhood (2 to 3)	Autonomy vs. Shame and Doubt	Toilet Training/ Dressing	Can I do things by myself or need I always rely on others?	Children require the development of a sense of personal control over their physical skills and the cultivation of independence. Achieving success in these areas fosters a feeling of autonomy, while encountering failure can lead to feelings of shame and doubt.
Preschool (3 to 5)	Initiative vs. Guilt	Exploration/ Play	Am I good or bad?	Children require the initiation of asserting control and power over their environment. Success in this pursuit engenders a sense of purpose. However, children who attempt to exert excessive power may face disapproval, which can lead to feelings of guilt.
School Age (6 to 11)	Industry vs. Inferiority	School/ Activities	How can I be good?	Children must learn to effectively manage new social and academic demands. Achieving success in these endeavors fosters a sense of competence, while encountering failure can lead to feelings of inferiority.
Adolescence (12 to 18)	Identity vs. Role Confusion	Relationships / Identity	Who am I and where am I going?	Adolescents require the development of a sense of self and personal identity. Success in this process results in an enhanced ability to maintain authenticity, while failure can lead to role confusion and a weakened sense of self.

Young Adult (19 to 40)	Infancy vs. Isolation	Initiate Relationships	Am I loved and wanted?	Young adults require the establishment of intimate, loving relationships with others. Achieving success in this endeavor fosters the development of strong relationships, while encountering failure can lead to feelings of loneliness and isolation.
Middle Adulthood (40 to 65)	Generativity vs. Stagnation	Work and Parenthood	Will I provide something of real value?	Adults require the creation or nurturing of things that will have a lasting impact, often achieved through having children or making positive changes that benefit others. Achieving success in these pursuits engenders feelings of usefulness and accomplishment, while encountering failure can lead to a sense of shallow involvement in the world.
Maturity (65 to death)	Ego Identity vs. Despair	Reflection on life	Have I lived a full life?	Older adults require the reflection on life and the experience of a sense of fulfillment. Achieving success in this stage fosters a feeling of wisdom, while encountering failure can lead to feelings of regret, bitterness, and despair.

Note. Erikson's stages summary chart was modified from Erikson's stages by SYCH-MENTAL HEALTH HUB. n.d. (https://pmhealthnp.com/eriksons-stages/).

The first stage of infancy concerns themes of trust versus mistrust. During the early years of life, the child is dependent on caregivers for stability and consistency. If the infant is well cared for, then he or she learns trust and hope. Erikson reminded us that the capacity for hope is the most essential virtue/quality in life that will remain lifelong (Erikson, 1964). During the second stage, the child encounters autonomy versus shame and develops personal will. The third stage involves play where themes of initiative versus guilt are experienced and purpose emerges. During the school years when the fourth stage emerges, themes of industry versus inferiority are faced. At the fifth stage, the adolescent experiences identity versus identity confusion as he or she enters adulthood. During the sixth stage young adulthood's struggle/crisis is that of intimacy versus isolation with love emerging. During the seventh stage, the individual undergoes generativity versus stagnation.

Erickson identified the final stage, the 8th stage in old age, as a stage of life review when adults confront "a renewed and old-age-specific willingness to remember and review earlier experiences" (Erikson et al., 1986, p. 40). During this eighth and final stage, the individual encounters integrity versus despair and disgust. Acceptance of one's past and present can develop a feeling of well-being and satisfaction with life (Clayton, 1975; Hearn et al., 2012)

while a negative resolution leads to depression and hopelessness (Erikson, 1963). Successful resolution results in wisdom and maturity which Erikson defines as an "informed and detached concern with life itself even in the face of death itself" (Erikson et al., 1986). In old age, the ability to maintain a sense of wholeness and coherence are essential for achieving integrity. Older adults will not only accept and find satisfaction in present life, but a sense of coherence and wholeness that will enable them to look forward and accept death without fear. (Erikson, 1964). Indeed, Erikson pointed out that the main components of maturity in life are the awareness, acceptance, and integration of knowledge of one's impending death (Erikson & Erikson, 1997).

The eighth stage of Erikson's theory of psychosocial development is most pertinent for my study of older adults. As described in chapter five, the seniors in my study were considering the value and meaning of their lives as they confronted their mortality. Erikson's theory is useful in helping to illustrate the ways in which this group could share their stories with each other to find community and resonance.

The concept of life review was first formulated by Robert Butler (1963) who posited that life review is a natural, universal, and normative process that helps one come to terms with past conflicts. Butler also stressed that reviewing, reviving, and reinterpreting experience is especially significant in old age as it is necessary in order to find acceptance as the individual faces death. Both Erikson and Butler supported achieving an integrated view of the past as necessary to finding acceptance and satisfaction at the end of life.

A similar construct to life review is reminiscence as used in therapeutic settings. Ross (1989) defines reminiscence as personal memories of a distant past, and personal long-term memories of events. As part of the process of recollecting memories, life review (or reminiscence) is related to death acceptance, identity construction, problem solving, boredom reduction, and intimacy maintenance, and is useful for conversation and instruction (Webster et al., 2010; Westerhof et al., 2010; Westerhof, 2018). Although a potentially negative aspect of this process may be production of feelings of rancor or resentment, life review can be effectively used to evoke positive memories, to enhance positive feelings, and to help stabilize identity (Westerhof et al., 2010; Wong & Watt, 1991). More recently, life review has been accepted as a

therapeutic intervention, an intervention therapy to support and promote the mental health and wellbeing of elderly people (Butler, 1963; Hendriks, 2019; Mastel-Smith et al., 2007).

Therapeutic uses of reminiscence have shown to be effective in treating a variety of afflictions in older adults, such as loneliness, anxiety, depression (Stinson et al., 2010), low self-esteem (Chao et al., 2006), loss of well-being (Chiang et al., 2010; Tam et al., 2021), intellectual disabilities (Korte et al., 2011), loss of cognitive functioning (Haslam et al., 2010; Pinquart & Forstmeier, 2012), dementia (Brooker & Duce, 2000; Fang-Yu et al., 2019), Alzheimer's disease (Cammisuli et al., 2022; Duru et al., 2016), and other severe mental disorders (Liu et al., 2021). However, reminiscence is not particular to old age, as people reminisce about and review their lives throughout their lives (Merriam & Cross 1982; Pasupathi et al., 2006; Webster & Gould 2007).

Life review may be differentiated from reminiscence as being a process which involves a systematic review of one's entire life span. While life review may refer to the process of integration and evaluation of both negative and positive memories with an evaluative component, reminiscence may refer, simply, to the mere recollection of past memories (Burnside & Haight, 1994; Butler, 1963, Gibson, 2011; Haight & Burnside, 1993; Webster et al., 2010). In practice, older adults who are able to integrate and evaluate their memories through a successful life review often come to a place of self-acceptance and meaning in their lives. However, those who are unsuccessful are likely to continue to struggle with past conflicts, resulting in regret and depression.

Studies have shown that ego integrity and despair do not only pertain to old age, but can occur throughout the lifespan (Pasupathi et al., 2006; Westerhof et al., 2010). Time spent reviewing one's life can result in the development of integrity and a sense of wholeness through an acceptance of the reality of past and present. Overall, life review is an important developmental task, and the management of one's development throughout the lifespan can mean the difference between wholeness and despair (Fivush et al., 2011; Pasupathi et al., 2006; Westerhof & Bohlmeijer, 2014).

Korean Community in Canada

According to the most recent data available, Korean-Canadians make up one of the smallest non-European ethnic groups in Canada, a minority among minorities (Statistics Canada, 2013). According to the 2021 Canadian Census, there are about 222,000 (0.6 %) Korean Canadians in Canada (Statistics Canada, 2022c). Like all other immigrant groups, most Koreans settled in one of Canada's major Census Metropolitan Areas (CMAs), namely Toronto or Vancouver. Toronto has the largest Korean community, accounting for 53,940 individuals (Statistics Canada, 2017a), where ethnic enclaves ("Korean towns") offer cultural resources and support. According to a 2021 Statistics Canada article, Korean-Canadians have one of the highest poverty rates among all visible minorities, a rate hovering between 27% and 32% (Statistics Canada, 2022b).

Members of the Korean community are less likely to live alone than are the members of other minorities. Among the population 15 years of age and older, just 6% of Canadians of Korean origin live alone; in 2001 this proportion was more than doubled to 13% (Lindsay, 2007). The trend is similar in the Korean senior population. Statistics Canada in 2016 reported that 17.3% of seniors of Korean origin live alone while 27.7% of all seniors in Canada live alone. Korean seniors are also more likely than other seniors to live with members of their extended families. For example, 13.7% of Korean seniors live with relatives, including the family of a son or a daughter, compared to 5.9% of all seniors in 2016 who live with relatives (Hou & Ngo, 2021). Previous research, however, shows that even those who are living with families can experience isolation; older Chinese and Korean immigrants who live with their spouses and adult children have lower overall psychological well-being than those who live alone (H. Han et al., 2007; Wong et al., 2007).

In terms of education, people of Korean origin are more likely to have a university degree than their Canadian counterparts. According to Statistics Canada, 37% of Korean Canadians are university graduates; only 15% of the overall population in 2001 were university graduates (Lindsay, 2007)⁷. While both men and women of Korean origin are more likely to have higher

⁷ Current statistics are unavailable at the time of writing.

education levels than their respective counterparts, men tend to be more highly educated than women. Despite the higher level of education, Korean Canadian employment rates are about ten percentage points lower than that of the overall Canadian population. Moreover, average income for this group tends to be lower than the national average, and 43% of the members of the Korean community have income levels below Statistics Canada's low-income cut-off, compared to 16% in the total Canadian population (Lindsay, 2007)⁸.

Korean seniors in particular are likely to have low incomes. According to the 2021 census, close to one in five (19.0%) Korean Canadians live in poverty in Canada, more than twice the national average and the highest rate among all racialized groups (Statistics Canada, 2022b). In 2000, about 80% of seniors who lived alone had incomes below the low-income cutoff, which is about double the rate within the Canadian senior population as a whole.

The Canadian government provides two major income security programs for seniors that many immigrants can access. The Old Age Security program provides a modest pension if immigrant seniors have been in Canada for ten years, regardless of their previous employment status in Canada (Durst, 2005). The Canada Pension Plan, however, is dependent upon participant contributions and number of years enrolled in the plan, so immigrant seniors may not be able to access this program as effectively as Canadian-born seniors. While there are international social security agreements between Canada and other countries which allow people to qualify for benefits in either country, these agreements are mostly with developed countries, leaving some of the seniors from visible minority group still vulnerable. Those seniors who immigrated late in life may find that they have to rely on their sponsored children for income, further contributing to vulnerability and compromising their financial independence (Han et al., 2007).

The vast majority of Koreans (over 82%) reported their mother tongue as Korean (Statistics Canada, 2022c). Based on the 2016 census of 22 main immigrant mother tongues in

⁸ It appears that rates of poverty are falling in Canada since this source was written, however more specific statistics for Korean Canadians are not available at this time.

Canada, the rate of retention (how frequently the mother tongue is spoken at home) is very high among ethnic Koreans. Complete retention—meaning the mother tongue is spoken most often at home—is almost 80% for the Korean mother tongue group (only three other groups exceed 80%: Mandarin, Punjabi, and Tamil) (Statistics Canada, 2016). This suggests that older Korean immigrants are more comfortable speaking Korean. This is likely true for earlier generations such as the first-generation immigrant seniors involved in this study.

Health Disparities and Research Needs of Visible Minority Groups in Canada

While increased decline in physical and mental health status is evident in older age, racial health disparities that exacerbate the problem have been extensively documented (Krieger et al., 2013; Nazroo, 2003; Williams & Collins, 1995; Williams & Sternthal, 2010). Much literature in the U.S. has focused on the health disadvantages of African Americans, with increasing interest in the Hispanic/Latino population; less attention, however, has been paid to minority populations from Asian countries, now the fastest growing minority populations in the U.S. (Markides & Rote, 2015) and in Canada (Khan et al., 2015).

A recent review of the research on the role of race or visible minority status on health in Canada identified a major gap in health information and research on this heterogeneous group. Khan et al. (1915) found that only five out of 99 studies have examined nationally representative data to compare visible minorities with their white counterparts on specific health conditions or behaviours. Even within the existing research, most studies fail to distinguish immigrant or foreign-born and Canadian-born visible minorities (Khan et al., 2015). Visible minority groups are often grouped with their racial identity (e.g., Asian or South Asian) although there exists a large variation of distinct ethnicities within the Pan-Asian groups which can result in conflation of the findings. Moreover, there are scarce data on visible minority seniors' health (Khan et al., 2015). Available research suggests that visible minority seniors in Canada have worse mental and self-rated physical health than their white counterparts or the senior population in general (Lai, 2004; Newbold, 2005).

In the area of gerontology, Durst (2005) found that most published social research has focused on single ethnic groups identified as Asian, Hispanic, African American, or Native American. Durst cautions against aggregating data from these disparate groups as this form of

reductionism invites misperceptions and requires assumptions. To fully appreciate participants' cultural, historical, and social backgrounds, the unique contextual factors of immigrant visible minority groups should be considered in any analysis of research data.

Most Korean seniors in Canada are either first-generation immigrants (81% of the total Korean population according to the 2016 census) or recent immigrants. Because the history of Canadian immigration is too short for the second generation of immigrants to belong to a senior age group, most if not all immigrant seniors will belong to a foreign-born category of visible minorities (Statistics Canada, 2017a). Determinants of health in visible minority groups consist of many well-known factors including socio-economic status (Bergeron et al., 2009; Dunn & Dyke, 2000; Lofters et al., 2010), social support (Reitmanova & Gustafson, 2009) and health literacy (Choi & Smith, 2004; Omariba et al., 2011), in addition to factors more relevant to foreign-born seniors such as immigrant status (Frisby, 2011; Koehn, 2009; Koehn et al., 2011) and linguistic proficiency (Fryer et al., 2012; Ng & Omariba, 2014).

Immigrants attempting to access care in both social and health areas experience barriers to services due both to the immigrants' lack of knowledge about navigating their new country's systems and to the service provider's inability to provide key practical resources in the language of the immigrants' choice (Dixon-Woods et al., 2005; Lindesay et al., 1997; Sadavoy et al., 2004). Cultural variations in how immigrant seniors utilize services also exist. A study comparing Chinese with Korean seniors in Canada found that Korean seniors tend to use fewer local services than their Chinese counterparts (Hwang, 2008). However, consistent findings from the Barriers to Access to Care for Ethnic Minority Seniors (BACEMS) study in Vancouver (Koehn, 2009) and other studies of visible minority seniors revealed that many ethnic seniors were uninformed about what services are available to them and unaware of where to go to access those services (Johnson et al., 2005; Lindesay et al., 1997; Sadavoy et al., 2004).

While older immigrants are more likely to suffer psychological issues than their Canadian-born counterparts, immigrant seniors face a number of challenges when they attempt to access services and support (Rudner, 2011). They tend to access services less frequently because of the cultural stigma associated with healthcare. They may decide to keep their symptoms to themselves, believing that talking about mental health issues is a sign of weakness

or a personal failing, behaviour that further contributes to their isolation (Guruge et al., 2015; Johnson et al., 2004; Koehn et al., 2016; Mental Health Commission of Canada [MHCC], 2013). Indeed, the recognition of symptoms of illness can be culture specific. In some cultures, depression, for example, may be viewed simply as stress or related to spirituality rather than as a medical concern, making it more difficult to provide much-warranted attention (Ahmad et al. 2004; Marwaha & Livingston 2002; Sadavoy et al., 2004).

Although it has not been examined extensively in Canada, links exist between poor health outcomes and discrimination (Khan et al., 2015). Analysis of data from the Longitudinal Survey of Immigrants to Canada showed that visible minorities who have experienced discrimination or unfair treatment are more likely to experience worsening of self-reported health status than their white counterparts (De Maio & Kemp, 2010). A 2009 General Social Survey reported discrimination or unfair treatment among visible minorities was almost double that of non-visible minorities in Canada during the five years preceding the survey (Chui & Meheux, 2011). Those visible minority individuals who suffered discrimination were more likely to experience worsening of self-reported health status than their white counterparts (DeMaio & Kemp, 2010).

Recent research based on crowdsourced data shows that since the beginning of the pandemic, some visible minority groups, particularly Korean, Chines, and Southeast Asians, experienced frequency in harassment based on race, ethnicity, and skin colour (Statistics Canada, 2020). Indeed, studies show that discrimination against immigrants in North America can be a great source of stress in the adjustment to new life and may contribute to psychological distress (Bernstein et al., 2011; Guo & Stensland, 2018; Leong et al., 2013; Noh et al., 2007; Tummala-Narra et al., 2012; Yip et al., 2008). While these factors warrant attention, barriers to health and social care may be just as likely compounded by limited language proficiency, cultural beliefs and practices, and immigrant status (Ahmad et al., 2016; Jung et al., 2018; A. H. Kim et al., 2012; Saldov & Chow, 1994; Weerasinghe & Numer, 2011). The ineffectiveness of many existing services for Korean immigrant seniors may also be attributed to the lack of culturally-relevant information in implementing appropriate services (Choi, 2001; Lee, 2006). Additionally, among immigrant groups, reliance on family, friends, and the ethnic community for assistance is rated highly, and ethnic immigrant seniors are more likely to gain useful information regarding health and social services when information is provided in their own languages (Dixon-Wood et

al., 2005; Koehn et al., 2016). Empirical findings are needed to substantiate the significance of culturally unique factors in relation to an immigrant's quality of life; such factors can then be used to inform the development of culturally appropriate programs and services for immigrant visible minority seniors, including Korean seniors in Canada. Delivering culturally sensitive services requires an awareness and respect for a person's contextual factors (Chen et al., 2010; Jafari et al., 2010; Tiwari & Wang, 2008). Ng et al (2011) indicate that this awareness should begin during the training of health practitioners, and social services workers.

Role of Ethnic Faith-based Organizations

Faith-based organizations such as Christian churches and Buddhist temples have played a valuable role in immigrant societies. In addition to meeting the spiritual needs of the immigrant population, they have provided social and emotional support by giving people a sense of belonging and safety, and a meaning or purpose to life (Hwang, 2008; W. Kim, 2012; Y. S. Kim, 2013; Min, 1992; Richards & Bergin, 1997; Schwarz & Cottrell, 2007; Yang & Ebaugh, 2001). Prayer, forgiveness, social support, and other coping mechanisms are some of the dimensions of religion that influence people's psychological well-being (Koenig et al., 2001; Krause, 2006; E. O. Lee, 2007; Roh et al., 2013). The role of faith-based ethnic organizations pertains not only to religious and social out-reach activities. These organizations also serve as vehicles for hard-to-reach visible minority populations in health promotion programs, education, and screening services offered by external or government agencies (Y. S. Kim, 2013; E. Kim, 2016; Kim-Goh & Baello, 2008; Jo et al., 2010).

Korean Church's Role in Welcoming Immigrants

Ethnic churches have played a crucial role in assisting immigrants with the settlement process and their integration into the hosting society by providing practical resources, social support, and venues for ethnic networks (Bankston & Zhou, 1996; Chen, 2006; Hurh & Kim, 1990; Lee & An, 2013; Min, 1992; Tirrito, & Choi, 2004). Historically, Korean immigrants to North America have built ethnic communities centering on their churches to provide major social resources not only for spirituality but also for other inter-personal dimensions of their lives. It is not surprising that, while about 25% of Koreans profess the Christian faith in Korea, this profession rises to 75% to 80% upon immigration to the U.S. (Min & Kim, 2002; Mui et al.,

2006; Warner, 2001). Similarly, in Canada, a greater proportion of Koreans (58.3%) belong to the Christian religion than to any other (Statistics Canada, 2022c, https://www150.statcan.gc.ca/n1/daily-quotidien/221026/dq221026b-eng.htm). In fact, previous studies report that a high percentage of immigrant Korean seniors (77% to 97%) maintain close ties with Korean faith-based organizations (Hurh & Kim, 1990; Lee et al., 2004; Lee & Hwang,

2014; Min & Kim, 2002; Mui, 2001; Roh et al., 2013).

As Korean immigrants adopt a new country, a different culture, and a foreign language, Korean churches have helped them to cope with changes and challenges by providing a place for them to socialize, share information, and build networks in a familiar setting where the Korean language and culture is the dominant form of interaction (Boddie et al., 2011; Choi, 2001; Y. S. Kim, 2013). Ethnic Korean churches have traditionally been well-established centres that are inclusive and accessible to all Koreans (Ayers et al, 2009). For example, Korean churches have taken up the responsibility of transmitting Korean culture and maintaining Korean identity in the host nation by offering Korean language schools for second- and third-generation Korean Canadians. Moreover, the church promotes senior programs targeting first-generation or recent immigrant seniors (Y. S. Kim, 2013; E. Kim, 2016). Such programs are especially useful for Korean seniors, who can participate using the mother-tongue language with which they are comfortable.

According to the Ethnic Diversity Survey by Statistics Canada (2003), first-generation immigrants are almost half as likely as second- or third-generation Korean Canadians to participate in sports clubs or service clubs or be involved in charitable organizations. Even for those who participate, however, the frequency of participation among first-generation Korean Canadians is less than the participation of subsequent generations (Statistics Canada, 2003). Furthermore, the proportion of first-generation immigrants involved in hobbies, art, dance, cultural groups, or community organizations is also lower than the proportion of second- or third-generation immigrants involved in such pursuits. The only areas in which participation rates are higher among first-generation immigrants are religious-affiliated groups and ethnic or immigrant associations. Government statistics highlight the importance of the existence of the Korean church and its associated services for immigrant seniors (Statistics Canada, 2003). These

Such findings are not surprising given that first-generation immigrants are more likely than other generations to rate their ancestry highly and state that maintaining the customs and traditions of their ethnic heritage are important to them (Statistics Canada, 2003). They also report having a stronger sense of belonging to their ethnic or cultural group than other generations. Indeed, research shows that older Korean adults have a strong desire to maintain ethnic ties and surround themselves with familiar cultural touchstones, including ethnic foods (Hurh & Kim, 1990; Lee, 2007; Son & Kim, 2006).

Since social interactions and network participation often decline when seniors retire from the workforce, participation in various activities that can contribute positively to seniors' quality of life and well-being become even more important. Previous research shows that, independent of socio-demographic and health factors, social relationships are significantly associated with health and well-being (Adams et al., 2011; Berkman & Syme, 1979; Escobar-Bravo et al., 2012; Fiori et al., 2006; Glass et al., 2006; Lund et al., 2010; Mendes de Leon et al., 2003; Wang et al., 2002; Wilkins, 2003). Additionally, for first-generation Korean immigrants, adherence to Korean culture and ties has a healthful effect. Having well-established ethnic ties is associated with better mental health outcomes (Kim, 1999b). While there is increasing evidence of benefits with such affiliations and program offerings, many questions have yet to be addressed. It remains unclear by what mechanism and to what extent the programs offered by ethnic organizations—programs like ethnic senior school—provide social, emotional, and health benefits to participating seniors.

Historical Background of Korean Immigrant Seniors

Korean immigration to Canada began in the mid-1910s (Yoo, 2002) when Canadian missionaries working in Korea sponsored a group of Korean students to help the missionaries in the field as they expanded their work in Korea. The first group of Western missionaries went to Korea at the end of the 19th century when Korea began to allow missionaries into the country. Western missionaries, including Canadian missionaries, devoted themselves to evangelical work and improving living standards. They established schools and hospitals and taught new agricultural methods (Yoo, 2002).

A Western educational system, which allowed women and the poor to be educated, was introduced in Korea through the establishment of modern mission schools in the 1890s. These schools included western music as a subject (Sohn, 1992; Son 1998; Underwood, 1951). Because very few western musical instruments were available at that time, music classes consisted wholly of the singing of translated hymns taught by missionaries and accompanied by small reed organs. Later, western folk songs were introduced with Korean lyrics. Examples include "The Battle Hymn of the Republic" and "Auld Lang Syne" (which was later sung during the March First Movement in 1919, an independence movement against the Japanese) (J. H. Kim, 2014; Min, 1966; Seth, 2011). Similar patriotic songs set to hymn or folk tunes were widely adapted and continued to be sung in secret during Japanese colonial rule (Lee, 1985; Min 1966).

During the Japanese annexation (1910-1945), the mission school system was controlled by the Japanese government as the chief means to assimilate Koreans into Japanese culture. The topic of Korean history and culture was removed from the schools' curricula, and later the use of the Korean language was forbidden in schools (H. Y. Lee et al., 1997; Seth, 2011, Tennant, 1996; Yoo, 1992). Similarly in music education, music textbooks consisted almost completely of Japanese songs focused on cultural assimilation, propaganda, and the military preparation of Koreans to serve in the Japanese army (J. H. Kim, 2014; Seth, 2011). Since songs taught in schools were solely western or Japanese melodies, a group of Korean composers began writing children's songs in the western style (Han, 1988). The content of the newly created children's songs included expected themes such as the beauty of nature; songs that reflected the reality of colonialized Korea were censored (Han, 1988; Lee, 1985; Ryoo & Ko, 1996). Many Koreans born during Japan's annexation of Korea (1910-1945), which ended when the Japanese surrendered to the Allies at the end of WWII (Seth, 2011, Tennant, 1996; Yoo, 1992), are now immigrant seniors, survivors and witnesses of the most turbulent time in Korean history.

A time of peace after the liberation of Korea from Japan was soon followed by the Korean War (1950-1953). Korea was divided by two opposing forces in 1948—communist-led Chinese and Russian troops occupied the territory north of the 38th parallel, and the southern part of the territory was supported by the United States military (Green, 1950; Tennant, 1996). The unresolved tension between the two regimes resulted in the Korean War (1950-1953) which began on the 25th of June, 1950, when North Korea, supported by China and the Soviet Union,

marched into South Korea (Cumings, 2005). The war ended with an armistice in 1953 but Korea still remains divided. Korea is still officially in a state of war. This devastating result of the war was evident in the war's four million casualties, about half of which were civilian (J. H. Kim, 2014). Many people lost their homes and families in the chaos of the war. The aftermath left survivors with many challenges and a lack of hope.

As the South Korean government sought to reconstruct the nation after this series of social upheavals, the South Korean people unified with a ferocious work ethic and the determination to revive the nation from the post-colonial ruins of the wars. The resulting industry induced unprecedented economic growth in the 1960-70s (Park, 2009). Diligence and a rigorous work ethic was matched with aggressive governmental programs that resulted in rapid economic growth. South Korea was recognized as one of the world's fastest growing economies from the early 1960s to the late 1990s. The per capita Gross Domestic Product (GDP) of \$73 in 1953 rose to \$21,695 in 2007 (Park, 2009). The group of elderly Korean immigrants featured in my study raised during this significant time in history are part of the great effort that helped to rebuild and shape South Korea.

The modern wave of Koreans arrived in Canada in the 1960s after a liberating reform in immigration policy was initiated by the Lester Pearson government in 1966. This policy opened the doors to skilled Asian immigrants, in line with the policy's emphasis on economic and population growth. The results of the reformed immigration policy began to show in the 1970s. The largest share of Korean immigrants to Canada, however, are relatively recent arrivals, with 60% of the current Korean population having arrived between 1991 and 2000 (Lindsay, 2007). The first wave of Korean immigrants came to Canada either through the skilled worker category or as sponsored by family (A.H. Kim et al, 2012).

Due to significant changes in immigration policy in the 1990s and 2000s, there was a shift in immigrants' demographic characteristics. The changes under the Immigration and Refugee Protection Act in the early 1990s increased the share of immigrants from diverse

⁹ The changes may include admission-class distribution, source regions, pre-landing Canadian work experience, and the cities or regions in which they settled) (Hou & Picot, 2016)

economic classes, and a revision of the system again in the early 2000s favored those who had higher levels of education and charter language proficiency. While these changes resulted in a decline in the family reunification and refuge classes, there was a significant increase in the concentration of immigrants entering through the economic class in occupations such as engineering and information-technology (Hou & Picot, 2016). This affected the share of immigrants from particular countries, favoring English speakers (Hou & Picot, 2016).

Korean Elder Identity

An examination of Korean elders' identity involves the exploration of self as situated in shifting socio-cultural contexts (Min & Kim, 2002; Schwartz et al., 2010). The ways that people contextualize, embody, and present themselves are necessarily fluid because identities are constructed through a flood of socio-cultural interactions, each with ever-changing personal, social, and cultural dimensions.

For the purposes of my study, it is essential to consider the ways that Koreans conceptualize ageing. Each culture's perceptions are embedded in its language. In Korean, the connotation of 'old' is 'always (all the time)' which also conveys the idea of 'staying the same.' Interestingly, whereas in English the word 'young' is the antonym of 'old' in English, in Korean 'young' is an adjective (describing a noun) (National Institute of Korean Language, 2020). Thus, the Korean word for old (neulgda) is an action verb that describes the process of getting older. This suggests that, in Korean culture, while one's thoughts and mind continue to develop no matter what one's age, one is not getting older, only wiser. This notion may help to explain how an on-going learning attitude, reflective of the Korean heritage, is embedded in the Korean language. Perhaps the Korean attitude toward ageing can be better illustrated by calling upon the wisdom of cultural knowledge.

As an illustration of the culturally embedded meaning of a word, the Korean understanding of 'old' can be traced to the ancient history of Korea. Koreans use a unique alphabet called Hangul, which was invented by King Sejoung during the Chosun Dynasty (1393-1910) in 1446. Until that time, Korean elites adopted Chinese characters (Hanja) for writing and bureaucratic purposes. Only the upper class and nobles were allowed to learn these characters. They maintained social control by forcing the common people to remain illiterate. It was King

Sejong's development of the Korean alphabet that democratized the Korean writing system (70% of Korean letters still derive from Chinese characters). Therefore, to better understand the meaning of a word like "old" (neulgda), one must refer to Chinese characters. Chinese characters for Koreans are analogous to Latin for Europeans. The origin of "neulgda, in Chinese characters, is "老", pronounced "Ro." This Chinese character illustrates a hunched old man leaning on his staff. It denotes ageing, slowness, getting used to something, experienced, and always (all the time). Although the word often has negative connotations, old men in agricultural societies are respected for their knowledge and the depth of wisdom that comes with experience (M. W. Kim, 2016).

Kim Si - Seuop (1435-1493), a scholar and a great poet who was also highly regarded by King Sejong, tells a story related to 'old' ('老' Ro). He was recognized as a genius, not only because he learned to read and recite literature at the age of three, but also for his creativity and poetic depth of understanding (S. Kim, 2015). Hearing of Kim's wisdom, a high official, Huh Joh, visited the poet when the boy was only five years old. To test Kim's wisdom and determine whether the lad's reputation was deserved, the aged scholar asked Kim, using the word, 'old' ('老' Ro), to recite a poem for him. Kim recited, "老木開花心不老" (read in Korean as - Ro Mok Gea Hwa Sim Bul Ro Moon), which translates as "an aged tree still flowers just like the heart that never grows old." Amazed at the boy's understanding, Huh Joh left, assured of Kim's reputation. Kim's abilities were reported to the Palace and eventually reached the King Sejong's ear (S. Kim, 2015).

Kim Si - Seiop's verse expresses a depth far beyond his age. In testing Kim, Huh might well have underestimated what would impress an aged and well-experienced man of intellect. It is also possible Huh reflected his own perception of ageing, as he was becoming an old man himself, and so was curious to know what a little boy would think of the meaning of 'old.' When Kim was able to answer Huh without hesitation, the boy challenged the scholar's expectations by redefining the meaning of the word to go beyond the obvious, physical realities of ageing. Kim's composition brings new insight by focusing on a vigorous mind. In other words, as long as one continues to gain knowledge and understanding, one is, despite outward appearances, still young (S. Kim, 2015).

The poem's evocation of a blooming mind gained through continuous learning also references Korean cultural heritage as deeply rooted in Confucianism through its prioritization of education (Im, 2021). Modern Korean society prizes educational capital as a means of furthering social mobility. One manifestation of this esteem is the system of learning centers, or Hagwon, which teach a variety of subjects. Even in old age, a similar fervour is evinced in the pursuit of active living and well-being, as the Korean song, "What's Age Got to Do with It," captures so well.

Like members of other Asian cultures, Koreans are family oriented, and traditional family values find expression through social relationships (Im, 2021; Lee & Holm, 2011). Influenced by the Confucian tradition of filial piety, elders are respected and valued for their wisdom. Respect is also embodied in the use of the formal, polite words which are spoken to the elderly. As the heads of families, seniors are consulted for advice and support and in decision-making and the resolution of family conflict. Older adults maintain a sense of self and control as decision-makers in their families. Social isolation and loneliness are prevalent risk factors for poor mental and physical health in older adult populations as many seniors must rely on increasing family support (Donovan & Blazer, 2020). However, as immigrant Korean elders adopt more and more of the western values of individualism and the egalitarian culture in Canada, more Korean seniors choose to live alone (A. H. Kim, 2010).

In a recent study reporting elderly abuse among four groups of elderly immigrants from different cultural backgrounds (Chinese, Korean, Tamil, and Punjabi) in the GTA, Korean older adults, with the lowest rating of English proficiency among the groups, reported social isolation as the most relevant risk factor (Guruge et al., 2021). While another study reported similar risk factors for elderly Koreans, it was also noted that family support and relationships with family members are crucial in preventing and ameliorating depression among elderly Korean immigrants (Lee & Holm, 2011).

A well-recognized, culturally bound symptom, Hwa-byung, is a Korean traditional emotional illness that has been noted and addressed for centuries (Min & Suh, 2010). Hwa-byung (hwa means 'fire and anger,' and byung means 'illness') is also identified as 'anger syndrome' which is caused by the suppression of anger and emotions. Suppressed emotions

include depression, worry, anger, fear, and shock, all of which are accompanied by various somatic and psychological symptoms such as sensations of heat, pressure in the chest, epigastric pressure, heart palpitations, headache, insomnia, and sadness (Donnelly, 2001; Min & Suh, 2010). Hwa-byung is a unique adaptive response to life's hardships and struggles and has long been studied as an established and documented disorder among Korean Americans (Lin et al., 1992; J. Lee et al., 2014). There is a strong association between hwa-byung and depressive disorders (Im, 2021). Many Koreans do not feel comfortable sharing their feelings; reticence is a culturally-valued disposition, especially among older Koreans, which helps to explain the prevalence of hwa-byung and the risk of depressive disorders among these elders (Min & Suh, 2010; Sin et al., 2004).

Lifelong Learning

The term lifelong learning was coined in a UNESCO report in 1972 (Friesen & Anderson, 2004). The report, entitled "Learning to Be," otherwise known as the Faure Report (Faure et al., 1972), proposed the concept of lifelong learning throughout one's life span as 1) vertical integration denoting lifelong learning continued throughout life, and 2) horizontal integration which is a life-wide concept (Faure et al., 1972). In addition, the parallel concept of life-wide learning emerged, referring to learning that occurs in all aspects of life, not specific to a specific time of life or age, but that is continuous through multiple contexts, situations, times, opportunities, and needs. It is inclusive of and impartial to a variety of ways of learning—formal, nonformal, and informal learning contexts (Field, 2001; Jones, 2009).

A policy document from The Commission of the European Communities emphasizes that "Lifelong learning is no longer just one aspect of education and training; it must become the guiding principle for provision and participation across the full continuum of learning contexts" (2000, p.3, emphasis in the original). Understanding lifelong learning entails an understanding that learning is continuous and ongoing throughout one's life. It views learning as an essential and necessary aspect of living if a person is to grow and adapt to rapidly changing society (Jarvis, 2006). Jarvis (2009) explained that, as an "intrinsic part of the process of living" (p. 10), learning is a part of living that occurs as an incidental, unplanned, and preconscious means of responding to everyday situations which provide diverse learning experiences within a

social context. It is suggested that learning begins with the fetus in a mother's womb and continues until death (Jarvis, 2009). This enhanced view of learning posits that life "itself is as much about learning" (Jarvis, 2006, p.10) as learning is about life (Jarvis, 1983; 2009, p. 19).

Learning and education are terms which are often used interchangeably. However, there are important distinctions between the two. The term education emphasizes the state's responsibility to provide learning opportunities, mostly in institutionalized, formal settings that involve expertise with a specific educational purpose or aim. It is narrow in scope and specific in terms of definition and content (Hayslip, 2014). In short, 'education' is intentional, and takes a systematic approach to learning provided through agencies; it does not recognize learners' responsibilities for their own learning.

Learning is broad in scope, holistic, diverse, and integrated with everyday living (Findsen & Formosa, 2011). Learning occurs throughout a life-span (education during every stage of life) and is life wide, occurring in a variety of contexts (Friesen & Anderson, 2004) which may relate to lifelong learning. The context of learning relates to a person's progress in gaining knowledge and varies in everyday life from the formal approaches of organized activities to a person's less methodical search for knowledge.

There are three major types of learning: formal learning, informal learning, and non-formal learning. Formal learning is the most familiar type of learning: organized, institutionalized, and chronologically graded within a hierarchical educational system. Informal learning occurs outside of the formal classroom environment. It refers to learning by experience or learning as the experience itself, reflecting John Dewey's (1999) broad take on informal experience as the key to learning. Similarly, Rogers (2004) also saw learning as a continuum and saw informal learning from experience as the foundation for other types of learning. Non-formal learning refers to the middle ground between formal and informal learning. It suggests any systematic and organized educational activity carried on outside the formal system; it includes selected types of learning among particular subgroups of a population, such as community-based sports/arts/music programs. Studies support that these leisure-type, non-formal courses are beneficial in the maintenance of older adult participants' psychological wellbeing (Jenkins, 2011; Jenkins & Mostafa, 2015).

Promotion of Lifelong Learning: World Trends

Promotion of lifelong learning is evidenced across the world. European countries have been particularly active in promoting lifelong learning as the key to bringing economic success in knowledge-based societies (Commission of the European Communities 2000, p. 5). Similarly, international policies also favour lifelong learning as a public good. Gorard et al. (2002) note that lifelong learning furthers national and international interests. Recent policy statements from international agencies such as the United Nations (UN), OECD, World Health Organization (WHO), UNESCO, World Bank, and the European Union promote lifelong learning and individual development in order to increase employability, economic benefit, and social cohesion, which are often linked to formal educational institutions (Field, 2011; Volles, 2016).

The 1997 report of the US Commission for a Nation of Lifelong Learning, for example, made a connection between learning and economic success (Merriam et al., 2007, p. 48). UNESCO (2016) promoted lifelong learning as a guiding principle for the future of education in societies that emphasize equality and human rights. In addition, the OECD's most recent report actively promoted lifelong learning as vital to adaptation and resilience in the dynamic and turbulent times of the COVID pandemic (OECD, 2021). As well, there is increasing recognition and exploitation of lifelong learning as a market driven commodity. Courses may be offered for profit as individuals upgrade their skills. This is a concerning development because when learning becomes mandatory for employees to ensure employability, compulsory courses may strain family finances (Jarvis, 2006).

Lifelong learning exists in educational and business settings, as it is both individual and institutional, a social movement and commodity, and a policy and practice, but one which requires a multidisciplinary approach if it is to be successful. Above all, lifelong learning must be sensitive to social and cultural concerns. Compared to many Western societies (whose cultures and mores have been shaped in the Judeo-Christian tradition), Asian countries with a Confucian background, like China, Hong Kong, Singapore, and Korea, are more familiar with lifelong learning and have a high regard for education (Han, 2007). The deeply embedded notion of learning recognizes life itself as learning and learning as integral to living. Confucian philosophy has prioritized learning as essential for morality; educational attainment and

intellectual achievements are considered essential to upward social mobility in meritocratic society (Kim & Merriam, 2004). Older adults like elderly immigrant Koreans and those who come from Confucian backgrounds are, thus, likely to display a positive attitude towards lifelong learning and pursue it for personal fulfillment (K. H. Kim, 2004). It is normative for Korean people to pursue their interest in and learning of various topics in order to update their knowledge.

Jarvis (2001) cautions against emphasis on lifelong education as a social phenomenon, which implies attendance in formal educational institutions. Moreover, researchers should take precautions against using the term 'lifelong' to indicate extended attendance in institutionalized learning settings through one's lifetime (European Commission, 2001; 2006, OECD, 2019). Educators should be especially careful in taking part in political discourse that sees education as a means solely to enhance economic competitiveness in the global economy (Merriam et al., 2007). In employing the term lifelong learning, one must ask what it means and what it entails in specific contexts and practices.

Despite a tendency to simply focus on lifelong learning within an economic perspective, research supports the notion that acquisition of knowledge contributes to wellness, promotes well-being, and develops resilience as well as enhancing cognitive functioning (Field, 2011). Learning can be a key component of active and healthy ageing in later life (Boulton-Lewis & Buys, 2015; Jenkins 2011; Jenkins & Mostafa, 2015; Leung & Liu 2011; Narushima, 2008). All learning activities are described as having the potential to contribute to physical, mental, and/or social well-being, whether they are undertaken for pleasure or for another purpose (Boulton-Lewis & Buys, 2015; Dench & Regan, 2000; Panayotoff, 1993). Learning is also socially empowering and meaningful when it enables participants to maintain social networks which sustain well-being over the long term (Boulton-Lewis & Buys, 2015; Dench & Regan, 2000; Findsen & Formosa, 2011; Lassen, 2015; Withnall, 2010).

A cross-sectional study with 416 older adults (aged 60 years and above and enrolled in a community-based lifelong learning program) reported a positive association between the duration of participation in a lifelong learning program and psychological well-being (Narushima et al., 2018). The study reported that self-esteem, self-efficacy, and resilience increased in one

third of the participants, despite a high level of vulnerability (measured across three domains of health, socio-economic status, and social support). Findings indicated that active participation in learning activities enhanced psychological well-being even among those with chronic conditions (Narushima et al., 2018).

In addition, Narushima (2008) interviewed fifteen older adult participants involved in a non-formal, leisure-related, public, continuing-education program in Canada. Findings revealed that participants were motivated to keep physically and mentally active. Participants experienced joy and a sense of mastery as well as increased social support. The study also suggested that non-formal, leisure-based learning can be more beneficial than other kinds of learning because it is more likely to sustain participation (Jenkins & Mostafa, 2015; Narushima, et al., 2018). The English Longitudinal Study of Ageing (Jenkins & Mostafa, 2013, 2015), which examined the relation between participation in later-life learning and wellbeing, reported similar findings.

Other benefits of learning include supporting older adults to gain financial, emotional, and social resources. Quality of life is enhanced through increasing enjoyment, self-confidence, self-concept/identity, self-satisfaction, and the ability to cope (Dench & Regan, 2000). Learning helps elders maintain and improve cognitive functioning by improving neuroplasticity, supporting new neural pathways, and facilitating neurogenesis (Valenzuela, 2009). Wolf (2009) found that older adults who engaged in continued learning reported continued growth in verbal knowledge and less decline in memory into their late seventies. For older adults, learning continues to be meaningful. Withnall's (2010) study of learning among older adults revealed that individuals have their own varied understandings of learning. Meaning making from learning is important for older adults as it is related to older adults' well-being (Fisher & Wolf, 2000). Learning experiences can expedite life review and self-reflection allowing a greater understanding of themselves and their individual insights (Withnall, 2000). Finding meaning gives life purpose and strength.

Older Adults' Perspectives of Learning

Boulton-Lewis and colleagues (2006) write that older adults have strong motivations for learning, while Withnall (2006) notes that it is important to address older adults' own perspectives on learning. Facilitators must be aware that seniors bring lifetimes of experience to

the classroom (Boyer et al., 2014; Wernstein, 2004). Older adults usually have clear ideas about what they want and how they want to learn. Knowles (1970) finds that members of this population tend to be internally motivated and that they are most often self-directed learners who bring life experiences and knowledge to their own learning.

Boulton-Lewis & Buys (2015) conducted a case study to explore the experiences and perspectives of older people involved in both formal and non-formal learning. They found that many older adult participants cited that learning for stimulation, interest, pleasure, curiosity, and simply to keep up with 'things' helped them feel alive and functioning. The social component of learning was important as elders interacted and shared learning experiences in a group. Relevancy of topic was held to be of major importance in this learning since it is often an individual learner's own responsibility to take the time and pay to travel to access learning. Unless the learning activity is relevant to their lives or gives them real pleasure, older people are less likely to participate. For older adults, lifelong learning related to the global economy is probably less about increasing employability, although more older adults are choosing to stay in workforce (Field, 2011), but more to do with the digital literacy that is increasingly essential for successfully living in an increasingly digitized world. For example, the older adults in Boulton-Lewis & Buys (1915)'s and Field (2011)'s studies placed high importance on learning about new technology. Such learning empowers them and enables them to feel alive as they gain knowledge and skills for themselves (Boulton-Lewis & Buys, 2015).

Learning Approach

Providing a supportive learning environment in which older adults can express their curiosity and explore personally meaningful interests is necessary in any successful lifelong learning endeavor (Boyer et al., 2014). Elderly participants prefer direct learning experiences in which they control all aspects of the learning process (Pipkin, 2012). Kang (2007) notes that learners like to shape and actively construct the learning context. Given the research, it is important that every learning activity be designed to be flexible and to take into consideration older adults' previous learning experiences. One key component is the social interaction amongst participants that can foster meaningful relationships. A supportive community-based program will be most successful when it encourages involvement and participation of older adults as well

as their input to both the content and delivery of the program (Boyer et al., 2014; Hayslip, 2014; Wernstein, 2004).

Although there is a positive shift in societal attitudes towards old age, challenges remain. Common stereotypes exist of seniors who are portrayed as more "feeble and miserable than they really are" (Sherman, 1993, p. 43). Physical and biological change encompasses both greying hair and hidden health concerns (Kaufman et al., 2004; Marsiske et al., 2001). Some seniors experience memory and intelligence loss (Marsiske et al., 2001), and the degree of such loss varies with some seniors experiencing improved memory with age (Honn Qualls & Abeles, 2000). Upon retirement, loss of income impacts social status and social roles, sometimes leading to social exclusion.

Older adults are also vulnerable to age-related discrimination: exclusion from services, resources, activities, and social relations that influence their quality of life and health. Estes (2006) reports an increasing rate of older adults becoming vulnerable to poverty and social isolation that negatively impacts their social functioning. Moreover, ethnic minority older adults are more disadvantaged than other seniors. Nazroo's study (2006) reported that ethnic minorities in Britain, especially Bangladeshi and Pakistani men, have poorer health and are more likely to suffer financially. Canadian seniors who are of ethnic minority report greater mental health needs because, due to linguistic deficiencies, financial difficulties, and lack of cultural competencies, they experience more barriers to health care services (Chow, 2010; Stewart et al., 2011).

Formal Older Adult Learning Programs

A variety of formal learning programs for older adults have been established and are available across North America. The most well-known, formal older adult learning programs are Universities of the Third Age [U3As] or U3As online (n.d.) (https://www.u3aonline.org.au), Lifelong Learning Institutes [LLI] (also called Osher Lifelong Learning Institutes [OLLIs] in North America), and Road Scholar. These programs were established to assist senior citizens to "remain integrated in society and form their own social environment while contributing to their intellectual potential and spiritual development" (Formosa, 2010, p. 198). At the same time, some of these programs have been criticized as middle-class biased, elitist, and female dominated.

U3As have expanded to establish learning centers across the world. There are 19,300 learning centres in China alone which provide courses to older adults (Thompson, 2002). Since U3As and LLIs are usually affiliated with higher education institutions, participants tend to be better educated and come from more affluent backgrounds (Formosa, 2012). Despite their declared intention to provide access to all, the U3As tend to draw an older adult population with better socioeconomic status (Mackowicz & Wnek-Gozdek, 2016). While the U3A programs offer similar benefits in improved life quality, these programs limit participation. They suggest that disadvantaged seniors be offered programs tailored for them. However, cultural situations may limit senior participation. Studies indicate that grandparents assume child rearing duties in Poland and that these obligations prevent some seniors from participating in the programs (Mackowicz & Wnek-Gozdek, 2016). Recognizing these pertinent issues, programs such as "Men's Sheds" (a community-based program that began in Australia) have been found to be successful in reaching older men from all socioeconomic statuses by offering programs tailored to the social health, well-being, and learning needs of retired men; these initiatives have spread and are now found in Scotland, Ireland, England, Finland, and Greece (Golding, 2011).

The theory of social capital has been used to explain how older adults' learning contributes to wellbeing (Biggs et al., 2012). Participation in learning activities increases social capital by developing social skills, building social networks, and promoting shared norms and tolerance of others (Field, 2009, p. 23). Older adult participants in U3As or UTA (Universities of the Third Age) programs cited outcomes such as making new friends and gaining social support as a major part of their involvement (Formosa, 2012). Furthermore, learning does not always mean participating in a program. Learning through participation in community activities and volunteering can have similar beneficial effects (Dench & Regan, 2000; Wister et al., 2010).

Lifelong Music Learning in Community Contexts

Lifelong learning is the ongoing and voluntary pursuit of knowledge continued through old age. Musical learning involves a variety of musical modalities and activities. Increasingly, music educators seek to foster-lifelong musicmaking because musical engagement can be an important form of participation associated with a wide range of benefits related to health and well-being. Studies support the idea that early exposure to musical activity, such as music

education in school, is directly linked to a person's preference for and appreciation of music in later life (Cho, 2018; Flowers & Murphy, 2001; Trevarthen, 2002).

Nazareth (1999) offers a perspective on lifelong music education using the idea of vertical and horizontal integration: "Educators should take musical cues from the community by drawing from and contributing to its diversity, heritage, and common practice, and provide music learning opportunities accordingly" (Nazareth, 1999, p. 18). With an emphasis on learning being integrated with life, Nazareth (1998) explains vertical integration as "coordination of learning experiences at different times of life," while horizontal integration is the "coordination of the diverse range of learning experiences" (Nazareth, 1998, pp. 11, 21-23). In vertical coherence, a program's entry points and learning opportunities are flexible so that anyone regardless of age or skill can begin musical engagement and learning (Bugos, 2014). Vertical orientation resonates with the notion of a "lifetime continuum of learning" or "lifespan engagement," which suggests how music-making may be available and open in a variety of settings (Myers 2007). Jones (2009) considers this approach through what he terms "lifewide music learning": formal, non-formal, or informal learning that takes place across the full array of life activities, which may include academic, personal, social, and professional pursuits, and which may occur at any stage in life.

Lifelong music education can refer to learning processes as well as longitudinal and continuing methodologies. Horizontal coherence concerns the quality of life of learners in that musical learning/music-making provides not only skill-related benefits but also brings personal and aesthetic fulfillment that enhances one's quality of life (Nazareth, 1998). Horizontal orientation corresponds with ISME's definition of Community Music Activity: "every individual should have the opportunity to be involved in musical activities reflective of the pluralistic nature of the society" (International Society for Music Education (ISME), n.d.) (https://www.isme.org/our-work/commissions/community-music-activity-commission-cma). Elderly participants' musical engagement can be successfully fostered through supportive environments in a community setting through community centers, ethnic organizations, or ethnic churches (like CESC, which is featured in the present study).

Myers perceives lifelong music learning as multifaceted: "the act of acquiring knowledge, understanding, and skill through high-quality opportunities made available across the life span" (Jones, 2009, p. 204). Terms and definitions of community are broad and vary across contexts. They may or may not fit every situation. My study is based on an ethnography of a community cultural program for elderly Koreans in a non-formal setting. The quality of the opportunities available to this group remains subjective. Myers' (2007) definition implies that specific types of opportunities are 'high-quality." Myer's definition would seem to distinguish between 'learning' activities with underlying production as opposed to endeavors which are simply 'activities.' The distinction is problematic as music educators know that what learners may experience in musical activities varies, individual by individual, and can be idiosyncratic as well as subjective. Without further clarification, there may be confusion: What is a "high-quality" opportunity? How does it compare to a "low-quality" opportunity? Myers' (2007) definition carries elitist implications with the potential to marginalize and thus exclude certain learners. My study embraces definitions of lifelong learning that fall under the umbrella of active ageing.

Understanding the challenges that older adults experience, including ageism and agerelated physiological, social, and psychological changes, demands awareness of the mechanisms
that promote engagement in musical activities. Evidence points to lifelong learning as supportive
of psychological well-being and resilience, and as preventative of cognitive decline in older
adults (Boulton-Lewis & Tam, 2012). Maintaining high cognitive functioning, which is
significant in active engagement, must be interrelated with physical and social functioning if one
is to live meaningfully.

Music educators should be careful when taking part in political-economic discourse that constructs education as a means of enhancing economic competitiveness in the global economy (Merriam et al., 2007). Previous studies have revealed differences in learning characteristic of adulthood as compared to childhood, and differences between a teacher-centered approach and a learner-centered approach in which learners take control in planning, setting goals, and determining activities (Bugos, 2014). It is also important to understand and address the learning needs that are relevant to older adults (Boulton-Lewis & Tam, 2012; Kang, 2007). Elders bring prior music-learning experience as well as motivation to the learning process (Lee, 2013).

Cho's (2018) narrative study with three elderly Korean immigrants indicates that songs from the past are valuable dimensions of their identity. Cho's exploration of everyday music among older Korean immigrants displays personal meanings paired with homeland songs and reveals emotions that continue to have impact in their old age. Tied to particular life events, the songs continue to be an important way for these immigrants to express, release, and reexperience emotions, while reaffirming who they are.

Less attention has been paid to cultural influences on older adults' learning. Tam (2012) has sought to fill this gap by offering a dichotomous view of lifelong learning in two different cultural contexts, the eastern and the western. The rhetoric of lifelong learning portrays learning as an obligation and not a choice, an obligation that causes persons to feel pressure to learn in order to survive in a society where learning is available throughout one's life (Hake, 1999). In Korea, most children attend after-school private instruction – whether in piano, English, or Arts – because they find no peers to play with unless they take these after-school classes. Lifelong learning has been inculcated as a "moral obligation" which individuals are required to accept as their responsibility (Edwards, 2008 p. 31).

Mantie (2012) has pointed out that one problem of lifelong learning rhetoric is that it changes the emphasis from choice to obligation, turning people from learners to doers. From this perspective, it is possible to see how learning enables people to do something in order to participate with others. For example, learners of any age learn to read notes (solfège) so they can play music. While learning to read music is useful, it is not a substitute for the social act of making music with others. If instruction (learning to read music notation) becomes the sole focus of learning, it valorizes individual achievement rather than social and musical engagement. As a result, both teachers and learners may miss, perhaps, the most important aspect of music learning: enjoyment of participating in and making music together.

Community Contexts for Lifelong Music Engagement

The emerging field of community music has expanded the discourse in music education to include direct application to lifelong music learning. It is, therefore, important to consider music making within the community since music education has previously been focused primarily on learning during childhood within school settings.

In Canada, the rich musical heritages of many immigrants are evidenced in the varieties of local music making found in communities (Bush & Krikun, 2013) where community schools and ensembles promote musical engagement. Unfortunately, there is little governmental sponsorship for this kind of micro-musical engagement in Canada. Local community centres and church programs serve the particular needs and interests of their communities. Self-funded and self-directed, these smaller organizations are usually staffed by volunteers. They are vulnerable to the changing needs and resources of their participants. More recently, especially since the pandemic and the growth of virtual networks (i.e., virtual learning, virtual meetings), social networking is increasingly replacing face-to-face interaction and learning (Anderson, 2016; Amit, 2002).

Community Music: Definitions and Negotiations

Williams (1976) described community as a 'warmly persuasive word" that is "never to be used unfavorably" (p. 76). The word community has positive connotations (Bauman, 2013), and is associated with hospitality (Higgins, 2007). The term community music has been used in different ways in different places and is an emerging field of research (Veblen, 2008).

Community music is premised as an ideal activity in which individuals choose to engage with others in music making (van der Merwe, 2017; Veblen, 2013). With inclusivity as a core value, researchers recognize music's power to create community (Clothier, 2002; Higgins & Willingham, 2017). Engagement creates a sense of belonging, intervention, and social transformation (Chadwick, 2011; Dunphy, 2018; Garrett, 2010; Goodrich, 2013; Moser, 2018; Rinde & Schei, 2017) as evidenced by community music practitioners in indigenous settings (Marais, 2021) and prisons (Cohen et al., 2021), and with children with disabilities (Mitchell, 2021), older adults (Coffman & Dabback, 2021; Creech, 2018), and minority older adults (Jo, Veblen, & Potter, 2021).

While definitions must be contextualized and may be contested (Elliott et al., 2008), some suggest that the emphasis should be on what community music does rather than what it is (Bartleet & Higgins, 2018). Bush & Krikun (2013) advise that community music is best understood through examining ways in which it is practiced globally. Music learning reflects the informal settings in which community music occurs. Active music making is the focus.

Participants are usually self-directed and responsible for their own learning. The relationship between community music worker/facilitator and participant is reciprocal as roles are fluid (Veblen, 2007).

Historical Background of Community Music

Community music in North America was highly influenced by music educator Peter Dykema (1873-1951). Dykema (1916) advocated community music which he defined as "socialized music" (p. 218) and expressed the importance of everyone being given the opportunity for "free and frequent participation in music" (p. 223). In 1919, Dykema (1919) compiled a song book entitled *Twice 55 Community Songs* which promoted national and humanistic views that surfaced after World War I. This song book and others like it included songs from many nationalities. At the end of World War II, in 1945, music making was seen as a positive social force. Many community music projects were funded at this time.

One notable researcher, Willem van de Wall, began his research on the role of music with wounded soldiers in hospitals during World War I. His research made major contributions to the field of music therapy in less observed institutional settings such as prisons, psychiatric wards, and general hospitals. Max Kaplan was a sociologist recognized for his research on music in local community music programs and practices.

In the United States, the largest school music education organization, the National Association for Music Education (NAfME), recognized community music and sponsored the Adult Continuing and Community Music Education (ACME) in the early 1990s. Currently, however, NAfME does not include community music in its organizational chart or on its website. https://nafme.org/about/

Community Music in Canada

In Canada, the rich music cultures of settlers gained attention in the second half of the twentieth century. Cagnon's historic 19th century collection of French-Canadian songs, *Chansons Populaires du Canada*, was printed in 1965 (Gagnon, 1965). Irish and Scottish settlers in Newfoundland and the Maritimes brought Celtic music. English immigrants and Loyalists from

the United States established singing schools throughout Southern Ontario while Ukrainians, Mennonites, and Icelanders brought singing traditions to the prairies (Bush & Krikun, 2013).

An early campaign called "Music in the Home" was promoted by the Canadian Piano and Organ Manufacturers' Association who sponsored essay contests, community music weeks, and local music festivals. A recent development reported by the North American Coalition for Community Music involves seventeen music educators meeting to discuss how to make music education more relevant and accessible to all (Bush & Krikun, 2013).

Community vocal ensembles can take many forms. The Barbershop Harmony Society claims nearly 30,000 members. In Canada, the Toronto Mendelssohn Choir is one of the oldest and most famous amateur chorus. Instrumental ensembles such as the New Horizons International Music Association which began in 1991 have become widespread, providing opportunities for seniors to learn and play band instruments. Currently, 17 New Horizons band groups in Ontario are listed on the New Horizons home page (https://newhorizonsmusic.org).

Social Networking in Community Music

Social networking as a medium for music making has become increasingly pertinent since the pandemic (see Waldron et al., 2020). Social interactions and relationships established through musical engagement are seen as one of the benefits of community music activity (Coffman, 2002; Myers & Diener, 1995; Veblen & Olsson, 2002). Depending on the needs and wants of the participants, community programs can offer creative outlets, especially vulnerable groups like older adults.

Missing Research in Community Music Literature

Although many well documented community programs have been designed for seniors who are of visible minorities, there is a distinct lack of research into programs offering music for these groups. The very successful New Horizons Programs may be attracting members of visible minorities, but this is not documented. As well the New Horizons program, at the time of this writing, do not feature cultural musics or instruction in any language besides English. As Cho (2018) shows, familiar childhood music is very important for older immigrants whose cultural milieu is markedly different from that of the host country. I found no specifically focused

community music programs for Korean Canadians. As well, I found very few programs offered in my subjects' mother tongue and cultural ways. With the advent of the pandemic, those programs that might have emerged were shut down, just as the numbers of elders of visible minorities, Korean Canadians among them, were increasing.

Chapter Three: Korean Immigrant Seniors' Music Making in an Ethno-Cultural Community Program in Canada¹⁰

H. Elisha Jo

Don Wright Faculty of Music, Western University

London, ON, Canada

Kari K. Veblen

Don Wright Faculty of Music, Western University

London, ON, Canada

Patrick J. Potter

Schulich School of Medicine & Dentistry, Western University

London, ON, Canada

This case study¹¹ explores music learning, sharing, and well-being among Korean immigrant seniors in a church-sponsored ethno-cultural community program in a Toronto suburb. The all-volunteer community program provides opportunities for Korean-Canadian first-generation immigrant seniors to interact with each other in their mother tongue. Our focus is on ways of engagement, most often through musical practices. Using ethnographic tool, we consider 1) music's various roles within the program, 2) group and individual identity in the Korean diaspora, 3) how shared culture may be nurtured through the program, with 4) the broader implications for wellness.

¹⁰ This paper was submitted for the proceedings from the International Society for Music Education (ISME) 2014 Seminar of the Commission for Community Music Activity in Salvador, Brazil. This research was mainly conceived by Elisha Jo and in the case of co-authorship, Elisha Jo contributed 60 % and co-authors, Kari Veblen and Patrick Potter contributed 40 %.

¹¹ This study is part of a doctoral project in music education exploring diasporas, identity, and musical knowing through the Korean immigrant experiences.

Key words: Music, Community Program, Lifelong Learning, Well-being, Identity, Asian Diaspora, Community Music, Senior, Ethnicity, Immigrant

Introduction

Demographics of a greying population have sparked demand for social activities for older adults, frequently offered through community organizations (e.g., community centers and religious organizations). Participating in such programs, particularly musical programs, can contribute to overall sense of belonging and well-being (Coffman, 2002; Cohen et al., 2006; 2007; Carucci, 2012; Clift et al., 2010; Creech et al., 2013; Sun et al., 2013b; Yinger, 2014). While more social opportunities are needed for seniors in general, immigrant seniors are more likely to be excluded from these services due to linguistic and cultural barriers (Amit & Litwin, 2010; Choi, 2001; Kuo et al., 2008; Y. Lee, 2007; Tendulkar et al., 2012).

With a significant global demographic shift resulting in rapidly increasing aging populations ^{12,} there is an increased need for targeted programs that serve community-living older adults to maximize their health and well-being ¹³. Canada Enoch Senior's College (CESC), an ethno-cultural community program offered by the Korean church in the Greater Toronto Area, established just such a program to meet the unique needs of immigrant seniors living in the community in 2003. For the past decade, the CESC program has been successful in reaching out to Korean seniors who are ethnic minorities in Canada. The all-volunteer community program provides opportunities for Korean seniors to interact with each other in Korean, mother tongue of seniors who are first generation immigrants. Particular attention is paid to music that connects

¹² The World Health Organization (WHO) reports that between 2000 and 2050, the percentage of world's population over 60 years will likely to double from 11% to 22%, that is approximately 605 million to 2 billion (WHO, 2014). Similarly, the percentage of Canada's senior population (aged 65 and over) is expected to increase from 13% to 25% (Statistics Canada, 2006) by 2036. Canada is ranked as one of top 10 countries with the highest proportion of older adults (aged 60 and over) by 2025 (WHO, 2002).

¹³ It is suggested that the physical and social structures of neighborhoods or communities can protect older adults from the risks of loneliness and social isolation documented by MacCourt, Wilson, & Tourigny-Rivard, (2011). WHO also recommends that community level actions are necessary to promote health and support well-being in older adults (WHO, 2007).

Korean seniors to their remembered homeland.¹⁴ Music making in this instance helps nurture cultural group identity and an exploration of their present and emerging senses of self. Revitalized through music making, these Korean diasporas embody essential social components of community music which includes, "identity, heritage, group solidarity, healing, bonding, celebration" (Veblen, 2013, p. 3).

Singing (art songs and popular songs), dancing (Korean traditional dance, worship dance, and line dance), harmonica ensemble, arts and crafts, sketching and drawing, calligraphy, and Chinese medicine courses attract over 150 participants every year. Although some course offerings change depending on seniors' interests and needs, music constitutes major part of the program and serves as a significant way in which individuals participate, engage and express themselves.

Attending Enoch College is important to these Korean seniors on many levels. Although Toronto is a dynamic and diverse Canadian metropolis, the seniors in this program may go for weeks within a Korean and pan-Asian immigrant bubble where the sights and sounds of their homeland are recreated so being able to interact and communicate in Korean in the Mississauga section of Toronto is especially comforting. Participants share the same ethnicity and a unique cohort history: liberation from Japanese annexation, Korean War, life pre-TV, and political/economic upheavals, as well as immigration. Homeland songs trigger memories of this shared past. Thus, the activities promoted through the CESC program are molded through tacit communal understandings, for example, songs which are chosen are participant-led and consensus-based.

Methodology

Our methodological framework combines a qualitative case study design with a wellness survey tool. Information was gathered from combination of field notes, as well as informal

¹⁴Music here is a powerful medium for remembering, for recreating, as also documented by other researchers (Batt-Rawden et al., 2005; Cohen et al., 2002; Halpern et al., 1996; McCaffrey, 2008; Bugos et al., 2007; Ferreri et al., 2014; Paquette & Goulet, 2014; Wang, 2013). It is important to note that seniors at Enoch College engage actively with music through singing, moving, playing, and creating song lyrics.

conversations with the participants, and through self-reflections (Denzin & Lincoln, 2005). Video recording of the activities supplemented other data collection methods (Bogdan & Biklen, 2003; Goldman- Segall, 1998). ¹⁵

Program Structure

Canada Enoch Senior's College (CESC) offers fall and spring terms that runs for nine to eleven weeks each year. Over 150 seniors are registered each term. Seniors who are over sixty years are accepted, however, the majority of students are in their seventies. This spring marked the twenty-third term since 2003 and 155 students have registered with 140 successfully completing the term. Every semester a variety of programs are offered. The classes are offered once a week with one session dedicated for a field trip.

Classes start at 10 am and ends at 2:30 pm with the time divided into four periods, somewhat like the formal Korean school system. To start the class, both the Korean and Canadian anthems are sung followed by the Enoch College anthem. Everyone attends a common class for the first two periods that covers a wide range of activities: singing with movement, heavenly message, art song singing with occasional Korean history or Korean literature class and special events. After a communal lunch, students take an elective course and then join their homeroom (or small group) for further social engagement.

Group membership is encouraged through the structural organization of CESC. Everyone is welcomed and valued throughout all aspects of the program. Registered seniors at the college are addressed as students and each student is assigned to a small group referred as homeroom

¹⁵ Additional data was collected from public records (newspaper articles) and visual documents (photos) as well as official documents such as course resources /materials, schedules and internal publications. Multiple sources of data collection will be used for triangulation, providing additional validity of findings. As well, we sought to triangulate participant observation with a health survey to measure participant perceptions of their emotional and physical wellbeing. The Medical Outcome Study-Short Form (SF-36v2TM Health Survey), originally developed for the Rand Corporation's Health Insurance Experiment (Ware & Sherbourne, 1992; Ware et al., 2003), is used here to measure health across eight domains.

¹⁶ In 2013-2014, a total of eleven elective courses are offered including line dance, Korean traditional dance, worship dance, aerobic dance, pop song singing, harmonica ensemble (intermediate level), calligraphy, arts and crafts, sketching and drawing, conversational English (beginner level), and Smartphone 101. The courses reflect current interests of seniors in order to optimize the learning experience.

("Ban"). This homeroom system somewhat mirrors Korean school system but negotiates power dynamics differently. For each homeroom at CESC, two representatives or leaders ("Ban-jang" or leader of class) are selected by vote to assume roles and responsibilities related to their homeroom such as taking attendance, relaying messages or announcements, and facilitating group activities. In a traditional Korean school setting, two leaders of hierarchical relationship ("Ban-jang" or leader of class and "Bu-ban-jang" or sub-leader of class) are voted from the list of nominated students on the basis of academic success and these roles are regarded as a privilege. By dismantling the traditional practice of selecting leaders based on merit, since anyone can volunteer for the role, CESC preclude excluding members while ensuring that all housekeeping duties get done. This homerooms system allows students to get to know one another at a more personal level. A student of Enoch College belongs to two small groups: the homeroom and an elective class. Some opt to be a member of Enoch Choir as well. During the term, students are identified with their homeroom. When individually addressed, homeroom name is called out in conjunction with student's own name. This identification reinforces their sense of membership with the group. An important component of the CESC program is these small groups. This ensures that everyone is included and active participation is encouraged. As a group, students eat lunch together and prepare for various performances such as Festival of Youth and end of school celebration. Other bonding experiences are planned throughout the program: an all-class picnic, poetry contest, singing contest, and Parents' Day celebration¹⁷. New friendships are frequently formed. Students exchanging phone numbers, going out for coffee, and extending invitation to each other's homes are observed.

Addressing seniors as students puts all students at the same level despite of their education levels, age, sex, and other social status and appears to evoke formal schooling days filled with possibilities. Many seniors were enthusiastic and felt empowered as they were no longer labeled as 'seniors' or 'retirees' but as students with opportunities for learning, growth, and development. This is reflected in the attitudes of students who have picked up new hobbies from taking elective courses at CESC. One student added that, "I picked up to play the Harmonica at

¹⁷ Instead of having a separate day to show appreciation to mother and father, both are celebrated together in Korea on May 8th of every year.

Enoch College as an elective course for the first time and now, I am playing with my husband at the intermediate level, who played Harmonica prior to Enoch College." The song titled "So what's age got to do with anything" summarizes the overall sentiment of students succinctly. Others also showed pride in attending Enoch College and satisfaction over belonging to such a "high class group". A student remarked:

I am proud to be a member of this outstanding group of students. It boosts my self-esteem to attend the renowned Enoch College. When I first attended Enoch, I was very impressed by the quality of the teachers and classes here. I've attended number of seniors' colleges back in Korea, but the quality of classes and activities here are by far, excellent, just like the reputation.

Moreover, a sense of belonging is reinforced from the "shared" culture of Korean seniors. Seniors greet each other in the Korean way by literally asking for one's health and well-being ("An-nyung-ha-sae-yo") with bowed heads. One student commented that, "where else can I receive such a warm welcome!" This may explain why seniors take great effort to come to Enoch, despite having to commute a long distance by public transit or having additional barriers such as mobility problems. Enoch students' commitment to the program is exemplified from a student who commutes three hours to attend Enoch. She shared:

I often wake up as early as 3 am as you lose sleep when you become aged, but instead of trying to go back to sleep on days when I have school (Enoch College), I spend extra time to get ready. For fear that I might oversleep and miss the bus that leaves at 6 am because then I won't be able to make it to Enoch on time.

Emphasizing how attending Enoch is a highlight of her week, she opted to celebrate her 80th birthday few months earlier so that she could celebrate with the rest of the group during the term, treating everyone to lunch.

Shared Musical Knowledge Plays a Significant Role in the Program

The students at Enoch College share a cultural background not only in ethnicity and mother tongue¹⁸ but also in their musical knowledge, especially in terms of songs from childhood and their elementary and secondary school years. The CESC program uses music to engage students in positive social interactions and participation. Instead of asking students to become acquainted with each other, familiar greeting songs of childhood interwoven with various activities such as hi-fiving with others or bowing heads (a Korean way of greeting) made potentially awkward or dry moments of introducing oneself to a complete stranger more fun and lively.

This familiarity with the music also plays an essential role in art song class, which covers a wide range of classical songs during each term¹⁹. Most students were already familiar with classical music and the Western canon, as they had encountered them in schooling. Students were commonly seen humming along the melody, nodding to the beat and singing from the memory with eyes closed, reminiscing the past. This afforded greater flexibility in developing musicianship of students as the teacher could focus on few challenging measures before mastering each score. It only took few classes for students to improve performance of a complete song providing aesthetic fulfillment and a greater sense of achievement. Being able to sing and re-learn songs of past appears to be a meaningful experience, recreating a sense of connectedness within the group.

Health and Wellness

Healthy aging and living are central to Enoch College's mission. Elective courses offer health-promoting activities such as singing and physical activities like Taekwondo and dancing. The day starts with an assembly singing familiar songs with actions. Once a week, songs are

¹⁸ These are more visible or audible form of culture.

¹⁹ The songs selected for this semester are: "Auf Flugeln des Gesanges" by Mendelsshon, "Annie Laurie" (a traditional (Old Scottish song), "Silver Threads Among the Gold" (an American folk song), "Heidenroslein" (a German lied) and "Geu-nae" or "Swing" (a Korean art song). All songs were sung in Korean.

followed with a twenty-minute Heavenly message, full of stories and healthy tips for living. The principal²⁰ addresses the whole school with inspiring or uplifting stories of older adults such as the amateur runner still clocking marathons at over 90 years of age. In between classes, students engage in physical activities such as massaging each other's shoulders as they sing a familiar children's folk song such as "Reminiscing older brother" and turn to the opposite direction in a row when a certain Korean alphabet is sung, making it like a game that is both physically and mentally stimulating.

Health is a frequent topic of conversations among students. Most seniors suffer from chronic health conditions like arthritis or back pain. Thus, ways to improve health is a common concern for many. Seniors frequently shared health-related information during breaks and even in classes and sometimes by instructors too, which was always welcomed. For example, when the pop song class instructor shared her traditional acupuncture knowledge related to alleviating shoulder pain, which involved pressing a small section of middle finger, everyone was eager to find out the exact pressure point and tried on themselves immediately.

"Well-being lunch" at Enoch

Enoch College's "well-being" meals are well accepted by seniors as they are nutritionally-balanced with grains, meat or fish, and vegetables and fruits as a dessert. Mixed grain rice which is healthier choice than white rice, is served. Seniors often commented that meals are "excellent" and mentioned that they look forward to having a good meal with friends every week. Some even commented that it is the best meal of their week as they live alone and do not like to cook for themselves alone. This suggests that social aspect of sharing meals is also valuable and may be emotionally fulfilling for seniors.

One student's description of her daily routine provides a glimpse of significant aspect of sharing Korean-style meal. She said:

²⁰ Here too the principal (also the pastor) sets a tone of welcome and equality by injecting humor in his address.

I usually get up at around 4 to 5 am when it is still dark outside so I would read a book till sunrise. Then I take my daily walk by the trail near the lake where I live for about two hours. After returning from the walk, I roast a coffee and drink it with a piece of toast. That will be my breakfast. I don't usually have cravings for lunch, so I often skip it. For supper, maybe I will cook something simple, like vegetable pancakes (Korean style) I made with fresh chives that my friend shared with me from her garden crops. Occasionally I will put an effort to cook something for my niece who lives alone to attend the university here away from her family. Other than that, I don't like to cook for myself. The meals are always excellent so I get treated very well when I come here.

Valuing and Belonging

A core belief that guides Enoch College is love of God, abiding by the commandment: "Honor your father and your mother." In keeping with Korean traditional values²¹ as well as Christian tenets, seniors are respected. This caring infuses every aspect of the school experience. Students often commented that the "Angel's team," (the name referring to all volunteers involved in the program) serves them "with a glad and cheerful heart" in the words of one senior. Furthermore, she noted that this approach is one of the core elements that distinguishes CESC from other similar programs she had attended. A heartfelt and genuine attitude of the team of volunteers and instructors foster one of kind experience that students look forward to year after year.

Discussion and Conclusions

At CESC, there is history, reminiscence, and joy. This group of Korean seniors is survivals of Korean historical turmoil, of difficult times through Japanese rule and the Korean War. They survived a move to a foreign land where they have faced many barriers. Having made

_

²¹ Respecting elders is embedded in the name of "Seniors' College". The seniors' program in Korea dates back as early as 1977, first established by Young-nak Presbyterian church in Seoul. The term used is "Kyung-lo Dae-hak": "kyung" means respect; "lo" means old age or elders; "dae-hak" means university or college, English equivalent of Seniors' College.

the choice to come to Canada and they freely state that Canada is a good place to live. Many undertook their immigration for the sake of their children's increased opportunities.

Most of the participants in CESC could be seen as having achieved a good life in Canadian society. Membership is drawn from the professions as well as blue collar occupations. But in the process of adjustment to the new country, it was at times necessary to leave their language and familiar customs behind. Attending Enoch College may have brought them back a chance to rediscover what was lost or forgotten in the busyness of life as these seniors experience firsthand a shift in roles, moving from full careers to transitioning to a place where they pursue enjoyable activities and enhance lives with meaningful engagement with others.

CESC presents these seniors with opportunities to come full circle, even as they are now integrated into Canadian society. This program is about seniors first, followed by the component of sharing traditions. But the cultural dimensions are very important, allowing participants to experience as Korean, in Korean language, in Korean culture. They meet, and come together as a group, creating a stronger, tighter bond by sharing, laughing, exercising, singing and eating.

Perhaps if Enoch College was geographically situated in Korea, what makes this program special would be taken for granted. However, these seniors obviously value this opportunity to selectively visit the Korea of their youth, or in some cases to reimagine and reinvent a collective vision, catalyzed and cherished through their music.

Reflection

Looking back, I recall that this experience did not initially unfold as a formal study. My initial engagement with the CESC program started with my intent to offer assistance as a volunteer, extending a helping hand where it was needed most. Through my involvement, I had the privilege of working closely with elderly Koreans, allowing me to become acquainted with their lives and stories. I started to contemplate the experiences of elderly Korean immigrants who lead active lives. I began to question whether this was partially attributed to the program and its influence on their social, physical, and emotional well-being. As I became more engaged in the program, my inquiries evolved into an examination of the importance of music and music-making for the elderly participants in the program.

Inquiring into the program's context and setting, I sought to understand how the CESC program was conceived and executed. Additionally, I pondered whether this program had the potential to actualize its intended goals, specifically in enhancing the quality of life and overall wellbeing of Korean elderly participants. This contemplation extended to exploring the effectiveness of music-making engagements as a means to achieve this objective.

Another primary objective guiding this initial study was to examine and gain insights into the successful practices of the CESC program. This endeavor aimed to facilitate the potential replication of these effective strategies in other comparable contexts, encompassing but not confined to, minority older adult populations.

As the study progressed, it not only enabled me to become competent in ethnographic research but also afforded me the opportunity to identify the pertinent issues concerning immigrant older adults. Simultaneously, I became acutely aware of the scarcity of resources available to address their needs adequately. These elderly Korean individuals are confronted with the challenges of social isolation. Those without familial or community support networks, such as those provided by the Korean community or institutions like CESC, often find themselves isolated due to cultural barriers that impede social participation and accessibility. For many of them, these gatherings served as a vital platform to foster social interactions and garner much-needed support.

I am honored to be a part of this study and to have the opportunity to assist them in their areas of need. My role as a volunteer enabled me to observe them with no expectations, free from the perceptions of older adults that I may have held about Korean elders. Consequently, I was able to take additional precautions to define what 'old' means to me and to also seek the elderly Koreans' own definitions and perceptions of the term. Being immersed in the setting enabled the opportunity to observe, listen, and engage in understanding what it means to be old, as it holds significance for them, both individually and collectively as a group. I also exercised careful attention to avoid making any assumptions in my journal and focused on maintaining a fresh perspective while collecting data.

Upon reflection, I now recognize that it would have been beneficial to inquire about how participants allocate their time beyond the program when they do not have access to CESC

offerings. Delving further into their choices and attitudes regarding health-promoting behaviors, as well as examining any changes before and after participation in CESC, would have provided a broader understanding of active aging and its implications for minority older adults.

References

- Adams, K. B., Leibbrandt, S., & Moon, H. (2011). A critical review of the literature on social and leisure activity and wellbeing in later life. *Ageing and Society*, *31*(4), 683-712. https://doi:10.1017/S0144686X100
- Batt-Rawden, K. B. DeNora, T., & Ruud, E. (2005). Music listening and empowerment in health promotion: A study of the role and significance of music in everyday life of the Longterm ill. Nordic *Journal of Music Therapy*, *14*(2), 120-136. https://doi.org/10.1080/08098130509478134
- Bogdan, R. C. & Biklen, S. K. (2003). Qualitative research for education: An introduction to theories and methods (4th ed.). Allyn & Bacon.
- Bugos, J. A., Perlstein, W. M., McCrae, C. S., Brophy, T. S., & Bedenbaugh, P. H. (2007). Individualized piano instruction enhances executive functioning and working memory in older adults. *Aging and Mental Health*, *11*(4), 464-471. https://doi.org/10.1080/13607860601086504
- Carucci, C. (2012). An investigation of social support in adult recreational music ensembles. *International Journal of Community* Music, 5(3), 237–252. https://doi.org/10.1386/ijcm.5.3.237_1
- Choi, N. G. (2001). Diversity within diversity: Research and social work practice issues Asian American elders. *Journal of Human Behavior in the Social Environment*, *3*(3/4), 301-319. https://doi.org/10.1300/J137v03n03_18
- Coffman, D. (2002). Music and quality of life in older adults. *Psychomusicology*, *18*(1–2), 76–88. https://doi.org/10.1037/h0094050
- Cohen, A., Bailey, B., & Nilsson, T. (2002). The importance of music to seniors. *Psychomusicology*, 18(1–2), 89–102. https://doi.org/10.1037/h0094049
- Cohen, G. D., Perlstein, S., Chapline, J., Kelly, J., Firth, K. M., & Simmens, S. (2006). The impact of professionally conducted cultural programs on the physical health, mental health, and social functioning of older adults. *The Gerontologist*, 46(6), 726–734. https://doi.org/10.1093/geront/46.6.726
- Cohen, G. D., Perlstein, S., Chapline, J., Kelly, J., Firth, K.M., & Simmens, S. (2007). The impact of professionally conducted cultural programs on the physical health, mental health and social functioning of older adults 2-year results. *Journal of Aging*,

- *Humanities and the Arts, 1*(1-2), 5–22. https://doi.org/10.1080/19325610701410791
- Creech, A., Hallam, S., McQueen, H., & Varvarigou, M. (2013). The power of music in the lives of older adults. *Research Studies in Music Education*, *35*(1), 87–102. https://doi.org/10.1177/1321103X13478862
- Denzin, N. K., & Lincoln, Y. S. (2005). *The SAGE handbook of qualitative research* (3rd ed.). Sage Publications.
- Ferreri, L., Bigand, E., Perrey, S., Muthalib, M., Bard, P., & Bugaiska, A. (2014). Less effort, better results: how does music act on prefrontal cortex in older adults during verbal encoding? An fNIRS study. *Frontiers in Human Neuroscience*, 8(301):1-11. https://doi.org/10.3389/fnhum.2014.00301
- Goldman-Segall, R. (1998). *Points of viewing children's thinking: A digital ethnographer's journey*. Lawrence Erlbaum.
- Halpern, A. R., Kwak, S., Bartlett, J. C., & Dowling, W. J. (1996). Effects of aging and musical experience on the representation of tonal hierarchies. *Psychology and Aging*, *11*(2), 235-246. https://doi.org/10.1037//0882-7974.11.2.235
- Kuo, B. C., Chong, V., & Joseph, J. (2008). Depression and its psychosocial correlates among older Asian immigrants in North America: A critical review of two decades' research.
 Journal of Aging and Health, 20(6), 615-652. https://doi.org/10.1177/0898264308321001
- Lee, Y. (2007). The immigration experience among elderly Korean immigrants. *Journal of Psychiatric and Mental Health Nursing*, 14(4), 403–410. https://doi.org/10.1111/j.1365-2850.2007.01098.x
- MacCourt, P, Wilson, K, Tourigny-Rivard, M. (2011). *Guidelines for comprehensive mental health services for older adults in Canada*. Mental Health Commission of Canada (MHCC) Seniors Advisory Committee. https://www.mentalhealthcommission.ca/wp-content/uploads/drupal/2017-09/mhcc_seniors_guidelines_0.pdf
- McCaffrey, R. (2008). Music listening: Its effects in creating a healing environment. *Journal of Psychosocial Nursing and Mental Health Services*, 46(10), 39-45. https://doi.org/10.3928/02793695-20081001-08
- Paquette, S., & Goulet, G. M. (2014). Lifetime benefits of musical training. *Frontiers in Neuroscience*, 8, 89. https://doi.org/10.3389/fnins.2014.00089

- Statistics Canada. (2006). *A portrait of seniors in Canada*.

 https://www150.statcan.gc.ca/n1/en/pub/89-519-x/89-519-x2006001-eng.pdf?st=ZHZP48BY
- Sun, J. Buys, N., & Tatow, D. (2013b). Using participative community singing programs to improve health behaviors. In J. Sun, N. Buys, & J. Merrick (Eds.), *Health promotion:*Community singing as a vehicle to promote health (pp. 3-14). Nova Science Publishers, Inc. https://ebookcentral.proquest.com/lib/west/detail.action?docID=3022886
- Tendulkar, S. A., Hamilton, R. C., Chu, C., Arsenault, L., Duffy, K., Huynh, V., Hung, W., Lee, E., Shwuling, J., & Friedman, E. (2012). Investigating the myth of the "model minority": A participatory community health assessment of Chinese and Vietnamese adults. *Journal of Immigrant Minority Health*, 14(5), 850–857. https://doi.org/10.1007/s10903-011-9517-y
- Veblen, K. K. (2013). The tapestry: Introducing community music. In K. K. Veblen, S. J. Messenger, M. Silverman, & D. Elliott (Eds.), *Community music today* (pp. 1–9). Rowman and Littlefield. ISBN 978-1-60709-320-
 - 6. https://ebookcentral.proquest.com/lib/west/detail.action?docID=1117171
- Wang, A. (2013). Cognitive effects of music: Working memory is enhanced in healthy older adults after listening to music [Doctoral dissertation, The University of Arizona] The University of Arizona College of Medicine Phoenix.

 https://repository.arizona.edu/handle/10150/281781
- Ware, J. E., & Sherbourne, C. D. (1992). The MOS 36-item short-form health survey (SF-36®): I. Conceptual framework and item selection. *Medical Care*, 30(6), 473-483.
- Ware, J. E., Jr., Kosinski, M., & Dewey, J.E. (2003). *Version 2 of the SF-36 Health Survey*. Quality Metric Incorporated.
- Ware, J. E., & Sherbourne, C. D. (1992). The MOS 36-item short-form health survey (SF-36®): I. Conceptual framework and item selection. *Medical Care*, 30(6), 473-483.
- World Health Organization. (2002). *Active ageing: A policy framework*. World Health Organization Press.

 https://apps.who.int/iris/bitstream/handle/10665/67215/WHO_NMH_NPH_02.8.pdf?sequence=1&isAllowed=y
- World Health Organization. (2007). Global age-friendly cities: a guide. World Health

Organization. https://apps.who.int/iris/handle/10665/43755

World Health Organization. (2014). *Ageing and life courses*. http://www.who.int/ageing/en/ Yinger, O. S. (2014). Adapting choral singing experiences for older adults: The implications of sensory, perceptual and cognitive changes. *International Journal of Music Education*, 32(2), 203-212. https://doi.org/10.1177/0255761413508064

Chapter Four: Enoch Senior's College for Korean Immigrant Seniors: Quality of Life Effects ²²

H. Elisha Jo,²³ Jungsun S. Jo,²⁴ Kari K. Veblen,² and Patrick J. Potter²⁵

Résumé

Cette recherche à méthode mixte a évalué l'impact d'un programme pour la communauté culturelle d'aînés immigrants coréens en examinant les bénéfices pour la qualité de vie des participants liée à leur santé (HR-QOL) et leur bien- être en général. Bien que la programmation communautaire soit de plus en plus reconnue comme un moyen de promouvoir la santé et la participation active des individus dans la sphère sociale, peu d'informations sont disponibles concernant l'utilisation et l'impact de tels programmes chez les immigrants aînés des minorités visibles. Dans cette étude, 79 participants ont répondu au questionnaire SF-36v2 à deux reprises dans le cadre de l'évaluation de l'impact du programme du Canada Enoch Senior's College (CESC) concernant le HR-QOL et leur bien-être. Des améliorations statistiquement significatives ont été observées sur les plans de la santé physique et mentale, notamment dans les dimensions liées à la douleur corporelle, au rôle émotionnel et aux limitations de rôle en raison de problèmes émotionnels. Les données qualitatives provenant des entretiens avec les participants ont confirmé les résultats positifs du sondage, par des améliorations aux niveaux de la vie sociale et de la santé des aînés. Ces résultats suggèrent que le programme CESC contribue à qualité de vie et au bien-être des aînés coréens participants. Cette étude permettra de soutenir des programmes communautaires culturels similaires.

²² This manuscript was published in Canadian Journal on Aging / La Revue canadienne du vieillissement, doi:10.1017/S0714980818000211. Elisha Jo conceived of the study. Writing was undertaken collaboratively with co-authors, with Elisha contributing 60 % and co-authors 40 %.

²³ Don Wright Faculty of Music, Western University

²⁴ Independent Consultant, Mississauga

⁴ Schulich School of Medicine & Dentistry, Western University

Abstract

A community-based program is increasingly recognized as promoting health and active social participation in one's life, yet information is lacking about the use and impact of such programs among immigrant visible minority seniors. This mixed-method research evaluated the impact of a cultural community program for Korean immigrant seniors by examining participants' health-related quality of life (HR-QOL) benefits and overall well-being. In this study, 79 participants completed the SF-36v2 questionnaire twice to assess the impact of Canada Enoch Senior's College (CESC) program on their HR-QOL and well-being. Statistically significant improvement in physical and mental health domains was observed: bodily pain and role limitations due to emotional problems. Qualitative data from participants' interviews supported the survey findings with positive contributions in health and social arenas of seniors' lives. These results suggest that the CESC program contributes to quality of life and well-being of Korean senior participants and supports similar community- based cultural programs.

Introduction

People aged 60 or older represent the fastest growing portion of the world's population, increasing at a rate of 3.25 per cent per year (United Nations Department of Economics and Social Affairs (UN DESA), 2015). In Canada, seniors will account for an increasingly larger share of the total population, growing from 15.6 per cent or over 6 million in 2014 to 23 per cent by 2030 or over 9.5 million as the majority of the "baby boom" generation (born between 1946 and 1965) reaches age 65 and older (Government of Canada, 2014). In con-junction with this aging population, the Canadian demographic landscape is changing rapidly with a significant increase in ethnocultural diversity. Immigrants accounted for more than 6.5 million or 20 per cent of the Canadian population in 2006, and this proportion is expected to continue to rise, reaching between 25 to 28 per cent of the population in 2031 (Statistics Canada, 2010).

As a result of recent changes in the immigration sources from traditional Anglo-Saxon European countries to non-European countries, the proportion of visible minority¹ persons among immigrants is also projected to increase to 71 per cent in 2031, compared to 54 per cent in 2006 (Statistics Canada, 2010). For example, the pro- portion of Asian-born immigrants has steadily increased from 14 per cent in 1981 to 41 per cent in 2006 while the proportion of

immigrants born in European countries has declined from 67 per cent to 37 per cent (Statistics Canada, 2010). Within two decades, more than two out of three immigrants living in Canada are projected to belong to a visible minority group with neither English nor French as their mother tongue (Statistics Canada, 2010). This change will have important implications in programs and policies that support growing ethnic senior populations.

Because advanced age is associated with the likelihood of chronic health conditions (Decady & Greenberg, 2014), improving older adults' quality of life while maintaining their health and functional independence has become a priority issue in both the public and private sectors (Public Health Agency of Canada, 2010; Ramage-Morin et al., 2010). In Canada, federal, provincial, and territorial governments have identified healthy aging as a policy focus which has necessitated the creation of supportive environments as a means to achieve this goal (Ndegwa, 2011; Plouffe & Kalache, 2011; Public Health Agency of Canada, 2010). To date, all 10 provinces have joined and have implemented the Age-Friendly Community Initiatives, developed by the World Health Organization in 2006, to promote and develop policies, services, and structures to support seniors to lead active, socially engaged, independent lives (Public Health Agency of Canada, 2015). Increased community engagement is sought through multiple streams, such as in-service provision and the promotion of local projects through government funding (Ndegwa, 2011; Public Health Agency of Canada, 2010; 2015).

These government and provincial initiatives are especially promising as ethnic organizations have a greater role to play in reaching out to vulnerable minority immigrant populations who face barriers to existing programs and services. Multiple barriers to accessing health and social care exist, most notably lack of fluency in English, cultural beliefs and practices, and immigrant status itself (Ahmad et al., 2004; Jang et al., 2005; Lai & Chau, 2007; Pourat et al., 2000; Weerasinghe, 2012). Such barriers not only hinder immigrant older adults' ability to seek and receive necessary care but also reduce social interactions (Kang et al., 2013; Y. S. Kim, 2013; Lai & Surood, 2008; Schaie & Pietrucha, 2000). These barriers and other stressors, such as acculturation, make elderly immigrants particularly vulnerable to poor psychological and physical health (Kang et al., 2009; Min et al., 2005; Mio et al., 2009). In particular, older Korean immigrants have been identified as a high-risk group with higher levels

of depression and suicidal ideation than other senior groups (Jang & Chiriboga, 2010; Min et al., 2005; Mui & Kang, 2006).

In light of recent government efforts through community involvement in developing supportive environments to enable healthy aging, there is a need to better under- stand the impact that community-based programs can have on the well-being of immigrant seniors (National Advisory Council on Aging, 2005). The role of ethnic communities is vital in achieving psychological and physical well-being in their new, adopted culture (Ahmad et al., 2004; Barrio et al., 2007; Lai & Surood, 2008; Stewart et al., 2011; Weerasinghe & Numer, 2011). The National Advisory Council on Aging (2005) has recommended proactive approaches to prevent social isolation, as cultural and linguistic barriers can be reduced through innovative programs to engage ethnic minority seniors more fully in group activities. Community-based programs targeting ethnic seniors have been effective in reaching out to various ethnic groups and have had numerous positive impacts on the participants' health outcomes (Hau et al., 2016; Y. S. Kim, 2013; E. Kim, 2016; Koehn et al., 2016). One noteworthy program is the Seniors Support Services for South Asian Community (S⁴AC) program implemented in collaboration with the non-profit local United Way agency in British Columbia to improve access to existing facilities for South Asian older adults.

In their case study of over a hundred Punjabi-speaking seniors, Koehn et al. (2016) found health benefits that accrued to seniors attending exercise classes offered by the S⁴AC program. The reported health benefits not only pertain to physical health such as reduced hypertension and improved balance and flexibility but also extended to mental health and well-being through enhanced social interactions resulting from the program (Koehn et al., 2016). Another community-based exercise and health education program, the Boston, Massachusetts-based Healthy Habits Program (Hau et al., 2016) targeting elderly Chinese immigrants, found clinically meaningful improvements in mobility and cognition of the participants at the end of the 6-month intervention period. As well, the researchers found statistically significant improvements in executive function, lessened depressive symptoms, and reduced perceived disability. More importantly, 78 per cent of Chinese seniors achieved the goal of performing exercises three or more times a week during a 6-month period (Hau et al., 2016).

As exemplary as these programs are, the majority of them are structurally implemented with an extensive collaboration with community partners and/or institutions. However, limited research has been conducted on the programs within grassroots from the ethnic community itself. One example of an ethnic community that developed its own program can be found in the work of Y. S. Kim (2013) who investigated Korean senior schools in North America. His qualitative study of a Korean senior school in the United States reported psychological benefits from the perspective of 10 Korean seniors (Y. S. Kim, 2013). Attending senior school positively influenced their psychological well-being by providing the participants opportunities to meet friends and helping them to deal with the psychological distress associated with immigration (Y. S. Kim, 2013).

There are many references to senior schools or "silver colleges" offered by Korean ethnic churches across North America (Hurh & Kim, 1990; Hwang, 2008; E. Kim, 2016; Lee & An, 2013; Sin & Hirsch, 2013), and yet the impact of senior schools has not been fully explored by both quantitative and qualitative means. To Korean immigrants, the ethnic churches are "a core source of various social interactions" (Hwang, 2008, p. 214) that provide social network and resources via a wide range of services and scheduled programs including senior schools that offer various educational classes and recreational activities to elderly Koreans (Y. S. Kim, 2013; E. Kim, 2016; Sin & Hirsch, 2013). The senior programs have a wider reach as they are not only for church congregations but also extend to Korean seniors in the community as a whole (Y. S. Kim, 2013; E. Kim, 2016). Thus, these sites can serve as a venue for further exploration into the community of Korean seniors.

Koreans accounted for 0.5 per cent of the total Canadian population or over 160,000 people in 2011 (Statistics Canada, 2013a). Of the total Korean population, seniors make up about 7 per cent (Statistics Canada, 2013b). Similar to other immigrant groups, most Koreans settled in Canada's three census metropolitan areas: Toronto, Montreal, and Vancouver. Toronto has by far the largest Korean community, accounting for 42 per cent of the total Korean population in Canada in 2011, where ethnic enclaves ("Korean towns") offer cultural resources and support (Lindsay, 2007). Across the Greater Toronto Area² alone, there are at least 11 Korean ethnic churches that offer senior colleges advertised through local Korean newspapers (Garden Kyohae ..., 2014; Gunganghago ..., 2016).

Korean communities may be a good case in point for research into the influence of cultural community pro- grams on the physical and psychological well-being of ethnic immigrant seniors. First, a relatively short immigration history of Koreans in Canada in the 1960s means that most, if not all, seniors are first-generation (Korean-born) immigrants (A. H. Kim et al., 2012; Lindsay, 2007). The findings will be highly relevant to foreign-born minority immigrants including a growing number of late-life immigrants who are adjusting to a new country with a culture and language distinctively different from their own. Second, by understanding the experience of elderly Korean immigrants as a singular population, researchers can better understand the ethnic diversity of minority senior populations and fill a gap in defining and analysing the context and cultural uniqueness of Koreans as an individual ethnic group (Hwang, 2008; National Advisory Council on Aging, 2005). Most of the published gerontology research has focused on a single ethnic group, mainly Chinese, or Asian as a single group (Durst, 2005), and yet variations in subgroups of Asian seniors exist as evidenced by their utilization of local services.

For example, a study in the Greater Vancouver Regional District found that Korean seniors are less likely to use local services and amenities than are Chinese seniors and rely on ethnic churches as a resource (Hwang, 2008). Research also shows that Korean seniors have a strong desire to maintain ethnic ties and surround themselves with familiar cultural touchstones, including in their choice of ethnic foods (Hurh & Kim, 1990; Lee, 2007; Son & Kim, 2006). Moreover, there are generational differences within the Korean population. According to the Ethnic Diversity Survey by Statistics Canada (2003), first-generation immigrants were less likely than second- or third-generation Korean-Canadians to participate in sports clubs or service clubs or to be involved in charitable organizations. The only areas in which participation rates were higher among first-generation immigrants, compared to subsequent generations, were religious-affiliated groups and ethnic or immigrant associations (Statistics Canada, 2003). These findings highlight the importance of these organizations and their associated services for Korean immigrant seniors.

This study stems from an ethnographic case study of Canada Enoch Senior's College (CESC) that explored the experience of Korean immigrant older adults in a culturally sensitive community program in the Greater Toronto Area (Jo, Veblen, & Potter, 2014). Our earlier

explorative study, utilizing a quantitative survey, suggested positive health effects of the program on participants. The purpose of this current investigation is to further evaluate the impact of CESC in improving the health-related quality of life (HR-QOL) of older adults using both quantitative and qualitative measures and to explore the nature of the outcomes and how they were achieved.

Background on the CESC

Canada Enoch Senior's College is a cultural community program for Korean immigrant older adults centred in a Korean ethnic church in the Greater Toronto Area. It was the first of its kind offered in Ontario for Korean seniors and has served approximately 3,000 seniors over 13 years and currently enrolls an average of 150 participants each term. The program is offered twice a year, in spring and fall, and each term lasts between 8 and 11 weeks. CESC is offered free-of-charge to all seniors and is run mostly by volunteers. Except for the director of the program, who is a part-time pastor at the church in which CESC operation is one of her responsibilities, all staff are volunteers. Staff consists of class instructors, kitchen staff, and others who help with day-to-day activities from class setup to cleanup at day's end. Many of the volunteers are also members of the church. The number of volunteers per term varies depending on the number of elective classes offered during the term or number of kitchen staff available to prepare meals in a given week. On average, there are about 12 to 14 instructors, about 8 to 10 kitchen staff, plus three extra staff for remaining activities. Some instructors are current or retired professionals sharing their skills or expertise with seniors, and their time commitment during the college term is one to two hours.

The CESC program runs weekly and is conducted in Korean. Each day is divided into four time slots, beginning at 10:00 a.m. and finishing at 2:30 p.m., with a traditional Korean style congregate meal at lunchtime. The morning session is organized as an assembly, consisting of singing and short lectures on topics such as Korean history, literature, and life stories with occasional special events, like birthday celebrations. The afternoon session is followed by an elective class and small group gathering. Typical electives include singing classes (art songs, popular songs, and folk songs), dancing (aerobics, Korean traditional dance, worship dance, and line dance), harmonica ensemble, arts and crafts, calligraphy, saxophone, smartphone 101, and

Chinese medicine. The electives can vary depending on seniors' interests and availability of instructors.³

Methods

This study used a mixed methodology to evaluate the effects of the CESC program on the HR-QOL and well-being in senior Korean participants. The evaluation of the program included documentation, qualitative inquiry, and a quantitative survey. We used semi- structured interviews with the CESC participants, self- reported health measure questionnaires, and field notes as our main data collection tools.

The quantitative evaluation used a repeated measures design with no control group. The baseline measures (Time 1) occurred during the first two weeks and the end line measures (Time 2) occurred during the last two weeks of each term. Although a true pre- and post- test design would have been ideal, based on the CESC schedule and director's recommendation, a decision was made on pragmatic grounds to give study participants an option to complete the questionnaire at home and return it the following week to allow for flexibility and reduce potential barriers to participation since the first week at the college is very busy with orientation. For similar reasons, we administered the post-test survey during the next-to-last week of the program. This means that there were some variations as to how many weeks participants would have participated in the program (i.e., 7 to 10 weeks) based on the timing of questionnaire completion.

The Short Form 36 version 2 (SF-36v2), a measure of health status derived from items used in the Medical Outcomes Study, was what we used to assess the HR- QOL of study participants as it has proven reliable and valid in a variety of populations (Walters et al., 2001; Ware & Gandek, 1998; Ware & Sherbourne, 1992), including older Korean adults (Han & Lee, 2009; Han et al., 2004). The SF-36 measures eight domains of HR-QOL, including physical function (PF), role-physical (RP), bodily pain (BP), general health (GH), vitality (VT), social function (SF), role-emotional (RE) and mental health (MH). These were scored using a scale ranging from 0 to 100, with higher scores indicating a higher HR-QOL. The SF-36v2 survey was translated into Korean by a researcher who is proficient in Korean and English, and the translation and adaptation procedures focused on maintaining conceptual equivalence while

remaining culturally appropriate (Wagner et al., 1998). For example, the questionnaire used the term "four weeks"; this was translated to "one month" to mirror the more common usage in Korean culture. After completion of the first draft, the translation was evaluated by a bilingual Korean speaker to ensure that the interpretation and translation of the survey items were clear and appropriate. The researcher distributed the questionnaires and was available to answer clarifying questions on three different occasions – in the morning before classes, during lunch, and after classes – in order to gather complete data from as many participants as possible. Some people chose to complete the questionnaire at home and brought it back the following week. The participation was voluntary, and ultimately, 79 CESC participants out of 165 eligible participants (25 in the spring term; 54 in the fall term) completed the SF-36v2 questionnaire on two occasions and provided informed consent to permit the use of their data for evaluation purposes.

In the spring term, we distributed the questionnaire to a subset of participants (25) as we originally conceived this study as an exploratory pilot. The survey was administered again in the fall term to increase sample size, so we distributed the questionnaire to all participants except those who had completed the survey in the spring term. The study sample consisted of the same population with little turnover between the terms (< 10%). Out of 150 participants in the fall term, 125 were eligible to participate excluding 25 participants who had already completed the survey beforehand. In total, 79 participants completed the questionnaire on the first occasion (Time 1), and 69 completed the questionnaire on the second occasion (Time 2). However, only 54 pairs from the fall term were included in the analysis due to missing identifiers or having completed the questionnaire only once. To analyse the data, paired-samples *t*-tests let us determine the statistical significance of the difference between Time 1 and Time 2 for each HR-QOL measure.

The qualitative inquiry component of this study drew upon ethnography of the CESC on the basis of fieldwork conducted during two consecutive terms between April and October 2014 by the researcher in both a participant and non-participant capacity. Additionally, the researcher conducted semi-structured interviews with CESC participants. Adherence to the ethnographic study necessitated that the researcher be immersed in the setting (Bresler, 1995; Creswell, 2007; Fetterman, 1998; Merriam, 2009). Through serving as a volunteer, the lead researcher was able to establish rapport with participants (Bell, 2005; Yin, 2009). The researcher observed all

sessions including lunch breaks and the short informal social gatherings that occurred at the end of each session, with a weekly rotation of elective classes and small group meetings. Multiple sources of information were used as data, including field notes written immediately after observation or participation in classrooms, informal conversations with the participants, and self-reflection (Bernard, 2011; Denzin & Lincoln, 2005). The data supplemented quantitative findings and aided in cross-checking.

A total of 11 participants were interviewed to provide more detailed accounts of the program's impact as perceived by the participants. The time and place of interviews were chosen by the participants, and most participants chose to meet at CESC before or after the program. The duration of each interview ranged from 15 minutes to two hours. For some participants who chose to meet during lunch hours, subsequent interviews were necessary to continue and conclude their previous conversations. On average, the total interview time per interviewee was an hour. In addition, on these occasions some of the individual interviews became a small group conversation when one or two members sitting nearby joined in to share their accounts. Conversations also continued throughout the term as participants offered their reflections and accounts. Informal conversations were recorded in the field notebook immediately after the conversation took place. The use of email made it possible for the researcher and participants to maintain contact even after the term ended. All interviews and email correspondences were conducted in Korean. Individual interviews were audio-recorded and transcribed verbatim, followed by translation into English. All email communications pertaining to this research were saved and translated into English.

The data were analysed using content analysis (Green & Thorogood, 2009), beginning with open coding of the transcripts followed by an iterative process of inference and analysis where emerging and recurrent topics were highlighted and combined or collapsed to a distinct and encapsulating theme. We ensured the quality and rigour of the data collection and analysis stages by employing a variety of strategies. Participants reviewed themes that emerged from data analysis by a member-checking process, which ensured that the data was interpreted correctly and in the appropriate context (Green & Thorogood, 2009). We maintained the credibility/reliability of the data by a back-translation process whereby the lead researcher, fluent in English and Korean, translated a random selection of paragraphs back to Korean. This was

followed by validation of the quality and conceptual equivalence of paragraphs by the researcher (Guillemin et al., 1993). The final stage of the research sought to capture the detailed context through compiling field notes to create a thick description of each theme, supported by verbatim quotes from the participants, as recommended by Green and Thorogood (2009). This research has been reviewed and approved by a Western University Ethics Review Board.

Results

Quantitative Evaluation

A total of 79 out of 165 eligible CESC participants completed both pre- and postquestionnaires. The mean age of the participants was 74.1 years, and 77 per cent were female (n = 61) and 23 per cent were male (n = 18). All participants were from the Greater Toronto Area and were first-generation (Korean-born) immigrants with Korean as the mother-tongue language. Analyses across the entire sample using one-tailed t-tests were employed since the previous explorative study indicated positive health effects of participants. Table 1 presents mean differences and standard deviations for each of the eight SF-36v2 HR- QOL domains (physical functioning, role-physical, bodily pain, general health, vitality, social functioning, roleemotional or role limitations due to emotional problems, and mental health). Significant differences were found for the two domains of bodily pain and role-emotional. The mean scores for bodily pain (p = .014) and role-emotional (p = .032) for the seniors were significantly higher at Time 2 or after 7 to 10 weeks of program participation. Higher scores refer to better self-rated QOL with regard to each subcategory. The respondents experienced less body pain and had less functional role limitations resulting from emotional issues after participation in the program. Additionally, physical functioning showed marginal, although not significant, improvement. There were no significant differences between Time 1 and Time 2 in terms of role limitation due to physical health problems, general health perceptions, general mental health, vitality, or social functioning in the group.

Table 1: Health impacts of the CESC program: Mean differences in HR-QOL of Korean immigrant seniors at Time 1 and Time 2

Health-Related Quality-of-Life Measure	Time 1 (n = 79) Mean (SD)	Time 2 (n = 79) Mean (SD)	<i>p</i> Value
Physical Functioning	77.1 (18.8)	78.9 (18.9)	0.089
Role-Physical	69.8 (23.7)	71 .4 (22.2)	0.250
Bodily Pain	74.0 (23.4)	78.2 (18.7)	0.014*
General Health	62.6 (21.5)	61.9 (19.8)	0.306
Mental Component			
Vitality	64.2 (22.3)	67.4 (21.4)	0.121
Social Functioning	83.1 (18.9)	83.2 (18.1)	0.444
Role-Emotional	75.2 (24.7)	79.6 (22.2)	0.032*
Mental Health	73.9 (17.5)	73.1 (18.8)	0.470

Note. HR-QOL = health-related quality of life; SD = standard deviation.

Qualitative Inquiry

We conducted a total of 11 interviews (six females and five males). The age range for these participants was 70 to 87 years. The following results are discussed in relation to the three main themes that emerged from the data analysis: (1) CESC as promoting health and vitality; (2) CESC as reducing social isolation and exclusion by fostering a renewed perspective on aging and self; and (3) CESC as providing social support and a sense of belonging.

CESC as Promoting Health and Vitality

Participation at CESC helped Korean older adults live healthier lives by increasing their activity levels and motivating them to take better care of themselves. Several of the participants spoke of how dance classes helped them become physically fit. One participant described the benefits of taking a line dance class:

So I started to take the dance class because our legs become weaker as we get older. At first, the steps were very confusing, and it took quite a bit of time for me to follow them correctly. And then there was music on top of it. So you can see the challenges, but it turned out to be good for my health. My legs have strengthened a lot since the class.

Participants also reported improvement in their appetite and energy levels. Several of the participants, who live alone or with their spouses without their families, stated a lack of

^{*} p < .05.

motivation or inspiration to cook for only one or two persons. One woman described her cooking and eating habits:

I don't usually feel like cooking much, so I have a cup of black coffee and a piece of toast in the morning and skip lunch. For supper, I usually make something simple. When I do cook, it is usually for my niece who lives alone away from her family to attend a university nearby.

However, she found herself cooking more frequently at home and adding more variety in her diet due to improved appetite since her participation at CESC. In this regard, the weekly lunch served at CESC may have also played a role. People spoke of getting ideas from the lunch menus and commented that they were eating better at home as a result. Referring to the positive experience of sharing meals with other Koreans, another participant commented, "The lunch here is the best meal of the week!" Several participants echoed her comment about how they enjoy the lunchtime at CESC and stated that their appetites had improved since the school started.

Through CESC, participants were motivated to take better care of themselves since the program provided them with social interaction. One senior mentioned how she ran into another CESC member at a hair salon just a day before CESC's fall term began; they were both getting "ready" for the school. Another participant described how she finds CESC as an opportunity to dress up. She explained that "there are less social occasions to dress up these days, so I pull out my best ensemble when I go to Enoch College." This attitude was noted by other participants as well. One senior explained: "Every week, both men and women look polished. You look around and see others paying attention to their looks and attire, so you also become more conscious as to how you look, in a good way." In this way, the program helped to restore vitality in the lives of older Korean adults.

CESC as Fostering Renewed Perspectives on Aging and Self

Elderly Koreans in the program spoke of increasing feelings of isolation and exclusion and their perceptions of the diminished role and value in society. One senior expressed that she felt like "a second-hand citizen" and another said that he often felt "out of place", not being able to keep up with the fast-changing society. One later-life immigrant (who migrated to Canada at age 66) pointed out:

Unlike in Korea where there are multiple places for retired seniors to hang out, as immigrants in Canada we have no place to go, no one to talk to, and nothing to do. So it's important that we have Enoch College where we can meet once a week and socialize with other Koreans.

Even those who have been active members of their com- munity also spoke about changing attitudes towards older adults. "I've always been volunteering in the kitchen, but I began feeling awkward amongst volunteers because of my age," one senior explained and continued: "Since I've become one of the eldest, I've noticed a younger group of volunteers being more self-conscious around me. I am starting to feel like I might be a burden to them and no longer sure of my place." Other participants also spoke of invisible tension, especially in a mixed-generation setting that impeded participation, further contributing to their diminished social circles and engagement. This also relates to common perceptions and attitudes shared by the participants. One senior explained, "It is a virtue of old people to step back and let the younger ones take the lead."

However, participants stated that this was not the case at CESC. Several participants mentioned that they were less restricted by their age and social status. One participant explained, "Here at Enoch College, every- one is the same." Another senior elaborated:

We all become like children, freed from the concerns of world. It is as if we go back in time to our child- hood when we played together regardless of our gender or social status. We have a great time enjoying activities together in which we are reminded of how grateful we are to be living.

This friendly atmosphere appears to encourage participants to explore. One senior described how she got involved in art class:

I've always wanted to study art, but never got a chance. In a way, I had given it up because I felt I was too old and embarrassed to start something. I was thinking to myself, 'What would be the use?' However, here at Enoch College, from watching others I've realized that it is never too late to try, so I took up my courage and have been taking art

for two consecutive terms. I found Enoch College to be the place for fulfilling my longlost dream.

A similar account was related by another participant who was taking a saxophone class; she explained that this was her very first lesson on a musical instrument. Others also spoke of how they expanded their horizons by getting out of their comfort zones, like singing in front of an audience for the first time in life or writing a memoir and getting it published. At CESC, Korean seniors are provided with opportunities to explore and extend their interest, which has helped promote self- awareness and insight in participating seniors.

The overall change in attitude of the group is expressed succinctly in a poem written by one participant:

I won't give up now just 'cause I am old
Joy comes from serving and sharing
My fellow friends at Enoch College
Let's shout to the world, our life begins at 80!

CESC as Providing Emotional Support and a Sense of Belonging

All participants expressed how they look forward to coming to CESC every week and rarely missed an opportunity to do so. CESC attendees suffer from various chronic illnesses including diabetes, arthritis, and cancer as well as age-related health declines such as vision and hearing loss. Health-related issues did not prevent them from attending CESC. One participant, who came despite having sprained her back the day before, explained:

As long as I am able, I will be here. Attending Enoch College is good for me, it makes me feel better. Now, I will have to take an extra precaution and won't be able to participate in some of the activities but I won't miss Enoch.

Participants described how they prioritize CESC. One senior commented, "I have a pretty busy schedule with several commitments with family and church, but I make sure that Thursday is free for my friends at Enoch College." Being able to communicate in Korean and meeting other Koreans appear to be important parts of CESC that older adults value. Another participant

remarked, "Where else can we meet Korean people?" Others also commented that attending CESC and seeing familiar faces alone is often enough to make them feel better and helps them forget about worries of life. Still another CESC student summed up the experience by saying, "We have such a good time together at Enoch College; we are busy laughing, learning, and sweating from exercising – the day goes by before you know it."

Through the CESC program, participants extended social ties and networks that went beyond the confines of school terms or activities. All of the participants described how they stay connected with other CESC members throughout the year in one form or another, many using cell phones and text messages. These connections were likely strengthened by small groups which may be unique to CESC. Referred to as home- room ("Ban" in Korean), each student participant was assigned to a small group where two representatives or leaders ("Ban-jang" in Korean) assumed responsibilities related to their homeroom such as taking attendance and facilitating homeroom/group activities. This system was modeled after the traditional primary and secondary school system in Korea. In the CESC setting, it allowed for better communication and created a sense of group belonging as well as opportunities for participants to get to know one another at a more personal level. For example, Banjang would follow up by a phone call when a member was absent the previous week. (See earlier exposition of CESC program; Jo et al., 2014.)

Participants also found enjoyment from sharing stories or YouTube videos with each other via email. One participant explained: "I keep a list of friends from Enoch College whom I often send or forward emails that contain either interesting stories or videos and useful living tips." Another participant added: "Occasionally we meet up with each other to exercise at a nearby mall and have a cup of coffee afterwards." These opportunities can also lead to meeting new people, such as when an elderly Korean was introduced to a local hiking club by a fellow CESC member.

Participation at CESC not only enhances social connections but also provides emotional support for older adults. One woman described a time that she came to CESC straight from an eye specialist's appointment despite blurred vision due to eye drops used in the examination: "I was planning to go home initially after the doctor's appointment, but I changed my mind. I knew

if I go home I would be alone and feel blue so I came here." Several participants also shared how they were able to overcome difficulties in life, such as bereavements, family issues, or suffering ill health, through their association with CESC. One participant, who suffered depression after a sudden loss of her husband, described her experience:

I remember my first day at Enoch College. Another participant, whom I've never met before, approached me to say hello as I was a new face here. For some reason, I told her about my husband. Without any words she held my hands tightly and I felt like she understood my pain. Her genuine gesture opened up my heart and since then I've tried to offer the same kind of support to others who lost their loved ones.

Supported by the care and friendship experienced at CESC, another participant described how he found strength to overcome the pain and loneliness that troubled him, which often kept him up all night. He stated, "Pain shed light out of my life but I regained hope and joy through Enoch College. It gave me something I look forward to."

Participants not only found solace through CESC relationships but also from various CESC activities. Specifically, group singing seems to have a positive impact on mood and enhances a sense of belonging in CESC members. One participant shared her experience: "When we sing songs from our schooling days, even though it was long ago and forgotten, quite surprisingly the songs come back to you. It brings us back to the joyful time when we were in school." Another senior shared a similar experience: "The art song class is my favourite class. Singing the songs from past is soothing and I find my mind at peace when we sing together." This psychological comfort is also noted in the attitudes of other participants. Another senior observed: "You can see people closing their eyes listening to the tunes and humming when we sing together because we are familiar with them. They remind us of good old days of our youth." Through reminiscence and strengthening bonds, music seems to be a source of comfort among older Koreans.

Discussion

This study sought to evaluate the impact of the CESC program on Korean immigrant older adults' health and well-being, and to explore, through the use of a mixed-methods approach, the nature of the outcomes and how they were achieved. The results of the quantitative

evaluation demonstrated that participation at CESC was associated with improvements in HR-QOL for physical and mental health domains. Statistically significant results were found for two variables, bodily pain and role-emotional or role limitations due to emotional problems. In the case of self-rated physical health, the program reduced perceptions of bodily pain among respondents and led to a marginally significant improvement on physical function. For the mental health domain, respondents had fewer issues or limitations in carrying out daily activities due to emotional status. The results of qualitative evaluation align strongly with these findings except for the social function variable. Overall, the CESC program encouraged Korean senior participants to live actively and helped them feel physically and emotionally stronger while reducing social isolation.

There are multiple ways in which the program could have influenced the positive health outcomes observed in the participants, including lessened bodily pain and improved roleemotional. One way is through increased activity levels. As a result of participating in the CESC, seniors reported that they were generally more active whether it was through program-related activities like dancing classes or social gatherings/outings outside of CESC. Previous research has shown that individuals who engage in higher levels of physical activity tend to have higher HR-QOL scores (Anokye et al., 2012; Hand et al., 2012; Moore-Harrison et al., 2009; Okamoto et al., 2007). In particular, moderate- intensity, low-impact aerobic activity, including dancing and brisk walking (similar to the exercise program at CESC), have also shown to reduce pain related to musculoskeletal conditions that are common in older adults, such as osteoarthritis, and to improve physical function, moods, and quality of life (Krampe et al., 2014; Hughes et al., 2004; Okamoto et al., 2007). It is interesting to note that physical functioning did not change despite many references to improvements. One explanation might be that the participants are already a high-functioning group. A comparison to the norm data shows that the physical function (PF) score for the group was above the population norm. In Canada, the norm for PF is 75.5 (SD 22.2) for adults aged 65 to 74 (Hopman et al., 2000), whereas the study group had an average of 77.1 (SD 18.8) at baseline examination and 78.9 (SD 18.9) at final examination (p =.089). This high baseline score may have created a ceiling effect, thus making any improvements difficult to observe.

Physical activity also may affect HR-QOL through the effects of self-esteem. Previous studies show that exercise is an effective means to enhance self-esteem in older adults (Li et al., 2002; McAuley et al., 2005; Opdenacker et al., 2009). One possible explanation is that physical activity can help maintain a sense of self-worth for those who have experienced a decline in perceived self-worth due to changes in health or physical abilities. Self-esteem, in turn, has a positive association with QOL.

A comparative study of QOL between Korean older adults and Korean-American older adults found that, for both groups, self-esteem and the ability to perform activities of daily living or instrumental activities of daily living (ADLs; IADLs) were strong predictors of QOL (S. Kim et al., 2009). Increased opportunities for self-achievement also contributes to self-esteem. Participants spoke of trying new activities/ exercises at CESC and how they were able to overcome various physical and mental challenges. One senior described how she gained confidence from taking an aerobics class for the first time:

It was so confusing at first because I've never done anything like it, moving so fast you know. I thought of giving it up, but slowly I followed the dancing sequences one by one and eventually I was able to do them all. It made me feel so good that I did it! I learned I can do anything once I put my mind to it.

This participant shared that she joined a choir for the first time prompted by her positive experience from aerobics, illustrating that success breeds other successes.

The CESC program contributed positively to Korean older adults' well-being through social support. The individual interviews found an emphasis on the social aspect of the program – for example, how important it was for participants to be able to interact with other Koreans and to communicate in Korean. Previous studies on arts-based community programs also found social connectedness or sense of community to be an important part of a program that contributed to participants' well-being and the main reason for continued participation by the seniors (Phinney et al., 2014; Teater & Baldwin, 2014). Similarly, the social aspect and learning were the central parts of the CESC experience which likely contributed to a high participation/retention rate as well as to participants' health. Previous research has suggested that interaction with members of the same ethnic group is positively associated with psychological

well-being among ethnic minorities (Frable et al., 1998; Sanchez & Garcia, 2009). Members of the same ethnicity can easily develop group cohesion and experience psychological comfort (Otten & Moskowitz, 2000) because of familiarities in customs and culture.

The field note data made broad-ranging reference to the emotional benefits experienced by the participants: People were generally happier; they stated that they felt appreciated and valued. Many seniors called Enoch College "heaven", reflecting their Christian beliefs, because of the joyful time they had together. They spoke of a supportive and encouraging environment, from the volunteers to instructors to participants, which helped create a positive experience and also put newcomers at ease. In particular, people spoke of music as something dear to them: bringing back childhood memories, comforting them, lightening their moods, increasing connectedness, and strengthening a sense of membership. Psychological comfort experienced by the participants can also be explained through music (Hays et al., 2002; Lally, 2009) helping to manage emotions, change moods, increase positive feelings, experience spirituality (Hays & Minichiello, 2005c; Lee, 2013; von Lob et al., 2010), improve bonding and co-operation, and create a sense of belonging (Bailey & Davidson, 2002; 2005; Blaine & Fels, 2003; Clift et al., 2010; Livesey, et al., 2012; Silber, 2005).

Although the program demonstrated a significant influence on the measure of emotional well-being, namely, role-emotional, there was no improvement in the mental health variable. It is possible that this relates to emotions at a deeper level perhaps requiring a longer duration and/or targeted intervention to make a difference. For Asian older adults in general, identifying depressive moods and anxiety carry a stigma (Baker et al., 2015; Conner et al., 2010; Gary, 2005). When identifying and rating self-health with psychological labeling such as "depression", Koreans are likely to refrain from expressing it⁴ (G. Kim, 2010; Kim et al., 2011b; Sin et al., 2004). Our participants also confirmed similar views towards mental health, referring to a feeling of uneasiness answering questions on the mental health domain. One commented: "Confronting that I've been depressed or whatever, makes me feel like I am depressed or I will get depressed." Moreover, questions associated with negative feelings such as nervousness, fatigue, and depression – as well as positive feelings like serenity, energy, and happiness – may be less relevant to older adults as inconsistent responses may result from a mis- interpretation of negative or positive questions (e.g., Did you feel worn out? Did you have a lot of energy?) (S. H.

Kim et al., 2013). This might help explain why the vitality scale did not show significant change in the participants, which also may have been affected by seasonal variations in mood and behaviour (Jia & Lubetkin, 2009).

Although there were many references to increased social contacts or relationships and ties within Enoch College community and beyond in the interviews, the social functioning measure did not show significant change over time. This may be due to cultural differences in conceptualization of social functioning in Korean seniors. Similar to the findings reported for other Asian countries (Gandek et al., 1998; Lim et al., 2008; Tseng et al., 2003; Wang et al., 2011). S. H. Kim et al.'s (2013) evaluation of the SF-36 in a Korean population suggested that the social functioning variable should be interpreted with caution. Compared to individualistic Western society, Asian cultures with collectivist values tend to conceptualize social roles and social functioning differently. For example, avoidance of social activities due to health problems may be less acceptable for Koreans than for people from other Western countries (S. H. Kim et al., 2013). Indeed, it was observed throughout the school year that weekly attendance to CESC was determined by self-motivation rather than having optimal health, as explained by high participation rates and interview data where participants chose to attend CESC despite having pains from injuries or other health issues.

Although this study provides support for the health and social benefits of the CESC program, there are limitations. Unlike a randomized controlled trial, our study design has known limitations related to the internal validity of results. These limitations include seasonal variations in health status and mood, lifestyle factors, and self-reporting and self-selection biases. For example, there may be an important difference between those who completed the questionnaire on two occasions versus those who did not. The reasons for non-participation were not sought out and would need to be considered in future studies. The detailed socio-demographic information on the study participants was not collected as pre-testing of the survey questionnaire with a sample of seniors found it unduly long, making it a barrier to participation. For future studies, this should be better planned. For a qualitative research component, we used a relatively small convenience sample of participants who volunteered to be interviewed but whose experience may not have been representative of all CESC participants. However, the results of this evaluation confirm findings of earlier research (Jo et al., 2014), and we have gone a step

further through the use of quantitative measures of health and well-being in the context of reallife settings.

In conclusion, this study provides evidence of the effectiveness of CESC in improving quality of life of Korean immigrant seniors. Considering that CESC is rooted in the ethnic community and run by volunteers, it has great potential – as an economical and effective program – for conferring health and social benefits to other elderly community-living adults. Partnership with ethnic organizations can be a valuable starting point as they may be better positioned to provide more culturally relevant and accessible services to minority immigrant seniors through the knowledge of shared cultural heritage and established networks. Future research might focus on (1) programmatic components of CESC such as an ethnic organization's capacity, commitment, and potential route for partnerships; and (2) an exploration of perspectives from the volunteers and organizers of this program. Because relatively few grassroots programs like this exist, it would be useful to identify essential implementation elements that con-tributed to the program's overall success in order to replicate the program in various cultural contexts. Similar research on the community-based cultural programs throughout the minority immigrant senior population would provide further methodological context to validate this evidence. Nevertheless, there is no doubt that CESC has made a lasting impact on the life of Korean immigrant seniors. In the words on one CESC member: "Many activities at Enoch College may seem a bit childish, but for us this is a loving nest that fills us with life's joy and happiness that are meaningful because we create our lasting memories together."

Reflection

Following the initial ethnography that introduced this study, the second and third studies emerged in close succession. As I formulated the questions for this dissertation, I reconsidered the approach and adjusted the order of the questions:

- 2) How does participating in the community program in the community program affect health and active aging among elderly Korean participants?
- 3) What does music mean and how important is it in the lives of elderly Koreans?

Therefore, the chapters are now organized in the order of the quantitative study, followed by the narrative study as the final piece.

Since these elements grew together and nested within an overarching theme, it seemed logical to employ a mixed methods approach. Greater insight can be gleaned from integrating two methods, qualitative and quantitative methodologies, into one study than from either approach alone (Creswell & Plano Clark, 2007). Moreover, I found that combining and linking quantitative and qualitative databases reduced personal researcher biases. The combined results serve to validate the data. Moreover, the narrative and statistical elements together create a more compelling story.

The study used a survey to measure health-related quality of life (HRQOL), which encompasses the ways in which health impacts an individual's subjective perception of health and well-being (Hays & Morales, 2001). The Short-Form Health Survey was chosen for several reasons:

- 1) The Medical Outcomes Study 36-item Short-Form Health Survey (SF-36) is one of the most widely studied and validated HRQOL instruments. It is utilized as a translated and tested generic measurement across the globe.
- 2) This survey is designed to be applicable across all diseases, conditions, and medical interventions, as well as a wide range of populations (Bullinger et al., 1998; Garrett et al., 2002; Ware & Gandek, 1998; Ware & Sherbourne, 1992).
- 3) The SF-36 is also considered the reference instrument by most researchers.

The study utilized a Korean version of the Short-Form Health Survey version 2.0 (SF-36v2) (Jang et al., 2018), specifically tested and validated for measuring HRQOL among Korean elderly individuals (Han et al., 2004). In adapting this self-reporting survey, I was guided by Dr. Potter, whose contribution significantly improved the study by enhancing the qualitative findings. Dr. Potter (MD, FRCP(C)) is an Emeritus Professor at Schulich School of Medicine. His specialties lie in rehabilitation, spinal cord injury, and medical issues affecting professional musicians. Additionally, he is a performing musician himself. Dr. Potter's expertise in medicine and wellness, coupled with his passion for music-making, provided a quantitative perspective to

the project. This expert believed that the SF-36 survey could gauge underlying physical and mental health domains to address the research questions.

There are two other generic HRQOL surveys that I have identified, which could enhance the findings if used in conjunction with the SF-36 survey. They are 1) the World Health Organization Quality of Life Assessment (WHOQOL) and 2) the European Quality of Life Questionnaire – 5 Dimensions (EQ-5D). Either option could have been viable, however, given that the Short-Form Health Survey (SF-36) was adapted for Korean participants, this survey appeared to be the most promising choice. For potential follow-up research, both of these surveys could be administered, as both are widely accepted across various cultures and populations.

1) the World Health Organization Quality of Life Assessment (WHOQOL)

The World Health Organization Quality of Life Assessment (the WHOQOL) -100, was developed collaboratively in diverse cultural settings to establish a reliable, valid, and responsive quality of life assessment applicable across diverse cultures (WHOQOL Group, 1994; 1998). The WHO defines quality of life as a subjective assessment rooted in cultural, social, and environmental contexts. With a focus on participants' perceptions of variables impacting an individual's quality of life, numerous field centers worldwide took part in developing an instrument using a cross-cultural approach that is genuinely international (WHOQOL Group, 1998). The WHQOL-100 is designed with 100 questions across six domains: physical capacity, psychological, level of independence, social relationships, environment, and spirituality/religion/personal beliefs. It measures HRQOL as well as overall QOL, encompassing domains such as "environment" and "spirituality" (WHOQOL Group, 1994, 1998). This section provides a space for individuals to express their personal and spiritual beliefs, as many turn to religion, personal beliefs, and spirituality for comfort, meaning, a sense of belonging, and purpose in life.

The World Health Organization Quality of Life Assessment would be useful in determining respondents' religious or spiritual orientations. For instance, Confucianism, renowned for its moral and ethical doctrines, as well as socio-political philosophy, holds significant prevalence in Korea (Ro, 2018). Alongside Confucianism, Buddhism has exerted a

vital influence as a national religion during the Koryo (918-1392) and the unified Shilla (668-935) periods, as well as in the Three Kingdoms era (57 B.C.E.-668 C.E.). In addition, Daoism is latent in Korean consciousness as an influence on Shamanism, which is a part of Korean indigenous culture (Jung, 2000).

2) European Quality of Life Questionnaire – 5 Dimensions (EQ-5D)

The European Quality of Life Questionnaire – 5 Dimensions (EQ-5D), developed by the EuroQol Group, has been translated into over 130 different language versions [www.euroqol.org]. It is also widely used for HRQOL evaluation, having been tested and utilized in both clinical trials and general population surveys. The health status is defined and measured across five dimensions: mobility, self-care, usual activities, pain/discomfort, and anxiety/depression (Brooks, 1996; Rabin & de Charro, 2001). Each dimension is scored using three levels of severity: no problems, some (moderate) problems, and severe problems. Designed for self-completion, respondents also rate their overall health between 0 (the worst imaginable health state) and 100 (the best imaginable health state) on the day of the interview using a vertical, graduated Visual Analogue Scale (VAS).

HRQOL studies have utilized more than one health survey tool to measure the subjective perspective of health and wellbeing. Both the SF-36 and the EQ-5D surveys are considered brief and easily manageable tools for assessing HRQOL in the general population (Javanbakht et al., 2015; Mayo et al., 2011; Nordlund et al., 2005).

Notes

- 1. The term *visible minorities* is defined in the Canadian Employment Equity Act, as "persons, other than Aboriginal peoples, who are non-Caucasian in race or non-white in colour" (Employment Equity Act, 1995). Statistics Canada's identification of visible minorities mainly includes the following groups: Chinese, South Asian, Black, Arab, West Asian, Filipino, Southeast Asian, Latin American, Japanese, and Korean (Statistics Canada, n.d.).
- The Greater Toronto Area is the metropolitan area surrounding and including the city of Toronto, Ontario, Canada. This includes four surrounding regional municipalities: Durham, Halton, Peel, and York.

- 3. See Jo, Veblen, and Potter's study (2014) for a detailed program structure and components.
- 4. Sometimes identified as "Hwa-Byung", a Korean term for anger syndrome caused by repression of emotions, can be found in Min and Suh's study (2010).

References

- Ahmad, F., Shik A., Vanza, R., Cheung, A. M., George, U., & Stewart, D. E. (2004). Voices of South Asian women: Immigration and mental health. *Women's Health*, 40(4), 113-130. https://doi:10.1300/J013v40n04_07
- Anokye, N. K., Trueman, P., Green, C., Pavey, T. G., & Taylor, R. S. (2012). Physical activity and health related quality of life. *BMC Public Health*, *12*(1), 624–624. https://doi: 10.1186/1471-2458-12-624
- Bailey, B.A., & Davidson, J.W. (2002). Adaptive characteristics of group singing: Perceptions from members of a choir for homeless men. *Musicae Scientiae*, 6(2), 221–256. https://doi.org/10.1177/102986490200600206
- Bailey, B. A., & Davidson, J. W. (2005). Effects of group singing and performance for marginalized and middle-class singers. *Psychology of Music*, *33*(3), 269–303. https://doi.org/10.1177/0305735605053734
- Baker, S., Marryshow, T., Poteau, S., & Sutin, A. (2015, May 15-17). Exploring the stigma of depression among older adults: Results from a community survey. *American Geriatrics Society* annual meeting, National Harbor, MD, U.S.A.

 https://med.fsu.edu/sites/default/files/userFiles/file/AGS%202015%20Depression%20Poster%20FINAL%205-7-15.pdf
- Barrio, C., Palinkas, L. A., Yamada, A., Fuentes, D., Criado, V., Garcia, P., & Jeste, D. V. (2007). Unmet needs for mental health services for Latino older adults: Perspectives from consumers, family members, advocates, and service providers. *Community Mental Health Journal*, 44(1), 57–74. https://doi.org/10.1007/s10597-007-9112-9
- Bell, J. (2005). Doing your research project: A guide for first-time researchers in education, health, and social science (4th ed.). Open University Press.
- Bernard, H. R. (2011). Research methods in anthropology: Qualitative and quantitative approaches (5th ed.). AltaMira Press.
- Blaine, T., & Fels, S. (2003). Collaborative musical experiences for novices. *Journal of New Music Research*, 32(4), 411–428. https://doi.org/10.1076/jnmr.32.4.411.18850
- Bresler, L. (1995). Ethnography, phenomenology and action research in music education. Quarterly Journal of Music Teaching and Learning, 6(3), 4–16.

- Clift, S., Nicols, J., Raisbeck, M., Whitmore, C., & Morrison, I. (2010). Group singing, wellbeing and health: A systematic mapping of research evidence. *UNESCO Observatory*, 2(1). http://www.abp.unimelb.edu.au/unesco/ejournal/pdf/clift-paper.pdf
- Conner, K. O., Copeland, V. C., Grote, N. K., Koeske, G., Rosen, D., Reynolds, C. F., & Brown, C. (2010). Mental health treatment seeking among older adults with depression: The impact of stigma and race. *The American Journal of Geriatric Psychiatry*, *18*(6), 531–543. https://doi.org/10.1097/JGP.0b013e3181cc0366
- Creswell, J. (2007). *Qualitative inquiry and research design: Choosing among five approaches* (2nd ed.). Sage Publications. ISBN 978-1-4129-1606-6
- Decady, Y., & Greenberg, L. (2014). Health at a glance: *Ninety years of change in life*expectancy. Statistics Canada, Health Statistics Division (Catalogue no. 82- 624-X).

 https://www150.statcan.gc.ca/n1/en/pub/82-624-x/2014001/article/14009-eng.pdf?st=6HX5CQO_
- Denzin, N. K., & Lincoln, Y. S. (2005). *The SAGE handbook of qualitative research* (3rd ed.). Sage Publications.
- Durst, D. (2005). Aging amongst immigrants in Canada: Population drift. *Canadian Studies in Population*, 32(2), 257–270. https://doi.org/10.25336/P6JK5Q
- Employment Equity Act. (1995, December 15). *Employment Equity Act*. (S.C. 1995, c. 44).

 Justice laws website, Government of Canada. https://laws-lois.justice.gc.ca/eng/acts/e-5.401/page-1.html
- Fetterman, D. M. (1998). Ethnography: Step by step. (2nd ed.). Sage Publications.
- Frable, D. E. S., Pratt, L., & Hoey, S. (1998). Concealable stigmas and positive self-perceptions: Feeling better around similar others. *Journal of Personality and Social Psychology*, 74(4), 909–922. https://doi.org/10.1037/0022-3514.74.4.909
- Gandek, B., Ware, J. E. Jr., Aaronson, N. K., Alonso, J., Apolone, G., Bjorner, J., ... Sullivan, M. (1998). Tests of data quality, scaling assumptions, and reliability of the SF- 36 in eleven countries: Results from the international quality of life assessment project. *Journal of Clinical Epidemiology*, 51(11), 1149-1158. http://dx.doi.org/10.1016/S0895-4356(98)00106-1

- Garden Kyohae Silver College Bom Hakgi Geagang [Garden Church Silver College at open for spring term]. (2014, April 2). *Korea Daily Toronto*. Retrieved from http://www.cktimes.net/board_read.asp?boardCode=board_it&boardNumber=1497
- Gary, F. A. (2005). Stigma: Barrier to mental health care among ethnic minorities. *Issues in Mental Health Nursing*, 26(10), 979–999. https://doi.org/10.1080/01612840500280638
- Government of Canada. (2014). *Government of Canada Action for seniors report*.

 https://www.canada.ca/en/employment-social-development/programs/seniors-action-report.html
- Green, J., & Thorogood, N. (2009). *Qualitative methods for health research* (2nd ed.). Sage Publications.
- Guillemin, F., Bombardier, C., & Beaton, D. (1993). Cross-cultural adaption of health-related quality of life measures: Literature review and proposed guidelines. *Journal of Clinical Epidemiology*, 46(12), 1417–1432.
- Gunganghago Aereumdawn Sangwhal Weehea. [For a Healthy Good-Quality Life], Toronto Korean Church Senior's College Guide. (2016, April 5). *The Korea Times Daily*. Retrieved from http://www.koreatimes.net/ ArticleViewer/Article/57310
- Han, C. W., Lee, E. J., Iwaya, T., Kataoka, H., & Kohzuki, M. (2004). Development of the Korean version of short-form 36-item health survey: Health related QOL of healthy elderly people and elderly patients in Korea. *The Tohoku Journal of Experimental Medicine*, 203(3), 189–194. https://doi.org/10.1620/tjem.203.189
- Han, C. W., & Lee, E. J. (2009). Use of structural equation modeling to test construct validity of the SF-36 health survey among community-dwelling elderly in Korea. *Journal Korea Academy of Care Management*, 4, 45–59.
- Hand, B. D., Cavanaugh, S., Forbes, W., Govern, J., & Cress, M. E. (2012). Changes in health-related quality of life and functional fitness with exercise training in older adults who attend senior centers. *Activities, Adaptation & Aging*, 36(1), 29-54. https://doi.org/10.1080/01924788.2011.647530
- Hau, C., Reid, K. F., Wong, K. F., Chin, R. J., Botto, T. J., Eliasziw, M., Bermudez, O. I., & Fielding, R. A. (2016). Collaborative evaluation of the healthy habits program: An effective community intervention to improve mobility and cognition of Chinese older

- adults living in the U.S. *The Journal of Nutrition, Health & Aging*, 20(4), 391–397. https://doi.org/10.1007/s12603-015-0623-8
- Hays, T., Bright, R., & Minichiello, V. (2002). The contribution of music to positive aging: A review. *Journal of Aging and Identity*, 7(3), 165–175. https://doi.org/10.1023/A:1019712522302
- Hays, T., & Minichiello, V. (2005). Older people's experience of spirituality through music. *Journal of Religion, Spirituality & Aging*, 18(1), 83–96.

 https://doi.org/10.1300/J496v18n01_06
- Hopman, W. M., Towheed, T., Anastassiades, T., Tenenhouse, A., Poliquin, S., Berger, C.,
 Joseph, L., Brown, J. P., Murray, T. M., Adachi, J. D., Hanley, D. A., &
 Papadimitropoulos, E. (2000). Canadian normative data for the SF-36 health survey.
 Canadian Medical Association Journal, 163(3), 265–271. PMID: 10951722; PMCID: PMC80287.
- Hughes, S. L., Seymour, R. B., Campbell, R., Pollak, N., Huber, G., & Sharma, L. (2004).
 Impact of the fit and strong intervention on older adults with osteoarthritis. *The Gerontologist*, 44(2), 217–228. https://doi.org/10.1093/geront/44.2.217
- Hurh, W. M., & Kim, K. C. (1990). Religious participation of Korean immigrants in the United States. *Journal for the Scientific Study of Religion*, 29(1), 19–34.
- Hwang, E. (2008). Exploring aging-in-place among Chinese and Korean seniors in British Columbia, Canada. *Ageing International*, *32*(3), 205–218. https://doi.org/10.1007/s12126-008-9014-3
- Jang, Y., & Chiriboga, D. A. (2010). Living in a different world: Acculturative stress among Korean American elders. *The Journals of Gerontology: Psychological Sciences*, 65B(1), 14–21. https://doi.org/10.1093%2Fgeronb%2Fgbp019
- Jang, Y., Kim, G., & Chiriboga, D. (2005). Health, health- care utilization, and satisfaction with services: Barriers and facilities for older Korean Americans. *Journal of the American Geriatrics Society*, *53*, 1613–1617. https://doi.org/10.1111/j.1532-5415.2005.53518.x
- Jia, H., & Lubetkin, E. I. (2009). Time trends and seasonal pat- terns of health-related quality of life among U.S. adults. *Public Health Report*, 124(5), 692–701. https://doi.org/10.1177/003335490912400511

- Jo, H. E., Veblen, K. K., & Potter, P. J. (2014). Korean immigrant seniors' music making in an ethno-cultural com- munity program in Canada. In M. L. Cohen (Ed.), *CMA XIV:*Listening to the world: Proceedings from the International Society for Music Education (ISME) 2014 seminar of the Commission for Community Music Activity (pp. 25–32).

 https://edoc.ku.de/id/eprint/19768/1/CMA_XIV_Full_Proceedings_FINAL_MC.pdf
- Kang, S., Basham, R., & Kim, Y. J. (2013). Contributing factors of depressive symptoms among elderly Korean immigrants in Texas. *Journal of Gerontological Social Work*, *56*(1), 67–82. https://doi.org/10.1080/01634372.2012.734369
- Kang, S., Domanski, M. D., & Moon, S. S. (2009). Ethnic enclave resources and predictors of depression among Arizona's Korean immigrant elders. *Journal of Gerontological Social Work*, 52(5), 489–502. https://doi.org/10.1080/01634370902983153
- Kim, G. (2010). Measuring depression in a multicultural society: Conceptual issues and research recommendations. *Hallym International Journal of Aging*, *12*(1), 27–46. http://dx.doi.org/10.2190/HA.12.1.c
- Kim, Y. S. (2013). Ethnic senior schools, religion, and psychological well-being among older Korean immigrants in the United States: A qualitative study. *Educational Gerontology*, *39*(5), 342-354. https://doi.org/10.1080/03601277.2012.700826
- Kim, E. (2016). Social service outreach programs using elderly Korean immigrants' church ties. *Journal of Social Service Research*, 42(1), 130–142. http://dx.doi.org/10.1080/0148
 8376.2015.1087448
- Kim, S., Jeon, E., Sok, S. R., Oh, H. K., & Kim, K. (2009). Quality of life of Korean and Korean American older adults: A comparison. *Journal of Gerontological Nursing*, *35*(6), 28. https://doi.org/10.3928/00989134-20090428-06
- Kim, G., DeCoster, J., Chiriboga, D. A., Jang, Y., Allen, R. S., & Parmelee, P. (2011).
 Associations between self-rated mental health and psychiatric disorders among older adults: Do racial/ethnic differences exist? *The American Journal of Geriatric Psychiatry*, 19(5), 416–422. https://doi.org/10.1097/JGP.0b013e3181f61ede
- Kim, A. H., Noh, M. S., & Noh, S. (2012). Introduction: Historical context and contemporary research. In S. Noh, A. H. Kim & M. S. Noh (Eds.), *Korean immigrants in Canada:*Perspectives on migration, integration, and the family (pp. 3–18). University of Toronto Press. https://doi.org/10.3138/9781442690387-004

- Kim, S. H., Jo, M., & Lee, S. (2013). Psychometric properties of the Korean short form-36 health survey version 2 for assessing the general population. *Asian Nursing Research*, 7, 60–66. https://doi.org/10.1016/j.anr.2013.03.001
- Koehn, S., Habib, S., & Bukhari, S. (2016). S4AC Case study: Enhancing underserved senior's access to health promotion programs. *Canadian Journal on Aging / La Revue Canadienne Du Vieillissement*, 35(1), 89-102. https://doi.org/10.1017/S0714980815000586
- Krampe, J., Wagner, J. M., Hawthorne, K., Sanazaro, D., Wong-Anuchit, C., Budhathoki, C., Lorenz, R. A., & Raaf, S. (2014). Does dance-based therapy increase gait speed in older adults with chronic lower extremity pain: A feasibility study. *Geriatric Nursing (New York)*, 35(5), 339–344. https://doi.org/10.1016/j.gerinurse.2014.03.008
- Lai, D. W. L., & Chau, S. B. (2007). Effects of service barriers on health status of older Chinese immigrants in Canada. *Social Work*, 52(3), 261–269. https://doi.org/10.1093/sw/52.3.261
- Lai, D. W. L., & Surood, S. (2008). Service barriers of Chinese family caregivers in Canada. *Journal of Gerontological Social Work*, 51(3–4), 315–336.

 https://doi.org/10.1080/01634370802039650
- Lally, E. (2009). The power to heal us with a smile and a song: Senior well-being, music-based participatory arts and the value of qualitative evidence. *Journal of Arts and Communities I*(1), 25–44. https://doi.org/10.1386/jaac.1.1.25_1
- Lee, E. O. (2007). Religion and spirituality as predictors of well-being among Chinese American and Korean American older adults. *Journal of Religion, Spirituality & Aging, 19*(3), 77-100. https://doi.org/10.1300/J496v19n03_06
- Lee, A. H. C. (2013). The musical experiences, motivations, and preferences of positive ageing in Taiwan. *International Journal of Community Music*, 6(1), 79–92. https://doi.org/10.1386/ijcm.6.1.79_1
- Lee, E. O., & An, C. H. (2013). Faith-based community support for Korean American older adults. *Social Work & Christianity*, 40(4), 446-459. https://search-ebscohost-com.proxy1.lib.uwo.ca/login.aspx?direct=true&db=asn&AN=92582199&site=ehost-live
- Li, F., Harmer, P., Chaumeton, N. R., Duncan, T. E., & Duncan, S. C. (2002). Tai chi as a means to enhance self-esteem: A randomized controlled trial. *The Journal of Applied Gerontology*, 21(1), 70–89. https://doi.org/10.1177/073346480202100105

- Lim, L. L., Seubsman, S., & Sleigh, A. (2008). Thai SF-36 health survey: Tests of data quality, scaling assumptions, reliability and validity in healthy men and women. *Health and Quality of Life Outcomes*, 6(52). https://doi.org/10.1186/1477-7525-6-52
- Lindsay, C. (2007). *Profiles of ethnic communities on Canada, The Korean community in Canada 2001* (Catalogue No. 89-621-XIE). Statistics Canada. http://www.statcan.gc.ca/pub/89-621-x/89-621-x2007014-eng.pdf
- Livesey, L., Morrison, I., Clift, S., & Camic, P. (2012). Benefits of choral singing for social and mental wellbeing: Qualitative findings from a cross-national survey of choir members.

 Journal of Public Mental Health, 11(1), 10–26.

 https://doi.org/10.1108/17465721211207275
- McAuley, E., Elavsky, S., Motl, R. W., Konopack, J. F., Hu, L., & Marquez, D. X. (2005). Physical activity, self-efficacy, and self-esteem: Longitudinal relationships in older adults. *The Journals of Gerontology. Series B, Psychological Sciences and Social Sciences*, 60(5), P268–P275. https://doi.org/10.1093/geronb/60.5.P268
- Merriam, S. B. (2009). *Qualitative research: A guide to design and implementation*. Jossey-Bass. https://ebookcentral.proquest.com/lib/west/detail.action?docID=1662771
- Min, J. W., Moon, A., & Lubben, J. E. (2005). Determinants of psychological distress over time among older Korean immigrants and non-Hispanic white elders: Evidence from a two-wave panel study. *Aging & Mental Health*, 9(3), 210–222. https://doi.org/10.1080/13607860500090011
- Min, S. K., & Suh, S. Y. (2010). The anger syndrome hwa- byung and its comorbidity. *Journal of Affective Disorders*, 124(1–2), 211–214. https://doi.org/10.1016/j.jad.2009.10.011
- Mio, J. S., Barker-Hackett, L., & Tumambing, J. 2009). *Multicultural psychology: Understading our diverse communities* (2nd ed.). McGraw-Hill Higher Education.
- Moore-Harrison, T. L., Johnson, M. A., Quinn, M. E., & Cress, M. E. (2009). An evidence-based exercise pro- gram implemented in congregate-meal sites. *Journal of Physical Activity and Health*, 6, 247–251. https://doi.org/10.1123/jpah.6.2.247
- Mui, A. C., & Kang, S. K. (2006). Acculturation stress and depression among Asian immigrant elders. *Social Work*, *51*(3), 243–255. https://doi.org/10.1093/sw/51.3.243
- National Advisory Council on Aging. (2005). Seniors on the margins: Seniors from ethnocultural minorities. Minister of Public Works and Government Services Canada.

- $\underline{http://seniorspolicylens.ca/Root/Materials/}$
- Seniors%20From%20Ethnocultural%20Minorities%20 NACA%202005.pdf
- Ndegwa, S. (2011). *Initiatives for healthy aging in Canada* [Environmental Scan Issue 17].

 Canadian Agency for Drugs and Technologies in Health.

 https://www.cadth.ca/media/pdf/Initiatives_on_ Healthy_Aging_in_Canada_es-17_e.pdf
- Okamoto, N., Nakatani, T., Morita, N., Saeki, K., & Kurumatani, N. (2007). Homebased walking improves cardiopulmonary function and health-related QOL in community-dwelling adults. *International Journal of Sports Medicine*, 28(12), 1040–1045. https://doi.org/10.1055/s-2007-965073
- Opdenacker, J., Delecluse, C., & Boen, F. (2009). The longitudinal effects of a lifestyle physical activity intervention and a structured exercise intervention on physical self-perceptions and self-esteem in older adults. *Journal of Sport & Exercise Psychology*, *31*(6), 743–760. https://doi.org/10.1123/jsep.31.6.743
- Otten, S., & Moskowitz, G. B. (2000). Evidence for implicit evaluative in-group bias: Affect-biased spontaneous trait inference in a minimal group paradigm. *Journal of Experimental Social Psychology*, 36(1), 77. https://doi.org/10.1006/jesp.1999.1399
- Phinney, A., Moody, E. M., & Small, J. A. (2014). The effect of a community-engaged arts program on older adults' well-being. *Canadian Journal on Aging / La Revue canadienne du vieillissement*, 33(3), 336–345. https://doi.org/10.1017/S071498081400018X
- Plouffe, L. A., & Kalache, A. (2011). Making communities age friendly: State and municipal initiatives in Canada and other countries. *Gaceta Sanitaria*, 25, 131–137. https://doi.org/10.1016/j.gaceta.2011.11.001
- Pourat, N., Lubben, H., Yu, H., & Wallace, S. (2000). Perceptions of health and use of ambulatory care: Differences between Korean and White elderly. *Journal of Aging and Health*, *12*(1), 112–134. https://doi.org/10.1177/089826430001200106
- Public Health Agency of Canada (PHAC). (2010). The chief public health officer's report on the state of public health in Canada, 2010: Growing older adding life to years.

 Government of Canada. https://www.canada.ca/content/dam/phac-aspc/migration/phac-aspc/cphorsphc-respeacsp/2010/fr-rc/pdf/cpho_report_2010_e.pdf

- Public Health Agency of Canada. (2015). *Age-friendly communities*. Government of Canada. https://www.canada.ca/en/public-health/services/health-promotion/aging-seniors/friendly-communities.html
- Ramage-Morin, P. L., Shields, M., & Martel, L. (2010). Health- promoting factors and good health among Canadians in mid- to late life. *Health Reports/Statistics Canada, Canadian Centre for Health Information = Rapports Sur La Santé/ Statistique Canada, Centre Canadien d'Information Sur La Santé*, 21(3), 45–53.
- Sanchez, D. T., & Garcia, J. A. (2009). When race matters: Racially stigmatized others and perceiving race as a biological construction affect biracial people's daily well-being. *Personality and Social Psychology Bulletin*, 35(9), 1154–1164. https://doi.org/10.1177/0146167209337628
- Schaie, W., & Pietrucha, M. (2000). *Mobility and transportation in the elderly*. Springer. ISBN: 0826113095
- Silber, L. (2005). Bars behind bars: The impact of a women's prison choir on social harmony. *Music Education Research*, 7(2), 251–271. https://doi.org/10.1080/14613800500169811
- Sin, M. K., LoGerfo, J., Belza, B., & Cunningham, S. (2004). Factors influencing exercise participation and quality of life among elderly Korean Americans. *Journal of Cultural Diversity*, 11, 139–145.
- Sin, M., & Hirsch, A. (2013). Health care utilization and hyper-tension management in community-based elderly Korean immigrants. *Journal of Immigrant Minority Health*, *15*(6), 1090–1095. https://doi.org/10.1007/s10903-012-9721-4
- Son, G., & Kim, H. (2006). Culturally familiar environment among immigrant Korean elders.

 *Research and Theory for Nursing Practice, 20(2), 159–171.

 https://doi.org/10.1891/rtnp.20.2.159
- Statistics Canada. (n.d.). *Definitions, data sources and methods: Visible minority of person*. https://www23.statcan.gc.ca/imdb/p3Var.pl?Function=DEC&Id=45152
- Statistics Canada. (2003). *Ethnic diversity survey: Portrait of a multicultural society* (Catalogue no. 89-593-XIE). https://www150.statcan.gc.ca/n1/en/pub/89-593-x/89-593-x2003001-eng.pdf?st=Zqp1Tc0v

- Statistics Canada. (2013a). *National Household Survey (NHS) Profile, Canada, 2011* (Catalogue No. 99- 010-X2011027). <a href="https://www12.statcan.gc.ca/nhs-enm/2011/dp-pd/dt-td/Rp-eng.cfm?LANG=E&APATH=3&DETAIL=0&DIM=0&FL=A&FREE=0&GC=0&GID=0&GK=0&GRP=0&PID=105470&PRID=0&PTYPE=105277&S=0&SHOWALL=0&SUB=0&Temporal=2013&THEME=95&VID=0&VNAMEE=&VNAMEF=
- Statistics Canada. (2013b). 2011 National Household Survey: Data tables (Catalogue No. 99-010-X2011030). <a href="https://www12.statcan.gc.ca/nhs-enm/2011/dp-pd/dt-td/Rp-eng.cfm?LANG=E&APATH=3&DETAIL=0&DIM=0&FL=A&FREE=0&GC=0&GID=0&GK=0&GRP=0&PID=105392&PRID=0&PTYPE=105277&S=0&SHOWALL=0&SUB=0&Temporal=2013&THEME=95&VID=0&VNAMEE=&VNAMEF=
- Stewart, M., Shizha, E., Makwarimba, E., Spitzer, D., Khalema, E. N., & Nsaliwa, C. E. (2011). Challenges and barriers to services for immigrant seniors in Canada: "You are among others but you feel alone." *International Journal of Migration, Health and Social Life*, 7(1), 16–32. https://doi.org/10.1108/17479891111176278
- Teater, B., & Baldwin, M. (2014). Singing for successful ageing: The perceived benefits of participating in the golden oldies community-arts programme. *British Journal of Social Work*, 44(1), 81–99. https://doi.org/10.1093/bjsw/bcs095
- Tseng, H. M., Lu, J. F. R., & Gandek, B. (2003). Cultural issues in using the SF-36 health survey in Asia: Results from Taiwan. *Health and Quality of Life Outcomes*, 1(72), 1-9. http://dx.doi.org/10.1186/1477-7525-1-72
- United Nations Department of Economic and Social Affairs, Population Division. (2015). World population prospects: The 2015 revision, key findings and advance tables. (Working Paper no. ESA/P/WP.241).
 - https://population.un.org/wpp/publications/files/key_findings_wpp_2015.pdf
- von Lob, G., Camic, P., & Clift, S. (2010). The use of singing in a group as a response to adverse life events. *International Journal of Mental Health Promotion*, *12*(3), 45–53. https://doi.org/10.1080/14623730.2010.9721818

- Wagner, A. K., Gandek, B., Aaronson, N. K., Acquadro, C., Alonso, J., Apolone, G., Bullinger, M., Bjorner, J., Fukuhara, S., Kaasa, S., Leplege, A., Sullivan, M., Wood-Dauphinee, S., & Ware, J. E. Jr. (1998). Cross- cultural comparisons of the content of SF-36 translations across 10 countries: Results from the IQOLA Project. *Journal of Clinical Epidemiology*, 51(11), 925–932. https://doi.org/10.1016/S0895-4356(98)00083-3
- Wang, R., Wu, C., Ma, X., Zhao, Y., Yan, X., & He, J. (2011). Health-related quality of life in Chinese people: A population-based survey of five cities in China. *Scandinavian Journal of Public Health*, *39*(4), 410–418. https://doi.org/10.1177/1403494810395817
- Walters, S. J., Munro, J. F., & Brazier, J. E. (2001). Using the SF- 36 with older adults: A cross-sectional community-based survey. *Age and Ageing*, 30(4), 337–343. https://doi.org/10.1093/ageing/30.4.337
- Ware, J. E., & Sherbourne, C. D. (1992). The MOS 36-item short-form health survey (SF-36®):

 I. Conceptual framework and item selection. *Medical Care*, 30(6), 473-483.
- Ware, J. E. Jr., & Gandek, B. (1998). Overview of the SF-36 Health Survey and the International Quality of Life Assessment (IQOLA) project. *Journal of Clinical Epidemiology*, 51(11), 903–912. https://doi.org/10.1016/S0895-4356(98)00081-X
- Weerasinghe, S. (2012). Inequities in visible minority immigrant women's healthcare accessibility. *Ethnicity and Inequalities in Health and Social Care*, *5*(1), 18–28. https://doi.org/10.1108/17570981211286750
- Weerasinghe, S., & Numer, M. (2011). A life-course exploration of the social, emotional and physical health behaviours of widowed South Asian immigrant women in Canada: Implications for health and social programme planning. *International Journal of Migration, Health and Social Care*, 6(4), 42–56. https://doi.org/10.5042/ijmhsc.2011.0153
- Yin, R. K. (2009). Case study research: Design and methods (4th ed.). Sage Publications.

Chapter Five: Remembering a Hard Life with Joy: Music-Making among Korean Elders²⁶

Elisha Jo Heesun, Kari K. Veblen, and Patrick J. Potter

The World Health Organization (WHO) estimates that the number of people aged sixty or older is expected to more than triple between 2000 and 2050, from approximately 605 million to 2 billion, representing 22 percent of the world's population (WHO, 2014). WHO has called upon stakeholders to enact policies and programs to promote active aging. Defined as "the process of optimizing opportunities for health, participation and security in order to enhance quality of life" (WHO, 2002), the term *active aging* is more inclusive than *healthy aging* because it acknowledges other dynamics in addition to health care. Recognizing that seniors can make meaningful contributions, the WHO chooses to use the term "active" because it entails continuous involvement in social, economic, cultural, spiritual, and civic affairs. In this context, opportunities for social engagement through lifelong learning, volunteering, and leisure activities become more important, given that older adults are less likely to continue in the workforce.

This research expands on an ongoing ethnographical case study of an ethnocultural community program for Korean immigrant seniors (Jo, Veblen, & Potter, 2014). Previous work found that shared musical knowledge plays a vital role in this program structurally and through fostering group membership. Our current study goes deeper into individual experiences to explore narratives of active aging through music.

Literature Review

Researchers suggest that a major deterrent to active aging is social isolation, which is a widespread health concern among older adults living in the community (Nicholson, 2012;

²⁶ This article has been published in *Community music at the boundaries* (2021) through Wilfrid Laurier University Press, edited by Lee Willingham. The writing was undertaken collaboratively with Kari Veblen and Patrick Potter, with Elisha contributing 60 % and co-authors 40 %.

https://books.scholarsportal.info/uri/ebooks/ebooks6/upress6/2021-05-13/1/9781771124591

Statistics Canada, 2006). Social isolation is associated with an array of negative health consequences, such as risk of cognitive decline (Beland et al., 2005) and physical risks such as an increased number of falls (Faulkner et al., 2003). Community programs for older adults, and community music programs in particular, have been found to promote active aging through multiple domains of health by generating opportunities for social engagement.

Group music-making – for example, New Horizons Bands in North America – makes a positive contribution to the overall health and well-being of seniors who participate (Carucci, 2012; Coffman, 2002; Jutras, 2011). Such events are also socially, culturally, and/or musically meaningful to the participants (Dabback, 2007; Creech et al., 2014; Kruse, 2007; Lally, 2009; Phinney, Moody, & Small, 2014) since they actively engage participants in their own interests and knowledge. In a comparative study, Cohen and colleagues (2006) found positive health outcomes for seniors who participated in a choir: fewer doctor visits, lower medication use, fewer falls, and less loneliness, as well as higher morale and activity involvement than among those who did not participate. Research focusing on musical engagement has demonstrated the positive influence of music on seniors' emotional, cognitive, physical, and social health and well-being (Clift et al., 2010; Creech et al., 2013; Hays & Minichiello, 2005a; 2005b; Solé et al., 2010; Sun et al., 2013).

Venues for social engagement are especially valuable for immigrant seniors from ethnic minority communities, who face a higher risk of isolation, for they often lack social support from others (Kim, 1999a, 1999b, 2013; Lee, 2007). Older Asian immigrant adults often suffer from psychological distress, resulting in poorer physical and psychological health than among their native-born counterparts (Choi, 2001; Kim et al., 2010; Tendulkar et al., 2012; Wong et al., 2007). Many community music programs offer benefits to seniors living in the community; however, immigrant seniors from backgrounds markedly different ethnically and linguistically from mainstream culture may face barriers to participation if the program or service is only offered in the dominant language, such as English (Emami et al., 2000; Kim et al., 2011; Mui et al., 2007; Stewart et al., 2011). This necessitates culturally sensitive programs such as CESC, designed for Korean immigrant seniors.

Method

Our research examines (1) individual perceptions of music-making within a community program, and (2) that program's influences on participants' health and well-being. The study employs a mixed method design, but at its core, it is a qualitative inquiry consisting of narrative case studies within a larger ethnography, augmented with quantitative data.

Narrative Approach

This phase of the research relies most heavily on a narrative approach that follows Clandinin and Connelly (2000) – "telling a life story, retelling a life story and reliving a life story" (p. 418) – to best understand how music may reveal identity construction through various stages of life. As a dialogic and collaborative practice, a narrative approach takes a sensitive approach to culture through listening, telling, and sharing (Bochner, 2001; Bochner & Ellis, 2003). Older adults' needs may vary, from training to continued or voluntary employment, to health and wellness, to digital literacy, to caregiving, to leisure and travel, to personal development (Merriam & Kee, 2014). However, the most significant learning needs in this stage of life may be the ones that focus on seeking the meaning and value of one's life, for a person's purpose in life is no longer delineated by his or her employment (Fisher & Wolf, 2000), nor as that person's identity.

I interviewed five participants for their life stories and musical experiences. These narratives serve as the heart of the data (Polkinghorne, 1995). The individuals who shared their stories were among the more than 150 participants in the Enoch College program; circumstances, natural affinity with the interviewer, largely due to sharing culture and mother tongue, and willingness to be part of the project were the determining factors in their initial selection.

Being aware of a possible bias by being drawn to people who share similar qualities to ourselves such as making assumptions, and impacting data analysis, I sought rigorous approach from data collection to analysis. For example, I transcribed interview and showed interviewees for them to check for facts, mistakes and or if they wished to add or subtract any information either via email, or personal communication. Semi-structured interviews, with open-ended questions, were conducted in an informal manner. Participants chose the location of the

interview – for example, a favorite coffee shop or park. Each interview lasted more than two hours, with follow-up conversations over the phone, via email, and in person during school hours at Enoch College. Interviews were recorded, transcribed into Korean, and then translated into English.

The actual interviews followed improvisatory routes as prompted by open- ended questions; later, though, the information was organized chronologically for analysis (Creswell, 2007; Josselson, 2010). Each narrative was fit into a narrative sequence, using a beginning, middle, and end, a process referred to as restorying (Creswell, 2007). Through reflective and repetitive reading of the texts, the key elements were determined (Green & Thorogood, 2009). More detailed sub-themes were drawn from field notes and additional sources such as publications and choral compositions.

Continuing Ethnographic Framework

Since CESC is integral to our contributors' worlds, we continued participant observation of the program at the Presbyterian Church in Mississauga. The principal researcher's proficiency in the Korean language and cultural understanding from a shared musical knowledge aided data collection. Findings were cross-checked for validity and possible bias from an emic perspective. Other sources included field notes, document analysis, observation of classes, informal conversations, and self-reflection (Denzin & Lincoln, 2005), with video recording to supplement data collection (Bogdan & Biklen, 2003; Goldman-Segall, 1998).

Five Narratives

1. Song's story: "Wind from a mountain and wind from a river."

Song was born in 1942, and his father died soon after that, leaving his mother as sole bread- winner. When the Korean War (1950–53) and Communist regime inflicted further hardship, the family fled Seoul, clinging to the rooftop of the train, as they could not afford to ride inside. They reached safety in Busan, at the southern tip of the country, but poverty continued to haunt them: "I think my childhood was greatly affected by the war, psychologically and emotionally."

Song remembers singing at his primary school for refugee children. It was not much of school, for it consisted of a marquee set up for children and teacher to gather around in the open air. Nevertheless, he recalls his singing with pride, for he had a good voice with "a clear and loud tone ... My teacher must have picked up my voice while we were singing as a class, because she selected me out of all classmates to sing for the class." Standing tall in front of his class, Song belted out his favourite song, "The wind from a mountain, and the wind from a river." This song expresses gratitude to the wind that blows and comforts hard-working laborers, refreshing their souls.

Another early memory is of the keyboard instruments at school: "I just loved the sound of the small organ and piano. Even though I didn't know how to play the piano, I would imitate and play the few notes and remember being so thrilled to hear the sound of it."

Song recalls early successes, such as being elected vice-president at his new school. This was a notable triumph, since only students with high marks were eligible to run for presidency and were then elected by popular vote. Despite these triumphs, some barriers were insurmountable – lack of funds meant that he never had formal music lessons, couldn't go on class field trips, and, worst of all, was rejected for class registration due to unpaid fees.

However, Song studied music on his own, and when he was eleven he found his outlet through choral singing in the Sunday school choir. This experience nurtured a love of music that led to him conducting his Sunday school choir only a decade later. His memories of this choir are so vivid that he can still recall each child's name: "When children sang a song in the open air near the coastline, people could not help but listen intently, peeking their heads out to search for the angelic voices echoing through the railway."

Under Song's direction, the children's choir won numerous competitions. He notes: "As a conductor, there's always part of me that longs for more, to make the best kind of music with my choir." His thirst for learning motivated him to become a very good musician himself. For example, although he had no professional training, he wrote a manual for his choristers.

Consisting of thirty-six pages of handwritten comprehensive instruction, this handbook includes vocal anatomy, pedagogy, basic theory, musical terms, history, and a philosophy of singing.

In 1993, after a life as a successful businessman who travelled to more than twenty countries, Song immigrated to Canada, where he again faced challenges in a foreign land. Here, too, music eased his mind. "I was having a quiet moment, deep in thought, standing by the window in my high-rise condo." he recalled. "I wrote lyrics, something like a poem. It was about asking why are you so anxious when God takes care of you? He will lead the way and so you should not worry." There, he confronted his fears, doubts, and worries and found comfort in God and expressed it through a song: "With this lyrics, I composed a simple melody and harmony for choral singing."

Today, at seventy-two, Song still feels a longing for the creative process. However, his physical condition prevents him from more active music-making. Six years ago, a triple bypass surgery forced him to retire from conducting, although he occasionally helps out. He is increasingly subject to exhaustion. Song comments: "I have unfulfilled aspirations as a conductor, and wish to make great music, and at the same time, I am also aware that I am aging."

Although physical infirmities slow him down, Song continues to make music. Most recently, he directed a small ensemble for the semester-end recital, with a short song with rhythmic clapping that he had composed. Song has recognized the need for programs like Enoch College for what he calls "Young Seniors," as he is acutely aware of the diminishing role for seniors, particularly immigrant seniors, in society. In fact, one reasons why he registered at Enoch was to observe the program for possible implementation in his own church.

2. Yong's story: Harmonica and "Heidenröslein"

Yong was born in 1932 to a doctor's family in rural Korea. His father declined a secure position in the Severance Hospital in Seoul to care for the underprivileged farmers in a rural area reachable only by miles of gravelled pathways. Yong remembers the country in turmoil, the Japanese annexation (1910–45), and the Korean War, when hundreds of thousands of South Koreans fled south after the North Korean army invaded.

Yong's early musical memories are of singing in the Sunday school choir until one day, he found a mysterious object in a desk drawer. When he blew air into the holes, he realized that it was a musical instrument, and thus began his a lifelong absorption with music-making,

experimenting with the sounds and playing his harmonica alone. Such an outlet was doubly welcomed since daily life was incredibly difficult at this time. Social and religious activities were restricted, for the Japanese occupiers were seeking to eradicate Korean culture and language.

Yong also remembers music in schools, such as his first day of high school German class. The new teacher had just returned from Germany with pedagogical ideas: "He told us that 'the best way to learn German, a foreign language, is to learn the song first,' and to illustrate, he began singing 'Heidenröslein' to the class." This Romantic *lied* from a poem by Goethe tells of unrequited love. Yong immediately fell in love with the song's sad beauty and learned it by heart. Since then, the song has resonated with Yong through his life, and he sings it now at Enoch College.

At age eighteen, Yong was forcefully recruited as a soldier into the Korean War, which cut short his dreams of becoming a medical doctor like his father. After the war, he continued in the army, working as an electronics technician. This brought opportunities to study in the United States for a year. Opportunities also came through Yong's wife Kay, whose profession as a nurse allowed her to go to Winnipeg, Canada, in 1966. After a few months, Yong followed her with their two young children. The family moved to Toronto in 1974 when Yong found a job there as an aviation technician for Air Canada.

After retiring, the couple began attending Enoch College and taking an active role in a harmonica ensemble, where he facilitates music learning and making. He continues to learn various musical instruments such as guitar, accordion, and clarinet, most of which he picked up after retiring, to play alone, in a duet, or in a group.

3. Kay's story: Joy of singing, thirst for learning.

Kay was born in 1932, the only child of a landowning family in the northern part of Korea. Her earliest memory of music is singing at school. There was one coveted musical instrument: "As kids, we would sneak into the classroom to play the piano. At times, the class would line up and take turn[s] to have the chance to play the piano."

Kay's most influential musical experiences started when she was at fresh- man at Yonsei University's College of Nursing. Daily chapel services were part of this university, established by Christian missionaries. Kay became a Chris- tian, joined a choir, and began what she calls her lifetime "joy of singing." In 1950, Kay's life changed forever when Communist troops invaded, separating her from her parents and kidnapping her classmates. Only eight of forty students in her freshman class survived to graduate. Worse still, she never saw her parents again.

Although only a student, Kay worked as a nurse, since there was a short- age of medical professionals during the war. She experienced trauma and brutality first-hand in hospitals and at UN prison camps. When the Korean War ended in 1953, she returned to her school, now relocated to a remote island in southern Korea. The reunited student choir began singing again as they had in the chapel. But now, as war survivors, the choir found new meanings of hope, strength, and resilience in the old songs.

Kay graduated into a responsible nursing career, teaching at the university. She also fell in love with and married Yong, who shared her passion for choral singing. When an opportunity arose for Kay to immigrate to Canada as a nurse at a hospital in Winnipeg, she took the job. Canada was still in the early years of accepting immigrants from Asian nations, having begun to ease national origin criteria in 1960s, and the young couple was among the first wave of Korean immigrants in Canada (A. H. Kim, 2012; Yoo, 2002).

Since there was no social or family support system in place, they shared responsibilities for their two young children. Kay took permanent evening shifts in the hospital, while Yong worked during the day. Although in Yong's words: "Leisure did not even occur to us," Kay and Yong joined the church choir and sang on Sundays when they could.

Now retired from nursing, Kay has time to pursue her many musical and artistic interests. She took a certification course for interior design from a local college, is a member of the Etobicoke Quilters' Guild, and is planning her first quilt show. Although very active, Kay is increasingly aware of the work required for her to participate in music at age eighty-two. She notes:

Last Christmas, I joined the choir for the fundraising benefit concert for North Korea. In addition to added choir rehearsals, I rewrote the words in large letters in thick black ink for easy reading. I marked difficult sections in the score and listened to the recordings constantly as the music for this event was a challenging Christmas Cantata sung in Latin. It required exceptional attention to detail both in the music and unfamiliar language.

Kay's strong work ethic and high level of professionalism are evident not only in her personal endeavours but in other areas as well. As a member of the Korean Canadian Women's Association since its inception in 1985, when the organization faced challenges that jeopardized its existence, she even switched her full-time position to part-time in order to take on additional responsibilities to provide support for the organization. This year, the KCWA will be celebrating its thirtieth year of supporting the lives of Korean Canadian women in Ontario. This is only a glimpse of Kay's achievements. She continues to share and take a leading role in music-making with her husband Yong, be it in choral or small ensemble settings. She continues to relish opportunities for learning. This year, Kay took up the saxophone, noting: "Not that I always wanted to play the saxophone ... I've been meaning to learn to play a new musical instrument, with no particular preference." In a recently published collection of Canadian immigrant experiences, Kay reflects: "I have only tried to live honestly and with diligence."

4. Kim's story: Music in a life for others

Born in 1940 in Busan, Korea, Kim comes from a poor but devout Christian family that was dominated by a very strict father. Kim was a shy and quiet child; his earliest musical memories are of singing hymns in church. By age fourteen he was already teaching at Sunday School: "There was always a shortage of Sunday School teachers, so I started to help out at a young age. As I become more involved, I felt a pressing need to teach children singing." To this end, he began looking for ways to strengthen his teaching: "There was a small organ in the church that no one played, so I decided that I will somehow figure out how to play."

Since no one in his family or church played an instrument, and Kim had no idea how to begin, he prayed to God for wisdom. Building on scant notation reading skills taught at school, he slowly worked himself toward playing note by note. He explained:

At first, God gave me the wisdom to begin with an easy hymn without sharps or flats. I picked "I Gave My Life for Thee," and it turned out to be a very good piece for me since the refrain also had the same melody/ repeats. I first began with my right hand only. Then I practised with left hand, and eventually added both hands together ... From then on, I began expanding my hymn repertoire from practising hymn with one flat, then moving on to two flats and so on. After about a year of diligent practice of building repertoire like that, I became good enough to manage to play the songs for the children.

Leading others in worshipful song filled Kim with joy, but economic necessity dictated that he work to maintain the family. Kim and his older brother established a business selling watches from a small booth at a dangerous inter- section in Busan. Shocked by the sight of frequent horrific car accidents, he remembers wondering: "If I was the one who died today, what would have been the meaning of my life?" It occurred to Kim, now in his early twenties, that he should not waste his life, as today is only given once.

As if in direct response to his internal questioning, a friend working with hospital ministry invited Kim to join him. "I said, 'Good! I want to devote my life for doing a meaningful and valuable work for others!" Straight away, he left the business to his older brother and went to Seoul. Resources were limited, and the hospital ministry provided only a meagre living, but it did not matter, for Kim was increasingly determined to live for a higher purpose.

While working in the hospital ministry, God called Kim to become a pas- tor. From then on, he answered his call without concern for language, national, or cultural barriers. Kim followed his inner compass to Germany, France, and eventually, Canada, responding to requests from small churches.

Music for Kim has been a calling, not his only calling but one of many to which he has cheerfully answered, "Yes!" This man is unassuming, quiet, and reserved, but his entire being becomes illuminated when he sings and plays his accordion, joining in the choir, leading small musical ensembles, and – as he has for most of his life – leading choirs. Kim is not always paid for his ministry and supports himself by working as a janitor at a local church. For this upcoming Sunday, he will be leading the choir in singing "There is a peace that the world never gave." As

he meditates on the words, Kim adds, "I hope that God will bless me to serve Him through music for as long as I live."

5. Min's story: Grief opens the heart

Born in 1944, Min has no childhood musical memories since, until recently, music has had "no relevance what so ever" to her life. While she worked as a secretary for the major Korean affiliate of Gulf Oil Corporation in Seoul, she was exposed to Western culture via the American Gulf employees. She believes that this influenced her decision to immigrate to Canada as an independent immigrant in 1973. She settled in Toronto, where she worked for the University of Toronto Alumni Association. In 1975, Min met her future husband at a neighbour's dinner party. Lightning struck, but the timing was off, since he was returning to Korea. The couple wrote to each other until he returned to marry her.

Min recalls the time when the newlywed couple took a dancing class together:

You know if one partner can lead, then the other will follow, but I got more confused because of him. We failed so miserably and were so embarrassed that after few try-outs we gave up. Since then, in all our life together, we sat through all the parties. We never got to enjoy dancing together.

Min also struggled at singing, particularly in a choir. "I will just sing which- ever part one sitting beside me sings. And it wouldn't matter whether the per- son is singing soprano, alto or tenor," she says.

Min's life revolved around her family, their small businesses, and her beloved husband until his sudden death four years ago. Soon after his passing, she looked into a mirror and could not recognize herself in the very old lady looking back (she had lost 14 pounds after his death). Her family was so concerned that her son moved in to be with her.

It was a year after her loss that music found her. She remembers the day vividly:

One morning my son had gone to work and I was taking a shower ... as bright sunlight shone through the window. Suddenly, a hymn, came into my heart, just like the ray of

sunshine ... Following my heart, I closed my eyes and began singing the song, then soon a wonderful peace overflowed in me like never before.

Since then, Min says that a song comes into her heart while walking along a park trail.

I've been singing meditatively like praying in my walks ever since. Every time I sing ... I am reminded that Jesus is with me, that I am not alone ... Sometimes I sing for hours, but not a complete song. Two verses from here, and two verses from other songs ... weaves into a whole when I sing alone.

A year later, Min found Enoch College and began to participate in singer's competitions, although her intention is to share joy rather than to compete. She explains:

Even though I am not good at singing at all, ... if the Holy Spirit enables me, then I will answer, "Yes, Lord" and sing. So, I overcame my fear of singing ... When you sing songs of praise, peace comes into your heart, and all your worries that seemed as enormous as the mountains turn to small pebbles – that's what I wanted to share.

In the process of claiming her musical self, Min has begun to memorize songs and now can sing two from memory for the first time in her life. She explains:

I write down the words in big letters on a piece of paper so that I can see it without my glasses and carry it with me during my walks ... Memorization is a challenge for me ... [but] this is indeed my struggle and my story. What I told you might not be significant to others, but this is the story of my life.

As Min shares her life, there are many memories of over thirty years with her husband, but now these memories bring joy instead of grief. Her story is one of finding freedom and peace through music.

Discussion

Each of these narratives tells of challenges and hardships through upheaval, wars, and immigration. Each person now experiences old age, a time of life frequently associated with fear and with loss of control, and each endures a measure of personal problems, such as physical or

emotional/mental health concerns, economic limitations, and lack of social position due to immigration.

Nonetheless, these seniors celebrate their lives through music. They enjoy an "encore life," one of revitalized sense of self and group. Song's early love for music continues to shape his life. Yong's lifelong involvement with music began with the solace of a harmonica in his adolescence and now emerges as he leads the harmonica ensemble. Yong also expands his musical endeavours through learning new instruments. Kay has begun studying the saxophone while planning her first quilt show at age eighty-two. In seeking to help children to sing, Kim began his musical journey and found his calling in God. He combines these passions as he devotes his life to a music ministry. Min found a source of hope through music in the depths of grief and now shares comfort and peace with others through singing hymns.

An obvious and uniting thread through these narratives is the role that CESC has played in bringing people together. Although this community program sponsors a wide range of courses including literature, history, and arts, even classes that are not focused on musical performance encourage daily music-making. The day begins and ends with prayer and songs. Students prefer to sing familiar Korean songs that resonate with a personal and collective past; school and children's songs are an essential component. In this way, students remember their lives and come to terms with the present (Moody, 1986).

While some students continue musical endeavours solely at home, others actively further their skills through supplemental singing or instrumental lessons elsewhere. Many seniors take advantage of CESC's opportunities to showcase their music learning. A regular performer, Kyung, shared her story, "It is from Enoch College I've found my love for singing because until then, I didn't know much about singing. I've been taking singing classes to learn new Korean songs and I practise them to sing it by heart so that I can sing them for my Enoch friends."

While musical engagement promotes active aging, there are seniors who cannot or do not participate. Age-related health issues, such as declining mobility, vision, or hearing, may limit social interaction (Mikkola et al., 2015; Public Health Agency of Canada, 2006; Sloan-Seale & Kops, 2008). Specific to this study, research indicates that older Koreans were found to have the highest psychological stress among other older Asian-American subgroups (G. Kim et al., 2010;

I. Kim et al., 2018). This may relate to Koreans having a shorter immigrant history in North America than other Asian cultures (Mui et al., 2006). Seniors who come to Canada through sponsorship by family members often lack social resources and can experience social isolation even while living with family (Koehn et al., 2014; Sadavoy et al., 2004). They are also less likely to participate in ethno specific programs like CESC due to time constraints or responsibilities to provide care for grandchildren (Choi et al., 2014). Further- more, transportation issues or a poor location may present further barriers to participation (Moody & Phinney, 2012; Sloane-Seale & Kops, 2008).

These barriers to participation seem to apply for some of Enoch College's students. Hyo, a quiet lady who loves Korean traditional music and has been known to write beautiful poems, said, "Although I knew about the CESC since its inception in 2003, it's been only few years since I could participate because I was helping out with grandchildren." Sook, another participant who immigrated to Canada with her husband to care for their grandchildren, shared her reluctance to participate in the next term, as her daughter had been driving them to the CESC. Enoch College is accessible by public bus, but many participants carpool together instead. Some participants, like Sue, rely on their grown-up children for rides. The use of public transit can be difficult, especially when a transfer is required or the weather is unfavourable. Indeed, a significant drop in attendance is observed on colder days.

There are many ways in which CESC addresses the usual barriers to participation. Concerns such as straitened finances (Koehn, 2009; Narushima, 2008) and cultural and linguistic barriers (Emami et al., 2000; Stewart et al., 2011) are ameliorated through this all-volunteer church- sponsored community program. Enoch College is free for all and provides home-cooked hot Korean specialties at lunch. The entire program is offered through the Korean language with Korean content.

A final and subtle barrier to participation is psychological. Korean elders at Enoch College, like older people everywhere, work to accept their age and the perceived limitations of aging. For example, Joon, who has been a member of CESC for four consecutive terms with his wife, shared his reluctance to participate in the beginning:

At first, I did not like to be surrounded by only old people. On your right, on your left, and even when you turn around, all you see is grey hair! Of course, I know that I am no different than they are. But we forget our age because our heart is still young as if no time has passed ... I have asked some friends to come with me to the CESC, they refused because they believe they are not old-aged enough to belong to a senior group.

This distancing (not wanting to be seen as old), but at the same time valuing programs that involve a similar age group, has been observed in other studies (Sharon et al., 1997; Stead et al., 1997; Yardley et al., 2006). While internalized negative age stereotypes may restrict behaviours (Chung & Yung, 2014; G. Kim et al., 2012; Yeom, 2014), it is possible to challenge and change participant perceptions (Fernández-Ballesteros et al., 2013). In this regard, the CESC program actively promotes affirmative views about aging. For example, if, instead of referring to participants as seniors, they are referred to as students, participants are empowered to adopt a positive attitude toward learning, for education spans one's lifetime and has no age limitations. Through a wide range of course offerings and opportunities for self-expression, students are encouraged to live life to the fullest. As Joon articulated, "Here at Enoch, there are joy, sharing, and feelings of gratitude with hope!"

Conclusion

This research presents a portrait of five seniors who find a renewed sense of self through music. Music has been woven into their lives in their moments of joy, as a friend and faithful companion, and as an answer to prayers or an affirmation of faith. The indispensability of music is not limited to these five elders only; it extends to all at Enoch College, for their days are filled with music-making. Although varying degrees of music accompany each individual's life, the sharing of music in this community setting bridges differences of age, gender, and the past to make the present meaningful.

Reflection

Qualitative research, particularly ethnography, enables participants' voices to be heard. While this section of the data supplements the overall findings, it is, in my view, the most enriching and evocative aspect of the study. Cultural familiarity emerged as a salient aspect of

the research. Since we shared the same cultural background, I was able to adopt both insider and outsider perspectives. Simultaneously, remaining mindful of potential biases, I employed a rigorous approach, including cross-checking findings not only with participants but also with the program director. Although unplanned, my role as a volunteer facilitated the development of rapport with elderly Koreans, resulting in their openness and willingness to discuss their life stories. Reflecting on this, volunteering at the program enabled me to establish trust, and they were not only eager to respond to my questions but also enthusiastic about participating in the survey.

I was deeply interested in exploring participants' perceptions and perspectives regarding the CESC experience, as well as its significance to them. Open-ended questions fulfilled this purpose by encouraging elderly participants to respond freely using their own words. However, the utilization of open-ended questions posed limitations on the number of interviews due to their time-consuming nature. Although I would have preferred to conduct more interviews, in-depth interviews demanded a significant investment of time and effort, ranging from pre-interview preparation to transcription and subsequent analysis. Given the circumstances, it was not feasible to undertake such an extensive process at that time.

The interviews lasted from one to two hours and took place at locations chosen by the participants themselves, often at a nearby coffee shop or their home. The questions centered around their musical experiences throughout their lives. For instance, inquiries like 'What's your memory of making-music as a child?' and "Could you describe your musical experiences in school?' were posed. Eliciting responses about their childhood music-making experiences provided participants with a comfortable platform to share openly, enabling them to tell stories that they considered significant or relevant.

Since I had been working as a volunteer, I was also able to address specific questions regarding what had occurred in the recent classes. These follow-up questions allowed participants to provide comprehensive accounts of their own experiences, without necessitating numerous detailed queries. Instead, I could offer a few comments and questions for clarification purposes. It also helped that I was genuinely interested in hearing their stories and patiently waited for the meanings to emerge rather than reinforce it from them. While the stories shared

might have appeared to hold less immediate significant meanings, it later became evident that these moments had a profound impact on their life choices and trajectories. For instance, Yong sang his favorite German lieder, "Heidenröslein," to his harmonica ensemble classmates in response to a classmate's request during the harmonica session. I happened to be present during the harmonica class that day, and I made a note of the incident. This allowed me to inquire about the song, referencing the recent event. Such shared experiences from the CESC environment facilitated Yong's willingness to openly discuss his stories during the interview. Furthermore, as Yong shared the story behind the song, it illuminated the significance that music held in his life, along with his lifelong commitment to music-making despite life changing events such as war, military training, immigration, and retirement.

Elderly Koreans also shared their experiences of major historical moments in Korea's history, such as the Korean War and immigration. As a guideline, these significant life-changing experiences alone were sufficient to prompt them to share tales of hardship, and more notably, of resilience. Kay mentioned that she could compile a book solely based on her experiences during the Korean War. Nevertheless, she recounted her stories in a manner that allowed us to share a laugh together. Despite enduring the loss of her parents, home, and life she knew due to the war, Kay faced the realities of her experiences with determination and fervor. Her attitude is one that chooses to harbor no regrets and instead extracts positive aspects from all situations, shaping her into the independent, strong and content woman she remains to this day.

These elderly Koreans stand as living witnesses to both Korea's past and present. They navigated through turmoil to emerge from the ruins, assuming responsibilities and diligently working to contribute to the Korea we know today. These major events facilitated interview preparation, guided analysis, and organization of the data into meaningful segments. They also aided in identifying themes for analysis. Though less explicitly stated at the time, the focus was honed on how these series of life events influenced their life choices and their present attitudes as seniors. The narratives were recounted in the order they chose; for instance, Kay's story unfolded chronologically more so than the others, mirroring her profession as a nurse and her analytical and competent nature.

As elderly Koreans engage in musical activities, they find more opportunities to re-define and renew their sense of identity, which holds significance for them. At the same time, their renewed sense of self is not entirely 'new', but rather involves accepting differences and reconciling with the past in their own way. While they are not without concerns such as their current medical condition, the pursuit of truth, aspirations, disappointments, and the prospect of mortality, they actively choose hope and opt to live each day meaningfully and well-attending the CESC program is one way they achieve this. Viewed in this light, this cultural community program for elderly Koreans offers a provision for social opportunities, but also opportunities to 'live out' through learning, performing, and playing, expressing what is meaningful to them in their lives.

Highlighting the contexts of music-making, I find myself wondering if the meanings that they attributed to musical engagement and social participation are articulated as they were deeply felt. Active aging, which frames this study, would then serve as an underlying concept, providing an opportunity to explore and re-define selves in this stage of their lives.

If I were to conduct a follow-up study, it would be ideal to collaborate with another researcher who is also Korean and culturally familiar. This collaboration could potentially lead to an increased number of participants, thereby gaining broader and deeper insights related to older adults' social participation through music and its meanings. In addition, in a future study, I would like to include the voices of non-participants. Investigating the barriers to participation would yield important findings that could impact the quality of life for the elderly population. What challenges do they face in participating? What choices do they make to live actively and gain social support? Investigating the barriers to participation would be particularly important in influencing the quality of life for the elderly population. A comparative analysis between participants and non-participants could be made to gain more understanding.

Another important discussion to derive from the participants would be to examine whether these barriers to participation are relevant to them. If so, understanding how they managed to overcome these barriers and challenges would be insightful. Gathering responses to these specific inquiries could offer valuable insights to community practitioners seeking to enhance the well-being of the elderly population.

References

- Beland, F., Zunzunegui, M. V., Alvarado, B., Otero, A., & Del Ser, T. (2005). Trajectories of cognitive decline and social relations. *The Journals of Gerontology. Series B, Psychological Sciences and Social Sciences, 60*(6), P320–P330. https://doi.org/10.1093/geronb/60.6.P320
- Bochner, A. P. (2001). Narrative's virtue. *Qualitative Inquiry*, 7(2), 135-157. https://doi.org/10.1177/107780040100700201
- Bochner, A. P. & Ellis, C. (2003). An introduction to the arts and narrative research: Art as inquiry. *Qualitative Inquiry*, 9(4), 506-514. https://doi.org/10.1177/1077800403254394.
- Bogdan, R. C. & Biklen, S. K. (2003). Qualitative research for education: An introduction to theories and methods (4th ed.). Allyn & Bacon.
- Carucci, C. (2012). An investigation of social support in adult recreational music ensembles. *International Journal of Community* Music, 5(3), 237–252. https://doi.org/10.1386/ijcm.5.3.237_1
- Choi, N. G. (2001). Diversity within diversity: Research and social work practice issues Asian American elders. *Journal of Human Behavior in the Social Environment*, *3*(3/4), 301-319. https://doi.org/10.1300/J137v03n03_18
- Choi, J., Kushner, K. E., Mill, J., & Lai, D. W. L. (2014). The experience of Korean immigrant women adjusting to Canadian society. *Journal of Cross-Cultural Gerontology*, 29(3), 277–297. https://doi.org/10.1007/s10823-014-9235-8
- Chung, S., & Yung, Y. (2014). Age norms for older adults among Koreans: Perceptions and influencing factors. *Ageing and Society*, *34*(8), 1335–1355. https://doi.org/10.1017/S0144686X13000111
- Clandinin, D. J. & Connelly, F.M. (2000) Narrative inquiry: Experience and story in qualitative research. Jossey-Bass.
- Clift, S., Nicols, J., Raisbeck, M., Whitmore, C., & Morrison, I. (2010). Group singing, wellbeing and health: A systematic mapping of research evidence. *UNESCO Observatory*, 2(1). http://www.abp.unimelb.edu.au/unesco/ejournal/pdf/clift-paper.pdf
- Coffman, D. (2002). Music and quality of life in older adults. *Psychomusicology, 18*(1–2), 76–88. https://doi.org/10.1037/h0094050

- Cohen, G. D., Perlstein, S., Chapline, J., Kelly, J., Firth, K. M., & Simmens, S. (2006). The impact of professionally conducted cultural programs on the physical health, mental health, and social functioning of older adults. *The Gerontologist*, 46(6), 726–734. https://doi.org/10.1093/geront/46.6.726
- Creech, A., Hallam, S., McQueen, H., & Varvarigou, M. (2013). The power of music in the lives of older adults. *Research Studies in Music Education*, *35*(1), 87–102. https://doi.org/10.1177/1321103X13478862
- Creech, A., Hallam, S., Varvarigou, M., Gaunt, H., McQueen, H., & Pincas, A. (2014). The role of musical possible selves in supporting subjective well-being in later life. *Music Education Research*, *16*(1), 32–49. https://doi.org/10.1080/14613808.2013.788143
- Creswell, J. (2007). *Qualitative inquiry and research design: Choosing among five approaches* (2nd ed.). Sage Publications. ISBN 978-1-4129-1606-6
- Dabback, W. (2007). *Toward a model of adult music learning as a socially-embedded phenomenon* (Unpublished doctoral dissertation). ProQuest 304917902
- Denzin, N. K., & Lincoln, Y. S. (2005). *The SAGE handbook of qualitative research* (3rd ed.). Sage Publications.
- Emami, A., Torres, S., Lipson, J. G., & Ekman, S. D. L. (2000). An ethnographic study of a day care center for Iranian immigrant seniors. *Western Journal of Nursing Research*, 22(2), 169–188. https://doi.org/10.1177/01939450022044340
- Faulkner, K. A., Cauley, J. A., Zmuda, J. M., Griffin, J. M., & Nevitt, M. C. (2003). Is social integration associated with the risk of falling in older community-dwelling women? *The Journals of Gerontology. Series A, Biological Sciences and Medical Sciences*, 58(10), M954–M959. https://doi.org/10.1093/gerona/58.10.M954
- Fernández-Ballesteros, R., Caprara, M., Schettini, R., Bustillos, A., Mendoza-Nunez, V., Orosa, T., Kornfeld, R., Rojas, M., López, M. D., Santacreu, L. M., Molina, M. Á., & Zamarrón, M. D. (2013). Effects of university programs for older adults: Changes in cultural and group stereotype, self-perception of aging, and emotional balance. *Educational Gerontology*, 39(2), 119–131. https://doi.org/10.1080/03601277.2012.699817
- Fisher, J. C., & Wolf, M. A. (2000). Older adult learning. In A. L. Wilson & E. R. Hayes (Eds.), *The handbook of adult and continuing education* (pp. 480–492). Jossey-Bass.

- Goldman-Segall, R. (1998). *Points of viewing children's thinking: A digital ethnographer's journey*. Lawrence Erlbaum.
- Green, J., & Thorogood, N. (2009). *Qualitative methods for health research* (2nd ed.). Sage Publications.
- Hays, T., & Minichiello, V. (2005a). The contribution of music to quality of life in older people: An Australian qualitative study. *Ageing and Society*, 25(2), 261–278. https://doi.org/10.1017/S0144686X04002946
- Hays, T., & Minichiello, V. (2005b). The meaning of music in the lives of older people: A qualitative study. *Psychology of music*, *33*(4), 437–451. https://doi.org/10.1177/0305735605056160
- Jo, H. E., Veblen, K. K., & Potter, P. J. (2014). Korean immigrant seniors' music making in an ethno-cultural com- munity program in Canada. In M. L. Cohen (Ed.), *CMA XIV:*Listening to the world: Proceedings from the International Society for Music Education (ISME) 2014 seminar of the Commission for Community Music Activity (pp. 25–32).

 https://edoc.ku.de/id/eprint/19768/1/CMA_XIV_Full_Proceedings_FINAL_MC.pdf
- Josselson, R. (2010). Narrative research. In N. J. Salkind (Ed.), *Encyclopedia of research design* (pp. 869-875). Sage. https://dx.doi.org/10.4135/9781412961288
- Jutras, P. J. (2011). The benefits of New Horizons Band participation as self-reported by selected New Horizons Band members. *Bulletin of the Council for Research in Music Education*, 187, 65–84. https://www.jstor.org/stable/41162324
- Kim, O. (1999a). Predictors of loneliness in elderly Korean immigrant women living in the United States of America. *Journal of Advanced Nursing*, 29(5), 1082–1088. https://doi.org/10.1046/j.1365-2648.1999.00993.x
- Kim, O. (1999b). Mediation effect of social support between ethnic attachment and loneliness in older Korean immigrants. *Research in Nursing and Health*, 22(2), 169–175. https://doi.org/10.1002/(SICI)1098-240X(199904)22:2<169::AID-NUR8>3.0.CO;2-F
- Kim, G., Chiriboga, D. A., Jang, Y., Lee, S., Huang, C., & Parmelee, P. (2010). Health status of older Asian Americans in California. *Journal of the American Geriatrics Society (JAGS)*, 58(10), 2003–2008. https://doi.org/10.1111/j.1532-5415.2010.03034.x
- Kim, G., Aguado Loi, C. X., Chiriboga, D. A., Jang, Y., Parmelee, P., & Allen, R. S. (2011a). Limited English proficiency as a barrier to mental health service use: A study of Latino

- and Asian immigrants with psychiatric disorders. *Journal of Psychiatric Research*, 45(1), 104–110. https://doi.org/10.1016/j.jpsychires.2010.04.031
- Kim, A. H., Noh, M. S., & Noh, S. (2012). Introduction: Historical context and contemporary research. In S. Noh, A. H. Kim & M. S. Noh (Eds.), *Korean immigrants in Canada:**Perspectives on migration, integration, and the family (pp. 3–18). University of Toronto Press. https://doi.org/10.3138/9781442690387-004
- Kim, G., Jang, Y., & Chiriboga, D. (2012). Personal views about aging among Korean American older adults: The role of physical health, social network, and acculturation. *Journal of Cross-Cultural Gerontology*, 27(2), 139–148. https://doi.org/10.1007/s10823-012-9165-2
- Kim, Y. S. (2013). Ethnic senior schools, religion, and psychological well-being among older Korean immigrants in the United States: A qualitative study. *Educational Gerontology*, *39*(5), 342-354. https://doi.org/10.1080/03601277.2012.700826
- Kim, I., Kang, S.Y., & Kim, W. (2018). The Effects of Religious Participation and Familial Assistance on Mental Health among Older Chinese and Korean Immi- grants: Multiple Mediator Analyses. *Journal of Cross-Cultural Gerontology 33*, 411–25. https://doi.org/10.1007/s10823-018-9355-7
- Koehn, S. (2009). Negotiating Candidacy: Ethnic minority seniors' access to care. *Ageing and Society*, 29(4), 585–608. https://doi.org/10.1017/S0144686X08007952
- Koehn, S. D., Jarvis, P., Sandhra, S., Bains, S., & Addison, M. (2014). Promoting mental health of immigrant seniors in community. *Ethnicity and Inequalities in Health and Social Care*, 7(3), 146–156. doi:10.1108/EIHSC-11-2013-0048
- Krause, N. (2007). Religion and health in late life. In J. E. Birren, K. W. Schaie, R. P. Abeles, M. Gatz, & T. A. Salthouse (Eds.), *Handbook of the psychology of aging* (6th ed., pp. 499–518). Elsevier Academic Press. ISBN: 0121012654
- Lally, E. (2009). The power to heal us with a smile and a song: Senior well-being, music-based participatory arts and the value of qualitative evidence. *Journal of Arts and Communities 1*(1), 25–44. https://doi.org/10.1386/jaac.1.1.25_1
- Lee, Y. (2007). The immigration experience among elderly Korean immigrants. *Journal of Psychiatric and Mental Health Nursing*, 14(4), 403–410. https://doi.org/10.1111/j.1365-2850.2007.01098.x

- Merriam, S., & Kee, Y. (2014). Promoting community wellbeing: The case for lifelong learning for older adults. *Adult Education Quarterly*, 64(2), 128–144. https://doi.org/10.1177/0741713613513633
- Mikkola, T. M., Porteigs, E., Rantakokko, M., Gagne, J., Rantanen, T., & Viljanen, A. (2015). Association of self-reported hearing difficulty of objective and perceived participation outside the home in older community-dwelling adults. *Journal of Aging and Health*, 27(1), 103–122. https://doi.org/10.1177/0898264314538662
- Moody, H. R. (1986). Late life learning in the information society. In D. A. Peter- son, J. E. Thornton, & J. E. Birren (Eds.), *Education and aging* (pp. 122–148). Prentice-Hall.
- Moody, E., & Phinney, A. (2012). A community-engaged art program for older people: Fostering social inclusion. *Canadian Journal on Aging / La Revue Canadienne Du Vieillissement*, 31(1), 55–64. https://doi.org/10.1017/S0714980811000596
- Mui, A. C., Nguyen, D. D., Kang, D. Y., & Domanski, M. D. (2006). Demographic profiles of Asian immigrant elderly residing in metropolitan ethnic enclave communities. *Journal of Ethnic and Cultural Diversity in Social Work*, 15(1–2), 193–214. https://doi.org/10.1300/J051v15n01_09
- Mui, A. C., Kang, S., Kang, D., & Domanski, M. D. (2007). English language proficiency and health-related quality of life among Chinese and Korean immigrant elders. *Health and Social Work*, 32(2), 119–127. https://doi.org/10.1093/hsw/32.2.119
- Narushima, M. (2008). More than nickels and dimes: The health benefits of a community-based lifelong learning programme for older adults. *International Journal of Lifelong Education*, 27(6), 673–692. https://doi.org/10.1080/02601370802408332
- Nicholson, N. R. (2012). A review of social isolation: An important but underassessed condition in older adults. *The Journal of Primary Prevention*, *33*(2–3), 137–152. https://doi.org/10.1007/s10935-012-0271-2
- Phinney, A., Moody, E. M., & Small, J. A. (2014). The effect of a community-engaged arts program on older adults' well-being. *Canadian Journal on Aging / La Revue canadienne du vieillissement*, 33(3), 336–345. https://doi.org/10.1017/S071498081400018X
- Polkinghorne, D. E. (1995). Narrative configuration in qualitative analysis. In J. A. Hatch & R. Wisniewski (Eds.), *Life history and narrative* (pp. 5-23). Taylor & Francis Group. https://ebookcentral.proquest.com/lib/west/detail.action?docID=167247

- Public Health Agency of Canada (PHAC). (2006). *Healthy aging in Canada: A new vision, a vital investment from evidence to action*. The Healthy Aging and Wellness Working Group of the Federal/Provincial/Territorial (F/P/T) Committee of Officials (Seniors). https://www.health.gov.bc.ca/library/publications/year/2006/Healthy_Aging_A_Vital_latest_copy_October_2006.pdf
- Sadavoy, J., Meier, R., & Ong, A. Y. M. (2004). Barriers to access to mental health services for ethnic seniors: The Toronto study. *Canadian Journal of Psychiatry / Revue Canadianne De Psychiatrie*, 49(3), 192. https://doi.org/10.1177/070674370404900307
- Sharon, B. H., Hennessy, C. H., Brandon, J., & Boyette, L. (1997). Older adults' experiences of a strength-training program. *Journal of Nutrition, Health, and Aging*, 1(2), 103–108.
- Sloane-Seale, A., & Kops, B. (2008). Older adults in lifelong learning: Participation and successful aging. *Canadian Journal of University Continuing Education*, *34*(1), 37–62. https://doi.org/10.21225/D5PC7R
- Solé, C., Mercadal-Brotons, M., Gallego, S., & Riera, M., (2010). Contributions of music to aging adults' quality of life. *Journal of Music Therapy*, 47(3), 264–281. https://doi.org/10.1093/jmt/47.3.264
- Stead, M., Wimbush, E., Eadie, D., & Teer, P. (1997). A qualitative study of older people's perceptions of ageing and exercise: The implications for health promotion. *Health Education Journal*, 56(1), 3–16. https://doi.org/10.1177/001789699705600102
- Sun, J., Buys, N., & Merrick, J. (Eds.). (2013a). *Health promotion: Community singing as a vehicle to promote health.* Nova Science.
- Statistics Canada. (2006). *A portrait of seniors in Canada*.

 https://www150.statcan.gc.ca/n1/en/pub/89-519-x/89-519-x2006001-eng.pdf?st=ZHZP48BY
- Stewart, M., Shizha, E., Makwarimba, E., Spitzer, D., Khalema, E. N., & Nsaliwa, C. E. (2011). Challenges and barriers to services for immigrant seniors in Canada: "You are among others but you feel alone." *International Journal of Migration, Health and Social Life*, 7(1), 16–32. https://doi.org/10.1108/17479891111176278
- Tendulkar, S. A., Hamilton, R. C., Chu, C., Arsenault, L., Duffy, K., Huynh, V., Hung, W., Lee,E., Shwuling, J., & Friedman, E. (2012). Investigating the myth of the "model minority":A participatory community health assessment of Chinese and Vietnamese adults. *Journal*

- Yardley, L., Bishop, F. L., Beyer, N., Hauer, K., Kempen, G. I., Piot-Ziegler, C., Todd, T., Cuttelod, T., Horne, M., Lanta, K., & Holt, A. R. (2006). Older people's views on falls-prevention intervention in six European countries. *The Gerontologist*, 46(5), 650–660. https://doi.org/10.1093/geront/46.5.650
- Yeom, H. (2014). Association among ageing-related stereotypic beliefs, self-efficacy, and health-promoting behaviors in elderly Korean adults. *Journal of Clinical Nursing*, 23(9–10), 1365–1373. https://doi.org/10.1111/jocn.12419
- Yoo, Y. (2002). Canada and Korea: A shared history. In R. W. L. Guisso, and Y. Yoo (Eds.), *Canada and Korea: Perspectives 2000* (pp. 9-43). University of Toronto Press.
- Wong, S. T., Yoo, G. J., & Stewart, A. L. (2007). An empirical evaluation of social support and psychological well-being in older Chinese and Korean immigrants. *Ethnicity and Health*, *12*(1), 43–67. https://doi.org/10.1080/13557850600824104
- World Health Organization. (2002). *Active ageing: A policy framework*. World Health Organization Press.

 https://apps.who.int/iris/bitstream/handle/10665/67215/WHO_NMH_NPH_02.8.pdf?sequence=1&isAllowed=y
- World Health Organization. (2014). Ageing and life courses. http://www.who.int/ageing/en/

Chapter Six: Summary and Conclusion

Chapter 6 ties together threads and themes from previous chapters. I present findings, summarize main outcomes and contributions of the research, address emerging topics, discuss implications, and offer recommendations for practice and future research.

Through the three studies contained in this dissertation, I explored the impact of musical activities (singing, performance, and group musical activities) on the lives of a cohort of older Korean immigrants who were participating in a Korean cultural community program in Canada. I sought to gain insight to the group as well as to get to know the members of the group individually because I believe their stories can teach us about what it means to grow older and about how identity evolves.

Revisiting Research Questions

The central focus of my investigation concerned how musical engagement shapes Korean elders' identity and how it influences wellbeing. I focused on the following research questions:

- 1. What are the factors that contribute to participation in the program?
- 2. How does participating in the community program influence health and active ageing among Korean elderly participants?
- 3. What is the meaning and importance of music in the lives of elderly Koreans?

This final chapter re-addresses these questions and summarizes the findings. I also explore the interconnections at the heart of the concepts of musical engagement, well-being, and cultural heritage. Three interlocking notions serve as my theoretical framework: 1) Active ageing is a main concept with 2) Erickson's 8th stage and 3) lifelong learning are subset concepts. Finally, I describe the current, post-pandemic state of the program, raising questions and making suggestions for further research.

Research Question 1) What Factors Contribute to Participation in the Program?

Chapter Three addresses the first research question through a study entitled "Korean Immigrant Seniors' Music making in and Ethno-Cultural Community Program in Canada."

Analysis in that study examined the features of the program that create a thriving and coherent community. The Canada Enoch Seniors' College (CESC) is an ethnocultural community program for elderly Korean immigrants which was established in 2003 by Pastor Kim. The program began with about 50 students and has become an on-going success with a current enrollment of 150 students each term. "Enoch Day," as the members of the CESC like to call every Thursday when the program is offered, is a place where participants get to re-live their past as students but now with the knowledge and experience of a lifetime.

Focus on Wellbeing and Healthy Ageing

The CESC takes the familiar form of a culturally embedded, formal type of education consisting of homerooms, elective classes, assemblies, and special events and performances. Since CESC's format mirrors formal schooling, elderly Korean participants have been able to ease quickly into the program. While group solidarity is important, individuality and independence are fostered through a variety of elective classes (mostly arts) to enable seniors to gain and develop skills in their areas of interest. The program has improved their physical activity, social connections, and networking, as well as independence, autonomy, and feelings of satisfaction and achievement through learning which has contributed to the overall health of the cohort. These positive impacts on participants' health and quality of life is supported by previous research on the benefits of musical engagement and learning (Coffman, 2002; Creech et al., 2013; Creech et al., 2014; Dabback, 2007; Fung & Lehmberg, 2016). Korean elders were also able to explore and express their renewed sense of self and cultural identity through musicking, singing, dancing, performing, and creating together as a group, leading them to feel a sense of home and to find comfort in activities that are meaningful to them (Clift et al., 2010; Clift & Hancox, 2010; Clift & Morrison, 2011; Lally, 2009).

Music's Part in the Program

This ethnography revealed the cultural significance of active musical engagement in the lives of Korean elderly immigrants. While one of the initial drivers of the program may have been to address participants' social needs, the key to its success was cultural; culture was the thread that connected theory to implementation. Music is one manifestation of cultural identity. Simply put, Korean elders love music; they engage easily with music and are highly satisfied

with their own and others' music-making. As one participant remarked, "Music works its wonders," a sentiment often expressed by CESC participants.

Korean Traditional Culture and Music

Music's positive role in Korean culture is assumed and welcomed. CESC's director commented that "music always has been an important part of Korean culture. We have always sung or chanted." Korea has been an agricultural society since ancient times; common people were farmers and sang together while working in the fields—plowing, planting, and cropping produce. The orally transmitted folk song, "Min-yo," was created to express the feelings that accompany a collective lifestyle. Singing strengthened unity and encouraged, supported, and expressed emotions and the hope for a better future as a result of labour (Cho, 1995). Consistent with life in a collectivist society, Koreans are committed to benefitting the group, even at the risk of losing individual rights. Social connection and interaction are important for their identity.

In a variety of occupations besides agriculture—fishing, civil engineering, transportation—and when women labor at housework or handicrafts—or simply whenever people work together—Koreans have sung songs to accompany arduous or tedious labor, unifying their actions to reduce the burden and contribute to a better result (Cho, 1995). Each song reflected the type of work during which it was sung, varying in dynamics, form, and content. For example, songs for harvesting barley and rice planting are sung in unison, reinforcing unified action, while songs for diversified work are call and response songs like the work itself. Songs for hard labour are simple and repeated, whereas songs sung during individual tasks are sung individually, in slow tempi, and are colourful and richly ornamented, even improvised as befits the work at hand (Cho, 1995). Interestingly, weaving, which is traditionally a woman's individual labour, is accompanied by long songs, suitable for accompanying long hours of weaving and with a prolonged introduction and a mellow sound. As they weave, women open up their hearts to tell their stories, so these songs are sincere, tender, and melancholy (Cho, 1995). For Korean elders, music is a link that connects one person to another, and fosters communication not only within the group but within the whole Korean culture. Singing songs has been the preferred means for elders to express themselves, and their emotions.

Largely because Korea is a collectivist society, the elderly Korean cohort that informed my studies experienced rapid social and political change in Korea. Following its independence from Japanese colonisation (1910-1945) and the Korean war (1950-1953) which left the country in ruins, Korea became one of the world's poorest nations. The 1960s was a period of rebuilding for the Korean economy. The government completely changed what had been an agricultural nation into an industrialized one, resulting in a remarkably rapid transformation and development of the economy in what has been called "the Miracle on the Han River" (J. D. Park, 2019). Collectivist values as well as a national education system and compulsory military training service encouraged people to be diligent, sacrificing individual interests for national gains and aiding in the rapid industrialization of the nation. In Korea during the 1960s, the per capita income of 82 USD, which was lower than most African countries at that time, jumped to 253 U.S. dollars in 1970 and continued to increase to 52,700 U.S. dollars by 1995 (S. H. Kim, 2007). The unemployment rate of 8.1% in 1963 fell to 4.4% in 1970 (J. D. Park, 2019). The seniors in the Enoch program were part of this time of nation rebuilding and they are used to hard work and being part of a team to achieve success. They appreciate a structured setting where they can grow and learn. A sense of membership is important for participants' active engagement, creating a safe place for them to socialize and participate. Being assigned to small groups in the program ensures that everyone is included and fosters a sense of belonging.

The director needed great knowledge of Korean culture to be able to provide relevant experiences to CESC members. Pastor Kim, whose former profession was as a middle-school English teacher before she came to Canada, took a great interest in Korean history and culture. Her devotion deeply affected her cultural approach to the program from its inception. Her husband's comment illustrates Pastor Kim's vision: "In our lives I never got bored of driving long hours, especially when we took vacation with kids, because she was full of stories to tell and could recite the kings and queens; it was never boring." Her deep understanding of Korean history and culture contributed to the successful implementation of the program and resulted in positive outcomes in health and well-being for Korean elders.

Shared knowledge allowed elders to engage freely with each other. Common age-related issues are deeply felt and can be shared in a safe place. Often, tacit understanding offers comfort. A newly widowed Korean member, Min, tells her story. It was her very first day, and Min came

to the program based upon a recommendation from a family member. Min was struggling with her loss and was suffering depression. Sitting next to a stranger, Min blurted out that she just lost her husband. At that moment, there was nothing else she could say. The lady listened to her, quietly reached out her hand and held Min's hand tightly. Min still vividly remembers that very first day at the program and what it meant for her.

For seniors, there are many occasions when no words are needed, but they understand each other because they share age-related issues and the realities of ageing. It is comforting for them to know that they are not alone and that everyone is going through similar life shifts. As Min found, Korean elders empathise with one another and feel free to open up and share their concerns, offering each other emotional support.

Myung, who recently lost her husband who had been paralyzed from a car accident for the past fourteen years, was not shy to show a picture of her husband. She gave her testimony about how her husband and she became a team, and how he had always been passionate about the Gospel and had been eager to share it with other patients in the hospital. She carried good memories of him and shared them with the group.

Some found it easier to talk to others who share similar experiences and challenges of old age. Many of these elders are concerned that sharing their challenges and difficulties may put on burden on their children. "I do not want my kids to worry about me. They are already concerned for my well-being, so I rather not tell them," was a widely held notion. However, at CESC participants feel that they can openly share their conditions and exchange information on nutritional advice, new exercise, or health tips they have picked up. "My doctor suggested this," is a frequent topic of conversation amongst the members since most are concerned about their health.

Korean elders are often eager to share new information not limited to health issues with one another. If one person starts to share, others listen, and soon what began as a conversation between a couple of people becomes a discussion in a group of people who have gathered around. Sometimes, depending on the topic of interest, the original group forms another group. If there is a testimony to share, the conversation becomes more vibrant and exciting. Lively communication fills the day with sounds of laughter and talking. Elderly Koreans' choice of

preference for acquiring new knowledge by asking others for information is possible because they are speaking in their native language and because the program involves a homogeneous cultural group with the same ethnicity and language.

Familiar Music

The learning sessions of this program are focused on refining previous knowledge the members learned mostly in elementary, middle, or high school in Korea. Many members stated that group singing was their favorite class in school, and they look forward to it at CESC every week. Reminiscence was one of the key elements that contributed to the positive outcome of the program.

Song selection played an important part in this regard. Well-known repertories were chosen that many participants were able to recall from their schooldays and youth. Memories of the past evoked while singing songs from their youth brought them happy memories of early years. Remembering the past positively, many smiled as they sang. It was as if they became teenage students again, sitting in a classroom belting out songs they liked and were familiar with. Familiarity was an integral element of the experience.

Some took pride in still being able to remember the words by heart, especially Italian or German songs. "I can still remember the words; once I hear them it all comes back to me," Soo commented proudly, and to show it, she closed her eyes while singing an Italian art song. "I sang these songs in school, so I know them." The elders clearly appreciate singing songs that they already know, singing complete songs, in full sound, and in tune with others. This was an aesthetically pleasing and satisfying experience for them.

Language Considerations

Along with cultural differences Korean elders readily cited the language barrier as one of the main challenges to living in Canada (Guruge et al. 2021; Wang & Kwak, 2015). Even if they were proficient in English from having had a successful profession in Canada, such as nurse, technician, or bank manager, their preferred language of communication when they joined the CESC was Korean, as it is most natural for them.

Jun said he believed that his level of English when he immigrated to Canada was good enough to make a smooth transition into Canadian society. His interest in English motivated him to study English throughout his career as a librarian, and thus he did not consider language would be an issue for him; he found, however, that he was mistaken. Although he is happy with his choice to immigrate, he named language as the first challenge of his immigration although the issue was not language itself but English idiomatic expressions. Jun explained, "Even though it is still English, terminology was different in Korea and in Canada. When I see a word, 'hardware,' for instance, in Korean, the word refers to computer hardware as opposed to software. Not long after I arrived in Canada, I called a number I found on a job posting site for the hiring of employees. The job was in 'hardware.' I called to see if they were still looking to hire. The man told me that I could come for an interview, so I asked him about the company. I said, 'Does this work involve working with computers?' He said, 'No. This is a hardware.' I did not get it. So, I asked, 'What do you mean by that? Isn't hardware a component of a computer?' He kept saying, 'This is a 'hardware.' I went there, and it was a Chulmuljeum (ironware store)! It was shocking."

Immigrants from non-western nations like Korea, where English is not an official language, are vulnerable and are often left alone to deal with challenges. Jun added that his wife feels more frustrated than he does, but she does not really have a choice except to try to deal with the frustration. Since communication is central to socialization and living an active life, language is an immediate concern that impacts everyday living.

Since the CESC program is offered in the elders' native language and with cultural markers, Korean elders can be eased from pressure and frustration and instead be integrated into the community more readily. These important factors enhance a sense of belonging and identity as they feel comfortable expressing themselves freely (Lebreton, 2011).

Cultural Understandings of Ageing

A sensitive cultural understanding of ageing is threaded through the program. The content of the program has been tailored for this cohort, and a deep understanding of the participants' cultural background and experiences played an important role in its success. Fluency in Korean culture, language, and practice fulfils the needs of Korean immigrant elders.

It is common for Koreans to ask another person's age when meeting a new person. When asked this question, a Korean will identify their age according to the Korean school system. For example, if the member is 82 years of age, he would say, "I am in eighth grade, class number two." (This is a common way for Korean students to identify their grade and class number as at the beginning of the school year students are assigned to different classes identified by a number. The student will be identified as a member of the assigned class (class number 1, 2, 3, etc.) until the end of that school year. Despite my familiarity with Korean culture, I did not understand what was meant when I heard it for the first time in the program. Once explained, I was soon able to catch up because that was how I had identified myself during schooling. When Koreans refer to their age in this way, it is a play on words and a recollection of earlier school days.

Summary of Research Question 1

These factors contribute to the success of the CESC community-based ethnocultural program for seniors: 1) the program director is knowledgeable and sympathetic; 2) there is a focus on wellness and healthy ageing; 3) classes are conducted in the participants' mother tongue, Korean; 4) the college is structured like the traditional Korean school system; 5) a variety of classes and activities are offered; 6) familiar music is an integral part of the program; and, 7) the program is offered under the auspices of the Korean church which creates a safe and welcoming space for participants.

Research Question 2) How Does Participating in the Community Program Affect Health and Active Aging among Elderly Korean Participants?

Research question two is the subject of the fourth chapter of this dissertation. Recognizing the importance of providing culturally relevant services for elderly Korean immigrants that meets their cultural and linguistic needs, the integrated article entitled "Enoch Senior's College for Korean Immigrant Seniors: Quality of Life Effects" uses a qualitative survey. I chose the research instrument, the Health-Related Quality of Life (HR-QOL) Short Form 36 version 2 (SF-36v2) survey with advice from Dr. Patrick Potter from Western University's Schulich School of Medicine and Dentistry.

Health Survey

This study focused on the evaluation of the health effects of the program on elderly Korean participants using a self-reported health measure questionnaire, the SF-36v2, which has proven to be a reliable and valid assessment of self-perceived health and quality of life. Addenda to field notes I made as I observed classes plus the content of semi-structured interviews with eleven participants supplemented the findings from the SF-36v2 survey. The survey was distributed twice: The first survey was completed at the beginning of the term, and the second was distributed later, at approximately the mid-point of the program (between seven and ten weeks) to assess the participants' health through eight domains: physical function (PF), role-physical (RP), bodily pain (BP), general health (GH), vitality (VT), social function (SF), role-emotional (RE), and mental health (MH).

Findings from Health Survey

The findings revealed that significant differences in two domains, bodily pain and roleemotional, are correlated with a positive impact on the health of elderly Korean participants in the CESC program. Overall, participation at CESC resulted in a positive effect on HR-QOL and well-being though shared cultural heritage in the participants' mother tongue.

Three main themes emerged from the data analysis: participation in CESC promotes health and vitality 1) increasing Korean elders' activity levels and motivating them to take better care of themselves; 2) reducing social isolation and exclusion among Korean elders; and 3) providing social support and a sense of belonging.

Analysis of the HR-QOL survey revealed some cultural differences that impacted the results. For instance, despite the participants' references to increased social support from participating in CESC, the social functioning measure did not show significant change. Culturally, Korean elders are less likely to miss social functions or commitments they have made previously unless it is absolutely necessary. Many shared how they overcame mobility issues, such as the number of transfers they had to take when using public transportation to attend the program. It was crucial to understand how context and cultural differences affected the results of the quantitative measurement. I was reminded of the importance of taking precautions in

interpretating the data and to be mindful of the cultural differences between the study participants and the instruments used to acquire data.

Discussion of Program Features

Director Pastor Kim was well aware of the deep cultural value of music for Korean elders, and she made singing central to the program from its inception. Making music by singing, whether through greeting songs with movements, singing games, choir, solo singing, or other variants of song, is a living and active constituent part of the program. The program recognizes the importance of group membership and bonding. A greeting song with movements, or songs with improvised movements to accompany them, are often practiced as exercises between classes, keeping the Korean elders involved and active. Physical exercise is an integral part of CESC, ensuring that members move and thereby gain all-around physical, mental, and emotional health.

Group singing was an effective way to create a welcoming and inviting environment for the participants while affording an aesthetically satisfying learning experience. Among elders, one of the most significant learning needs concerns the meaning and value of one's life, for an elderly person is seldom defined by their employment (Fisher & Wolf, 2000). Old age is a time when a person can reflect on their past and re-evaluate what is meaningful to them. With expanded opportunities for social engagement, CESC promotes healthy living and activities through a combination of community music and lifelong learning which meets the needs of a marginalized elderly immigrant group.

Pastor Kim had long envisioned a culturally sensitive program for Korean elders. She says she is an old soul:

Since I was little, I liked to hang out with people much older than I. I even liked the smell of my grandparents, sitting by them, watching them, and listening to their conversation. I have always been more fascinated by them than people of my age. I felt kin to old people and wanted to do something for seniors.

Pastor Kim saw the needs of Korean elders and began the CESC as the first of its kind in Ontario. With a keen eye and heart for seniors, Pastor Kim's cultural competence helped to

shape a program tailored for Korean elders, a program dealing with real issues relating to ageing and culture. She said, "I always thought that music would be a big part of the program, simply because Koreans have always liked to sing. Music engages people and is entertaining, making the time joyful, which is good for health."

Many CESC members have commented on their enjoyment of music. Given that many Korean elders do not come from academic musical backgrounds, this finding is significant and suggestive. The influence of music in the program is clear, engaging the participants in an informal setting which allows them to participate and experience positive outcomes. These Korean elders felt better about themselves, both physically and emotionally, and they knew that, rain or shine, in health or in sickness, if they came out to the CESC, joined the rest of the group, and carried on with them, they would have a place which felt like home. Findings from this study are triangulated with the findings of the ethnographic and case studies also conducted during this investigation.

Research Question 3) What Does Music Mean and How Important is it in the Lives of Elderly Koreans?

The published paper in Chapter five addresses this research question. Entitled 'Remembering a Hard Life with Joy: Music Making Among Korean Elders,' this chapter explores the impact of lifelong learning and musical engagement in a community context from the narratives of a cohort of five Korean elderly participants. As these Korean elders reminisced and reviewed their lives during the interviews, music was intricately woven into their stories, enriching, engaging, and encouraging them in their lives in profound and unexpected ways.

Erik Erikson's Eighth Stage of Psychosocial Development is useful in understanding older adults as they evolve and redefine their sense of self. Deeply felt but not always articulated was the reality of death. Erickson (1968) posits that a healthy identity is gained by coming to terms with the past. Many of this cohort of Korean elders demonstrated this acceptance through their attitudes toward death and personal contentment. One unifying characteristic of the cohort was that every member of CESC is a survivor and has demonstrated resilience in the face of uncommon change. Particular to the cohort was their shared experience of Korean War and the aftermath of war that left their homeland, South Korea, in ruin. However, their generation

worked together with a rigorous work ethic to bring about a sea change in South Korea's economy while preserving traditional Korean values.

Shared Cultural Identity

This cohort expressed a strong cultural identity through shared musical knowledge, songs from childhood, and schooling. Recreating the time of their youth by singing, playing, and performing allowed these elderly Koreans to be immersed into the CESC setting, providing opportunities for self-exploration and self-expression. Reminiscing, remembering, and reviewing the past through music making contributed to these elderly Koreans' search for meaning and coherence in life review (Cohen, 2005).

Music Making

Elderly Koreans' love of music and music-making is seen in the program. They appreciated opportunities to make music, singing together cherished old songs as they reminisced and gained new meaning from the experience. Old songs took on new meaning; meanings that these elders had taken for granted were seen in new light.

Life Narratives

Five Korean elders shared their own narratives about music in their lives. Their life stories illuminate how music, sometimes in quiet, unassuming ways and at other times profoundly, has been part of elderly Koreans' lives. The positive influence of music-making in their lives is summed up by their successful participation in the program. Their genuine appreciation for music was apparent.

An honest work ethic and determination distinguishes Kay, a natural leader, who played a pioneering role in Korean immigration history. For Kay, music is like a best friend, someone who has accompanied her throughout her life, enriching her life, like her partner and best supporter, Yong. Together, as a couple, Kay and Yong enjoy singing in a choir. For Yong, playing music was like a home for his heart, where he found solace in his quiet time. Like Yong, his music is gentle and dignified. He found playing musical instruments for recreation was a way to experience the world in his own quiet way.

A tall and quiet man, Pastor Kim transforms into a confident performer when he picks up his accordion or stands on a podium to conduct his choir. To him, music is his identity, a sure calling that set him on a steadfast way of life.

Min's personality blossomed as she shared her new-found identity. Never in her life did she imagine that she would sing or compete in a singing festival. A song, like a spirit, came to her, drawing her out of deep grief after her husband's sudden death. She found hope and a new self and is now heartened to share a new life with others through singing.

Jun explains the ways that songs spoke to him now in older age: "It may have to do with age. I was too young and busy to appreciate it. The song lyrics are like poetry, you know. It gives us wisdom about life, but I was too young, and too ambitious, and too proud to see that then. But now that I come here and sing it again, then suddenly!" Jun was surprised to realize how the song speaks to him, reflecting his own life. Jun continued:

Bongseonhwa (Balsam flower), the song we just sang (in the group singing class), was a song that I always knew, a long time ago. I first learned it during elementary and never really appreciated it, nor did I give any thought about it. I thought that music was irrelevant to me. However, nowadays, I sing it alone at home and meditate on the words. The song is originally about oppression during Japanese annexation, but I realized that it tells the story of our lives. Especially the third verse speaks to my heart. As we get old, and lose strength, our flesh perishes, but soul will remain. I meditate the words softly at night lying on my bed.

Bongseonhwa²⁷

The lyrics (a lyrical poem, meaning Balsam flower, was written as propaganda underlining a hidden message of a cry for freedom from oppression, the lyrics describe the sorrow and agony of suppression by the Japanese)

1. Bongseonhwa flowers under the fence look pitiful. When you were in bloom in a long, long summer days, Pretty girls were playing with you for nail coloring.

울밑에선 봉선화야 네 모양이 처량하다 길고 긴 날 여름철에 아름답게 꽃필적에 어여쁘신 아가씨들 너를 반겨 놀았도다

²⁷ Translated in English by Whon-il Park (2022). http://www.koreanlii.or.kr/w/index.php/Korean_lyric_songs?ckattempt=1

2. Now the summer is gone with cool wind blowing. So the beautiful petals are encroached by cold. As those flowers disappeared, you look so pitiful.

어언간에 여름 가고 가을바람 솔솔 불어 아름다운 꽃송이를 모질게도 침노하니 낙화로다 늙어졌다 네 모양이 처량하다

3. Tho' you've perished in freezing winter, here is your spirit dreaming of a peaceful dream. I wish you revive again in a splendid spring breeze.

북풍한설 찬바람에 네 형체가 없어져도 평화로운 꿈을 꾸는 너의 혼이 예 있나니 화창스런 봄바람에 환생키를 바라노라

Song lyrics, once sung without thought at the time, are revealed anew as an insightful reflection of one's own life. It is as if these elders grew up with the song, growing into it to express their own stories. Music is central to their identity, shaping, redefining, and renewing who they are. For example, as he reflected on the song lyrics of Bongseonhwa which he learned as a boy, Jun has discovered that the song is a story of his own life.

Music assigns Meaning and Value to Life

In old age, one of the most significant learning needs is about assigning meaning and value to one's life, for a person is no longer defined by employment (Fisher & Wolf, 2000). Mr. Song shared his experience about finding solace through music. Long a devotee of music, music making has always had a special place in his heart. His devotion to and love for music was not like the attitude towards music of others in the group. From the first time he sang a song before his classmates, a lifetime relationship with music began. Although a businessman by trade, choir music fascinated him whether he was conducting, composing, or singing. Musicality, making a perfectly harmonized sound with balanced dynamics, tempo, and tone, was important for him. Despite a heart condition, he cherished making great music, and, despite ageing, he shared his aspirations with mixed emotions. Music had always been an important part of his identity, something in which he took pride. Unlike some other members of the cohort, he enthusiastically shared his passion for music making despite disappointments. Mr. Song was most vocal about his emotions and concerns. His identity as a musician was always very clear, which was unusual among the members of the group.

For most members of the cohort, music was an important outlet for emotions. It was elderly Koreans' most accessible method for expressing their identity. Singing is especially valued because it has accompanied Korean lives throughout history and continues to do so today; it is an important cultural continuum and echo of the Korean *Min-yo* folk song tradition. Further, Korean elders voiced their emotions through singing and this act created a feeling of social connection which is important for wellbeing and active ageing.

Hwa-Byung and Need for Social Connection

There is a condition specific to Korea termed *Hwa-Byung*. *Hwa-Byung* is an illness caused by the suppression of emotions. Koreans are reportedly vulnerable to psychological distress and social isolation. There is a strong association between *hwa-byung* and depressive disorders (Im, 2021). Many Koreans do not feel comfortable sharing their feelings; reticence is a culturally valued disposition, especially among older Koreans, which helps to explain the prevalence of *hwa-byung* and the risk of depressive disorders among elders (Min & Suh, 2010; Park, 2014; Shin et al., 2007). It is commonly felt by Koreans that social connection such as that afforded through music making can prevent and alleviate such conditions.

Korean elders have experienced war, social upheaval, and the re-building of a nation. They are a resilient group of people who have rich life experiences; as immigrants, they have a unique story of their own to tell. As Kay said, "I could write a book about my experience of war alone." Every member with whom I had a chance to talk had a remarkable life story, with experiences that are not like those of anyone else. Each member of the cohort chose a path in life and lived to an old age, and is now confronting the realities of ageing, deteriorating health, and changes in status, as well as death.

What does music mean to them? There seemed to be an unstated and unquestioned assumption and belief among the study participants that music making is healthy and good and fundamental to social engagement. While these elders' life stories do not center on music, music had played an important part in their lives.

Further Study Findings

In the previous pages I have reviewed the findings that directly emerged from the three research questions that propelled this dissertation. In this next section, I will discuss seven other topics that emerged:

ageing and the quality of life;
wellness and active ageing;
importance of social support;
COVID-19 Pandemic and social isolation;
senior Korean immigrants in North America;
government initiatives and local stopgap measures; and
Canadian lifelong learning programs for seniors

Ageing and Quality of Life

As this study and a growing body of research documented in the literature survey in Chapter 2 indicate, wellness and active ageing are increasingly important components of a healthy population. Social isolation is a major and prevalent issue for older adults, something which negatively impacts their health, well-being, and quality of life, regardless of their nationality (Cornwell et al., 2008; Coyle & Dugan, 2012; Nicholson, 2012; Syed et al., 2017; Taylor, 2019; Tomaka et al., 2006; Veazie et al, 2019). Old age is a time when life events like retirement from work, increased care and caregiving responsibilities, and the loss of spouse, family, or friends occur, making elders increasingly socially isolated. This prevalence of social isolation is concerning because of its connection to an increased rate of morbidity and mortality in elderly populations (Berg, & Cassells, 1992; Freedman, 2020; Holt-Lunstad et al., 2015).

A meta-analytic review that examined associations between social relationships and the risk of mortality found that this risk rate is like that of smoking and alcohol use, surpassing that of obesity and lack of exercise (Holt-Lunstad et al., 2015). The preponderance of evidence demonstrates that social isolation and loneliness have a detrimental impact on physical and psychological outcomes, including high blood pressure, obesity, and cardiovascular, inflammatory, and neuroendocrine disorders, and anxiety, depression, cognitive deterioration,

Alzheimer's disease, and even increased mortality (Bhatti & Haq, 2017; Holt-Lunstad et al., 2010; Nicholson, 2012; Steptoe et al., 2013; Xia & Li, 2018).

Wellness and Active Ageing

The World Health Organization (WHO) defines active ageing as "the process of optimizing opportunities for health, participation and security in order to enhance quality of life as people age" (WHO, 2002, p. 12). WHO's active aging model is based on the three key domains of health, participation, and security. This model represents a comprehensive life-course approach to ageing, addressing both personal and environmental factors (Foster & Walker, 2015; Marsillas et al., 2017; Walker, 2002). The WHO active aging model of active ageing takes a more comprehensive perspective than other models such as 'successful ageing' (Rowe & Kahn, 1998), 'productive ageing', or 'healthy ageing' (Walker, 2002; World Health Organization, 2002). It provides opportunities for the physical, social, and mental well-being of the elderly population, in addition to their participation in society (International Longevity Centre, 2015). If 'successful ageing' or 'healthy ageing' focuses more on the achievement of clinical standards, 'active ageing' highlights actions, not only at the individual level but also at organizational and social levels. This 'active ageing' model has been widely adopted in Europe to promote policy strategies which increase the involvement of older adults in society (European Commission, 2013; Foster & Walker, 2015; International Longevity Centre, 2015).

The 'active ageing' framework is utilized with older adults for their physical, psychological, and social well-being as participants in society by providing continuous or lifelong opportunities for health, community participation, and security (World Health Organization, 2002). Health, in terms of the WHO definition, including the promotion and protection of healthy lifestyles and access to health and social services information, is important. Participation refers to engagement in various cognitive, social, recreational, economic, and cultural activities which provide a sense of meaning and belonging. Security involves the areas of food provision, finance, and social services, as well as physical and environmental security.

Three components are important for active ageing: social activities, social support, and physical activities (Pruchno et al., 2010). Social activities include social and community participation, including community and leisure activities; social support highlights the

measurement of social relationships by social satisfaction and networking; physical activities involve active participation (Holt-Lundstad et al., 2010). Increasing opportunities for community participation in physical activities also increases social contact just as improvement in social skills reduces feelings of depression, loneliness, and functional decline (International Longevity Centre, 2015). It is probable that seniors in better health will have more opportunities for social participation, which will improve mental health.

My dissertation follows musical engagement and wellness in a single community program in Canada. However, programs such as Enoch (CESC) provide a personal voice and intensive focus on immigration and the changing demographics in Canada (and to some extent in the rest of North America) as will be seen in the sections that follow.

Specifically, and most pertinent to this study, elderly Korean immigrants' strong ethnic attachment contributes positively to their overall health and well-being by reducing the loneliness and isolation that have caused major challenges among immigrant seniors of ethnic minorities (O, Kim, 1999b). By eating ethnic food, communicating in Korean, listening to, and watching Korean programs (TV, YouTube, Music), and joining together to make and enjoy the music of their youth, elderly Korean immigrants maintained their cultural identity and sense of belonging (Son & Kim, 2006).

Importance of Social Support

Social support can offer a means of addressing maladaptive conditions. A longitudinal national ageing study (Dahlberg et al., 2018) reported that social support was negatively associated with loneliness (i.e., perceptions or negative feelings of social isolation). Lai and colleagues (L. J. Lai et al., 2020) conducted a peer support intervention program involving 60 elderly Chinese immigrants in Canada by assigning them to either an experimental or a control group in a randomized control parallel trial design. Twenty-four volunteers engaged in the eightweek study, providing social support through home visits and telephone calls to the experimental group. The experimental group experienced a significant reduction in loneliness, demonstrating that participation in social groups and activities with an educational or training component was effective.

Health practitioners are increasingly aware of the need for supportive social settings to aid seniors in keeping actively engaged in life. The World Health Organization (WHO) and various other organizations have made recommendations to help seniors remain mentally and physically healthy. These organizations include the American College of Sports

Medicine (ACSM), American Heart Association (AHA), American Physical Association
(APTA), International Association for Physical Therapists Working with Older People (IPTOP)
(WHO, 2020a). Canada's National Seniors Council adopted WHO's (2007) age-friendly communities model, which uses a positive and active-ageing approach to promote active lifestyles—because older adults with the most social interaction within their communities had the slowest rates of memory decline (Ertel et al., 2008). WHO's model addresses eight domains that promote living with security, good health, and full participation in society: transportation, housing, social participation, respect and social inclusion, civic participation and employment, communication and information, community support and health services, and outdoor spaces and buildings.

Participating in a variety of activities can increase level of physical activity as well as degree of interaction with others, which enriches quality of life for seniors. Social participation, according to WHO (2001), involves active participation in various areas including sports, cultural, recreational, and religious groups, and political and volunteer community organizations.

A review of the effectiveness of North American-based health promotion interventions (Cattan et al., 2005) found that the most effective interventions were educational or training group activities targeting specific populations such as women, caregivers, the widowed, and people with serious mental problems. Such programs are beneficial in prompting individuals to assert their independence though joining social groups. Health is holistic, not merely limited to dietary prescriptions or an exercise routine or visits to a doctor. Whether a person consciously realizes that musical engagement is what they want, it has proven to be important for their health and well-being.

COVID-19 Pandemic and Social Isolation

Beginning in 2019, Coronavirus disease-2019 (COVID-19) outbreaks caused great concern in more than 223 countries and regions (Priyadarshini, et al., 2020; Cucinotta, &

Vanelli, 2020; WHO, 2020b). The ongoing global pandemic has led to mandatory social distancing and stay-at home or similar quarantine orders to prevent the spread of virus. Older adults are identified as being at a higher risk of infection with high mortality rates reported in seniors aged 65 and older (WHO, 2020b). Seniors, who are already exposed to high rates of social isolation, are particularly susceptible to the adverse effects of social distancing protocols (Berg & Cassells, 1992; Chang, et al., 2021; Sayin Kasar, & Karaman, 2021; Roy et al., 2020; Smith et al., 2020; Taylor et al., 2018).

There is a strong correlation between social isolation and negative health-related quality of life outcomes (Hawton et al., 2011; Victor et al., 2000). Social isolation has adverse effects on overall well-being; it is associated with higher levels of depression and a decrease in feelings of belonging and the social connectedness that is related to life satisfaction in older adults (Courtin & Knapp, 2017; Dahlberg et al., 2018; Lee & Cagle, 2018). Social distancing and stay-at-home orders such as those imposed during COVID outbreaks contribute negatively to well-being (Harasemiw et al., 2018; Zavaleta, et al., 2017; Yang et al., 2020). A wide body of literature supports the ameliorating effects of social support on social isolation and loneliness among older adults (Chen et al., 2019; Hawkley & Kocherginsky, 2018; Wang et al., 2011). Researchers note that positive and rewarding social relationships are important contributors to positive psychological health (Myers & Diener, 1995; Uchino et al., 1999).

Studies investigating the effects of the pandemic and will continue to emerge. Although direct data-gathering for my studies had ceased before COVID-19 began, I am still in touch with with some CESC participants and Pastor Kim who has since retired. These participants noted that the pandemic has imposed hardships on many people, compounded by the limitations and boundaries that all immigrants face, even after years of successfully negotiating their new culture. A number of Korean seniors are returning to Korea to find the security of a national identity, a mother tongue, and familiar foodways. Those who remain in Canada find themselves isolated. Some have died, and I speculate that social isolation speeded their demise.

Senior Korean Immigrants in North America

Korean senior immigrants in North America are likely to experience language barriers. They may be unprepared or unable to access health care and social service systems, services that are freely available under South Korea's universal health coverage (Terrazas & Batog, 2010; Song, 2009).

Korean churches have historically provided needed support as well as a sense of community for Korean seniors. Pastors take the responsibility to address elders' needs and to provide for their congregation. Some 43% of the Korean churches provided social services and 37% provided healthcare needs (E. Kim, 2016). Approximately 80% of Korean immigrants attend church in United States while only 14%-30% of Koreans in South Korea attend church on a regular basis (Chang, 2005; W. Kim, 2012; Y.S. Kim, 2013; H. B. Lee et al., 2008; Min & Kim, 2002).

With the growth in numbers of Korean immigrants, Korean Senior Citizens Society of Toronto offered educational, and recreational programs, although these were temporarily closed during the COVID-19 pandemic. Eight similar seniors' programs are hosted by Korean Canadian churches across the greater Toronto area; however, music is not a focus for these programs. As well, the CESC has a history of high attendance and retention that serves as a model for other programs. In addition to church-based programs, other community-based lifelong learning program for older adults across Toronto documented support positive impact from participation (Jenkins & Mostafa, 2015; Narushima, Liu, & Diestelkamp, 2018).

Often, Asian immigrant seniors rely on family and friends for support, and these connections have positively contributed to their overall health and well-being. Researchers noted that providing a culturally relevant and appropriate social support within community may be significant for ethnic minority seniors (Lee & Chan, 2009; J. W. Min et al., 2005; Wong et al., 2007).

Government Initiatives and Local Stopgap Measures

The Canada Ministry of Health and Long-Term care has established Community Care Access Centres (CCACs). In Ontario, fourteen CCAC centers are available to seniors who need healthcare and home care services. Seniors can contact their local Community Care Access Centre for access to services such as homemaking, nursing, physiotherapy, speech-language therapy, social work, and case management/co-ordination. Health related services such as

provision of medical supplies and wound dressings, transportation to other health care services, and drug coverage under the Ontario drug benefit plan are also available. Home and Community Care Support Services provided care to more than 644, 000 patients in 2021-2022 (Home and Community Care Support Services, 2022). The service, however, requires communication in English and only a limited number of elderly Korean immigrants are proficient enough to communicate with the care coordinator.

Besides the government initiatives, there are local stopgap measures in place to help specific cultural groups through ethnic centers featuring native language and culture. For instance, the Yee Hong Center established in 1994 provides services in six locations for elderly Chinese immigrants. Yee is the largest non-profit nursing home in Canada. As well, Yee Hong Center offers housing services, and community care services (Yee Hong Centre for Geriatric Care, 2017)

Concerned about the needs of Korean seniors, a group of second generation Korean Canadian doctors and lawyers came together in partnership with 12 Korean religious institutions in an effort to offer health related services in the Korean language and culture. Their collaboration resulted in the Arirang Age-Friendly Community Center, established in 2015. Although a seniors' day program is offered in partnership with Mil-Al Korean church, it is entirely funded by donations. Unfortunately for the program, few people have chosen to participate. Lack of government funding makes it difficult to offer such cultural community-based programs.

Canadian Lifelong Learning Programs for Seniors

Lifelong learning programs for seniors have been documented in Canadian contexts, although these programs are not music centered. Narushima (2008) interviewed 15 older adult participants involved in non-formal, leisure related public continuing education community programs. He reported that participants were motivated to keep physically and mentally active while experiencing increased social support. This research suggests that non-formal, leisure-based types of learning can be beneficial because they are more likely to sustain participation in lifelong learning courses (Jenkins & Mostafa, 2015; Narushima, 2008).

More recently, a cross-sectional study examined the relationship between duration of participation in lifelong learning programs and psychological wellbeing with 416 older adults (aged 60 years and above) enrolled in community-based program. Findings indicated positive effects from continuous engagement in lifelong learning activity (Narushima et al., 2018). The study's results also support a positive association between lifelong learning and the health and wellbeing of older adults with severe chronic conditions such as diabetes, high blood pressure, arthritis. As well, the study's findings support lifelong learning activities as providing an important resource for self-esteem, self-efficacy, and resilience, and for sustaining psychological wellbeing and health in older adults (Narushima et al., 2018). However, there are no studies that focus on music in the community-based setting for visible minority seniors despite the high rate of immigrants in Canada.

Researcher's Stance

During my school days, part of the high school entrance exam was a physical test. The last part of the physical test, long distance running, fell in the early evening after a full day of testing. We were required to run multiple laps of the school field. To my surprise, the teacher blew the whistle to gather everyone in the class together; we were made to line up four abreast. Together we jogged alongside our teacher who encouraged slower runners to keep up with the group. Amazingly, everyone who stayed in line succeeded while those who fell out of the line did not pass the final test. There I learned the value of teamwork; the experience made a lasting impression. Never before had I appreciated a physical education teacher, nor had I ever had a chance to run along with others to make the final line together. As I look back to times I spent with Korean elders, I am reminded of jogging laps. Even though no one said a word, I know that I was cheering for my classmates and physical-education teacher.

How has meeting Korean elders impacted my life? I have gained a new insight about older adults and no longer see them as old but as exuberant with life that I never knew until I lent my heart to them. As I heard their stories, I learned that their stories are my story, too. I realized that the best I could do was to tell their stories plainly, accepting them as they are. Perception is already biased. How could I avoid judgement? This question directed me to see more clearly.

I remember one elderly participant saying, "I couldn't sleep because I was in pain, but I am here." She mentioned it as a matter of fact, sitting up straight, focused, and ready to live the moment. Similarly, many others shared painful health conditions. They coped, and their suffering was not visible unless they opened up about it. Through ethnographic observation as well as interview, I became more aware of their physical conditions. At the same time, most of them were capable and represented a high functioning group. Then one day, I realized that I need to re-test my views and re-examine dominant perceptions related to ageing population.

I am saddened as well as happy to have done this research -- saddened because I've lost dear people to death. In my memory they are still living and active, playing the harmonica and singing the songs, and my heart goes out to them. This project has been dear to me because I was forced into a journey which has had outcomes and effects I could not have anticipated. I am no longer the same person I used to be, scared, ashamed, and little. People used to step all over me. Once I got yelled at for bumping into a woman although I had never seen her before. I said I was sorry and wondered what I had done wrong. Was it my existence that I should feel sorry about? I wondered, many times. If I were to bump into you and happen to get in your way, I would apologize but not because I am sorry that I am alive; I would apologize because I admit that I am a person of many faults and inadvertently got in your way.

Implications

The study I have reported here is useful for those seeking to understand the changing demographics of ageing minority populations in Canada. It offers insights for those seeking to guide government policies, as well as provincial and local initiatives. In addition, music educators, community musicians, researchers, community program directors, and those involved in health fields may find the study valuable.

Recommendations for Practice and Future Research

Music has been shown to play an integral role in people's everyday lives in the construction, negotiation, and expression of identity (Hargreaves et al., 2002). Immigrants face challenges to redefine who they are as they encounter unfamiliar ways of life in a new culture. Music education has been shown to support the acculturation processes of immigrant students

(Karlsen & Westerlund, 2010; Marsh, 2013; O'Hagin & Harnish, 2006). One study documents how immigrant students listened and sang homeland songs that connected them with positive times from their past (Karlsen, 2013). In school contexts, musical activities can foster a safe environment for immigrant children (Marsh, 2013). Other studies explored the meaning of music in the lives of young immigrants in Scandinavia (Karlsen, 2012; 2013) and of immigrant students in Australia (Marsh, 2012; 2013). However, few studies have documented older immigrants' musical experience and its impact on their wellbeing. In North America, Cho's (2018) research explored the long-term effects of musical experience on an "unheard group" of older immigrants who focused on homeland music. Cho's (2018) study reveals the diverse meanings and roles of music through subjective musical experiences of three Korean older immigrants, in California, United States. Her findings resonate with the narratives in this study, yet Cho's study is, to my knowledge, the only research thus far into the musical engagement in Korean older immigrants in a North American context besides my own work. Clearly there are opportunities for further work in this field.

My research offers insights into lifelong music learning and engagement although it is limited. Corresponding studies into lifelong learning (Cai et al., 2018; Duay & Bryan, 2008; Findsen & Formosa, 2011; Hori & Cusack, 2006; Maulod & Lu, 2020). There are studies that take lifelong learning wholistically, as an important process, highlighting that learning occurs across the lifespan (Lindman, 1989). In this dissertation I present a three-pronged investigation into music making among seniors in a specific cultural setting. The strength of this research rests in its specificity as Cho's (2018) study was able to illustrate the long-term impact and meaning of 'homeland' music in everyday lives of Korean older adults in American context. Likewise, this is a limitation since any group of people and any other cultural setting might yield multiple outcomes.

This study used a mixed methods approach. The ethnographic components were especially revealing. The individuals who shared their life stories with me bring vivid pictures of their long, hard, but joyous lives, particularly evoked through their musical memories. Future researchers might find similar rich and valuable narratives. As well, the health survey was valuable since it served as a counterpoint to individual interviews and other ethnographic data.

I am a native Korean speaker, an invaluable asset for working with immigrant Koreanborn elders. It is important, especially in an increasingly diverse world and especially in pluralistic societies such as Canada, to be aware and appreciative of the strengths of many cultures. Therefore, future researchers who are from minority cultural backgrounds might find their research valuable in addressing under-represented populations.

As this investigation concludes, the pandemic continues, and so will its effects. Future studies will be needed to document enduring outcomes from COVID. As well, there will be a need to research social media and technology as a way to ease isolation.

Summary

In the study, seniors described listening to songs as important if sometimes bittersweet. They reported that music making was a meaningful activity that connected them with others. Some noted ways it helped them express their spirituality. My dissertation adds to a body of research on the role of active musical engage in lives of seniors (Hays & Minichiello, 2005; Cohen et al., 2005; Carucci, 2012; Coffman, 2002; Creech et al., 2011).

At the same time, it is important to realize that these populations are at risk of alienation and marginalization (Kim, 1999; Kim, 2013). Cultural unfamiliarity can pose further challenges to immigrants especially ethnic minority elders vulnerable to psychological distress with lack of information or misinformation of existing support and healthcare systems (Lai & Chau, 2007; Marin, 2009; Mui & Kang 2006; Choi & Smith, 2004). These obstacles are real but may be ameliorated through directed community programs, as shown in this research.

My research considers the value of music in engaging and promoting wellness and active ageing among Korean Canadian seniors. Having a strong sense of meaning and purpose as can be afforded by music is not only associated with well-being in old age, but also relates positively to successful coping with both pain and illness.

Although limited to a specific population in a particular time and context, this study contributes to growing body of research, filling an unaddressed concern. Voices of Korean Canadian seniors offer valuable insights for music educators, community musicians, researchers, community program directors, health care providers as well as those guiding government policy.

"The story of my life," as older adults often began, was never dull, not because the stories were happy but because they were testimonies to their lives.

References

- Adams, K. B., Leibbrandt, S., & Moon, H. (2011). A critical review of the literature on social and leisure activity and wellbeing in later life. *Ageing and Society*, *31*(4), 683-712. https://doi:10.1017/S0144686X100
- Ahmad, F., Shik A., Vanza, R., Cheung, A. M., George, U., & Stewart, D. E. (2004). Voices of South Asian women: Immigration and mental health. *Women's Health*, 40(4), 113-130. https://doi:10.1300/J013v40n04_07
- Ahmed, S., Shommu, N. S., Rumana, N., Barron, G. R. S., Wicklum, S., & Turin, T. C. (2016).

 Barriers to Access of Primary Healthcare by Immigrant Populations in Canada: A

 Literature Review. *Journal of Immigrant and Minority Health*, 18(6), 1522–1540.

 https://doi.org/10.1007/s10903-015-0276-z
- Anderson, B. (2016). *Imagined communities: Reflections on the origin and spread of nationalism* (Revised ed.). Verso. ISBN: 9781844670864
- Amit, V. (2002). Reconceptualizing community. In V. Amit (Ed.), *Realizing community: Concepts, social relationships and sentiments* (pp. 1-20). Routledge.
- Amit, K., & Litwin, H. (2010). The subjective well-being of immigrants aged 50 and older in Israel. *Social Indicators Research*, 98(1), 89–104. https://doi.org/10.1007/s11205-009-9519-5
- Anokye, N. K., Trueman, P., Green, C., Pavey, T. G., & Taylor, R. S. (2012). Physical activity and health related quality of life. *BMC Public Health*, *12*(1), 624–624. https://doi: 10.1186/1471-2458-12-624
- Asquith, N. (2009). Positive ageing, neoliberalism and Australian sociology. *Journal of Sociology*, 45(3), 255–269. https://doi.org/10.1177/1440783309335650
- Ayers, J. W., Hofstetter, C. R., Usita, P., Irvin, V. L., Kang, S., & Hovell, M. F. (2009). Sorting out the competing effects of acculturation, immigrant stress, and social support on depression: A report on Korean women in California. *Journal of Nervous and Mental Disease*, 197(10), 742–747. https://doi.org/10.1097/NMD.0b013e3181b96e9e
- Bailey, B.A., & Davidson, J.W. (2002). Adaptive characteristics of group singing: Perceptions from members of a choir for homeless men. *Musicae Scientiae*, 6(2), 221–256. https://doi.org/10.1177/102986490200600206

- Bailey, B. A., & Davidson, J. W. (2005). Effects of group singing and performance for marginalized and middle-class singers. *Psychology of Music*, *33*(3), 269–303. https://doi.org/10.1177/0305735605053734
- Baker, S., Marryshow, T., Poteau, S., & Sutin, A. (2015, May 15-17). Exploring the stigma of depression among older adults: Results from a community survey. *American Geriatrics Society* annual meeting, National Harbor, MD, U.S.A.

 https://med.fsu.edu/sites/default/files/userFiles/file/AGS%202015%20Depression%20Poster%20FINAL%205-7-15.pdf
- Bankston, C. L., & Zhou, M. (1996). The ethnic church, ethnic identification, and the social adjustment of Vietnamese adolescents. *Review of Religious Research*, *38*(1), 18-37. https://doi.org/10.2307/3512538
- Bauman, Z. (2013). Community: Seeking safety in an insecure world. Polity Press.
- Barrio, C., Palinkas, L. A., Yamada, A., Fuentes, D., Criado, V., Garcia, P., & Jeste, D. V. (2007). Unmet needs for mental health services for Latino older adults: Perspectives from consumers, family members, advocates, and service providers. *Community Mental Health Journal*, 44(1), 57–74. https://doi.org/10.1007/s10597-007-9112-9
- Bartleet, B.-D., & Higgins, L. (2018). (Eds.). *The Oxford handbook of community music*. Oxford University Press. https://doi.org/10.1093/oxfordhb/9780190219505.001.0001
- Batt-Rawden, K. B. DeNora, T., & Ruud, E. (2005). Music listening and empowerment in health promotion: A study of the role and significance of music in everyday life of the Longterm ill. Nordic *Journal of Music Therapy*, *14*(2), 120-136. https://doi.org/10.1080/08098130509478134
- Beland, F., Zunzunegui, M. V., Alvarado, B., Otero, A., & Del Ser, T. (2005). Trajectories of cognitive decline and social relations. *The Journals of Gerontology. Series B, Psychological Sciences and Social Sciences, 60*(6), P320–P330. https://doi.org/10.1093/geronb/60.6.P320
- Bell, J. (2005). Doing your research project: A guide for first-time researchers in education, health, and social science (4th ed.). Open University Press.
- Berg, R. L., & Cassells, J. S. (Eds.). (1992). Social isolation among older individuals: The relationship to mortality and morbidity. In *The second fifty years: Promoting health and*

- *preventing disability* (pp. 243-261). National Academies Press. https://www.ncbi.nlm.nih.gov/books/NBK235604/
- Bergeron, P., Auger, N., & Hamel, D. (2009). Weight, general health and mental health: Status of diverse subgroups of immigrants in Canada. *Canadian Journal of Public Health* 100(3), 215-220. DOI: 10.1007/bf03405544
- Berkman, L. F., & Syme, L. (1979). Social networks, host resistance, and mortality: A nine-year follow-up study of Alameda County residents. *American Journal of Epidemiology*, 109(2), 186-204. https://doi.org/10.1093/oxfordjournals.aje.a112674
- Bernard, H. R. (2011). Research methods in anthropology: Qualitative and quantitative approaches (5th ed.). AltaMira Press.
- Bernstein, K. S., Park, S. Y., Shin, J., Cho, S., & Park, Y. (2011). Acculturation, discrimination and depressive symptoms among Korean immigrants in New York City. *Community Mental Health Journal*, 47, 24–34. http://dx.doi.org/10.1007/s10597-009-9261-0
- Bhatti, A., & Haq, A. U. (2017). The pathophysiology of perceived social isolation: effects on health and mortality. *Curēus (Palo Alto, CA)*, *9*(1), e994–e994. https://doi.org/10.7759/cureus.994
- Biggs, S., Carstensen, L., & Hogan, P. (2012). Social capital, lifelong learning and social innovation. In J.R. Beard, S. Biggs, D. E. Bloom, L. P. Fried, P. Hogan, A. Kalache, & J. Oshansky (Eds.) Global population ageing: Peril or promise? (pp. 39–41). World Economic Forum, 2011.
- https://www3.weforum.org/docs/WEF_GAC_GlobalPopulationAgeing_Report_2012.pdf
- Blaine, T., & Fels, S. (2003). Collaborative musical experiences for novices. *Journal of New Music Research*, 32(4), 411–428. https://doi.org/10.1076/jnmr.32.4.411.18850
- Bochner, A. P. (2001). Narrative's virtue. *Qualitative Inquiry*, 7(2), 135-157. https://doi.org/10.1177/107780040100700201
- Bochner, A. P. & Ellis, C. (2003). An introduction to the arts and narrative research: Art as inquiry. *Qualitative Inquiry*, *9*(4), 506-514. https://doi.org/10.1177/1077800403254394.
- Boddie, S. C., Hong, P. Y. P., Im, H., & Chung, S. (2011). Korean-American churches as partners in community development. *Social Work & Christianity*, *38*(4), 395-416.
- Bogdan, R. C. & Biklen, S. K. (2003). Qualitative research for education: An introduction to theories and methods (4th ed.). Allyn & Bacon.

- Born, G., & Hesmondhalgh, D. (2000). Western music and its others: Difference, representation, and appropriation in music. University of California Press.
- Boudiny, K. (2013). 'Active ageing': From empty rhetoric to effective policy tool. *Ageing and Society*, *33*(6), 1077-1098. https://doi.org/10.1017/S0144686X1200030X.
- Boulton-Lewis, G., Buys, L., & Lovie-Kitchin, J. (2006). Learning and active aging. *Educational Gerontology*, 32(4), 271–282. https://doi.org/10.1080/03601270500494030
- Boulton-Lewis, G., & Tam, M. (Eds.). (2012). *Active Ageing, Active Learning: Issues and Challenges*. Springer Netherlands. https://doi.org/10.1007/978-94-007-2111-1
- Boulton-Lewis, G. M., & Buys, L. (2015). Learning choices, older Australians and active ageing, *Educational Gerontology*, 41(11), 757-766.

 https://doi.org/10.1080/03601277.2015.1039455
- Boyer, S., Edmondson, D., Artis, A., & Fleming, D. (2014). Self-directed learning: A tool for lifelong learning. *Journal of Marketing Education*, *36*, 20–32. https://doi.org/10.1177/0273475313494010
- Bresler, L. (1995). Ethnography, phenomenology and action research in music education. *Quarterly Journal of Music Teaching and Learning*, 6(3), 4–16.
- Brooker, D., & Duce, L. (2000). Wellbeing and activity in dementia: A comparison of group reminiscence therapy, structured goal-directed group activity and unstructured time.

 *Aging & Mental Health, 4(4), 354–358. https://doi.org/10.1080/713649967
- Brooks, R. (1996). EuroQol: The current state of play. Health Policy, 37, 53–72.
- Bugos, J. A., Perlstein, W. M., McCrae, C. S., Brophy, T. S., & Bedenbaugh, P. H. (2007). Individualized piano instruction enhances executive functioning and working memory in older adults. *Aging and Mental Health*, *11*(4), 464-471. https://doi.org/10.1080/13607860601086504
- Bugos, J. A. (2014). Community music as a cognitive training programme for successful ageing. *International Journal of Community Music*, 7(3), 319–331. https://doi.org/10.1386/ijcm.7.3.319_1
- Burnside, I., & Haight, B. (1994). Reminiscence and life review: therapeutic interventions for older people. *The Nurse practitioner*, 19(4), 55–61. https://doi.org/10.1097/00006205-199404000-00011

- Bush, J. E., & Krikun, A. (2013). Community music in North America: Historical foundations In K. K. Veblen & S. J. Messenger, M. Silverman, & D. J. Elliott (Eds.), *Community music today* (pp. 13-24). Rowman & Littlefield Publishers Inc. ISBN 978-1-60709-320-6
- Butler, R. N. (1963). The life review: an interpretation of reminiscence in the aged. *Psychiatry*, 26(1), 65–76. https://doi.org/10.1080/00332747.1963.11023339
- Cammisuli, Cipriani, G., Giusti, E. M., & Castelnuovo, G. (2022). Effects of reminiscence therapy on cognition, depression and quality of life in elderly people with Alzheimer's disease: A systematic review of randomized controlled trials. *Journal of Clinical Medicine*, 11(19), 5752—. https://doi.org/10.3390/jcm11195752
- Carucci, C. (2012). An investigation of social support in adult recreational music ensembles. *International Journal of Community* Music, 5(3), 237–252. https://doi.org/10.1386/ijcm.5.3.237_1
- Carrière Y., Martel L., Légaré J., & Picard J.-F. (2016). The contribution of immigration to the size of ethnocultural diversity of future cohorts of seniors, Insights on Canadian society (Catalogue no. 75-006-X). Ottawa: Minister of Industry. http://www.statcan.gc.ca/pub/75-006-x/2016001/article/14345-eng.pdf
- Cattan, M., White, M., Bond, J., & Learmouth, A. (2005). Preventing social isolation and loneliness among older people: A systematic review of health promotion interventions. *Ageing and Society*, 25(1), 41-67. doi:10.1017/S0144686X04002594
- Chadwick, S. (2011). Lift every voice and sing: Constructing community through culturally relevant pedagogy in the University of Illinois Black chorus. *International Journal of Community Music*, 4(2), 147–162.https://doi.org/10.1386/ijcm.4.2.147_1
- Chai, L. (2023). Unpacking the association between length of residence and health among immigrants in Canada: A moderated mediation approach. *Journal of Immigrant and Minority Health*, 25(1), pp. 38-49. https://doi.org/10.1007/s10903-022-01377-x
- Chang. C. T. (2005). A history of the Korean immigrant Baptist church movement in the United States: the growth of ethnic minority churches related to the Southern Baptist convention is exploding. *Baptist History and Heritage*, 40(1), 58–.
- Chang, L.-C., Dattilo, J., Hsieh, P.-C., & Huang, F.-H. (2021). Relationships of leisure social support and flow with loneliness among nursing home residents during the COVID-19 pandemic: An age-based moderating model. *Geriatric Nursing*, 42(6), 1454–1460.

- https://doi.org/10.1016/j.gerinurse.2021.08.014
- Chao, S.-Y., Liu, H.-Y., Wu, C.-Y., Jin, S.-F., Chu, T.-L., Huang, T.-S., Clark, M. J., (2006). The effects of group reminiscence therapy on depression, self esteem, and life satisfaction of elderly nursing home residents. *Journal of Nursing Research 14*(1), 36-45. https://doi.org/10.1097/01.JNR.0000387560.03823.c7
- Chen, C. (2006). From filial piety to religious piety: Evangelical Christianity reconstructing Taiwanese immigrant families in the United States. *International Migration Review*, 40(3), 573–602. https://doi.org/10.1111/j.1747-7379.2006.00032.x
- Chen, A. W., Kazanijian, A., Wong, H., & Goldner, E. M. (2010). Mental Health Service Use by Chinese Immigrants with Severe and Persistent Mental Illness. *Canadian Journal of Psychiatry*, 55(1), 35-42. https://doi.org/10.1177/070674371005500106
- Chen, L., Alston, M., & Guo, W. (2019). The influence of social support on loneliness and depression among older elderly people in China: Coping styles as mediators. *Journal of Community Psychology*, 47(5), 1235–1245. https://doi.org/10.1002/jcop.22185
- Chiang, K. J., Chu, H., Chang, H.-J., Chung, M.-H., Chen, C.-H., Chiou, H.-Y., & Chou, K.-R. (2010). The effects of reminiscence therapy on psychological well-being, depression, and loneliness among the institutionalized aged. *International Journal of Geriatric Psychiatry*, 25(4), 380–388. https://doi.org/10.1002/gps.2350
- Cho, E. (2018). Songs of their lives: A narrative study of three older Korean immigrants in Los Angeles. *Research Studies in Music Education*. 40(2), 157-175. https://doi.org/10.1177/1321103X18774346
- Cho, D. I. (1995). *Minyo*. Encyclopedia of Korean Culture. http://encykorea.aks.ac.kr/Contents/Item/E0020212
- Choi, N. G. (2001). Diversity within diversity: Research and social work practice issues Asian American elders. *Journal of Human Behavior in the Social Environment*, *3*(3/4), 301-319. https://doi.org/10.1300/J137v03n03_18
- Choi, J., Kushner, K. E., Mill, J., & Lai, D. W. L. (2014). The experience of Korean immigrant women adjusting to Canadian society. *Journal of Cross-Cultural Gerontology*, 29(3), 277–297. https://doi.org/10.1007/s10823-014-9235-8
- Choi, N.G., & Smith, J. (2004). Reaching out to racial/ethnic minority older persons for elderly nutrition programs. *Journal of Nutrition for the Elderly*. 24(1), 89-104.

- https://doi.org/10.1300/J052v24n01_07
- Chow H. P. (2010). Growing old in Canada: Physical and psychological well-being among elderly Chinese immigrants. *Ethnicity & Health*, *15*(1), 61-72. https://doi.org/10.1080/13557850903418810.
- Chui, T., & Meheux, H. (2011). *Visible minority women* (Catalogue no. 89-503-X). http://www.statcan.gc.ca/pub/89-503-x/2010001/article/11527-eng.pdf
- Chung, S., & Yung, Y. (2014). Age norms for older adults among Koreans: Perceptions and influencing factors. *Ageing and Society*, *34*(8), 1335–1355. https://doi.org/10.1017/S0144686X13000111
- Clandinin, D. J., & Connelly, F. M. (1999). *Shaping a professional identity: Stories of educational practice*. Althouse Press. ISBN: 0920354475
- Clandinin, D. J. & Connelly, F.M. (2000) Narrative inquiry: Experience and story in qualitative research. Jossey-Bass.
- Clarke, A., & Warren, L. (2007). Hopes, fears and expectations about the future: What do older people's stories tell us about active ageing? *Ageing & Society*, 27(4), 465-488. https://doi.org/10.1017/S0144686X06005824
- Clayton, V. (1975). Erikson's theory of human development as it applies to the aged: Wisdom as contradictive cognition. *Human development*, 18(1-2), 119-128. https://doi.org/10.1159/000271479
- Clift, S., Hancox, G., Morrison, I., Hess, B., Kreutz, G., & Stewart, D. (2010). Choral singing and psychological wellbeing: Quantitative and qualitative findings from English choirs in a cross-national survey. *Journal of Applied Arts and Health*, *1*(1), 19–34. https://doi.org/10.1386/jaah.1.1.19/1
- Clift, S., Nicols, J., Raisbeck, M., Whitmore, C., & Morrison, I. (2010). Group singing, wellbeing and health: A systematic mapping of research evidence. *UNESCO Observatory*, 2(1). http://www.abp.unimelb.edu.au/unesco/ejournal/pdf/clift-paper.pdf
- Clift, S., & Hancox, G. (2010). The significance of choral singing for sustaining psychological wellbeing: Findings from a survey of choristers in England, Australia and Germany.

 Music Performance Research, 3(1), 79–96.

- Clift, S., & Morrison, I. (2011). Group singing fosters mental health and wellbeing: Findings from the East Kent 'Singing for Health' network project. *Mental Health and Social Inclusion*, *15*(2), 88–97. https://doi.org/10.1108/20428301111140930
- Clothier, J. (2002). Interview with Peter Moser. *Sounding Board*. http://www.mormusic.net/info/about/articles/12_02_Sounding_Board.html
- Coffman, D. (2002). Music and quality of life in older adults. *Psychomusicology*, *18*(1–2), 76–88. https://doi.org/10.1037/h0094050
- Coffman, D. D., & Dabback, W. M. (2021). Time to Face the Music? Possibilities for Community Music with Older Adults, In L Willingham (Ed.). *Community Music at the Boundaries* (pp. 155-170). Wilfrid Laurier University Press. https://books-scholarsportal-info.proxy1.lib.uwo.ca/uri/ebooks/ebooks6/upress6/2021-05-13/1/9781771124591
- Cohen, G. D. (2005). The mature mind: The positive power of the aging brain. Basic Books. ISBN 0465012035
- Cohen, A., Bailey, B., & Nilsson, T. (2002). The importance of music to seniors. *Psychomusicology*, 18(1–2), 89–102. https://doi.org/10.1037/h0094049
- Cohen, G. D., Perlstein, S., Chapline, J., Kelly, J., Firth, K. M., & Simmens, S. (2006). The impact of professionally conducted cultural programs on the physical health, mental health, and social functioning of older adults. *The Gerontologist*, 46(6), 726–734. https://doi.org/10.1093/geront/46.6.726
- Cohen, L., Manion, L., & Morrison, K. (Eds.) (2007). *Research Methods in Education* (6th ed.). Routledge. IBSN 978-0-415-36878-0
- Cohen, G. D., Perlstein, S., Chapline, J., Kelly, J., Firth, K.M., & Simmens, S. (2007). The impact of professionally conducted cultural programs on the physical health, mental health and social functioning of older adults 2-year results. *Journal of Aging, Humanities and the Arts, 1*(1-2), 5–22. https://doi.org/10.1080/19325610701410791
- Cohen, M., Kana, J., & Winemiller, R. (2021). Life within These Walls: Community Music-Making as a Bridge of Healing and Transformation in Prison Contexts, In L Willingham (Ed.). *Community Music at the Boundaries* (pp. 383-401). Wilfrid Laurier University Press. https://books-scholarsportal-info.proxy1.lib.uwo.ca/uri/ebooks/ebooks6/upress6/2021-05-13/1/9781771124591

- Connelly, F. M., & Clandinin, D. J. (1990). Stories of experience and narrative Inquiry. *Educational Researcher*, 19(5), 2–14. https://doi.org/10.3102/0013189X019005002
- Conner, K. O., Copeland, V. C., Grote, N. K., Koeske, G., Rosen, D., Reynolds, C. F., & Brown, C. (2010). Mental health treatment seeking among older adults with depression: The impact of stigma and race. *The American Journal of Geriatric Psychiatry*, *18*(6), 531–543. https://doi.org/10.1097/JGP.0b013e3181cc0366
- Commission of the European Communities (2000). *A Memorandum on lifelong learning*. https://arhiv.acs.si/dokumenti/Memorandum_on_Lifelong_Learning.pdf
- Courtin, E., & Knapp, M. (2017). Social isolation, loneliness and health in old age: a scoping review. *Health & Social Care in the Community*, 25(3), 799–812. https://doi.org/10.1111/hsc.12311
- Cornwell, B., Laumann, E. O., & Schumm, L. P. (2008). The social connectedness of older adults: A national profile. *American Sociological Review*, 73(2), 185–203. https://doi.org/10.1177/000312240807300201
- COSTI Immigration Services (2022). http://costi.org
- Coyle, C. E., & Dugan, E. (2012). Social Isolation, loneliness and health among older adults. *Journal of Aging and Health*, 24(8), 1346–1363. https://doi.org/10.1177/0898264312460275
- Creamer, E. (2018). *An introduction to fully integrated mixed methods research*. (Vols. 1-0). Sage Publications. https://doi.org/10.4135/9781071802823
- Creech, A. (2018). Community-Supported Music-Making as a Context for Positive and Creative Ageing. In B.-L. Bartleet, & L. Higgins, (Eds.). *The Oxford handbook of community music (pp. 89-108)*. Oxford University Press. https://doi.org/10.1093/oxfordhb/9780190219505.013.13
- Creech, A., Hallam, S., McQueen, H., & Varvarigou, M. (2013). The power of music in the lives of older adults. *Research Studies in Music Education*, *35*(1), 87–102. https://doi.org/10.1177/1321103X13478862
- Creech, A., Hallam, S., Varvarigou, M., Gaunt, H., McQueen, H., & Pincas, A. (2014). The role of musical possible selves in supporting subjective well-being in later life. *Music Education Research*, *16*(1), 32–49. https://doi.org/10.1080/14613808.2013.788143

- Creswell, J. (2003). *Research design: Qualitative, quantitative, and mixed method approaches* (2nd ed.). Sage Publications.
- Creswell, J. (2007). *Qualitative inquiry and research design: Choosing among five approaches* (2nd ed.). Sage Publications. ISBN 978-1-4129-1606-6
- Creswell, & Creswell, J. D. (2018). *Research design: Qualitative, quantitative, and mixed methods approaches* (5th ed.). Sage Publications.
- Creswell, J.W., & Hirose, M. (2019). Mixed methods and survey research in family medicine and community health. *Family Medicine and Community Health*, 7(2), e000086–e000086. https://doi.org/10.1136/fmch-2018-000086
- Creswell, J. W., & Plano Clark, V. L. (2018). *Designing and conducting mixed methods* research (3rd ed.). Sage Publications.
- Cucinotta, D., & Vanelli, M. (2020). WHO Declares COVID-19 a Pandemic. *Acta biomedica:*Atenei Parmensis, 91(1), 157–160. https://doi.org/10.23750/abm.v91i1.9397
- Cumings, B. (2005). *Korea's place in the sun: A modern history* (Updat. ed.). W.W. Norton & Company. ISBN: 978-0-393-32702-1
- Cumming, E., & Henry, W. E. (1961). Growing old: the process of disengagement. Basic Books.
- Dabback, W. (2007). *Toward a model of adult music learning as a socially-embedded phenomenon* (Unpublished doctoral dissertation). ProQuest 304917902
- Dahlberg, L., Andersson, L., & Lennartsson, C., (2018). Long-term predictors of loneliness in old age: results of a 20-year national study. *Aging & Mental Health*, 22(2), 190–196. https://doi.org/10.1080/13607863.2016.1247425
- Davey, J.A. (2002). Active Ageing and education in mid and later life. *Ageing and Society*, 22(1), 95–113. https://doi.org/10.1017/S0144686X02008528
- Decady, Y., & Greenberg, L. (2014). Health at a glance: *Ninety years of change in life*expectancy. Statistics Canada, Health Statistics Division (Catalogue no. 82- 624-X).

 https://www150.statcan.gc.ca/n1/en/pub/82-624-x/2014001/article/14009-eng.pdf?st=6HX5CQO
- Dench, S., & Regan, J. (2000. Learning in later life: Motivation and impact (Research Brief No. 183). England: Department for Education and Employment. https://www.education.gov.uk/publications/eOrderingDownload/RB183.pdf

- De Maio, F. G., & Kemp, E. (2010). The Deterioration of health status among immigrants to Canada. *Global Public Health*, 5(5), 462-478. https://doi.org/10.1080/17441690902942480
- Denzin, N. K., & Lincoln, Y. S. (2005). *The SAGE handbook of qualitative research* (3rd ed.). Sage Publications.
- Dewey, J. (1999). Experience and education. Simon & Schuster.
- Dixon-Woods. M., Kirk, M. D., Agarwal, M. S., Annandale, E., Arthur, T., Harvey, J., Hsu, R., Katbamna, S., Olsen, R., Smith, L., Riley, R., & Sutton, A. (2005). *Vulnerable groups and access to health care: A critical interpretive review*. National Co-ordinating Centre for NHS Service Delivery and Organisation R & D (NCCSDO). https://www.menshealthforum.org.uk/sites/default/files/pdf/sdovulnerablegroups2005.pdf
- Donovan, N. J., & Blazer, D. (2020). Social isolation and loneliness in older adults: Review and commentary of a national academies report. *The American Journal of Geriatric Psychiatry*, 28(12), 1233–1244. https://doi.org/10.1016/j.jagp.2020.08.005
- Duke, N. K., & Beck, S. W. (1999). Education should consider alternative formats for the dissertation. *Educational Researcher*, 28(3), 31-36. https://doi.org/10.2307/1177255
- Dunn, J. R., & Dyck, I. (2000). Social determinants of health in Canada's immigrant population: results from the National Population Health Survey. *Social Science and Medicine*, *51*(11), 1573-1593. https://doi.org/10.1016/S0277-9536(00)00053-8
- Dunphy, K. (2018). Theorizing Arts Participation as a Social Change Mechanism, In Bartleet, B.-L. Bartleet, & L. Higgins (Eds.). *The Oxford Handbook of Community Music* (pp.301-322). Oxford University Press. https://doi.org/10.1093/oxfordhb/9780190219505.013.16
- Durst, D. (2005). Aging amongst immigrants in Canada: Population drift. *Canadian Studies in Population*, 32(2), 257–270. https://doi.org/10.25336/P6JK5Q
- Duru Aşiret, G., & Kapucu, S. (2016). The effect of reminiscence therapy on cognition, depression, and activities of daily living for patients with Alzheimer disease. *Journal of Geriatric Psychiatry and Neurology*, 29(1), 31–37.

 https://doi.org/10.1177/0891988715598233
- Dykema, P.W. (1916). The spread of the community music idea. *Annals of the American Academy of Political and Social Science*, 67, 218-228.
- Dykema, P.W. (1919). Twice 55 community songs. C.C. Birchard.

- Edwards, R. (2008). Actively seeking subjects? In K. Nicoll & A. Fejes (Eds.), *Foucault and lifelong learning: Governing the subject* (pp. 21-33). Routledge. ISBN:978-0-415-42402-8
- Emami, A., Torres, S., Lipson, J. G., & Ekman, S. D. L. (2000). An ethnographic study of a day care center for Iranian immigrant seniors. *Western Journal of Nursing Research*, 22(2), 169–188. https://doi.org/10.1177/01939450022044340
- Employment Equity Act. (1995, December 15). *Employment Equity Act*. (S.C. 1995, c. 44).

 Justice laws website, Government of Canada. https://laws-lois.justice.gc.ca/eng/acts/e-5.401/page-1.html
- Elder, G. H., Jr., & Giele, J. Z. (2009). Life Course Studies: An Evolving Field. In G.H. Elder Jr., & J. Z. Giele (Eds.), *The Craft of Life Course Research* (pp.1-162). Guilford Publications.
- Elliott, D., Higgins, L., & Veblen, K. (Eds.). (2008). Editorial. *International Journal of Community Music*, 1(1), 3–4. https://doi.org/10.1386/ijcm.1.1.3_2
- Erikson, E. H. (1963). Childhood and society. W. W. Norton.
- Erikson, E. H. (1964). *Insight and responsibility: lectures on the ethical implications of psychoanalytic insight*. W. W. Norton.
- Erikson, E. H. (1968). *Identity, youth, and crisis*. W. W. Norton.
- Erikson, E. H., Erikson, J. M., & Kivnick, H. (1986). Vital involvement in old age. Norton.
- Erikson, E. H., & Erikson, J. M. (1998). *The life cycle completed* (Extended version / with new chapters on the ninth stage of development by Joan M. Erikson.). W.W. Norton.
- Ertel, K. A., Glymour, M. M., & Berkman, L. F. (2008). Effects of Social Integration on preserving memory function in a nationally representative US elderly population. *American Journal of Public Health* (1971), 98(7), 1215–1220. https://doi.org/10.2105/AJPH.2007.113654
- Escobar-Bravo, M., Puga-González, D., & Martín-Baranera, M. (2012). Protective effects of social networks on disability among older adults in Spain. *Archives of Gerontology and Geriatrics*, *54*(1), 109-116. https://doi.org/10.1016/j.archger.2011.01.008
- European Commission (EC). (2001). Communication from the commission: Making a European area of lifelong learning a reality (COM [2006] 678 Final). Commission of the European

- Communities. http://eur-
- lex.europa.eu/LexUriServ/LexUriServ.do?uri=COM:2001:0678:FIN:EN:PDF
- European Commission (EC). (2006). *Communication for the commission: Adult learning: It is never too late to learn* (COM [2006] 614 Final). Commission of the European communities. https://op.europa.eu/en/publication-detail/-/publication/07ca9c53-3b2e-4b70-8ef5-efaad639bf76
- European Commission (EC). (2013). European year for active ageing and solidarity between generations 2012. http://europa.eu/ey2012.
- Fang-Yu, H., Yang, Y., Feng-Ping, L., Chi-Jane, W., & Wang, J. (2019). Effects of group reminiscence therapy on agitated behavior and quality of life in Taiwanese older adults with dementia. *Journal of Psychosocial Nursing & Mental Health Services*, 57(8), 30-36. https://doi.org/10.3928/02793695-20190315-01
- Faulkner, K. A., Cauley, J. A., Zmuda, J. M., Griffin, J. M., & Nevitt, M. C. (2003). Is social integration associated with the risk of falling in older community-dwelling women? *The Journals of Gerontology. Series A, Biological Sciences and Medical Sciences*, 58(10), M954–M959. https://doi.org/10.1093/gerona/58.10.M954
- Faure, E., Herrera, F., Kaddoura, A.-R., Lopes, H., Petrovsky, A. V., Rahnema, M., & Ward, F.
 C. (1972). Learning to be: The world of education today and tomorrow.
 UNESCO/Harrap. ISBN 92-3-101017-4, 0-245-51900-7
- Fernández-Ballesteros, R., Caprara, M., Schettini, R., Bustillos, A., Mendoza-Nunez, V., Orosa, T., Kornfeld, R., Rojas, M., López, M. D., Santacreu, L. M., Molina, M. Á., & Zamarrón, M. D. (2013). Effects of university programs for older adults: Changes in cultural and group stereotype, self-perception of aging, and emotional balance. *Educational Gerontology*, 39(2), 119–131. https://doi.org/10.1080/03601277.2012.699817
- Ferreri, L., Bigand, E., Perrey, S., Muthalib, M., Bard, P., & Bugaiska, A. (2014). Less effort, better results: how does music act on prefrontal cortex in older adults during verbal encoding? An fNIRS study. *Frontiers in Human Neuroscience*, 8(301):1-11. https://doi.org/10.3389/fnhum.2014.00301
- Fetterman, D. M. (1998). Ethnography: Step by step. (2nd ed.). Sage Publications.
- Field, J. (2001) Lifelong education. *International Journal of Lifelong Education*, 20(1-2), 3–15. https://doi.org/10.1080/09638280010008291

- Field, J. (2009). Well-being and happiness. Inquiry into the future for lifelong learning. Thematic Paper 4, National Institute of Adult Continuing Education.
- Field, J. (2011). Adult learning, health and well-being changing lives. Adult Learner: *The Irish Journal of Adult and Community Education*, (0790-8084), 13–25. http://www.aontas.com/download/pdf/adult_learner_2011.pdf
- Findsen, B., & Formosa, M. (2011). *Lifelong learning in later life: A handbook on older adult learning*. Sense Publishers. DOI: 10.1177/0741713613496614
- Fiori, K. L., Antonucci, T. C., & Cortina, K. S. (2006). Social network typologies and mental health among older adults. *Journals of Gerontology Series B Psychological Sciences and Social Sciences*, 61(1), P25-P32. https://doi.org/10.1093/geronb/61.1.P25
- Fitzpatrick, T. R., Gitelson, R. J., Andereck, K. L., & Mesbur, E. S. (2005). Social support factors and health among a senior center population in southern Ontario, Canada. *Social Work in Health Care*, 40(3), 15-37. https://doi.org/10.1300/J010v40n03_02
- Fisher, J. C., & Wolf, M. A. (2000). Older adult learning. In A. L. Wilson & E. R. Hayes (Eds.), *The handbook of adult and continuing education* (pp. 480–492). Jossey-Bass.
- Fivush, R., Habermas, T., Waters, T. E. A., & Zaman, W. (2011). The making of autobiographical memory: Intersections of culture, narratives and identity. *International Journal of Psychology*, 46(5), 321–345. https://doi.org/10.1080/00207594.2011.596541
- Flowers, P., & Murphy, J. (2001). Talking about music: Interviews with older adults about their music education, preferences, activities, and reflections. *Update: Applications of Research in Music Education*, 20(1), 26-32. https://doi.org/10.1177/875512330102000106
- Formosa, M. (2010). Universities of the third age: A rationale for transformative education in later life. *Journal of Transformative Education*, 8(3), 197–219. https://doi-org.proxy1.lib.uwo.ca/10.1177/1541344611419857
- Formosa, M. (2012). Education and older adults at the university of the third age. *Educational Gerontology*, 38(2), 114-126. https://doi.org/10.1080/03601277.2010.515910
- Foster, L. (2010). Towards a new political economy of pensions? The implications for women. *Critical Social Policy*, *30*(1), 27–47. https://doi.org/10.1177/0261018309350807
- Foster, L., & Walker, A. (2013). Gender and active ageing in Europe. *European Journal of Ageing*, 10(1), 3-10. https://doi.org/10.1007/s10433-013-0261-0

- Foster, L., & Walker, A. (2015). Active and successful aging: A European policy perspective. *The Gerontologist*, 55(1), pp. 83-90. https://doi.org/10.1093/geront/gnu028
- Frable, D. E. S., Pratt, L., & Hoey, S. (1998). Concealable stigmas and positive self-perceptions: Feeling better around similar others. *Journal of Personality and Social Psychology*, 74(4), 909–922. https://doi.org/10.1037/0022-3514.74.4.909
- Freeman, M. (1997). Why narrative? hermeneutics, historical understanding, and the significance of stories. *Journal of Narrative and Life History*, 7,169-176.
- Freedman, A., & Nicolle, J. (2020). Social isolation and loneliness: the new geriatric giants: Approach for primary care. *Canadian Family Physician*, 66(3), 176–182.
- Friesen, N., & Anderson, T. (2004). Interaction for lifelong learning. *British Journal of Educational Technology*, *35*(6), 679–687. https://doi.org/10.1111/j.1467-8535.2004.00426.x
- Frisby W. (2011). Promising physical activity inclusion practices for Chinese immigrant women in Vancouver, Canada. *Quest.* 63(1), 135-147. https://doi.org/10.1080/00336297.2011.10483671
- Fryer C., Mackintosh S., Batchelor F., Hill K., & Said C. (2012). The effect of limited English proficiency on falls risk and falls prevention after stroke. *Age and Ageing*. *41*(1), 104-107. https://doi.org/10.1093/ageing/afr127
- Fung, C. V., & Lehmberg, L. J. (2016). *Music for life: Music participation and quality of life of senior citizens*. Oxford University Press. https://doi-org.proxy1.lib.uwo.ca/10.1093/acprof:oso/9780199371686.001.0001
- Gagnon, E. (1965) *Chansons Populaires du Canada*. Ministère des Affaires culturelles Bibliothèque nationale du Québec.
- Gandek, B., Ware, J. E. Jr., Aaronson, N. K., Alonso, J., Apolone, G., Bjorner, J., ... Sullivan, M. (1998). Tests of data quality, scaling assumptions, and reliability of the SF- 36 in eleven countries: Results from the international quality of life assessment project. *Journal of Clinical Epidemiology*, 51(11), 1149-1158. http://dx.doi.org/10.1016/S0895-4356(98)00106-1
- Garden Kyohae Silver College Bom Hakgi Geagang [Garden Church Silver College at open for spring term]. (2014, April 2). *Korea Daily Toronto*. Retrieved from http://www.cktimes.net/board_read.asp?boardCode=board_it&boardNumber=1497

- Garrett, S. (2010). The role of community music in helping disadvantaged young people in South Wales to confront social exclusion. *International Journal of Community Music*, *3*(3), 371–377. https://doi.org/10.1386/ijcm.3.3.371_1
- Garratt, A. M., Schmidt, L., Mackintosh, A., & Fitzpatrick, R. (2002). Quality of life measurement: bibliographic study of patient assessed health outcome measures. *British Medical Journal*, 324(7351), 1417-1419.
- Gary, F. A. (2005). Stigma: Barrier to mental health care among ethnic minorities. *Issues in Mental Health Nursing*, 26(10), 979–999. https://doi.org/10.1080/01612840500280638
- Geertz, C. (1973). The interpretation of cultures. Basic Books.
- Gibson, F. (2011). Reminiscence and life story work: A practice guide. Jessica Kingsley.
- Glass, T., Mendes de Leon, C., Bassuk, S., & Berkman, L. (2006). Social engagement and depressive symptoms in late life: Longitudinal findings. *Journal of Aging and Health*, *18*(4), 604-628. https://doi.org/10.1177/0898264306291017
- Golding, B. G. (2011). Social, local, and situated: Recent findings about the effectiveness of older men's informal learning in community contexts. *Adult Education Quarterly*, 61(2), 103–120. https://doi.org/10.1177/0741713610380437
- Goldman-Segall, R. (1998). *Points of viewing children's thinking: A digital ethnographer's journey*. Lawrence Erlbaum.
- Goodrich, A. (2013). Health musicing in a community orchestra. *International Journal of Community Music*, 6(1), 45–63. https://doi.org/10.1386/ijcm.6.1.45_1
- Gorard, S., Rees, G., & Selwyn, N. (2002). The "Conveyor belt effect": A re-assessment of the impact of national targets for lifelong learning. *Oxford Review of Education* 28(1), 75-89. https://doi.org/10.1080/03054980120113643
- Government of Canada. (2014). Government of Canada Action for seniors report.

 https://www.canada.ca/en/employment-social-development/programs/seniors-action-report.html
- Government of Canada (2021, April 18). 2020 Annual Report to Parliament on Immigration (ISSN 1706-3329). Immigration, Refugees and Citizenship Canada.

 https://www.canada.ca/en/immigration-refugees-citizenship/corporate/publications-manuals/annual-report-parliament-immigration-2020.html
- Green, E. G. (1950). The epic of Korea. Public Affairs Press.

- Green, J., & Thorogood, N. (2009). *Qualitative methods for health research* (2nd ed.). Sage Publications.
- Greene, J. C. (2007). Mixed methods in social inquiry. Wiley.
- Guillemin, F., Bombardier, C., & Beaton, D. (1993). Cross-cultural adaption of health-related quality of life measures: Literature review and proposed guidelines. *Journal of Clinical Epidemiology*, 46(12), 1417–1432.
- Gunganghago Aereumdawn Sangwhal Weehea. [For a Healthy Good-Quality Life], Toronto Korean Church Senior's College Guide. (2016, April 5). *The Korea Times Daily*. Retrieved from http://www.koreatimes.net/ ArticleViewer/Article/57310
- Guo, M., & Stensland, M. (2018). A systematic review of correlates of depression among older Chinese and Korean immigrants: what we know and do not know. *Aging & Mental Health*, 22(12), 1535–1547. https://doi.org/10.1080/13607863.2017.1383971
- Guruge, S., Thomson, M. S., & Seifi, S. G. (2015). Mental health and service issues faced by older immigrants in Canada: A scoping review. *Canadian Journal on Aging / La Revue Canadienne Du Vieillissement*, 34(4), 431-444.
- Guruge, S., Sidani, S., Man, G., Matsuoka, A., Kanthasamy, P., & Leung, E. (2021). elder abuse risk factors: perceptions among older Chinese, Korean, Punjabi, and Tamil Immigrants in Toronto. *Journal of Migration and Heath*, *4*(100059). https://doi.org/10.1016/j.jmh.2021.100059
- Haight, B. K., & Burnside, I. (1993). Reminiscence and life review: explaining the differences. *Archives of psychiatric nursing*, 7(2), 91–98. https://doi.org/10.1016/s0883-9417(09)90007-3
- Hake, B. J. (1999). Lifelong learning in late modernity: The challenges to society organizations, and individuals. *Adult Education Quarterly*, 49(2), 79-90.
 https://doi.org/10.1177/074171369904900201
- Halpern, A. R., Kwak, S., Bartlett, J. C., & Dowling, W. J. (1996). Effects of aging and musical experience on the representation of tonal hierarchies. *Psychology and Aging*, 11(2), 235-246. https://doi.org/10.1037//0882-7974.11.2.235
- Han, Y. (1988). *Hanguk dong-yo eumaksa [The history of Korean children songs]*. Sekwang eumak chulpansa.
- Han, C. W., Lee, E. J., Iwaya, T., Kataoka, H., & Kohzuki, M. (2004). Development of the

- Korean version of short-form 36-item health survey: Health related QOL of healthy elderly people and elderly patients in Korea. *The Tohoku Journal of Experimental Medicine*, 203(3), 189–194. https://doi.org/10.1620/tjem.203.189
- Han, S. (2007). Asian lifelong learning in the context of a global knowledge economy: A task revisited. *Asia Pacific Education Review*, 8(3), 478–486. https://doi.org/10.1007/BF03026475
- Han, H., Kim, M., Lee, H. B., Pistulka, G., & Kim, K. B. (2007). Correlates of depression in the Korean American elderly: Focusing on personal resources of social support. *Journal of Cross-Cultural Gerontology*, 22(1), 115-127. https://doi.org/10.1007/s10823-006-9022-2
- Han, C. W., & Lee, E. J. (2009). Use of structural equation modeling to test construct validity of the SF-36 health survey among community-dwelling elderly in Korea. *Journal Korea Academy of Care Management*, 4, 45–59.
- Han, C. W., Lee, E. J., Iwaya, T., Kataoka, H., & Kohzuki, M. (2004). Development of the Korean version of short-form 36-item health survey: Health related QOL of healthy elderly people and elderly patients in Korea. *The Tohoku Journal of Experimental Medicine*, 203(3), 189–194. https://doi.org/10.1620/tjem.203.189
- Hand, B. D., Cavanaugh, S., Forbes, W., Govern, J., & Cress, M. E. (2012). Changes in health-related quality of life and functional fitness with exercise training in older adults who attend senior centers. *Activities, Adaptation & Aging*, 36(1), 29-54. https://doi.org/10.1080/01924788.2011.647530
- Harasemiw, O., Newall, N., Mackenzie C. O., Shooshtari, S. & Menec, V. (2019) Is the association between social network types, depressive symptoms and life satisfaction mediated by the perceived availability of social support? A cross-sectional analysis using the Canadian Longitudinal Study on *Aging*, *Aging* & *Mental Health*, 23:10, 1413-1422, DOI: 10.1080/13607863.2018.1495176
- Hargreaves, D. J., Miell, D., & MacDonald, R. A. R. (2002). What are musical identities, and why are they important? In R.A.R. MacDonald, D.J. Hargreaves, & D. Miell (Eds.), *Musical identities* (pp. 1–20). Oxford University Press.
- Haslam, C., Haslam, S. A., Jetten, J., Bevins, A., Ravenscroft, S., & Tonks, J. (2010). The social treatment: The benefits of group interventions in residential care settings. *Psychology and Aging*, 25(1), 157–167. https://doi.org/10.1037/a0018256

- Hau, C., Reid, K. F., Wong, K. F., Chin, R. J., Botto, T. J., Eliasziw, M., Bermudez, O. I., & Fielding, R. A. (2016). Collaborative evaluation of the healthy habits program: An effective community intervention to improve mobility and cognition of Chinese older adults living in the U.S. *The Journal of Nutrition, Health & Aging*, 20(4), 391–397. https://doi.org/10.1007/s12603-015-0623-8
- Hawkley, L. C., & Kocherginsky, M. (2018). Transitions in loneliness Among older adults: A 5-year follow-up in the national social life, health, and aging project. *Research on Aging*, 40(4), 365–387. https://doi.org/10.1177/0164027517698965
- Hawton, A., Green, C., Dickens, A. P., Richards, S. H., Taylor, R. S., Edwards, R., Greaves, C. J., & Campbell, J. L. (2011). The impact of social isolation on the health status and health-related quality of life of older people. *Quality of Life Research*, 20(1), 57–67. https://doi.org/10.1007/s11136-010-9717-2
- Hays, T., Bright, R., & Minichiello, V. (2002). The contribution of music to positive aging: A review. *Journal of Aging and Identity*, 7(3), 165–175. https://doi.org/10.1023/A:1019712522302
- Hays, T., & Minichiello, V. (2005a). The contribution of music to quality of life in older people: An Australian qualitative study. *Ageing and Society*, 25(2), 261–278. https://doi.org/10.1017/S0144686X04002946
- Hays, T., & Minichiello, V. (2005b). The meaning of music in the lives of older people: A qualitative study. *Psychology of music*, *33*(4), 437–451. https://doi.org/10.1177/0305735605056160
- Hays, T., & Minichiello, V. (2005c). Older people's experience of spirituality through music. *Journal of Religion, Spirituality & Aging*, 18(1), 83–96.

 https://doi.org/10.1300/J496v18n01_06
- Hays, R. D., & Morales, L. S. (2001). The RAND-36 measure of health-related quality of life. *Annals of Medicine (Helsinki)*, *33*(5), 350–357. https://doi.org/10.3109/07853890109002089
- Hayslip Jr. B. (2014). Lifelong learning: a dyadic ecological perspective. *Journal of Intergenerational Relationships*, *12*, 368-380. https://doi.org/10.1080/15350770.2014.961826

- Health Canada. (2002). Workshop on healthy aging November 28-30, 2001. Part I: Aging and health practices (Catalogue No. H39-612/2002-1E). Government of Canada. http://publications.gc.ca/collections/Collection/H39-612-2002-1E.pdf
- Hearn, S., Saulnier, G., Strayer, J., Glenham, M., Koopman, R., & Marcia, J. E. (2012). Between integrity and despair: Toward construct validation of Erikson's eighth stage. Journal of Adult Development, 19(1), 1-20. https://doi.org/10.1007/s10804-011-9126-y
- Hendriks, L., Veerbeek, M. A., Volker, D., Veenendaal, L., & Willemse, B. M. (2019). Life review therapy for older adults with depressive symptoms in general practice: results of a pilot evaluation. *International Psychogeriatrics*, 31(12), 1801–1808. https://doi.org/10.1017/S1041610219000218
- Higgins, L. (2007). Acts of hospitality: The community in community music. *Music Education Research*, 9(2), 281–292. https://doi.org/10.1080/14613800701384441
- Higgins, L., & Willingham, L. (2017). *Engaging in community music: An introduction*. Routledge. ISBN 9781138638174
- Holt-Lunstad, J., Smith, T. B., & Layton, J. B. (2010). Social relationships and mortality risk: a meta-analytic review. *PLoS medicine*, 7(7), e1000316. https://doi.org/10.1371/journal.pmed.1000316
- Holt-Lunstad, J., Smith, T. B., Baker, M., Harris, T., & Stephenson, D. (2015). Loneliness and social isolation as risk factors for mortality: a meta-analytic review. *Perspectives on psychological science: a journal of the Association for Psychological Science*, 10(2), 227–237. https://doi.org/10.1177/1745691614568352
- Home and Community Care Support Services (2022). https://healthcareathome.ca
- Honn Qualls, S., & Abeles, N. (2000). *Psychology and the aging revolution: How we adapt to longer life.* American Psychological Association Books. ISBN 978-1-55798-707-5
- Hopman, W. M., Towheed, T., Anastassiades, T., Tenenhouse, A., Poliquin, S., Berger, C., Joseph, L., Brown, J. P., Murray, T. M., Adachi, J. D., Hanley, D. A., &

- Papadimitropoulos, E. (2000). Canadian normative data for the SF-36 health survey. *Canadian Medical Association Journal*, *163*(3), 265–271. PMID: 10951722; PMCID: PMC80287.
- Hou, F., & Picot, G. (2016). Analytical studies branch research paper series: Changing immigrant characteristics and entry earnings (Catalogue No. 11Foo19M-No. 374).
 Statistics Canada. http://www.statcan.gc.ca/pub/11f0019m/11f0019m2016374-eng.pdf
- Hou, F., & Ngo, A. (2021). Differences in living arrangements of older seniors by mother tongue. Statistics Canada. https://www150.statcan.gc.ca/n1/pub/36-28-0001/2021005/article/00003-eng.htm
- Hughes, S. L., Seymour, R. B., Campbell, R., Pollak, N., Huber, G., & Sharma, L. (2004).
 Impact of the fit and strong intervention on older adults with osteoarthritis. *The Gerontologist*, 44(2), 217–228. https://doi.org/10.1093/geront/44.2.217
- Hung, L., Kempen, G. I. J. M, & De Vries, N. K. (2010). Cross-cultural comparison between academic and lay views of healthy ageing: A literature review. *Ageing and Society, 30*(8), 1373-1391. https://doi.org/10.1017/S0144686X10000589
- Hurh, W. M., & Kim, K. C. (1990). Religious participation of Korean immigrants in the United States. *Journal for the Scientific Study of Religion*, 29(1), 19–34.
- Hwang, E. (2008). Exploring aging-in-place among Chinese and Korean seniors in British Columbia, Canada. *Ageing International*, *32*(3), 205–218. https://doi.org/10.1007/s12126-008-9014-3
- Im, E. O. (2021). People of Korean heritage. In L. Purnell & E. Fenkl (Eds.), *Textbook for transcultural health care: A Population approach*. Springer, Cham. https://doi.org/10.1007/978-3-030-51399-3_23
- International Longevity Centre, B. (2015). *Active ageing: A policy framework in response to the longevity revolution*].

 https://www.matiainstituto.net/en/publicaciones/active-ageing-policy-framework-response-longevity-revolution
- International Society for Music Education (ISME) Community Music Activity (CMA). (n.d.). https://www.isme.org/our-work/commissions/community-music-activity-commission-cma

- Jackson, A. (2016, January 10). *Fighting seniors poverty* (Weblog). http://www.broadbentinstitute.ca/andrew ajackson/fighting seniors poverty
- Jafari, S., Baharlou, S., & Mathias, R. (2010). Knowledge of determinants of mental health among Iranian immigrants of BC, Canada: "A Qualitative study". *Journal of Immigrant Minor Health*, 12, 100-106. https://doi.org/10.1007/s10903-008-9130-x
- Jang, Y., & Chiriboga, D. A. (2010). Living in a different world: Acculturative stress among Korean American elders. *The Journals of Gerontology: Psychological Sciences*, 65B(1), 14–21. https://doi.org/10.1093%2Fgeronb%2Fgbp019
- Jang, Y., Kim, G., & Chiriboga, D. (2005). Health, health- care utilization, and satisfaction with services: Barriers and facilities for older Korean Americans. *Journal of the American Geriatrics Society*, *53*, 1613–1617. https://doi.org/10.1111/j.1532-5415.2005.53518.x
- Jang, E. S., Kim, Y. S., Kim, K.-A., Lee, Y. J., Chung, W. J., Kim, I. H., Lee, B. S., & Jeong, S.-H. (2018). Factors Associated with Health-Related Quality of Life in Korean Patients with Chronic Hepatitis C Infection Using the SF-36 and EQ-5D. *Gut and Liver*, 12(4), 440–448. https://doi.org/10.5009/gnl17322
- Jarvis, P. (1983). Education and the Elderly. *Adult Education*, 55(4), 343-350.
- Jarvis, P. (2001). *Learning in later life: An introduction for educators & carers*. Kogan Page. https://doi.org/10.4324/9781315042435
- Jarvis, P. (2006). *Lifelong learning and the learning society: Towards a comprehensive theory of human learning*. Routledge. ISBN 9780415355414
- Jarvis, P. (2009). Learning to be a person in society: Learning to be me. In K. Illeris (Ed.), Contemporary theories of learning: Learning theorists ... in their own words (pp. 21–34). Routledge. ISBN 9781138550490
- Javanbakht, M., Keshtkaran, A., Shabaninejad, H., Karami, H., Zakerinia, M., & Delavari, S. (2015). Comparison of Blood Transfusion Plus Chelation Therapy and Bone Marrow Transplantation in Patients with β-Thalassemia: Application of SF-36, EQ-5D, and Visual Analogue Scale Measures. *International Journal of Health Policy and Management*, 4(11), 733–740. https://doi.org/10.15171/ijhpm.2015.113
- Jenkins, A. (2011). Participation in learning and wellbeing among older adults. *International Journal of Lifelong Education*, 30(3), 403-420. https://doi.org/10.1080/02601370.2011.570876

- Jenkins, A., & Mostafa, T. (2013). Learning and wellbeing trajectories among older adults in England (DoQSS Working Paper. No. 13-02). repec.ioe.ac.uk/REPEc/pdf/qsswp1302.pdf
- Jenkins A. & Mostafa T. (2015). The effects of learning on wellbeing for older adults in England. *Ageing & Society*, *35*(10), 2053–70. https://doi.org/10.1017/S0144686X14000762
- Jia, H., & Lubetkin, E. I. (2009). Time trends and seasonal pat- terns of health-related quality of life among U.S. adults. *Public Health Report*, 124(5), 692–701. https://doi.org/10.1177/003335490912400511
- Jo, A. M., Maxwell, A. E., Yang, B., & Bastoni, R. (2010). Conducting health research in Korean American churches. *Journal of Community Health*, 35(2), 156-164. https://doi.org/10.1007/s10900-009-9213-1
- Jo, H. E., Veblen, K. K., & Potter, P. J. (2014). Korean immigrant seniors' music making in an ethno-cultural com- munity program in Canada. In M. L. Cohen (Ed.), *CMA XIV:*Listening to the world: Proceedings from the International Society for Music Education (ISME) 2014 seminar of the Commission for Community Music Activity (pp. 25–32).

 https://edoc.ku.de/id/eprint/19768/1/CMA_XIV_Full_Proceedings_FINAL_MC.pdf
- Jo, H. E., Veblen, K. K., & Potter, P. J. (2021). Remembering a hard life with joy: Music-making among Korean elders, In L Willingham (Ed.). *Community Music at the Boundaries* (pp. 498-510). Wilfrid Laurier University Press. https://books-scholarsportal-info.proxy1.lib.uwo.ca/uri/ebooks/ebooks6/upress6/2021-05-13/1/9781771124591
- Johnson, M. L., Bengtson, V. L., Coleman, P. G., & Kirkwood, T. B. L. (Eds.). (2005). *The Cambridge handbook of age and ageing*. Cambridge University. https://books-scholarsportal-info.proxy1.lib.uwo.ca/uri/ebooks/ebooks2/cambridge/2010-04-12/1/0511201233
- Johnson, J. L., Bottorff, J. L., Browne, A. J., Grewal, S., Hilton, B. A., & Clarke, H. (2004).

 Othering and being othered in the context of health care services. *Health Communication*, 16(2), 255–71. https://doi.org/10.1207/S15327027HC1602_7
- Jones, P. (2009). Lifewide as well as lifelong: broadening primary and secondary school music education's service to students' musical needs. *International Journal of Community Music*, 2(2&3), 201-214. https://doi.org/10.1386/ijcm.2.2-3.201_1

- Josselson, R. (2006). Narrative research and the challenge of accumulating knowledge. *Narrative Inquiry:NI*, 16(1), 3-10. https://doi.org/10.1075/ni.16.1.03jos
- Josselson, R. (2010). Narrative research. In N. J. Salkind (Ed.), *Encyclopedia of research design* (pp. 869-875). Sage. https://dx.doi.org/10.4135/9781412961288
- Jung, J-S. (2000). Daoism in Korea. In L. Kohn (Ed.), *Daoism Handbook* (pp. 792-820). Brill. https://doi.org/10.1163/9789004391840_028
- Jung, M. Y., Holt, C. L., Ng, D., Sim, H. J., Lu, X., Le, D., Juon, H., Li, J., & Lee, S. (2018).
 The Chinese and Korean American immigrant experience: a mixed-methods examination of facilitators and barriers of colorectal cancer screening. *Ethnicity & Health*, 23(8), 847–866. https://doi.org/10.1080/13557858.2017.1296559
- Jutras, P. J. (2011). The benefits of New Horizons Band participation as self-reported by selected New Horizons Band members. *Bulletin of the Council for Research in Music Education*, 187, 65–84. https://www.jstor.org/stable/41162324
- Kalache, A., & Kickbusch, I. (1997). A global strategy for healthy ageing. World Health, 4, 4-5.
- Kang, D. (2007). Rhizoactivity: Toward a postmodern theory of lifelong learning. *Adult Education Quarterly, (American Association for Adult and Continuing Education)*, *57*(3), 205–220. https://doi.org/10.1177/0741713606297445
- Kang, S., Basham, R., & Kim, Y. J. (2013). Contributing factors of depressive symptoms among elderly Korean immigrants in Texas. *Journal of Gerontological Social Work*, *56*(1), 67–82. https://doi.org/10.1080/01634372.2012.734369
- Kang, S., Domanski, M. D., & Moon, S. S. (2009). Ethnic enclave resources and predictors of depression among Arizona's Korean immigrant elders. *Journal of Gerontological Social Work*, 52(5), 489–502. https://doi.org/10.1080/01634370902983153
- Karlsen, S. (2013). Immigrant students and the "homeland music": Meanings, negotiations and implications. *Research Studies in Music Education*, *35*(2), 161–177.
- Karlsen, S., & Westerlund, H. (2010). Immigrant students' development of musical agency: Exploring democracy in music education. *British Journal of Music Education*, 27(3), 225–239.
- Kaufman, R. S., Shin, J. K., & Russ, A. J. (2004). Revisiting the biomedicalisation of aging: Clinical trends and ethnical challenges. *Gerontologist*, 44(6), 731-738. https://doi.org/10.1093/geront/44.6.731

- Kembhavi, R. (2012). *Research Note- Canadian seniors: A demographic profile*. http://www.elections.ca/res/rec/part/sen/pdf/sen_e.pdf
- Khan, M., Kobayashi, K., Lee, S. M., & Vang, Z. (2015). (In)Visible minorities in Canadian health data and research. Population change and lifecourse strategic knowledge cluster discussion paper series/ Un Réseau stratégique de connaissances changements de population et parcours de vie Document de travail, 3(1), 5. Retrieved from: http://ir.lib.uwo.ca/pclc/vol3/iss1/5
- Kim, O. (1999a). Predictors of loneliness in elderly Korean immigrant women living in the United States of America. *Journal of Advanced Nursing*, 29(5), 1082–1088. https://doi.org/10.1046/j.1365-2648.1999.00993.x
- Kim, O. (1999b). Mediation effect of social support between ethnic attachment and loneliness in older Korean immigrants. *Research in Nursing and Health*, 22(2), 169–175. https://doi.org/10.1002/(SICI)1098-240X(199904)22:2<169::AID-NUR8>3.0.CO;2-F
- Kim, K. H. (2004). An attempt to elucidate notions of lifelong learning: Analects-based analysis of Confucius' ideas about learning. *Asia Pacific Education Review*, *5*(2), 117–126. https://doi.org/10.1007/BF03024949
- Kim, S. H. (2007). Finance and Growth of the Korean Economy from 1960 to 2004. Seoul Journal of Economics, 20(4), 377-418. https://s-space.snu.ac.kr/bitstream/10371/1392/1/v20n4_377.pdf
- Kim, A. H. (2010). Filial piety, financial independence, and freedom: explaining the living arrangements of older Korean immigrants. In D. Durst & M. J. MacLean (Eds.), *Diversity and aging among immigrant seniors in Canada: Changing faces and greying temples* (pp. 387–408). Detselig Enterprises Ltd. ISBN: 9781550594072
- Kim, G. (2010). Measuring depression in a multicultural society: Conceptual issues and research recommendations. *Hallym International Journal of Aging*, *12*(1), 27–46. http://dx.doi.org/10.2190/HA.12.1.c
- Kim, W. (2012). How gender and religion influence alcohol use in elderly Korean immigrants. *Journal of Applied Gerontology*, 31(2), 173–192. https://doi.org/10.1177/0733464810384115
- Kim, Y. S. (2013). Ethnic senior schools, religion, and psychological well-being among older Korean immigrants in the United States: A qualitative study. *Educational Gerontology*,

- 39(5), 342-354. https://doi.org/10.1080/03601277.2012.700826
- Kim, J. H. (2014). Rethinking colonialism: Korean primary school music education during the Japanese colonial rule of Korea, 1910-1945. *Journal of Historical Research in Music Education*, 36(1), 23-42. https://doi.org/10.1177/153660061403600103
- Kim, S. (2015, October 1). *The Pronoun of fidelity, Meweldang Kim Si-Seup*, Cultural heritage administration. https://www.cha.go.kr/cop/bbs/selectBoardArticle.do?nttId=27686&bbsId=27686&bbsId=27686&mn=NS_01_09_01
- Kim, E. (2016). Social service outreach programs using elderly Korean immigrants' church ties. *Journal of Social Service Research*, 42(1), 130–142. http://dx.doi.org/10.1080/0148
 8376.2015.1087448
- Kim, M. W. (2006. January 6). "Since it's a New Year, it would be also good to act your age..."

 Pressian. https://www.pressian.com/pages/articles/210#0DKU
- Kim, A., & Merriam, S. (2004). Motivations for learning among older adults in a learning in retirement institute. *Educational Gerontology*, 30(6), 441–455. https://doi.org/10.1080/03601270490445069
- Kim, S., Jeon, E., Sok, S. R., Oh, H. K., & Kim, K. (2009). Quality of life of Korean and Korean American older adults: A comparison. *Journal of Gerontological Nursing*, *35*(6), 28. https://doi.org/10.3928/00989134-20090428-06
- Kim, G., Chiriboga, D. A., Jang, Y., Lee, S., Huang, C., & Parmelee, P. (2010). Health status of older Asian Americans in California. *Journal of the American Geriatrics Society (JAGS)*, 58(10), 2003–2008. https://doi.org/10.1111/j.1532-5415.2010.03034.x
- Kim, G., Aguado Loi, C. X., Chiriboga, D. A., Jang, Y., Parmelee, P., & Allen, R. S. (2011a). Limited English proficiency as a barrier to mental health service use: A study of Latino and Asian immigrants with psychiatric disorders. *Journal of Psychiatric Research*, 45(1), 104–110. https://doi.org/10.1016/j.jpsychires.2010.04.031
- Kim, G., DeCoster, J., Chiriboga, D. A., Jang, Y., Allen, R. S., & Parmelee, P. (2011b).
 Associations between self-rated mental health and psychiatric disorders among older adults: Do racial/ethnic differences exist? *The American Journal of Geriatric Psychiatry*, 19(5), 416–422. https://doi.org/10.1097/JGP.0b013e3181f61ede

- Kim, G., Jang, Y., & Chiriboga, D. (2012). Personal views about aging among Korean American older adults: The role of physical health, social network, and acculturation. *Journal of Cross-Cultural Gerontology*, 27(2), 139–148. https://doi.org/10.1007/s10823-012-9165-2
- Kim, A. H., Noh, M. S., & Noh, S. (2012). Introduction: Historical context and contemporary research. In S. Noh, A. H. Kim & M. S. Noh (Eds.), *Korean immigrants in Canada:**Perspectives on migration, integration, and the family (pp. 3–18). University of Toronto Press. https://doi.org/10.3138/9781442690387-004
- Kim, S. H., Jo, M., & Lee, S. (2013). Psychometric properties of the Korean short form-36 health survey version 2 for assessing the general population. *Asian Nursing Research*, 7, 60–66. https://doi.org/10.1016/j.anr.2013.03.001
- Kim, I., Kang, S.Y., & Kim, W. (2018). The Effects of Religious Participation and Familial Assistance on Mental Health among Older Chinese and Korean Immi- grants: Multiple Mediator Analyses. *Journal of Cross-Cultural Gerontology 33*, 411–25. https://doi.org/10.1007/s10823-018-9355-7
- Kim-Goh, M., & Baello, J. (2008). Attitudes toward domestic violence in Korean and Vietnamese immigrant communities: Implications for human services. *Journal of Family Violence*, 23(7), 647–654. https://doi.org/10.1007/s10896-008-9187-2
- Knowles, M. S. (1970). *The modern practice of adult education: Andragogy versus pedagogy*. Association Press.
- Koehn, S. (2009). Negotiating Candidacy: Ethnic minority seniors' access to care. *Ageing and Society*, 29(4), 585–608. https://doi.org/10.1017/S0144686X08007952
- Koehn, S., Jarvis, P., & Kobayashi, K. (2011). *Taking care of chronic disease: Realizing approaches for Canada's aging ethnic population: A workshop final report*. Vancouver, BC: Centre for Healthy Aging at Providence Health Care (CHAP).

 http://www.centreforhealthyaging.ca/documents/TakingCareofChronicDiseaseReport-FINAL.pdf
- Koehn, S. D., Jarvis, P., Sandhra, S., Bains, S., & Addison, M. (2014). Promoting mental health of immigrant seniors in community. *Ethnicity and Inequalities in Health and Social Care*, 7(3), 146–156. DOI:10.1108/EIHSC-11-2013-0048
- Koehn, S., Habib, S., & Bukhari, S. (2016). S4AC Case study: Enhancing underserved senior's access to health promotion programs. *Canadian Journal on Aging / La Revue*

- Canadienne Du Vieillissement, 35(1), 89-102. https://doi.org/10.1017/S0714980815000586
- Koenig, H., McCullough, M., & Larson, D. (2001). *Handbook of religion and health*. Oxford University Press. ISBN: 0195118669
- Korte, J., Bohlmeijer, E. T., Westerhof, G. J., & Pot, A. M. (2011). Reminiscence and adaptation to critical life events in older adults with mild to moderate depressive symptoms. *Aging & Mental Health*, *15*(5), 638–646. https://doi.org/10.1080/13607863.2010.551338
- Krampe, J., Wagner, J. M., Hawthorne, K., Sanazaro, D., Wong-Anuchit, C., Budhathoki, C., Lorenz, R. A., & Raaf, S. (2014). Does dance-based therapy increase gait speed in older adults with chronic lower extremity pain: A feasibility study. *Geriatric Nursing (New York)*, 35(5), 339–344. https://doi.org/10.1016/j.gerinurse.2014.03.008
- Krause, N. (2006). Religion and health in late life. In J. E. Birren, K. W. Schaie, R. P. Abeles, M. Gatz, & T. A. Salthouse (Eds.), *Handbook of the psychology of aging* (6th ed., pp. 499–518). Elsevier Academic Press. ISBN: 0121012654
- Krieger, N., Waterman, P.D., Kosheleva, A., Chen, J.T., & Smith, K.W. (2013). Racial discrimination and cardiovascular disease risk: My body my story study of 1005 US-born black and white community health center participants (US). *PloS One*, 8(10), e77174–e77174. https://doi.org/10.1371/journal.pone.0077174.
- Krippendorf, K. (1980). Content analysis: An introduction to its methodology. Sage: Newbury Park, CA.
- Kruse, N. B. (2007). Andragogy and music: Canadian and American models of music learning among adults (Order No. 3264178) [Doctoral dissertation. Michigan State University]. ProQuest Dissertations & Theses Global 304843147.
- Kuo, B. C., Chong, V., & Joseph, J. (2008). Depression and its psychosocial correlates among older Asian immigrants in North America: A critical review of two decades' research. *Journal of Aging and Health*, 20(6), 615-652. https://doi.org/10.1177/0898264308321001
- Lai, D.W. L. (2004). Impact of culture on depressive symptoms of elderly Chinese immigrants.

 *Canadian Journal of Psychiatry, 49(12), 820-827.

 https://doi.org/10.1177/070674370404901205
- Lai, D. W. L., & Chau, S. B. (2007). Effects of service barriers on health status of older Chinese immigrants in Canada. *Social Work*, 52(3), 261–269. https://doi.org/10.1093/sw/52.3.261

- Lai, D. W. L., & Surood, S. (2008). Service barriers of Chinese family caregivers in Canada. *Journal of Gerontological Social Work*, 51(3–4), 315–336.

 https://doi.org/10.1080/01634370802039650
- Lai, L. J., Ou, X., & Li, C. Y. P. (2020). Effectiveness of a peer-based intervention on loneliness and social isolation of older Chinese immigrants in Canada: a randomized controlled trial. *BMC Geriatrics*, 20(1), 356–356. https://doi.org/10.1186/s12877-020-01756-9
- Lally, E. (2009). The power to heal us with a smile and a song: Senior well-being, music-based participatory arts and the value of qualitative evidence. *Journal of Arts and Communities I*(1), 25–44. https://doi.org/10.1386/jaac.1.1.25_1
- Lassen, A. J. (2015). Keeping disease at arm's length how older Danish people distance disease through active ageing. *Ageing & Society*, *35*(7), 1364-1383. https://doi.org/10.1017/S0144686X14000245
- Lebreton, M. (2011). Language as a sense of belonging among international school students: The case of the international baccalaureate primary years programme. *Education and Belonging*, *13*(1), 183-194.
- Lee, Y. (1985). *Hangukyang-ak baeknyeonsa* [The 100-year history of Western music in Korea]. Eumak chunchusa.
- Lee, H. Y. (2006). Social and cultural construction of elder mistreatment: Definition, perception, and help-seeking behaviour among elderly Korean immigrants (Order No. 3247421). [Doctoral dissertation, University of California]. ProQuest Dissertations & Theses Global (305339895).
- Lee, E. O. (2007). Religion and spirituality as predictors of well-being among Chinese American and Korean American older adults. *Journal of Religion, Spirituality & Aging, 19*(3), 77-100. https://doi.org/10.1300/J496v19n03_06
- Lee, Y. (2007). The immigration experience among elderly Korean immigrants. *Journal of Psychiatric and Mental Health Nursing*, 14(4), 403–410. https://doi.org/10.1111/j.1365-2850.2007.01098.x
- Lee, A. H. C. (2013). The musical experiences, motivations, and preferences of positive ageing in Taiwan. *International Journal of Community Music*, 6(1), 79–92. https://doi.org/10.1386/ijcm.6.1.79_1

- Lee, H. Y., Yoon, J. H., & Ryoo, B. R. (1997). *Hankook keundae hakkvo kvovook baeknvunsa vunkoo (II) [A study on 100-years history of Korean modem school education (II)]*. Korean Educational Development Institute.
- Lee, H. Y., Moon, A., & Knight, B. G. (2004). Depression among elderly Korean immigrants: Exploring socio-cultural factors. *Journal of Ethnic & Cultural Diversity in Social Work,* 13(4), 1-26. https://doi.org/10.1300/J051v13n04_01
- Lee, H. B., Hanner, J. A., Cho, S. J., Han, H. R., & Kim, M. T. (2008). Improving access to mental health services for Korean American immigrants: Moving toward a community partnership between religious and mental health services. *Psychiatry Investigation*, *5*(1), 14–20. https://doi.org/10.4306/pi.2008.5.1.14
- Lee Y., & Holm, K. (2011). Family relationships and depression among elderly Korean immigrants. *International Scholarly Research Network Nursing*, 429249-7. https://doi.org/10.5402/2011/429249
- Lee, E. O., & An, C. H. (2013). Faith-based community support for Korean American older adults. *Social Work & Christianity*, 40(4), 446-459. https://search-ebscohost-com.proxy1.lib.uwo.ca/login.aspx?direct=true&db=asn&AN=92582199&site=ehost-live
- Lee, J., Wachholtz, A., & Choi, K. H. (2014). A review of the Korean cultural syndrome Hwa Byung: Suggestions for theory and intervention. *Asia T'aep'yŏngyang Sangdam Yŏn'gu*, 4(1), 49–64. https://doi.org/10.18401/2014.4.1.4
- Lee, K. H., & Hwang, M. J. (2014). Private religious practice, spiritual coping, social support, and health status among older Korean adult immigrants. *Social work in Public Health*, 29(5), 428–443. https://doi.org/10.1080/19371918.2013.858234428-443
- Lee, J., & Cagle, J. G. (2018). Social exclusion factors influencing life satisfaction among older adults. *The Journal of Poverty and Social Justice: Research, Policy, Practice*, 26(1), 35–50. https://doi.org/10.1332/175982717X15127351091521
- Leung, D. S. Y., & Liu, B. C. P. (2011). Lifelong education, quality of life and self- efficacy of Chinese older adults. *Educational Gerontology*, 37, 11, 967-981. https://doi.org/10.1080/03601277.2010.492732
- Leong, F., Park, Y. S., & Kalibatseva, Z. (2013). Disentangling immigrant status in mental health: Psychological protective and risk factors among Latino and Asian American

- immigrants. *American Journal of Orthopsychiatry*, 83(2-3), 361–371. https://doi.org/10.1111/ajop.12020
- Li, F., Harmer, P., Chaumeton, N. R., Duncan, T. E., & Duncan, S. C. (2002). Tai chi as a means to enhance self-esteem: A randomized controlled trial. *The Journal of Applied Gerontology*, *21*(1), 70–89. https://doi.org/10.1177/073346480202100105
- Lim, L. L., Seubsman, S., & Sleigh, A. (2008). Thai SF-36 health survey: Tests of data quality, scaling assumptions, reliability and validity in healthy men and women. *Health and Quality of Life Outcomes*, 6(52). https://doi.org/10.1186/1477-7525-6-52
- Lin, K. M, Lau, J. K. C., Yamamoto, J., Zheng, Y. P., Kim, H. S., Cho, K. H., & Nakaski, G. (1992). Hwa- Byung: A community study of Korean Americans. *The Journal of Nervous and Mental Disease*, 180(6), 386-391. https://doi.org/10.1097/00005053-199206000-00008
- Lindsay, C. (2007). *Profiles of ethnic communities on Canada, The Korean community in Canada 2001* (Catalogue No. 89-621-XIE). Statistics Canada. http://www.statcan.gc.ca/pub/89-621-x/89-621-x2007014-eng.pdf
- Lindesay, J., Jagger, C., Hibbett, M. J., Peet, S. M., & Moledina, F. (1997). Knowledge, uptake and availability of health and social services among Asian Gujarati and white elderly persons. *Ethnicity and Health*, 2(1-2), 59–69. https://doi.org/10.1080/13557858.1997.9961815
- Liu, Z., Yang, F., Lou, Y., Zhou, W., & Tong, F. (2021). The effectiveness of reminiscence therapy on alleviating depressive symptoms in older adults: A systematic review. *Frontiers in psychology*, *12*, 709853. https://doi.org/10.3389/fpsyg.2021.709853
- Livesey, L., Morrison, I., Clift, S., & Camic, P. (2012). Benefits of choral singing for social and mental wellbeing: Qualitative findings from a cross-national survey of choir members.

 Journal of Public Mental Health, 11(1), 10–26.

 https://doi.org/10.1108/17465721211207275
- Lofters, A. K., Hwang, S. W., Moineddin, R., & Glazier, R. H. (2010). Cervical cancer screening among urban immigrants by region of origin: A population-based cohort study. *Preventive Medicine*, *51*(6), 509-516. https://doi.org/10.1016/j.ypmed.2010.09.014
- Lund, R., Nilsson, C. J., & Avlund, K. (2010). Can the higher risk of disability onset among older people who live alone be alleviated by strong social relations? A longitudinal study

- of non-disabled men and women. *Age and Ageing*, *39*(3), 319-326. https://doi.org/10.1093/ageing/afq020
- MacCourt, P, Wilson, K, Tourigny-Rivard, M. (2011). *Guidelines for comprehensive mental health services for older adults in Canada*. Mental Health Commission of Canada (MHCC) Seniors Advisory Committee. https://www.mentalhealthcommission.ca/wp-content/uploads/drupal/2017-09/mhcc_seniors_guidelines_0.pdf
- Mackowicz, J., & Wnek-Gozdek, J. (2016). "It's never too late to learn"- How does the Polish U3A change the quality of life for seniors? *Educational Gerontology*, 42(3), 186-197. https://doi.org/10.1080/03601277.2015.1085789
- Malone, J.C., Liu, S. R., Vaillant, G.E., Rentz, D. M., & Waldinger, R. J., (2016). Midlife Eriksonian psychosocial development: Setting the stage for late-life cognitive and emotional health. *Developmental Psychology*, *52*(3), 496-508.
- Mantie, R. (2012). Learners or Participants? The Pros and Cons of 'Lifelong Learning.' *International Journal of Community Music*, 5(3), 217-235.

 https://doi.org/10.1386/ijcm.5.3.217_1
- Marais, G. (2021). The Sacred canoe: The Search for truth and reconciliation through cultural partnership, In L Willingham (Ed.). *Community Music at the Boundaries* (pp. 274-288). Wilfrid Laurier University Press. https://books-scholarsportal-info.proxy1.lib.uwo.ca/uri/ebooks/ebooks6/upress6/2021-05-13/1/9781771124591
- Markides, K. S., & Rote, S. (2015). Aging, minority status, and disability, *Generations*, 38(4), 19-24. https://search-ebscohost-com.proxy1.lib.uwo.ca/login.aspx?direct=true&db=rlh&AN=101914514&site=ehost-live
- Marsh, K. (2012). "The beat will make you be courage": The role of a secondary school music program in supporting young refugees and newly arrived immigrants in Australia. *Research Studies in Music Education*, *34*(2), 93–111.
- Marsh, K. (2013). Music in the lives of refugee and newly arrived immigrant children in Sydney, Australia. In P. S. Campbell & T. Wiggins (Eds.), *The Oxford handbook of children's musical cultures* (pp. 492–509). New York, NY: Oxford University Press.
- Marsillas, S., De Donder, L., Kardol, T., van Regenmortel, S., Dury, S., Brosens, D., Smetcoren, A. S., Brana, T., & Varela, J. (2017). Does active ageing contribute to life satisfaction for

- older people? Testing a new model of active ageing. *European Journal of Ageing*, *14*(3), 295–310. https://doi.org/10.1007/s10433-017-0413-8.
- Marsiske, M., Franks, M. M., & Mast, B. T. (2001). Psychological perspectives on aging. In L. Morgan & S. Kunkel (Eds.), *Aging: The social context* (2nd ed., pp. 145-182). Pine Forge Press.
- Marwaha, S., & Livingston, G. (2002). Stigma, racism or choice: why do depressed ethnic elders avoid psychiatrists? *Journal of Affective Disorders*, 72(3), 257–65. https://doi.org/10.1016/S0165-0327(01)00470-0
- Mastel-Smith, B. A., McFarlane, J., Sierpina, M., Malecha, A., & Haile, B. (2007). Improving depressive symptoms in community-dwelling older adults: A psychosocial intervention using life review and writing. *Journal of Gerontological Nursing*, *33*(5), 13-19.
- Matsuoka, A., Guruge, S., Koehn, S., Beaulieu, M., & Ploeg, J. (2013). Prevention of abuse of older women in the post-migration context in Canada. *Canadian Review of Social Policy*, 68-69, 107-120. https://link-gale-com.proxy1.lib.uwo.ca/apps/doc/A334087418/AONE?u=lond95336&sid=bookmark-AONE&xid=b65530ef
- Mayo, N.E., Moriello, C., Asano, M., van der Spuy, S., & Finch, L. (2011). The extent to which common health-related quality of life indices capture constructs beyond symptoms and function. *Quality of Life Research*, 20(5), 621–627. https://doi.org/10.1007/s11136-010-9801-7
- McAuley, E., Elavsky, S., Motl, R. W., Konopack, J. F., Hu, L., & Marquez, D. X. (2005). Physical activity, self-efficacy, and self-esteem: Longitudinal relationships in older adults. *The Journals of Gerontology. Series B, Psychological Sciences and Social Sciences*, 60(5), P268–P275. https://doi.org/10.1093/geronb/60.5.P268
- McCaffrey, R. (2008). Music listening: Its effects in creating a healing environment. *Journal of Psychosocial Nursing and Mental Health Services*, 46(10), 39-45. https://doi.org/10.3928/02793695-20081001-08
- Mendes, F. R. (2013). Active ageing: A right or a duty? *Health Sociology Review*, 22(2), 174-185. https://doi.org/10.5172/hesr.2013.22.2.174

- Mendes de Leon, C. F., Glass, T. A., & Berkman, L. F. (2003). Social engagement and disability in a community population of older adults: The new haven EPESE. *American Journal of Epidemiology*, 157(7), 633-642. https://doi.org/10.1093/aje/kwg028
- Mental Health Commission of Canada. (2013). *Mental health commission of Canada issue: Seniors*. http://www.mentalhealthcommission.ca/English/issues/seniors?terminitial
- Merriam, S. B., & Cross, L. H. (1982). Adulthood and reminiscence: A descriptive study. *Educational Gerontology*, 8(3), 275–290. https://doi.org/10.1080/0360127820080307
- Merriam, S. B., Caffarella, R. S., & Baumgartner, L. (2007). Learning in adulthood (3rd ed.). Jossey-Bass.
- Merriam, S. B. (2009). *Qualitative research: A guide to design and implementation*. Jossey-Bass. https://ebookcentral.proquest.com/lib/west/detail.action?docID=1662771
- Merriam, S., & Kee, Y. (2014). Promoting community wellbeing: The case for lifelong learning for older adults. *Adult Education Quarterly*, 64(2), 128–144. https://doi.org/10.1177/0741713613513633
- Mikkola, T. M., Porteigs, E., Rantakokko, M., Gagne, J., Rantanen, T., & Viljanen, A. (2015). Association of self-reported hearing difficulty of objective and perceived participation outside the home in older community-dwelling adults. *Journal of Aging and Health*, 27(1), 103–122. https://doi.org/10.1177/0898264314538662
- Min, P. K. (1992). The structure and social functions of Korean immigrant churches in the United States. *International Migration Review*, 26(4), 1370-1394. https://doi.org/10.2307/2546887
- Min, W. (1966). *Gaehwagiui eumakgyoyuk [Music education in the Enlightenment period]*. Hanguk Munhwa Yeonguwon.
- Min, P. G. & Kim, J. H. (2002). *Religions in Asian America: Building faith communities*. AltaMira Press. ISBN: 0759100837
- Min, J. W., Moon, A., & Lubben, J. E. (2005). Determinants of psychological distress over time among older Korean immigrants and non-Hispanic white elders: Evidence from a two-wave panel study. *Aging & Mental Health*, 9(3), 210–222. https://doi.org/10.1080/13607860500090011
- Min, S. K., & Suh, S. Y. (2010). The anger syndrome hwa- byung and its comorbidity. *Journal of Affective Disorders*, 124(1–2), 211–214. https://doi.org/10.1016/j.jad.2009.10.011

- Mio, J. S., Barker-Hackett, L., & Tumambing, J. 2009). *Multicultural psychology:Understading our diverse communities* (2nd ed.). McGraw-Hill Higher Education.
- Mitchell, A. (2021). And the Melody Lingers On: Building Social Capital through Community Music Participation, In L Willingham (Ed.). *Community Music at the Boundaries* (pp. 171-184). Wilfrid Laurier University Press. https://books-scholarsportal-info.proxy1.lib.uwo.ca/uri/ebooks/ebooks6/upress6/2021-05-13/1/9781771124591
- Moody, H. R. (1986). Late life learning in the information society. In D. A. Peter- son, J. E. Thornton, & J. E. Birren (Eds.), *Education and aging* (pp. 122–148). Prentice-Hall.
- Moody, E., & Phinney, A. (2012). A community-engaged art program for older people: Fostering social inclusion. *Canadian Journal on Aging / La Revue Canadienne Du Vieillissement*, 31(1), 55–64. https://doi.org/10.1017/S0714980811000596
- Moore-Harrison, T. L., Johnson, M. A., Quinn, M. E., & Cress, M. E. (2009). An evidence-based exercise pro- gram implemented in congregate-meal sites. *Journal of Physical Activity and Health*, 6, 247–251. https://doi.org/10.1123/jpah.6.2.247
- Moser, P. (2018). Growing Community Music Through a Sense of Place. In B.-L. Bartleet, & L. Higgins, (Eds.). *The Oxford Handbook of Community Music*, (pp. 213-228). Oxford University Press. https://doi.org/10.1093/oxfordhb/9780190219505.013.26
- Mui, A. C. (2001). Stress, coping, and depression among elderly Korean immigrants. *Journal of Human Behavior in the Social Environment*, *3*(3-4), 281-299. https://doi.org/10.1300/J137v03n03_17
- Mui, A. C., & Kang, S. K. (2006). Acculturation stress and depression among Asian immigrant elders. *Social Work*, *51*(3), 243–255. https://doi.org/10.1093/sw/51.3.243
- Mui, A. C., Kang, S., Kang, D., & Domanski, M. D. (2007). English language proficiency and health-related quality of life among Chinese and Korean immigrant elders. *Health and Social Work*, 32(2), 119–127. https://doi.org/10.1093/hsw/32.2.119
- Mui, A. C., Nguyen, D. D., Kang, D. Y., & Domanski, M. D. (2006). Demographic profiles of Asian immigrant elderly residing in metropolitan ethnic enclave communities. *Journal of Ethnic and Cultural Diversity in Social Work*, 15(1–2), 193–214. https://doi.org/10.1300/J051v15n01_09
- Myers, D. E. (2007). Freeing music education from schooling: toward a lifespan perspective on music learning and teaching. *International Journal of Community Music*, *1*(1), 49–61.

- https://doi.org/10.1386/ijcm.1.1.49_1
- Myers, D. G., & Diener, E. (1995). Who is happy? *Psychological Science*, *6*(1), 10-19. https://doi.org/10.1111/j.1467-9280.1995.tb00298.x
- Narushima, M. (2008). More than nickels and dimes: The health benefits of a community-based lifelong learning programme for older adults. *International Journal of Lifelong Education*, 27(6), 673–692. https://doi.org/10.1080/02601370802408332
- Narushima, M., Liu, J., & Diestelkamp, N. (2018). I learn, therefore I am: A phenomenological analysis of meanings of lifelong learning for vulnerable older adults. *The Gerontologist*, 58(4), 696–705. https://doi.org/10.1093/geront/gnx044
- National Advisory Council on Aging. (2005). Seniors on the margins: Seniors from ethnocultural minorities. Minister of Public Works and Government Services Canada. http://seniorspolicylens.ca/Root/Materials/
 Seniors%20From%20Ethnocultural%20Minorities%20 NACA%202005.pdf
- Ndegwa, S. (2011). *Initiatives for healthy aging in Canada* [Environmental Scan Issue 17].

 Canadian Agency for Drugs and Technologies in Health.

 https://www.cadth.ca/media/pdf/Initiatives on Healthy Aging in Canada es-17 e.pdf
- National Institute of Korean Language (2020. January,16). *'Neulgda'*, *'Keuda' pumsa* ['Old', 'Big' a part of speech]. National Institute of Korean Language (republic of Korea). https://www.korean.go.kr/front/mcfaq/mcfaqView.do?mn_id=217&mcfaq_seq=8726
- Nazareth, T. (1998). Achieving lifelong music education: Cooperation, coordiation, and coherence. *General Music Today*, 11(2), 21-23.
- Nazareth, T. (1999). A Coherent framework for lifelong music education. *General Music Today*, 12(2), 17-19.
- Nazroo, J. (2003). The structuring of ethnic inequalities in health: economic position, racial discrimination, and racism. *American Journal of Public Health 93*(2), 277-284. https://doi.org/10.2105/AJPH.93.2.277
- Nazroo, J. (2006). *Health and social research in multiethnic societies*. Taylor & Francis Group. https://ebookcentral.proquest.com/lib/west/detail.action?docID=273753
- Negus, K., & Velazquez, P. R. (2002). Belonging and detachment: musical experience and the limits of identity. *Poetics*, 30(1), 133–145. https://doi.org/10.1016/S0304-422X(02)00003-7

- Newbold, K.B. (2005). Self-rated health within the Canadian immigrant population: risk and the healthy immigrant effect. *Social Science and Medicine*, 60(6), 1359-1370. https://doi.org/10.1016/j.socscimed.2004.06.048
- Ng, E., Pottie, K., & Spitzer, D. (2011). Official language proficiency and self-reported health among immigrants to Canada (Catalogue no. 82-003-XPE). Health Reports 22(4), 1-9. Statistics Canada. https://www150.statcan.gc.ca/n1/en/pub/82-003-x/2011004/article/11559-eng.pdf?st=2EvXaMG1
- Ng, E., & Omariba, W. (2014). Immigration, generational status and health literacy in Canada. *Health Education Journal*, 73(6), 668-682. https://doi.org/10.1177/0017896913511809
- Nicholson, N. R. (2012). A review of social isolation: An important but underassessed condition in older adults. *The Journal of Primary Prevention*, *33*(2–3), 137–152. https://doi.org/10.1007/s10935-012-0271-2
- Noh, S., Kaspar, V., & Wickrama, K. A. S. (2007). Overt and subtle racial discrimination and mental health: Preliminary findings for Korean immigrants. *American Journal of Public Health*, 97(7), 1269-1274. https://doi.org/10.2105/AJPH.2005.085316
- Nordlund, A., Ekberg, K., M. Kristenson, & Linquest Group. (2005). EQ-5D in a general population survey: A description of the most commonly reported EQ-5D health states using the SF-36. *Quality of Life Research*, *14*(4), 1099–1109. https://doi.org/10.1007/s11136-004-3062-2
- OECD (2019). *Recognition of non-formal and informal Learning*. Organisation for Economic Co-operation and Development Publishing. https://www.oecd.org/education/skills-beyond-school/recognitionofnon-formalandinformallearning-home.htm
- OECD (2021). *OECD Skills Outlook 2021: Learning for Life*. OECD Publishing. https://doi.org/10.1787/0ae365b4-en.
- O'Hagin, I. B., & Harnish, D. (2006). Music as a cultural identity: A case study of Latino musicians negotiating tradition and innovation in northwest Ohio. *International Journal of Music Education*, 24(1), 56–70. https://doi.org/10.1177/0255761406063107
- Okamoto, N., Nakatani, T., Morita, N., Saeki, K., & Kurumatani, N. (2007). Homebased walking improves cardiopulmonary function and health-related QOL in community-dwelling adults. *International Journal of Sports Medicine*, 28(12), 1040–1045. https://doi.org/10.1055/s-2007-965073

- Omariba, D., Rasugu, W., & Ng, E. (2011). Immigration, generation and self-rated health in Canada: On the role of health literacy. *Canadian Journal of Public Health*, 102(4), 281-85. https://doi.org/10.1007/BF03404049
- Ontario Council of Agencies Serving Immigrants (OCASI). (2020). Settlement.org.: Welcome to Ontario. Immigration, Refugees and Citizenship Canada (IRCC) and the Ministry of Children, Community and Social Services (MCSS). https://settlement.org
- Opdenacker, J., Delecluse, C., & Boen, F. (2009). The longitudinal effects of a lifestyle physical activity intervention and a structured exercise intervention on physical self-perceptions and self-esteem in older adults. *Journal of Sport & Exercise Psychology*, 31(6), 743–760. https://doi.org/10.1123/jsep.31.6.743
- Otten, S., & Moskowitz, G. B. (2000). Evidence for implicit evaluative in-group bias: Affect-biased spontaneous trait inference in a minimal group paradigm. *Journal of Experimental Social Psychology*, 36(1), 77. https://doi.org/10.1006/jesp.1999.1399
- Panayotoff, K.G. (1993) The impact of continuing education on the health of older adults. *Educational Gerontology*, 191(1), 9–20. https://doi.org/10.1080/0360127930190102
- Paquette, S., & Goulet, G. M. (2014). Lifetime benefits of musical training. *Frontiers in Neuroscience*, 8, 89. https://doi.org/10.3389/fnins.2014.00089
- Pengel, L.H., Herbert, R. D., Maher, C. G., & Refshauge, K. M. (2003). Acute low back pain: systematic review of its prognosis. *British Medical Journal*, *327*(7410), 323-325. https://doi.org/10.1136/bmj.327.7410.323
- Park, S. (2009). Analysis of Saemaul Undong: A Korean rural development programme in the 1970s. *Asia-Pacific Development Journal*, *16*(2), 113-140. https://doi.org/10.18356/3898e112-en
- Park. H. (2014). Living with "Hwa-byung": The psycho-social impact of elder mistreatment on the health and well-being of older people. *Aging & Mental Health*, *18*(1), 125–128. https://doi.org/10.1080/13607863.2013.814103
- Park, J. D. (2019). Korea's path of development in retrospect. In *Re-Inventing Africa's development: Linking Africa to the Korean development model* (pp. 177-205). Palgrave Macmillan. https://doi.org/10.1007/978-3-030-03946-2
- Park, W.- I. (2022, October 24). *Korean lyric songs*. Korean LII

 http://www.koreanlii.or.kr/w/index.php/Korean_lyric_songs?ckattempt=1

- Pasupathi, M., Weeks, T., & Rice, C. (2006). Reflecting on life: Remembering as a major process in adult development. *Journal of Language and Social Psychology*, 25(3), 244–263. https://doi.org/10.1177/0261927X06289425
- Peel, N., Bartlett, H., & McClure, R. (2004). Healthy ageing: how is it defined and measured? *Australasian Journal on Ageing*, 23(3), 115–119. https://doi.org/10.1111/j.1741-6612.2004.00035.x
- Phinney, A., Moody, E. M., & Small, J. A. (2014). The effect of a community-engaged arts program on older adults' well-being. *Canadian Journal on Aging / La Revue canadienne du vieillissement*, 33(3), 336–345. https://doi.org/10.1017/S071498081400018X
- Pipkin, A. (2012). Personal learning networks for retirees. *Knowledge Quest*, 41(2), 64–6.
- Pinquart, M., & Forstmeier, S. (2012). Effects of reminiscence interventions on psychosocial outcomes: A meta-analysis. *Aging & Mental Health*, *16*(5), 541–558. https://doi.org/10.1080/ 13607863.2011.651434
- Plouffe, L. A., & Kalache, A. (2011). Making communities age friendly: State and municipal initiatives in Canada and other countries. *Gaceta Sanitaria*, 25, 131–137. https://doi.org/10.1016/j.gaceta.2011.11.001
- Polkinghorne, D. E. (1995). Narrative configuration in qualitative analysis. In J. A. Hatch & R. Wisniewski (Eds.), *Life history and narrative* (pp. 5-23). Taylor & Francis Group. https://ebookcentral.proquest.com/lib/west/detail.action?docID=167247
- Pourat, N., Lubben, H., Yu, H., & Wallace, S. (2000). Perceptions of health and use of ambulatory care: Differences between Korean and White elderly. *Journal of Aging and Health*, *12*(1), 112–134. https://doi.org/10.1177/089826430001200106
- Priyadarshini, I., Mohanty, P., Kumar, R., Son, L. H., Chau, H., Nhu, V. H., Thi Ngo, P. T., & Tien Bui, D. (2020). Analysis of Outbreak and Global Impacts of the COVID-19. *Healthcare (Basel, Switzerland)*, 8(2), 148. https://doi.org/10.3390/healthcare8020148
- Pruchno, R. A., Wilson-Genderson, M., & Cartwright, F. (2010). A two-factor model of successful aging. *The Journals of Gerontology Series B, Psychological Sciences and Social Sciences*, 65(6), 671–679. https://doi.org/10.1093/geronb/gbq051
- Public Health Agency of Canada (PHAC). (2006). *Healthy aging in Canada: A new vision, a vital investment from evidence to action*. The Healthy Aging and Wellness Working Group of the Federal/Provincial/Territorial (F/P/T) Committee of Officials (Seniors).

- https://www.health.gov.bc.ca/library/publications/year/2006/Healthy_Aging_A_Vital_latest_copy_October_2006.pdf
- Public Health Agency of Canada (PHAC). (2010). The chief public health officer's report on the state of public health in Canada, 2010: Growing older adding life to years.

 Government of Canada. https://www.canada.ca/content/dam/phac-aspc/migration/phac-aspc/cphorsphc-respcacsp/2010/fr-rc/pdf/cpho_report_2010_e.pdf
- Public Health Agency of Canada. (2015). *Age-friendly communities*. Government of Canada. https://www.canada.ca/en/public-health/services/health-promotion/aging-seniors/friendly-communities.html
- Rabin, R., & de Charro, F. (2001). EQ-5D: A measure of health status from the EuroQol Group. Annals of Medicine, 33, 337–343.
- Ranzijn, R. (2010). Active ageing —Another way to oppress marginalized and disadvantaged elders?: Aboriginal elders as a case study. *Journal of Health Psychology*, *15*(5), 716–723. https://doi.org/10.1177/1359105310368181
- Ramage-Morin, P. L., Shields, M., & Martel, L. (2010). Health- promoting factors and good health among Canadians in mid- to late life. *Health Reports/Statistics Canada, Canadian Centre for Health Information = Rapports Sur La Santé/ Statistique Canada, Centre Canadien d'Information Sur La Santé*, 21(3), 45–53.
- Reitmanova, S., & Gustafson, D.L. (2009). Mental health needs of visible minority immigrants in a small urban center: Recommendations for policy makers and service providers.

 **Journal of Immigrant and Minority Health, 11(1), 46-56. https://doi.org/10.1007/s10903-008-9122-x
- Richards, P. S., & Bergin, A. E. (1997). *A spiritual strategy for counseling and psychotherapy*. American Psychological Association Press.
- Rinde, F. B., & Schei, T. B. (2017). Towards an understanding of community music in Norway. *International Journal of Community Music*, 10(1), 1931. https://doi.org/10.1386/ijcm.10.1.19_1
- Rogers, A. (2004). *Looking again at non-formal and informal education—towards a new paradigm*. https://infed.org/mobi/looking-again-at-non-formal-and-informal-education-towards-a-new-paradigm/

- Ro, Y. C. (2018). Spirituality, Spontaneity, and moral charisma in Korean confucianism. *Journal of Korean Religions*, 9(2), 55–82. https://doi.org/10.1353/jkr.2018.0012
- Roh, S., Lee, K. H., & Yoon, D. P. (2013). General well-being of Korean immigrant elders: The significance of religiousness/spirituality and social support. *Journal of Social Service Research*, *39*(4), 483-497. https://doi.org/10.1080/01488376.2012.709451
- Rohr, J. M., Spears, K. L., Geske, J., Khandalavala, B., & Lacey, M. J. (2019). Utilization of Health Care Resources by the Amish of a Rural County in Nebraska. *Journal of community health*, 44(6), 1090–1097. https://doi.org/10.1007/s10900-019-00696-9
- Ross, M. (1989). Relation of implicit theories to the construction of personal histories. *Psychological Review*, 96(2), 341-357. https://doi.org/10.1037/0033-295X.96.2.341
- Rowe, J. W., & Kahn, R. L. (1997). Successful aging. *Gerontologist*, *37*(4), 433-440. https://doi.org/10.1093/geront/37.4.433
- Roy, J., Jain, R., Golamari, R., Vunnam, R., & Sahu, N. (2020). COVID-19 in the geriatric population. *International Journal of Geriatric Psychiatry*, *35*(12), 1437–1441. https://doi.org/10.1002/gps.5389
- Rudner, A. (2011, March 23-26). What are the data telling us about immigrant seniors' health? [Paper presentation]. National Metropolis Conference: Immigration: Bringing the world to Canada, Vancouver, BC.
- Ryoo, D. H., & Ko, S. H. (1996). *Hankook dong-vo baldalsa [The history of Korean children songs]*. Hansung Publishing Co.
- Sanchez, D. T., & Garcia, J. A. (2009). When race matters: Racially stigmatized others and perceiving race as a biological construction affect biracial people's daily well-being. *Personality and Social Psychology Bulletin*, 35(9), 1154–1164.

 https://doi.org/10.1177/0146167209337628
- Sadavoy, J., Meier, R., & Ong, A. Y. M. (2004). Barriers to access to mental health services for ethnic seniors: The Toronto study. *Canadian Journal of Psychiatry /Revue Canadianne De Psychiatrie*, 49(3), 192. https://doi.org/10.1177/070674370404900307
- Saldov, M., & Chow, P. (1994). Ethnic elderly in metro Toronto hospitals, nursing homes, and homes for the aged: communication and health care. *International Journal of Aging and Human Development*, 38(2), 117-35. https://doi.org/10.2190/MK8X-H1HC-QADK-F77J

- Sanchez, D. T., & Garcia, J. A. (2009). When race matters: Racially stigmatized others and perceiving race as a biological construction affect biracial people's daily well-being. Personality and Social Psychology Bulletin, 35(9), 1154–1164.

 https://doi.org/10.1177/0146167209337628
- Sayin Kasar, K., & Karaman, E. (2021). Life in lockdown: Social isolation, loneliness and quality of life in the elderly during the COVID-19 pandemic: A scoping review. *Geriatric nursing (New York, N.Y.)*, 42(5), 1222–1229. https://doi.org/10.1016/j.gerinurse.2021.03.010
- Schaie, W., & Pietrucha, M. (2000). *Mobility and transportation in the elderly*. Springer. ISBN: 0826113095
- Schwarz, L., & Cottrell, F. (2007). The value of spirituality as perceived by elders in long-term care. *Physical & Occupational Therapy in Geriatrics*, 26(1), 43–62. https://doi.org/10.1300/J148v26n01_04
- Schwartz, S. J., Zamboanga, B. L., Weisskirch, R. S., & Wang, S. C. (2010). The relationships of personal and cultural identity to adaptive and maladaptive psychosocial functioning in emerging adults. *The Journal of Social Psychology, 150*(1), 1–33. https://doi.org/10.1080/00224540903366784
- Seals, C., Clanton, K., Agarwal, R., Doswell, F., & Thomas, C. (2008). Lifelong learning:

 Becoming computer savvy at a later age. *Educational Gerontology*, *34*(12), 1055–1069.

 https://doi.org/10.1080/03601270802290185
- Seth, M. J. (2011). A history of Korea: From antiquity to the present. Rowman & Littlefield. ISBN: 9780742567153
- Sharon, B. H., Hennessy, C. H., Brandon, J., & Boyette, L. (1997). Older adults' experiences of a strength-training program. *Journal of Nutrition, Health, and Aging*, *1*(2), 103–108.
- Sherman. E. (1993). Mental health and successful adaptation in later life. *Generations: Journal of the American Society on Aging*, 17(1), 43-46. http://www.jstor.org/stable/44877151
- Shin, H. S., Han, H.-R., & Kim, M. T. (2007). Predictors of psychological well-being amongst Korean immigrants to the United States: A structured interview survey. *International Journal of Nursing Studies*, 44(3), 415–426. https://doi.org/10.1016/j.ijnurstu.2006.04.007

- Silber, L. (2005). Bars behind bars: The impact of a women's prison choir on social harmony. *Music Education Research*, 7(2), 251–271. https://doi.org/10.1080/14613800500169811
- Sin, M. K., LoGerfo, J., Belza, B., & Cunningham, S. (2004). Factors influencing exercise participation and quality of life among elderly Korean Americans. *Journal of Cultural Diversity*, 11, 139–145.
- Sin, M., & Hirsch, A. (2013). Health care utilization and hyper- tension management in community-based elderly Korean immigrants. *Journal of Immigrant Minority Health*, *15*(6), 1090–1095. https://doi.org/10.1007/s10903-012-9721-4
- Sloane-Seale, A., & Kops, B. (2008). Older adults in lifelong learning: Participation and successful aging. *Canadian Journal of University Continuing Education*, *34*(1), 37–62. https://doi.org/10.21225/D5PC7R
- Smith, R. J., Lehning, A. J., & Dunkle, R. E. (2013). Conceptualizing age-friendly community characteristics in a sample of urban elders: an exploratory factor analysis. *Journal of gerontological social work*, 56(2), 90–111. https://doi.org/10.1080/01634372.2012.739267
- Smith, M. L., Steinman, L. E., & Casey, E. A. (2020). Combatting Social Isolation Among Older
 Adults in a Time of Physical Distancing: The COVID-19 Social Connectivity
 Paradox. Frontiers in Public Health, 8, 403–. https://doi.org/10.3389/fpubh.2020.00403
- Sohn, I. S. (1992). Hankook keundae kvovooksa: 1885-1945 [The modem history of Korean education: 1885-1945]. Yonsei University Press.
- Solé, C., Mercadal-Brotons, M., Gallego, S., & Riera, M., (2010). Contributions of music to aging adults' quality of life. *Journal of Music Therapy*, 47(3), 264–281. https://doi.org/10.1093/jmt/47.3.264
- Son, I. (1998). Hanguk gyoyuksa yeongu [History of Korean education]. Mun-eumsa.
- Son, G., & Kim, H. (2006). Culturally familiar environment among immigrant Korean elders. *Research and Theory for Nursing Practice*, 20(2), 159–171. https://doi.org/10.1891/rtnp.20.2.159
- Song, Y. J. (2009). The South Korean health care system. *Japan Medical Association Journal*, 52(3), 206–209. https://www.med.or.jp/english/journal/pdf/2009_03/206_209.pdf
- Stake, R. (1995). The art of case study research. Sage Publications. ISBN: 0803957661

- Statistics Canada. (n.d.). *Definitions, data sources and methods: Visible minority of person*. https://www23.statcan.gc.ca/imdb/p3Var.pl?Function=DEC&Id=45152
- Statistics Canada. (2003). *Ethnic diversity survey: Portrait of a multicultural society* (Catalogue no. 89-593-XIE). https://www150.statcan.gc.ca/n1/en/pub/89-593-x/89-593-x2003001-eng.pdf?st=Zqp1Tc0v
- Statistics Canada. (2006). *A portrait of seniors in Canada*.

 https://www150.statcan.gc.ca/n1/en/pub/89-519-x/89-519-x2006001-eng.pdf?st=ZHZP48BY
- Statistics Canada. (2012). *Immigrant languages in Canada. Language, 2011 census of population* (Catalogue no. 98-314-X2011003). http://www12.statcan.gc.ca/census-recensement/2011/as-sa/98-314-x/98-314-x2011003_2-eng.pdf
- Statistics Canada. (2013a). *National Household Survey (NHS) Profile, Canada, 2011* (Catalogue No. 99- 010-X2011027). <a href="https://www12.statcan.gc.ca/nhs-enm/2011/dp-pd/dt-td/Rp-eng.cfm?LANG=E&APATH=3&DETAIL=0&DIM=0&FL=A&FREE=0&GC=0&GID=0&GK=0&GRP=0&PID=105470&PRID=0&PTYPE=105277&S=0&SHOWALL=0&SUB=0&Temporal=2013&THEME=95&VID=0&VNAMEE=&VNAMEF=
- Statistics Canada. (2013b). 2011 National Household Survey: Data tables (Catalogue No. 99-010-X2011030). <a href="https://www12.statcan.gc.ca/nhs-enm/2011/dp-pd/dt-td/Rp-eng.cfm?LANG=E&APATH=3&DETAIL=0&DIM=0&FL=A&FREE=0&GC=0&GID=0&GK=0&GRP=0&PID=105392&PRID=0&PTYPE=105277&S=0&SHOWALL=0&SUB=0&Temporal=2013&THEME=95&VID=0&VNAMEE=&VNAMEF=
- Statistics Canada (2016). Linguistic diversity and multilingualism in Canadian homes: Census of population, 2016 (Catalogue no. 98-200-X2016010). https://www12.statcan.gc.ca/census-recensement/2016/as-sa/98-200-x/2016010/98-200-x2016010-eng.pdf
- Statistics Canada. (2017a). *Immigration and ethnocultural diversity: Key results from the 2016 census*. https://www150.statcan.gc.ca/n1/daily-quotidien/171025/dq171025b-eng.htm

- Statistics Canada (2017b). Focus on geography series, 2016 census (Catalogue no. 98-404-X2016001). https://www12.statcan.gc.ca/census-recensement/2016/as-sa/fogs-spg/Facts-can-eng.cfm?Lang=Eng&GK=CAN&GC=01&TOPIC=7
- Statistics Canada (2020). Experiences of discrimination during the COVID-19 pandemic. https://www150.statcan.gc.ca/n1/en/daily-quotidien/200917/dq200917a-eng.pdf?st=iYqBgesl
- Statistics Canada. (2022a). *Canada's population Estimates, First Quarter 2022*. https://www150.statcan.gc.ca/n1/daily-quotidien/220622/dq220622d-eng.htm
- Statistics Canada. (2022b). *Census in brief. Disaggregated trends in poverty from the 2021 Census of Population*. https://www12.statcan.gc.ca/census-recensement/2021/as-sa/98-200-x/2021009/98-200-x2021009-eng.pdf
- Statistics Canada (2022c). *The Canadian census: A rich portrait of the country's religious and ethnocultural diversity*. https://www150.statcan.gc.ca/n1/en/daily-quotidien/221026/dq221026b-eng.pdf?st=-zXVLqzn
- Stead, M., Wimbush, E., Eadie, D., & Teer, P. (1997). A qualitative study of older people's perceptions of ageing and exercise: The implications for health promotion. *Health Education Journal*, *56*(1), 3–16. https://doi.org/10.1177/001789699705600102
- Steptoe, A., Shankar, A., Demakakos, P., & Wardle, J. (2013). Social isolation, loneliness, and all-cause mortality in older men and women. *Proceedings of the National Academy of Sciences PNAS*, 110(15), 5797–5801. https://doi.org/10.1073/pnas.1219686110
- Stewart, M., Shizha, E., Makwarimba, E., Spitzer, D., Khalema, E. N., & Nsaliwa, C. E. (2011). Challenges and barriers to services for immigrant seniors in Canada: "You are among others but you feel alone." *International Journal of Migration, Health and Social Life*, 7(1), 16–32. https://doi.org/10.1108/17479891111176278
- Stinson, C. K., Young, E. A., Kirk, E., & Walker, R. (2010). Use of a structured reminiscence protocol to decrease depression in older women. *Journal of Psychiatric and Mental Health Nursing*, *17*(8), 665–673. https://doi.org/10.1111/j.1365-2850.2010.01556.x
- Sun, J., Buys, N., & Merrick, J. (Eds.). (2013a). *Health promotion: Community singing as a vehicle to promote health.* Nova Science.
- Sun, J. Buys, N., & Tatow, D. (2013b). Using participative community singing programs to improve health behaviors. In J. Sun, N. Buys, & J. Merrick (Eds.), *Health promotion:*

- *Community singing as a vehicle to promote health* (pp. 3-14). Nova Science Publishers, Inc. https://ebookcentral.proquest.com/lib/west/detail.action?docID=3022886
- Syed, M., McDonald, L., Smirle, C., Lau, K., Mirza, R., & Hitzig, S. (2017). Social isolation in Chinese older adults: Scoping review for age-friendly community planning. *Canadian Journal on Aging / La Revue Canadienne Du Vieillissement*, *36*(2), 223-245. doi:10.1017/S0714980817000101
- Tam, M. (2012). East-West perfectives on elder learning. *Educational Gerontology*, *38*(10), 661-665. https://doi.org/10.1080/03601277.2011.595346
- Tam, W., Poon, S. N., Mahendran, R., Kua, E. H., & Wu, X., V. (2021). The effectiveness of reminiscence-based intervention on improving psychological well-being in cognitively intact older adults: A Systematic review and meta-analysis. *International Journal of Nursing Studies*, 114, 103847, https://doi.org/10.1016/j.ijnurstu.2020.103847
- Taylor, H. O. (2019). Social Isolation's Influence on Loneliness Among Older Adults. *Clinical Social Work Journal*, 48(1), 140–151. https://doi.org/10.1007/s10615-019-00737-9
- Taylor, R. J., Nguyen, A. W., & Chatters, L. (2018). Social Isolation, Depression, and Psychological Distress Among Older Adults. *Journal of Aging and Health*, *30*(2), 229-246. https://doi.org/10.1177/0898264316673511
- Teater, B., & Baldwin, M. (2014). Singing for successful ageing: The perceived benefits of participating in the golden oldies community-arts programme. *British Journal of Social Work*, 44(1), 81–99. https://doi.org/10.1093/bjsw/bcs095
- Tendulkar, S. A., Hamilton, R. C., Chu, C., Arsenault, L., Duffy, K., Huynh, V., Hung, W., Lee, E., Shwuling, J., & Friedman, E. (2012). Investigating the myth of the "model minority": A participatory community health assessment of Chinese and Vietnamese adults. *Journal of Immigrant Minority Health*, *14*(5), 850–857. https://doi.org/10.1007/s10903-011-9517-y
- Tennant, R. (1996). A history of Korea. Kegan Paul International. ISBN 9781138863569
- Terrazas, A., & Batog, C. (2010). Korean Immigrants in the United States. Migration Information Source.
 - http://www.migrationinformation.org/USFocus/display.cfm?IDD793#1
- Thompson J. (2002, October). *The amazing university of the third age in China today*. http://www.worldu3a.org/worldpapers/u3a-china.htm

- Tirrito, T., & Choi, G. (2004). Faith organizations and ethnically diverse elders: A community action model. *Journal of Religious Gerontology*, *16*(1-2), 123–142. https://doi.org/10.1300/J078v16n01_08
- Tiwari, S. K., & Wang, J. L. (2008). Ethnic differences in mental health service use among White, Chinese, South Asian and South East Asian populations living in Canada. *Social Psychiatry Epidemiol*, 43(11), 866-871. https://doi.org/10.1007/s00127-008-0373-6
- Tomaka, J., Thompson, S., & Palacios, R. (2006). The relation of social isolation, loneliness, and social support to disease outcomes among the elderly. *Journal of aging and health*, *18*(3), 359–384. https://doi.org/10.1177/0898264305280993
- Tseng, H. M., Lu, J. F. R., & Gandek, B. (2003). Cultural issues in using the SF-36 health survey in Asia: Results from Taiwan. *Health and Quality of Life Outcomes*, 1(72), 1-9. http://dx.doi.org/10.1186/1477-7525-1-72
- Trevarthen, C. (2002). Origins of musical identity: Evidence from infancy for musical social awareness, In R.A.R. MacDonald, D.J. Hargreaves, & D. Miell (Eds.), *Musical identities* (pp. 21-40). Oxford University Press. ISBN: 9780199679485
- Tummala-Narra, P., Alegria, M., & Chen, C. (2012). Perceived discrimination, acculturative stress, and depression among South Asians: Mixed findings. *American Journal of Psychology*, *3*(1), 3–16. http://dx.doi.org/10.1037/a0024661
- Uchino, B. N., Uno, D., & Holt-Lunstad, J. (1999). Social support, physiological processes, and health. *Current Directions in Psychological Science*, 8(5), 145-148. https://doi.org/10.1111/1467-8721.00034
- Underwood, H. H. (1951). Tragedy and faith in Korea. Friendship Press.
- UNESCO (2016). Third global report on adult learning and education (GRALE III). UNESCO Institute of Lifelong Learning. https://uil.unesco.org/adult-education/global-report/third-global-report-adult-learning-and-education-grale-3
- U3A Online (n.d.). U3A online. https://www.u3aonline.org.au/home
- United Nations Department of Economic and Social Affairs, Population Division. (2015). *World population prospects: The 2015 revision, key findings and advance tables*. (Working Paper no. ESA/P/WP.241). https://population.un.org/wpp/publications/files/key_findings_wpp_2015.pdf

- United Nations (2020). *International Migration 2020 Highlights* (ST/ESA/SER.A/452).

 Department of Economic and Social Affairs, Population Division. United Nations.

 https://www.un.org/development/desa/pd/sites/www.un.org.development.desa.pd/files/un_desa_pd_2020_international_migration_highlights.pdf
- Vaillant, G. E. (2012). *Triumphs of experience: The men of the Harvard grant study*. Belknap Press of Harvard University Press. https://doi.org/10.4159/harvard.9780674067424
- Valenzuela, M. J. (2009). It's never too late to change your mind. ABC Books.
- van der Merwe, J. (2017). "We make a song": Moving beyond active music-making in the Field Band Foundation. *International Journal of Community Music*, 10(2), 121–138. https://doi.org/10.1386/ijcm.10.2.121_1
- Veazie, S., Gilbert, J., Winchell, K., Paynter, R., & Guise J.-M. (2019). *Addressing social isolation to improve the health of older adults: A rapid review* (AHRQ Publication No. 19-EHC009-EF) [Rapid evidence product]. Agency for Healthcare Research and Quality. https://doi.org/10.23970/AHRQEPC-RAPIDISOLATION
- Veblen, K. K. (2007). The many ways of community music. *International Journal of Community Music 1*(1), 5–21. https://doi.org/10.1386/ijcm.1.1.5_1
- Veblen, K. K. (2013). The tapestry: Introducing community music. In K. K. Veblen, S. J. Messenger, M. Silverman, & D. Elliott (Eds.), *Community music today* (pp. 1–9). Rowman and Littlefield. ISBN 978-1-60709-320-
 - 6. https://ebookcentral.proquest.com/lib/west/detail.action?docID=1117171
- Veblen, K.K., & Olsson, B. (2002). Community music: Towards an international perspective. In R. Colwell & C. Richardson (Eds.), *The New Handbook of Research on Music Teaching and Learning* (pp. 730-753). Oxford University Press.
- Victor, C., Scambler, S., Bond, J., & Bowling, A. (2000). Being alone in later life: loneliness, social isolation and living alone. *Reviews in Clinical Gerontology*, *10*(4), 407–417. https://doi.org/10.1017/S0959259800104101
- Volles, N. (2016). Lifelong learning in the EU: Changing conceptualisations, actors and policies. *Studies in Higher Education*, 41(2), 343–363.

 https://doi.org/10.1080/03075079.2014.927852

- von Lob, G., Camic, P., & Clift, S. (2010). The use of singing in a group as a response to adverse life events. *International Journal of Mental Health Promotion*, *12*(3), 45–53. https://doi.org/10.1080/14623730.2010.9721818
- Wagner, A. K., Gandek, B., Aaronson, N. K., Acquadro, C., Alonso, J., Apolone, G., Bullinger, M., Bjorner, J., Fukuhara, S., Kaasa, S., Leplege, A., Sullivan, M., Wood-Dauphinee, S., & Ware, J. E. Jr. (1998). Cross- cultural comparisons of the content of SF-36 translations across 10 countries: Results from the IQOLA Project. *Journal of Clinical Epidemiology*, 51(11), 925–932. https://doi.org/10.1016/S0895-4356(98)00083-3
- Waldron, J. L., Horsley, S., & Veblen, K.K. (Eds.). (2020). The Oxford Handbook of social media and music learning, Oxford University
 Press. https://doi.org/10.1093/oxfordhb/9780190660772.001.0001
- Walker, A. (2002). A strategy for active ageing. *International Social Security Review*, 55(1), 121-139. https://doi.org/10.1111/1468-246X.00118
- Walker, A. (2015). Active ageing: Realising its potential. *Australasian Journal on Ageing*, 34(1), 2-8. https://doi.org/10.1111/ajag.12219
- Walters, S.J., Munro, J., & Brazier, J. E. (2001). Using the SF-36 with older adults: a cross-sectional community-based survey. *Age & Ageing*, *30*(4), 337-343. https://doi.org/10.1093/ageing/30.4.337
- Wang, A. (2013). Cognitive effects of music: Working memory is enhanced in healthy older adults after listening to music [Doctoral dissertation, The University of Arizona] The University of Arizona College of Medicine Phoenix. https://repository.arizona.edu/handle/10150/281781
- Wang, L., & Kwak, M.-J. (2015). Immigration, barriers to healthcare and transnational ties: A case study of South Korean immigrants in Toronto, Canada. *Social Science & Medicine* (1982), 133, 340–348. https://doi.org/10.1016/j.socscimed.2014.11.039
- Wang, H., Karp, A., Winblad, B., & Fratiglioni, L. (2002). Late-life engagement in social and leisure activities is associated with a decreased risk of dementia: A longitudinal study from the kungsholmen project. *American Journal of Epidemiology*, 155(12), 1081-1087. https://doi.org/10.1093/aje/155.12.1081

- Wang, R., Wu, C., Ma, X., Zhao, Y., Yan, X., & He, J. (2011). Health-related quality of life in Chinese people: A population-based survey of five cities in China. *Scandinavian Journal of Public Health*, *39*(4), 410–418. https://doi.org/10.1177/1403494810395817
- Walters, S. J., Munro, J. F., & Brazier, J. E. (2001). Using the SF- 36 with older adults: A cross-sectional community-based survey. *Age and Ageing*, *30*(4), 337–343. https://doi.org/10.1093/ageing/30.4.337
- Ware, J. E., & Sherbourne, C. D. (1992). The MOS 36-item short-form health survey (SF-36®):

 I. Conceptual framework and item selection. *Medical Care*, 30(6), 473-483.
- Ware, J. E. Jr., & Gandek, B. (1998). Overview of the SF-36 Health Survey and the International Quality of Life Assessment (IQOLA) project. *Journal of Clinical Epidemiology*, 51(11), 903–912. https://doi.org/10.1016/S0895-4356(98)00081-X
- Ware, J. E., Jr., Kosinski, M., & Dewey, J.E. (2003). *Version 2 of the SF-36 Health Survey*. Quality Metric Incorporated.
- Warner, R. S. (2001). The Korean immigrant church as case and model. In H.-Y. Kwon, K. C. Kim, & R. S. Warner (Eds.), *Korean Americans and their religions: Pilgrims and missionaries from a different shore* (pp. 25–52). Pennsylvania State University Press. ISBN 0271020725
- Webster, J. D., & Gould, O. (2007). Reminiscence and vivid personal memories across adulthood. *International Journal of Aging & Human Development*, 64(2), 149–170. https://doi.org/10.2190/Q8V4-X5H0-6457-5442
- Webster, J. D., Bohlmeijer, E. T., & Westerhof, G. J. (2010). Mapping the future of reminiscence: a conceptual guide for research and practice. *Research on Aging*, 32(4), 527–564. https://doi.org/10.1177/0164027510364122
- Weerasinghe, S. (2012). Inequities in visible minority immigrant women's healthcare accessibility. *Ethnicity and Inequalities in Health and Social Care*, *5*(1), 18–28. https://doi.org/10.1108/17570981211286750
- Weerasinghe, S., & Numer, M. (2011). A life-course exploration of the social, emotional and physical health behaviours of widowed South Asian immigrant women in Canada: Implications for health and social programme planning. *International Journal of Migration, Health and Social Care*, 6(4), 42–56.

 https://doi.org/10.5042/ijmhsc.2011.0153

- Wernstein, L. (2004). Lifelong learning benefits older adults. *Activities, Adaptation and Aging,* 28(4), 1-12. https://doi.org/10.1300/J016v28n04_01
- Westerhof, G. J., Bohlmeijer, E. T., & Webster, J. D. (2010). Reminiscence and mental health: A review of recent progress in theory, research and interventions. *Ageing and Society*, 30(4), 697–721. https://doi.org/10.1017/S0144686X09990328
- Westerhof, G. J., & Bohlmeijer, E. T. (2014). Celebrating fifty years of research and applications in reminiscence and life review: State of the art and new directions. *Journal of Aging Studies*, 29, 107–114. https://doi.org/10.1016/j.jaging.2014.02.003
- Westerhof, G. J. (2018). Life review: Lifespan development, meaning processes, and interventions. In F. Gibson (Ed.), *International perspectives on reminiscence, life review, and life story work* (pp. 312–326). Jessica Kingsley Publishers.
- WHOQOL Group. (1994). Development of the WHOQOL: Rationale and current status. *International Journal of Mental Health*, 23(3), 24–56. http://www.jstor.org/stable/41344692
- WHOQOL Group. (1998). The World Health Organization Quality of Life Assessment (WHOQOL): Development and general psychometric properties, *Social Science Medicne*, 46(12), 1569-1585.
- Wilkins K. (2003). Social support and mortality in seniors. *Health Reports*, *14*(3), 21-34. https://search.proquest.com/scholarly-journals/social-support-mortality-seniors/docview/207490887/se-2
- Williams, R. (1976). Keywords: A vocabulary of culture and society. Oxford University Press.
- Williams, D.R. & Collins, C. (1995). US socioeconomic and racial differences in health: Patterns and explanation. *Annual Review of Sociology 21*(1), 349-386. https://doi.org/10.1146/annurev.so.21.080195.002025
- Williams, D.R., & Sternthal, M. J. (2010). Understanding racial-ethnic disparities in health: Sociological contributions. *Journal of Health and Social Behavior 51*(1_suppl), S15-S27. https://doi.org/10.1177/0022146510383838
- Williams, M., Unrau, Y. A., & Grinnell, R. M. (2003). Research methods for social workers: *A generalist approach for BSW students*. Eddie Bowers Publishing.
- Withnall, A. (2000). The debate continues: Integrating educational gerontology and lifelong learning. In F. Glendenning & S. A. Cusack (Eds.), *Teaching and Learning in Later Life:*

- Theoretical implications (pp. 87–98). Ashgate Arena.
- Withnall, A. (2006) Exploring influences on later life learning. *International Journal of Lifelong Education*, 25(1), 29–49. https://doi.org/10.1080/02601370500309477
- Withnall, A. (2010). *Improving learning in later life*. Routledge.
- Wister, A. W., Malloy-Weir, L. J., Rootman, I., & Desjardins, P. (2010). Lifelong educational practices and resources in enabling health literacy among older adults. *Journal of Aging and Health*, 22(6), 827–854. https://doi.org/10.1177/0898264310373502
- Wong, P. T., & Watt, L. M. (1991). What types of reminiscence are associated with successful aging? *Psychology and Aging*, *6*(2), 272–279. https://doi.org/10.1037/0882-7974.6.2.272
- Wong, S. T., Yoo, G. J., & Stewart, A. L. (2007). An empirical evaluation of social support and psychological well-being in older Chinese and Korean immigrants. *Ethnicity and Health*, 12(1), 43–67. https://doi.org/10.1080/13557850600824104
- World Health Organization. (1986, November 17-21) Ottawa charter for health promotion: An international conference on health promotion. Ottawa, Ontario, Canada. https://www.who.int/teams/health-promotion/enhanced-wellbeing/first-global-conference
- World Health Organization. (1997). The Jakarta Declaration: on leading health promotion into the 21st century = Déclaration de Jakarta sur la promotion de la santé au XXIe siécle. World Health Organization. WHO_HPR_HEP_4ICHP_BR_97.4_eng.pdf (557.1Kb)
- World Health Organization. (2001). *Health and ageing: A discussion paper*. World Health Organization. https://apps.who.int/iris/handle/10665/66682
- World Health Organization. (2002). *Active ageing: A policy framework*. World Health Organization Press.

 https://apps.who.int/iris/bitstream/handle/10665/67215/WHO_NMH_NPH_02.8.pdf?sequence=1&isAllowed=y
- World Health Organization. (2007). *Global age-friendly cities: a guide*. World Health Organization. https://apps.who.int/iris/handle/10665/43755
- World Health Organization. (2014). Ageing and life courses. http://www.who.int/ageing/en/
- World Health Organization. (2020a). *WHO guidelines on physical activity and sedentary behaviour*. World Health Organization. https://apps.who.int/iris/bitstream/handle/10665/336656/9789240015128-eng.pdf

- World Health Organization (2020b). Corona virus disease 2019 (COVID-19): Situation report-90. World Health Organization.
 - https://apps.who.int/iris/bitstream/handle/10665/331686/nCoVsitrep02Apr2020-eng.pdf
- Wolf, M. A. (2009). Older adulthood. In P. Jarvis (Ed.), *The Routledge international handbook of lifelong learning* (pp.56-64). Routledge.
- Xia, L., & Li, H. (2018). Loneliness, social isolation, and cardiovascular health. *Antioxidants & Redox Signaling*, 28(9), 837–851. https://doi.org/10.1089/ars.2017.7312
- Yang, F. & Ebaugh, H. (2001). Transformations in new immigrant religions and their global implications. *American Sociological Review*, 66(2), 269-288.
- Yang, J., Zheng, Y., Gou, X., Pu, K., Chen, Z., Guo, Q., Ji, R., Wang, H., Wang, Y., & Zhou, Y. (2020). Prevalence of comorbidities and its effects in patients infected with SARS-CoV-2: a systematic review and meta-analysis. *International Journal of Infectious Diseases*, 94, 91–95. https://doi.org/10.1016/j.ijid.2020.03.017
- Yardley, L., Bishop, F. L., Beyer, N., Hauer, K., Kempen, G. I., Piot-Ziegler, C., Todd, T., Cuttelod, T., Horne, M., Lanta, K., & Holt, A. R. (2006). Older people's views on falls-prevention intervention in six European countries. *The Gerontologist*, 46(5), 650–660. https://doi.org/10.1093/geront/46.5.650
- Yee Hong Centre for Geriatric Care. (2022, April, 20). *Yee Hong Foundation*. https://www.yeehong.com
- Yeom, H. (2014). Association among ageing-related stereotypic beliefs, self-efficacy, and health-promoting behaviors in elderly Korean adults. *Journal of Clinical Nursing*, 23(9–10), 1365–1373. https://doi.org/10.1111/jocn.12419
- Yin, R. K. (2009). Case study research: Design and methods (4th ed.). Sage Publications.
- Yinger, O. S. (2014). Adapting choral singing experiences for older adults: The implications of sensory, perceptual and cognitive changes. *International Journal of Music Education*, 32(2), 203-212. https://doi.org/10.1177/0255761413508064
- Yip, T., Gee, G. C., & Takeuchi, D. T. (2008). Racial discrimination and psychological distress: The impact of ethnic identity and age among immigrant and United States-born Asian adults. *Developmental Psychology*, 44(3), 787–800. http://dx.doi.org/10.1037/0012-1649.44.3.787

- Yoo, B. H. (1992). Hankook kvovook kwaiungsa vunkoo [The history of Korean school curricular]. Kyohak Yunkoosa.
- Yoo, Y. (2002). Canada and Korea: A shared history. In R. W. L. Guisso, and Y. Yoo (Eds.), *Canada and Korea: Perspectives 2000* (pp. 9-43). University of Toronto Press.
- Yount, W. (2009). Transcendence and Aging: The Secular Insights of Erikson and Maslow. *Journal of Religion, Spirituality & Aging, 21*(1-2), 73–87. https://doi.org/10.1080/15528030802265361
- Zavaleta, D., Kim, S. & Mills, C. T. (2017). Measures of social isolation. *Social Indicators of Research*, *131*(1), 367–391. https://doi.org/10.1007/s11205-016-1252-2
- Zeidan, A. J., Smith, M., Leff, R., Cordone, A., Moran, T. P., Brackett, A., & Agrawal, P. (2023). Limited English proficiency as a barrier to inclusion in emergency medicine-based clinical stroke research. *Journal of Immigrant and Minority Health*, 25, 181-189. https://doi.org/10.1007/s10903-022-01368-y

Appendix A: Major Terminology

MAJOR TERMINOLOGY

This thesis utilizes existing terms as defined in government policies and documents and/or established by WHO. The meanings of some terms may be modified to provide operational definitions for this study.

- The term "senior" is defined, depending upon the governmental or non-governmental agency, as a person aged 50 (e.g. Canadian Association of Retired Persons) to 60 (the United Nations definition of older persons) or 65 (Eligibility for Old Age Security and the Guaranteed Income Supplement). For the purpose of this dissertation, the terms "senior," "older adult" and "elder" or "elderly" are used interchangeably, and all refer to a persons aged 65 or older unless defined otherwise in the body of sentence.
- The term "visible minorities" is defined in Canadian legislation, specifically the Employment Equity Act, as "persons, other than Aboriginal peoples, who are non-Caucasian in race or non-white in colour." (Employment Equity Act, 1995) The visible population consists mainly of the following groups: Chinese, South Asian, Black, Arab, West Asian, Filipino, Southeast Asian, Latin American, Japanese, and Korean (Statistics Canada, n.d., a).
- The term "immigrant" is defined as a person who is or has ever been a landed immigrant/permanent resident who has been granted the right to live in Canada permanently by immigration authorities, including those who have resided in Canada for a number of years and those who have arrived recently; whereas, a person from another country who has a work or study permit or who is a refugee claimant or any non-Canadian-born family member living in Canada with them is referred as a non-permanent resident (Statistics Canada, n.d., b)

In this study, I focus on visible minority immigrant seniors, specifically those who are the first generation Korean immigrants to Canada. It is important to note that visible minority immigrants are different from Canadian-born visible minorities in terms of official language proficiency, cultural background, and familiarity with Canadian society and institutions (Khan, Kobayashi, Lee, & Vang, 2015), which together likely have a substantial impact on how they experience the

world and, subsequently, on their health and well-being. According to the 2011 National Household Survey, Canadian-born visible minorities comprise 30% of the visible minority population (Statistics Canada, 2013a).

- The term "first generation" refers to people who were born outside Canada. In 2011, 22% (about 7.2 million of the total population) representing about 200 countries of birth. The majority (93.9%) are immigrants and 4.9% are non-permanent residents who have work or study permits or are refugee claimants. This population also includes a small group of people (1.2% of the first generation) who are Canadian citizens by birth (Statistics Canada, 2013b).
- The term "second generation" refers to people who were born in Canada and had at least one parent born outside Canada. In 2016, these individuals accounted for 17.7% of the total population (Statistics Canada, 2016).
- The term "third generation" or more refers to people who are born in Canada with both parents born in Canada. In 2011, these individuals accounted for 60.7% of the total population (Statistics Canada, 2013b).
- The term "healthy aging" is described by Health Canada as a lifelong process of optimizing opportunities for improving and preserving health and physical, social, and mental wellness, independence, quality of life, and enhancing successful life-course transitions (Health Canada, 2002). While some may have adapted the term "healthy aging" to have a narrower focus on health, such as focusing on functional ability or health care system, the concept utilized by Health Canada is comparable to the WHO's adoption of active aging as both are rooted in the public health framework and have a wider focus on health and quality of life, including the environment. Based on the discussion, active aging and healthy aging are used interchangeably throughout this thesis. Regardless of which term is used and the subtleties associated with terms, they represent a common goal toward enabling expansion of well-being and health of growing older population.
- The term "active aging" is defined by the World Health Organization as "the process of optimizing opportunities for health, participation and security in order to enhance quality

of life as people age" (WHO, 2002, p. 12). "Active" refers to lifelong participation in social, economic, cultural, spiritual, and civic areas and does not only pertain to a person's employability or ability to remain physically active. In defining "good health", the WHO adopted and kept a holistic view, stating that "good health is not merely the absence of illness or infirmity, but a state of complete physical, mental and social well-being" (as cited in Shields & Martel, 2006, p. 9).

- The term "quality of life" can be defined as "an individual's perception of his or her position in life in the context of the culture and value system where they live, and in relation to their goals, expectations, standards and concerns" (WHO, 1998, p.17). It is a broad concept that incorporate one's physical and psychological health, functional independence, social relationships, personal beliefs and relationship to salient features in the environment (WHO, 1998).
- The term "lifelong learning" can be broadly referred to as "All learning activity undertaken throughout life, with the aim of improving knowledge, skills and competences within a personal, civic, social and/or employment-related perspective" (European Commission [EC], 2001, p. 9). More recent view adds lifewide concept to include, all three types of learning, formal, non-formal, and/or informal learning that takes place across the full range of life activities, which is embedded in all life contexts (Jones, 2009; Laal, 2011). Relevant to this study, Confucius heritage cultural belief that learning is an effort to be made throughout one's life to instill a good moral character (Kim, 2004), is reflected in the group's attitude of learning.
- The term "life review" refers to the recollection and attribution of meaning to past memories throughout one's lifespan. The concept of life review was first advanced by Robert Butler (1963) who proposed that this process is a natural, universal, and normative process that helps one to come to terms with past conflicts.

Appendix B: The SF-36v2 Health Survey



The SF-36v2™ Health Survey

Instructions for Completing the Questionnaire

Please answer every question. Some questions may look like others, but each one is different. Please take the time to read and answer each question carefully by filling in the bubble that best represents your response.

EXAMPLE

This is for your review. Do not answer this question. The questionnaire begins with the section **Your Health in General** below.

For each question you will be asked to fill in a bubble in each line:

How strongly do you agree or o	disagree wit	th each of th	e following st	atements?	
	Strongly agree	Agree	Uncertain	Disagree	Strongly disagree
a) I enjoy listening to music.	0			0	0
b) I enjoy reading magazines.	•	0	0		0

Please begin answering the questions now.

Your Health in General

1. In general, would you say your health is:

Excellent	Very good	Good	Fair	Poor
O ₁	O ₂	O ₃	O ₄	○5

GH01

2.	Compared to one	year ago, how would	you rate your	r health in	general <u>now</u>	?
----	-----------------	---------------------	---------------	-------------	--------------------	---

Much better now than one year ago	Somewhat better now than one year ago	About the same as one year ago	Somewhat worse now than one year ago	Much worse now than one year ago
01	O ₂	\bigcirc_3	O ₄	\bigcirc_5

нт

Please turn the page and continue.

3. The following questions are about activities you might do during a typical day. Does <u>your health</u> now limit you in these activities? If so, how much?

iow little you in these activities: It so, now much:				
	Yes, limited a lot	Yes, limited a little	No, not limited at all	
Vigorous activities, such as running, lifting heavy objects, participating in strenuous sports	01	O ₂	O ₃	F
Moderate activities, such as moving a table, pushing a vacuum cleaner, bowling, or playing golf	O ₁	O ₂	O ₃	F
c) Lifting or carrying groceries	01	O ₂	O ₃	F
d) Climbing several flights of stairs	O ₁	O ₂	O ₃	'
e) Climbing one flight of stairs	O ₁	O ₂	O ₃	'
f) Bending, kneeling, or stooping	01	O ₂	. O ₃	
g) Walking more than a mile	01	O ₂	O ₃	
h) Walking several hundred yards	01	O ₂	O ₃	
i) Walking one hundred yards	01	O ₂	O ₃	
j) Bathing or dressing yourself	01	02	O ₃	
	I	1	1	_

4. During the <u>past 4 weeks</u>, how much of the time have you had any of the following problems with your work or other regular daily activities <u>as a result of your physical health?</u>

	All of the time	Most of the time	Some of the time	A little of the time	None of the time	
a) Cut down on the amount of time you spent on work or other activities	O ₁	O ₂	○3	04	O ₅	RP01
b) Accomplished less than you would like	0,	O_2	○3	○4	O_5	RP02
Were limited in the kind of work or other activities	01	O ₂	O ₃	O ₄	O ₅	RP03
d) Had difficulty performing the work or other activities (for example, it took extra effort)	01	○2	○3	O ₄	O ₅	RP04

BP01

BP02

5. During the <u>past 4 weeks</u>, how much of the time have you had any of the following problems with your work or other regular daily activities <u>as a result of any emotional problems</u> (such as feeling depressed or anxious)?

×

depiessed of afficus;						
,	All of the time	Most of the time	Some of the time	A little of the time	None of the time	
a) Cut down on the amount of time you spent on work or other activities	01	O ₂	O ₃	O ₄	O ₅	RE01
b) Accomplished less than you would like	01	O ₂	○3	O ₄	O ₅	RE02
c) Did work or other activities less carefully than usual	01	O ₂	○3	O ₄	05	RE03

6. During the <u>past 4 weeks</u>, to what extent has your physical health or emotional problems interfered with your normal social activities with family, friends, neighbors, or groups?

Not at all	Slightly	Moderately	Quite a bit	Extremely	
O ₁	O_2	O ₃	. 04	○5	SF01

8. During the past 4 weeks, how much did pain interfere with your normal work (including both work

outside the non	ie and nouseworky:				1
Not at all	A little bit	Moderately	Quite a bit	Extremely	
01	O_2	\bigcirc_3	O ₄ .	O ₅	

9. These questions are about how you feel and how things have been with you <u>during the past 4 weeks</u>. For each question, please give the one answer that comes closest to the way you have been feeling. How much of the time during the <u>past 4 weeks</u>...

	All of the time	Most of the time	Some of the time	A little of the time	None of the time	
a) did you feel full of life?	01	O ₂	O ₃	04	O ₅	VT01
b) have you been very nervous?	01	O ₂	O ₃	O ₄	O ₅	MH01
c) have you felt so down in the dumps that nothing could cheer you up?	01	O ₂	O ₃	O ₄	O ₅	MH02
d) have you felt calm and peaceful?	O ₁	O ₂	O ₃	O ₄	O ₅	мн03
e) did you have a lot of energy?	. O ₁	O ₂	O ₃	O ₄	O ₅	VT02
f) have you felt downhearted and depressed?	01	02	○3	O ₄	O ₅	MH04
g) did you feel worn out?	01	O ₂	\bigcirc_3	O ₄	O ₅	VT03
h) have you been happy?	01	O ₂	O ₃	O ₄	O ₅	MH05
i) did you feel tired?	01	O ₂	O ₃	O ₄	O ₅	VT04

10. During the <u>past 4 weeks</u>, how much of the time has your <u>physical health or emotional problems</u> interfered with your social activities (like visiting friends, relatives, etc.)?

All of the time	Most of the time	Some of the time	A little of the time	None of the time	
O ₁ ·	O_2	\bigcirc_3	O ₄	○5	SF

11. How TRUE or FALSE is each of the following statements for you?

HOW TRUE OF PALSE IS EACH OF THE	tonowing st	atemento io	you.	,		
er en	Definitely true	Mostly true	Don't know	Mostly false	Definitely false	
a) I seem to get sick a little easier than other people	O ₁	O ₂	\bigcirc_3	O ₄	○5	•
b) I am as healthy as anybody I know	O ₁	O ₂	O ₃	O ₄	O ₅	
c) I expect my health to get worse	O ₁	\bigcirc_2	○3	O ₄	O ₅	•
d) My health is excellent	01	O ₂	\bigcirc_3	O ₄	O ₅	G

Curriculum Vitae – Elisha Jo

EDUCATIONAL BACKGROUND

In progress	Ph.D. Candidate, Music Education, The University of Western Ontario
2010	Masters of Music, Music Education, The University of Western Ontario
2007	Masters of Music, Organ Performance, McGill University
2004	Bachelor of Music, Organ Performance, McGill University

PUBLICATIONS

Chapter in an edited book

Jo, H. E., Veblen, K. K., & Potter, P. J. (2021). Remembering a hard life with joy: Music-making among Korean elders, In L Willingham (Ed.). *Community Music at the Boundaries* (pp. 498-510). Wilfrid Laurier University Press. https://books-scholarsportal-info.proxy1.lib.uwo.ca/uri/ebooks/ebooks6/upress6/2021-05-13/1/9781771124591

Veblen, K., Jo, H. E., & Messenger, S. J. (2016). Teaching the supreme art: Pre-service perceptions of creative opportunities in the higher education music class. In E. Haddon & P. Burnard (Eds.), *Creative teaching for creative learning in higher education* (pp.93-105). Abingdon, Oxon: Routledge.

Book Review

Jo, E. (2012). Musicians as lifelong learners: Discovery through biography, by Rineke Smilde (2009). *International Journal of Community Music* 5(3), 305-307.

Online publication and Conference proceeding

Jo, E. & Linton, L. (2010). "Musical Identity and Culture: Exploring the Korean Diaspora through the Lens of Piano Pedagogy" PowerPoint presentation at the *Leading Music Education International Conference* published through the Scholarship@Western website: http://ir.lib.uwo.ca/lme/May31/Program/12/

Jo, H. E., Veblen, K. K., & Potter, P. J. (2014, July). *Korean immigrant seniors' music making in an ethno-cultural community program in Canada*, Paper presented at the ISME Community Music Activity Commission Seminar, Salvador Bahia, Brazil.

PRESENTATIONS, PAPERS, AND POSTERS

Elisha Jo & Kari Veblen. "Remembering a Hard Life with Joy: Music-making among Korean elders." Paper presentation at the international community music conference. Wilfrid Laurier University in Waterloo, ON Canada. May 11-13, 2017

Elisha Jo & Kari Veblen. "Remembering a Hard Life with Joy: Musicing Among Korean Elders" Paper presentation at the *Suncoast Music Education Research Symposium (SMERS X): Music Learning Across the Lifespan.* Florida, U.S.A. February 4-7 2015

Elisha Jo & Kari Veblen. "Creative Opportunities in the Elementary Music Education Class: Perceptions of Students-Turning-Music Teachers." Poster Presentation at the *International Society for Music Education (ISME) World Conference*. Porte Alegre, Brazil. July 20-25, 2014.

Elisha Jo & Kari Veblen. "Korean Immigrant Seniors' Music Making in an Ethno-Cultural Community Program in Canada." Paper presentation at the *International Society for Music Education (ISME) Community Music Activity (CMA) Commission*. Salvador Bahia, Brazil. July 15-18, 2014.

Elisha Jo & Leslie Linton. (2011, May) "Musical Identity and Culture; Exploring the Korean Diaspora Through the Lens of Piano Pedagogy." Paper Presentation at the *Leading Music Education: International Conference*. Western University, ON, Canada. May 29-June 1, 2011.

Elisha Jo & Leslie Linton. "Piano Pedagogy and Identity: Diasporic Perception of Stability and Fluidity through Music Training in Canada and Korea." Paper Presentation at the *Multidisciplinary Research in Music Pedagogy: 2010 International Conference*. University of Ottawa, ON, Canada. May 26-29, 2010.

Jo, H. E. (2015, Winter). Strengthening community through music. *Ensemble*, p. 7.

TEACHING EXPERIENCE

Teaching/Research Assistantship - Don Wright Faculty of Music, University of Western Ontario Winter 2014 **Psychology of Music**: 4th year undergraduate course

- Delivered a 2-hour lecture on the topic of "Transfer of Learning"
- Graded exams and papers and provided comments for improvement
- Advised students for potential research topics and relevant resource

Fall 2013, Fall 2011 Introduction to Music Education: required undergraduate course

- Conducted occasional lectures and led class discussions
- Graded assignments and term papers to provide comments for improvement
- Counseled students experiencing difficulties with the course during office hours

Winter 2013 **Elementary Music Education and Canadian Content**: 4th year undergraduate course for Music Education majors

- Conducted occasional lectures and led class discussions
- Evaluated students' performances including presentations and in-class activities
- Guided and counseled students for potential research topics on class presentation and final project
- Developed class assignments and projects with the primary course instructor

Fall 2013. Music, Education and Culture: undergraduate course

- Lectured on music across diverse cultures, musical concepts, and arts curriculums in discussions with the primary course instructor
- Led class discussions and activities

Fall 2011 - Winter 2012. **Elementary Music Education**: undergraduate course for Music

Education majors

- Conducted occasional short lectures and facilitated class discussions
- Developed assignments with the primary course instructor
- Provided guidance to students for final projects and research ideas
- Graded assignments and provided comments for improvement
- Conducted a creativity project with the primary course instructor
 - o Researched and reviewed literature related to perception of creativity
 - o Interviewed students to collect and documented data

Fall 2010 - Winter 2011. Choral Conducting: undergraduate course

- Assisted as a piano accompanist for the class
- Provided feedback during individual practice sessions and collaborated as a piano accompaniment for in-class conducting evaluations
- Practiced requested repertories for individual students as well as for class activities

Fall 2009 - Winter 2011. Vocal Methods: required 1st year undergraduate course

- Coached individual students during practice sessions and collaborated as a piano accompaniment for the class performance
- Practiced requested repertories for individual students as well as for class activities